

Kent Academic Repository

Oppong, Isaac (2021) A Quantum Deformation of the Second Weyl Algebra: Its derivations and poisson derivations. Doctor of Philosophy (PhD) thesis, University of Kent,.

Downloaded from https://kar.kent.ac.uk/92766/ The University of Kent's Academic Repository KAR

The version of record is available from https://doi.org/10.22024/UniKent/01.02.92766

This document version UNSPECIFIED

DOI for this version

Licence for this version CC BY (Attribution)

Additional information

Versions of research works

Versions of Record

If this version is the version of record, it is the same as the published version available on the publisher's web site. Cite as the published version.

Author Accepted Manuscripts

If this document is identified as the Author Accepted Manuscript it is the version after peer review but before type setting, copy editing or publisher branding. Cite as Surname, Initial. (Year) 'Title of article'. To be published in *Title of Journal*, Volume and issue numbers [peer-reviewed accepted version]. Available at: DOI or URL (Accessed: date).

Enquiries

If you have questions about this document contact <u>ResearchSupport@kent.ac.uk</u>. Please include the URL of the record in KAR. If you believe that your, or a third party's rights have been compromised through this document please see our <u>Take Down policy</u> (available from <u>https://www.kent.ac.uk/guides/kar-the-kent-academic-repository#policies</u>).

A QUANTUM DEFORMATION OF THE SECOND WEYL ALGEBRA: ITS DERIVATIONS AND POISSON DERIVATIONS

By

Isaac Oppong

Supervised by Professor Stéphane Launois

School of Mathematics, Statistics and Actuarial Science

University of Kent

A THESIS SUBMITTED TO THE UNIVERSITY OF KENT FOR THE AWARD OF DOCTOR OF PHILOSOPHY (IN MATHEMATICS)

July 2021

Acknowledgements

First and foremost, my sincere gratitude goes to the Almighty God for his protection over me throughout my studies at the University of Kent. I am very thankful to my supervisor Prof. Stéphane Launois for taking up the full responsibility of supervising me and ensuring that I came out with the best in my research work. Prof, as I usually call you, thank you so much, you have deeply inspired me with your patience, dedication and commitment. I feel honoured to have you as my PhD supervisor. To the School of Mathematics, Statistics and Actuarial Science, you made my study possible by providing me with the needed funding, I heartily appreciate the financial support.

Again, my profound gratitude goes to my examiners—Samuel Lopes and Chris Woodcock—for spending the time to read through my thesis and providing me with useful comments/suggestions. I am forever grateful. I also had some interesting discussions with Lewis Topley on Lie algebras. Many thanks to you Lewis for your eagerness to always help. To Alex Roger (my predecessor:)), thank you so much for reading through my work and providing me with helpful comments. You did beyond expectations, I truly appreciate your efforts. Whoever comes into contact with Claire Carter in the department would be marvelled at her commitment and dedication to work. Claire, I am very grateful to you for all that you did for me and inspiring me with your enthusiasm and professionalism.

To my wife Abigail, you have been so supportive in this journey. You were always curious to understand what I was doing on my laptop, however, maths never showed you her beauty. May the good Lord continue to bless you and our little baby girl, Christabel. I also thank my Dad, Mum and all my siblings (Nicho, Bea, Mike, Eunice, Sam and Evans) for your prayers and always calling on phone to check up on me. Finally, to Stratford Church of Christ, I say a big thank you for the fellowship we had throughout my studies. You have been more than a family, God bless you all.

Abstract

Since the introduction of quantum algebras in the 1980's, many have introduced quantum deformations of the Weyl algebras. Two such examples are the quantum Weyl algebras and the Generalized Weyl algebras. In this thesis, we use a different approach to find deformations of (a quadratic extension of) the second Weyl algebra $A_2(\mathbb{C})$, and compare some properties of these deformations to those of $A_2(\mathbb{C})$.

Let n be a nilpotent Lie algebra, $U(\mathfrak{n})$ the enveloping algebra of n and Q a primitive ideal of $U(\mathfrak{n})$. Dixmier [12] proved that the factor algebra $U(\mathfrak{n})/Q$ is isomorphic to an n^{th} Weyl algebra $A_n(\mathbb{C})$, where $n \in \mathbb{N}_{\geq 1}$. This isomorphism gives a route to construct potential deformations for any Weyl algebra. Let $\mathfrak{g} = \mathfrak{g}^- \oplus \mathfrak{h} \oplus \mathfrak{g}^+$ represent a simple Lie algebra. Now, Dixmier's result holds for $\mathfrak{n} = \mathfrak{g}^+$. Since $U_q^+(\mathfrak{g})$ is a q-deformation of $U(\mathfrak{g}^+)$, it is natural to consider $U_q^+(\mathfrak{g})/P$, where P is a primitive ideal of $U_q^+(\mathfrak{g})$, as a potential deformation of the Weyl algebras.

This thesis focuses on the case where $\mathfrak{g} = G_2$. We find a family of primitive ideals $(P_{\alpha,\beta})_{(\alpha,\beta)\in\mathbb{C}^2\setminus(0,0)}$ of $U_q^+(G_2)$ whose corresponding quotients $A_{\alpha,\beta} := U_q^+(G_2)/P_{\alpha,\beta}$ are simple noetherian domains of Gelfand-Kirillov dimension 4. In view of Dixmier's result, we consider $A_{\alpha,\beta}$ as a q-deformation of (a quadratic extension of) $A_2(\mathbb{C})$. The derivations of the Weyl algebras are all known to be inner derivations [5]. Motivated by this, we also study the derivations of $A_{\alpha,\beta}$ and compare them to those of the Weyl algebras. The final part of the thesis studies a Poisson derivation of a semiclassical limit $\mathcal{A}_{\alpha,\beta}$ of $A_{\alpha,\beta}$. Interestingly, the Poisson derivations of $\mathcal{A}_{\alpha,\beta}$ and the derivations of $A_{\alpha,\beta}$ are congruent.

Contents

Ac	Acknowledgements						
Abstract							
0	Intr	roduction					
1	liminaries	9					
	1.1	Root systems	9				
	1.2	Quantized enveloping algebras	13				
	Ore extension	17					
1.4 Localization and rings of fractions							
	1.5	Quantum affine space and quantum torus	20				
	1.6	Prime spectrum	21				
		1.6.2 Rational torus action.	23				
		1.6.5 \mathcal{H} -Stratification.	24				
	1.7	Deleting derivations algorithm (DDA)	26				
2	Que	est for height two maximal ideals of $U_q^+(G_2)$	30				
	2.1	The algebra $U_q^+(G_2)$	31				
	2.2	2.2 DDA and the center of $U_q^+(G_2)$					
		2.2.1 DDA of $U_q^+(G_2)$	34				
		2.2.3 The center of $U_q^+(G_2)$.	36				
2.3 Proof of the completely primeness of $\langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$							

	2.4	Height	two maximal ideals of $U_q^+(G_2)$	47				
		2.4.1	$\mathcal{H}-Spec(A)$.	47				
		2.4.3	$\mathcal{H}-$ strata corresponding to $\langle 0 \rangle$, $\langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$	49				
3	Sim	tients of $U_q^+(G_2)$ and their relation to the second Weyl algebra	a 56					
	3.1	Gelfand	I-Kirillov dimension of $A_{\alpha,\beta}$	58				
	3.2	Linear I	basis for $A_{lpha,eta}$	60				
	3.3	$A_{lpha,eta}$ as	a q -deformation of a quadratic extension of $A_2(\mathbb{C})$	68				
4	Derivations of the simple quotients of $U_q^+(G_2)$ 7							
	naries	73						
	4.2	Derivat	ions of $A_{\alpha,\beta}$	84				
		4.2.1	Derivations of $A_{\alpha,\beta}$ $(\alpha,\beta \neq 0)$.	84				
		4.2.5	Derivations of $A_{\alpha,0}$ and $A_{0,\beta}$.	103				
5	Sem	iclassic	al limit of the simple quotients of $U_q^+(G_2)$	112				
5.1 Preliminaries								
		5.1.1	Definitions and examples.	113				
		5.1.4	Semiclassical limit.	114				
		5.1.6	Poisson prime spectrum.	116				
		5.1.8	Poisson deleting derivation algorithm (PDDA).	118				
	5.2	Semicla	issical limit of $U_q^+(G_2)$	121				
		5.2.1	PDDA of \mathcal{A} .	124				
		5.2.5	Poisson $\mathcal{H}-invariant$ prime ideals of \mathcal{A} of at most height 2	129				
		5.2.14	\mathcal{H} -strata corresponding $\langle 0 \rangle, \langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$	136				
	issical limit of $A_{lpha,eta}$	139						
6	Poi	sson de	rivations of the semiclassical limits of $A_{lpha,eta}$	148				
	6.1 Poisson derivations of Poisson group algebras							

6.3	B Poisson derivations of $\mathcal{A}_{lpha,eta}$					
	6.3.1	Poisson derivations of $\mathcal{A}_{\alpha,\beta}$ $(\alpha,\beta\neq 0)$.	161			
	6.3.5	Poisson derivations of $\mathcal{A}_{\alpha,0}$ and $\mathcal{A}_{0,\beta}$.	178			
Appendices						
А	Comput	vations in $U_q^+(G_2)$	183			
В	Computations in $\mathcal{A} = \mathbb{C}[X_1, \cdots, X_6]$					
С	Definitio	on of scalars used	196			
References						

Chapter 0

Introduction

One particular class of algebras in non-commutative algebras that has been widely studied since its inception is the class of Weyl algebras. Dirac is well noted to be the one who introduced the Weyl algebras in the field of quantum mechanics. Since then, the notion and concept of the Weyl algebras has diffused into other areas of non-commutative algebras from different perspectives. For example, one can understand the Weyl algebras from the perspective of differential operators, quotient of free algebras, and of course quantum mechanics. In line with this, we will provide the definition of Weyl algebras from three different perspectives. Before we do that, it is imperative to also add that the Weyl algebras come in series. That is, for every $n \in \mathbb{N}_{>0}$, one can associate a Weyl algebra denoted by $A_n(\mathbb{C})$. Now, $A_n(\mathbb{C})$ is naturally called the n^{th} Weyl algebra. The smallest of this series of algebras is called the first Weyl algebra $A_1(\mathbb{C})$, and it is the building block for any n^{th} Weyl algebra $A_n(\mathbb{C})$. That is, given n copies of $A_1(\mathbb{C})$, one can define $A_n(\mathbb{C})$ as follows:

$$A_n(\mathbb{C}) = A_1(\mathbb{C}) \otimes A_1(\mathbb{C}) \otimes \cdots \otimes A_1(\mathbb{C})$$
 (*n* copies).

As a result of this, it is sufficient to define the first Weyl algebra $A_1(\mathbb{C})$. We are now ready to understand $A_1(\mathbb{C})$ from the following three points of view.

In theoretical physics (particularly, quantum mechanics), one can describe $A_1(\mathbb{C})$ as

follows. Take the momentum operator \hat{p} and position operator \hat{q} , it is well known that $[\hat{p}, \hat{q}] := \hat{p}\hat{q} - \hat{q}\hat{p} = i\hbar$, where \hbar is called Plank's constant. One can therefore define $A_1(\mathbb{C})$ as an algebra of quantum mechanics generated by the operators \hat{g} and \hat{q} subject to the relation $\hat{g}\hat{q} - \hat{q}\hat{g} = 1$, where $\hat{g} = -\frac{i}{\hbar}\hat{p}$.

Secondly, in the ring of differential operators, $A_1(\mathbb{C})$ can be described in the following context. Let $\mathbb{C}[x]$ be a polynomial ring and, \hat{x} and $\hat{y} := \frac{d}{dx}$ be operators of $\mathbb{C}[x]$ with \hat{x} defined as $\hat{x}(f) = xf$, $\forall f \in \mathbb{C}[x]$. Observe that $\hat{y}\hat{x}(f) = \frac{d}{dx}(xf) = x\frac{df}{dx} + f = \hat{x}\hat{y}(f) + f$. This implies that $\hat{y}\hat{x} - \hat{x}\hat{y} = 1$. As a result, $A_1(\mathbb{C})$ can be defined as a \mathbb{C} -subalgebra of $\operatorname{End}_{\mathbb{C}}(\mathbb{C}[x])$ generated by the operators \hat{x} and \hat{y} (subject to the relation $\hat{y}\hat{x} - \hat{x}\hat{y} = 1$).

Algebraically, one can also define $A_1(\mathbb{C})$ as follows. Let $R := \mathbb{C}\langle X, Y \rangle$ be a free algebra generated by X and Y. Now, $I := \langle XY - YX - 1 \rangle$ is a two-sided ideal of R. The factor algebra R/I satisfies the relation xy - yx - 1 = 0, where x := X + I and y := Y + I. Therefore, $A_1(\mathbb{C})$ can be defined as a \mathbb{C} -algebra generated by x and y subject to the relation xy - yx = 1.

Above are three different contexts in which $A_1(\mathbb{C})$ can be understood. Generally, every n^{th} Weyl algebra $A_n(\mathbb{C})$ can precisely be described as a \mathbb{C} -algebra generated by $x_1, \dots, x_n, y_1, \dots, y_n$ subject to the following defining relations: $[x_i, x_j] = [y_i, y_j] = 0$ and $[x_i, y_j] = \delta_{ij}$, where δ_{ij} is the Kronecker symbol. The Weyl algebras $A_n(\mathbb{C})$ are simple, noetherian domains, and have a Gelfand-Kirillov dimension of 2n [30, Chapter 8]. Dixmier has also studied the automorphism group of $A_1(\mathbb{C})$, and concluded in [11, Theorem 8.10] that the automorphism group is generated by two families of automorphisms $\phi_{s,\mu}$ and $\phi'_{s,\mu}$ of $A_1(\mathbb{C})$ defined as follows: $\phi_{s,\mu}(x) = x$, $\phi_{s,\mu}(y) = y + \mu x^s$ and $\phi'_{s,\mu}(x) =$ $x + \mu y^s$, $\phi'_{s,\mu}(y) = y$, where $\mu \in \mathbb{C}$ and $s \in \mathbb{N}$. For n > 1, to the best knowledge of the author, the automorphism groups of $A_n(\mathbb{C})$ are not known. The center of the Weyl algebras is reduced to scalars. Another property of the Weyl algebras that is worth mentioning is that all its derivations are inner [5, Lemma 1], and the first Hochschild cohomology group is of dimension zero. What has even made the Weyl algebras very famous and instigated a lot of research in non-commutative algebras are the Jacobian Conjecture [28], and the Dixmier Conjecture [11]. These two Conjectures are stated below.

Conjecture 1 (Jacobian, 1939). Let $F : \mathbb{C}^n \to \mathbb{C}^n$ be a polynomial map. If the Jacobian determinant of F is a non-zero constant, then F must have an inverse polynomial map function.

The case where n = 1 is considered trivial since one can easily show that the associated polynomial map is a linear map, hence, has an inverse map which is also linear. To achieve this, let $f : \mathbb{C} \to \mathbb{C}$ be a polynomial map defined by $x \mapsto f(x)$. Suppose that df/dx is a non-zero constant, then clearly f(x) is of the form $f(x) = a_0 + a_1 x$, where $a_1 \in \mathbb{C}^*$ and $a_0 \in \mathbb{C}$. Since $a_1 \neq 0$, one can easily confirm that the inverse polynomial map of f is also of the form $f^{-1}(x) = -a_0 a_1^{-1} + a_1^{-1} x$ as desired. For n > 1, to the author's best knowledge, the Jacobian Conjecture remains open.

Conjecture 2 (Dixmier, 1968). Every algebra endomorphism of Weyl algebras $A_n(\mathbb{C})$ is an automorphism.

Dixmier's Conjecture remains open even for the case where n = 1. Tsuchimoto [43], and Belov-Kanel and Kontsevich [6] independently proved that Conjecture 1 and Conjecture 2 are stably equivalent. That is, the Jacobian Conjecture implies the Dixmier Conjecture and vice versa.

With the appearance of quantum groups in the 1980's, people have introduced various quantum deformations (or analogues) of $A_n(\mathbb{C})$. The notable deformations are the quantum Weyl algebras. For example, the first quantum Weyl algebra is defined as $A_1^{(q)}(\mathbb{C}) := \mathbb{C}\langle x, y \mid xy - qyx = 1 \rangle$. Clearly, $A_1^{(1)}(\mathbb{C}) = A_1(\mathbb{C})$. As a result of this, we say that $A_1^{(q)}(\mathbb{C})$ is a quantum deformation (or q-deformation for short) of $A_1(\mathbb{C})$. By extension, the quantum Weyl algebras $A_n^{(q)}(\mathbb{C})$ are q-deformations of the Weyl algebras $A_n(\mathbb{C})$ for each $n \in \mathbb{N}_{>0}$. Note, the properties of the quantum Weyl algebras do not always reflect the properties of the Weyl algebras. For example, the first quantum Weyl algebra is not simple when q is not a root of unity, however, the first Weyl algebra is simple. Are there any other deformations of $A_n(\mathbb{C})$? Yes, Bavula [3] has introduced a generalization of the Weyl algebras called the generalized Weyl algebras (GWAs), but they are also not simple in general. As a result, their properties too do not always reflect the properties of the Weyl algebras. Let D be a \mathbb{C} -algebra with a central element a, and σ denote an automorphism of D. The first generalized Weyl algebra $D(\sigma, a)$ is simply defined as an algebra generated over D by the indeterminates x and y subject to the following relations:

$$yx = a$$
, $xy = \sigma(a)$, $xd = \sigma(d)x$, $yd = \sigma^{-1}(d)y$, for all $d \in D$.

Let $q \in \mathbb{C}^*$ and $\lambda' \in \mathbb{C}$. From [41, Proposition 2.1.1], we have that the GWA $\mathbb{C}[H'](\sigma', a')$, where $\sigma(H') = qH' - \lambda'$, is isomorphic to one of the following:

- i. $\mathbb{C}[H](\mathsf{id}, a)$,
- ii. $\mathbb{C}[H](\sigma, a)$ with $\sigma(H) = H 1$,
- iii. $\mathbb{C}[H](\sigma, a)$ with $\sigma(H) = qH$ and $q \in \mathbb{C} \setminus \{0, 1\}$.

The third family (iii) is usually called the *quantum GWA* provided q is not a root of unity, and the second family (ii) is well known to be isomorphic to the first Weyl algebra $A_1(\mathbb{C})$ (for example, see [2, §4]). Other specific examples of GWAs are the quantum plane and quantum Weyl algebra $A_1^{(q)}(\mathbb{C})$ (see [2, §4] for further details).

In this thesis, we used a different approach to find a deformation of (a quadratic extension of) $A_2(\mathbb{C})$. Let \mathfrak{n} be a finite dimensional nilpotent Lie algebra over \mathbb{C} , $U(\mathfrak{n})$ be the enveloping algebra of \mathfrak{n} , and Q a primitive ideal of $U(\mathfrak{n})$. Dixmier has shown that the factor algebra $U(\mathfrak{n})/Q$ is isomorphic to an n^{th} Weyl algebra $A_n(\mathbb{C})$ [12, Theorem 4.7.9].

Let \mathfrak{g} be a simple Lie algebra. Then, the enveloping algebra $U(\mathfrak{g})$ has a quantum deformation, introduced independently by Drinfeld [13] and Jimbo [27], called *the quantized enveloping algebra* $U_q(\mathfrak{g})$ of \mathfrak{g} . There is a triangular decomposition for \mathfrak{g} as $\mathfrak{g}^- \oplus \mathfrak{h} \oplus \mathfrak{g}^+$; where \mathfrak{h} is a Cartan subalgebra of \mathfrak{g} , and \mathfrak{g}^- and \mathfrak{g}^+ are the negative and positive nilpotent subalgebras of \mathfrak{g} respectively. As a result of this triangular decomposition, there are corresponding decompositions for $U(\mathfrak{g})$ and $U_q(\mathfrak{g})$ as $U(\mathfrak{g}^+) \otimes U(\mathfrak{h}) \otimes U(\mathfrak{g}^-)$ and $U_q^+(\mathfrak{g}) \otimes U^0(\mathfrak{h}) \otimes U_q^-(\mathfrak{g})$ respectively. It is well known that $U_q^+(\mathfrak{g})$ is isomorphic to $U_q^-(\mathfrak{g})$ (for example, see [45, §2]). Now, Dixmier's theorem applies to $\mathfrak{n} = \mathfrak{g}^+$. That is, $U(\mathfrak{g}^+)/Q \cong A_n(\mathbb{C})$. Since $U_q^+(\mathfrak{g})$ is a q-deformation of $U(\mathfrak{g}^+)$, it is natural to ask whether there exists a primitive ideal P of $U_q^+(\mathfrak{g})$ such that $U_q^+(\mathfrak{g})/P$ is also a q-deformation of $U(\mathfrak{g}^+)/Q \cong A_n(\mathbb{C})$? If yes, then this gives another deformation of $A_n(\mathbb{C})$. In [33], Launois described a family of simple quotients of $U_q^+(B_2)$, and under certain conditions, retrieved the first Weyl algebra from these simple quotients when q = 1. In line with Dixmier's result (i.e. $U(\mathfrak{g}^+)/Q \cong A_n(\mathbb{C})$) and motivated by Launois' example, we also a imed to find a family of simple quotients of a quantized enveloping algebra whose deformation gives (a quadratic extension of) $A_2(\mathbb{C})$ and then compare some properties of these simple quotients to those of $A_2(\mathbb{C})$. This is the main aim of the thesis which is organized as follows.

Chapter 1 focuses on studying preliminary materials, and Chapter 2 studies some properties of the positive part of the quantized enveloping algebra of type G_2 , $U_q^+(G_2)$. The algebra $U_q^+(G_2)$ is a \mathbb{C} -algebra generated by the indeterminates E_1 and E_2 subject to the quantum Serre relations (omitted for now). It is a noetherian domain and can be written as an iterated Ore extension over \mathbb{C} . The center of $U_q^+(G_2)$ is the polynomial ring $\mathbb{C}[\Omega_1, \Omega_2]$, where Ω_1 and Ω_2 are central elements of $U_q^+(G_2)$. The torus $\mathcal{H} = (\mathbb{C}^*)^2$ acts by automorphism on $U_q^+(G_2)$ via $(\alpha_1, \alpha_2) \cdot E_i = \alpha_i E_i$; i = 1, 2; for all $(\alpha_1, \alpha_2) \in \mathcal{H}$. As a result, one can use the \mathcal{H} -stratification of Goodearl and Letzter [21] to study the prime spectrum of this algebra, and identify primitive ideals. The \mathcal{H} -stratification partitions the prime spectrum of $U_q^+(G_2)$ into disjoint strata and these partitions are indexed by the \mathcal{H} -invariant prime ideals. Now, the ideals $\langle 0 \rangle$, $\langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$ are all \mathcal{H} -invariant prime ideals of $U_q^+(G_2)$ of at most height one, with the following strata:

• $\langle 0 \rangle$ -stratum of $U_q^+(G_2) = \{ \langle 0 \rangle \} \cup \{ P(\Omega_1, \Omega_2) \mid P(\Omega_1, \Omega_2) \in \mathcal{P}, P(\Omega_1, \Omega_2) \neq \Omega_1, \Omega_2 \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle \mid \alpha, \beta \in \mathbb{C}^* \}$. Note, \mathcal{P} is the set of all unitary irreducible polynomials of $\mathbb{C}[\Omega_1, \Omega_2]$.

•
$$\langle \Omega_1 \rangle$$
-stratum of $U_q^+(G_2) = \{ \langle \Omega_1 \rangle \} \cup \{ \langle \Omega_1, \Omega_2 - \beta \rangle \mid \beta \in \mathbb{C}^* \}.$

• $\langle \Omega_2 \rangle$ -stratum of $U_q^+(G_2) = \{ \langle \Omega_2 \rangle \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 \rangle \mid \alpha \in \mathbb{C}^* \}.$

For each $(\alpha,\beta) \in \mathbb{C}^2 \setminus \{(0,0)\}$, the prime ideal $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$ is maximal in its respective stratum, as a result, it is a primitive ideal of $U_q^+(G_2)$ [7, Theorem II.8.4]. Moreover, $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$ is a height 2 maximal ideal of $U_q^+(G_2)$. Given this family of height 2 maximal ideals of $U_q^+(G_2)$, Chapter 3 focuses on studying the simple quotients of $U_q^+(G_2)$, which are the main algebras of interest in this thesis. The simple quotient $A_{\alpha,\beta} := U_q^+(G_2)/\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$ is a noetherian domain of GKdim 4, with a trivial center (i.e. central elements are all scalars). One could observe that $A_{\alpha,\beta}$ shares some common properties with $A_2(\mathbb{C})$. That is, they are both simple, noetherian domains, have GKdim of 4, and have trivial central elements. Could it be that $A_{\alpha,\beta}$ can be deformed to obtain $A_2(\mathbb{C})$? Generally, this is not guaranteed. However, in this particular case, at appropriate choices of α and β , when q = 1, we have that $A_{\alpha,\beta}$ is isomorphic to (a quadratic extension of) $A_2(\mathbb{C})$.

In Chapter 4, we explore the derivations of $A_{\alpha,\beta}$. Note, to the best of the author's knowledge, there is no known general results for the derivations of the simple quotients of $U_q^+(\mathfrak{g})$, with the exception of the case where $\mathfrak{g} = \mathfrak{sl}_3$ (also known as the quantum Heisenberg algebra (see for instance [1, §2.2])). In this case, the simple quotients are known to be isomorphic to the GWAs [2, §4] whose derivations have been studied by Kitchin [29, Chapter 7], and are known to be the sum of inner and scalar derivations (note, scalar derivations act on the set of generators of an algebra by multiplication by scalars). More precisely, the first Hochschild cohomology group of the GWAs is of dimension one. We therefore take interest in knowing the derivations of $A_{\alpha,\beta}$, and comparing them to those of $A_2(\mathbb{C})$. Recall that the derivations of $A_2(\mathbb{C})$ are known to be inner. Since $A_{\alpha,\beta}$ and $A_2(\mathbb{C})$ share some common properties, could it be that the derivations of $A_2(\mathbb{C})$, the derivations of $A_{\alpha,\beta}$ are all inner. However, if either α or β is zero, then the derivations of $A_{\alpha,\beta}$ are the sum of inner and scalar derivations. Precisely, the first Hochschild cohomology group of a and β are both non-

zero, and dimension one when either α or β is zero. To achieve these results, we use Cauchon's theory of *deleting derivations algorithm* [8] to successively embed $A_{\alpha,\beta}$ into a suitable quantum torus. Now, every derivation of the quantum torus, through the work of Osborn and Passman [40], is known to be the sum of an inner and a scalar/central derivation. Since $A_{\alpha,\beta}$ can be embedded into a quantum torus, we extend every derivation of $A_{\alpha,\beta}$ to a derivation of the quantum torus via localization, and then "pull back" the derivations of the quantum torus to $A_{\alpha,\beta}$, a process Launois and Lopes [36] called *restoring derivations algorithm*. We conclude that every derivation of $A_{\alpha,\beta}$ is inner when α and β are non-zero, and the sum of an inner and a scalar derivation when either α or β is zero.

The final part of the thesis (Chapters 5 and 6) study the Poisson derivations of a semiclassical limit $\mathcal{A}_{\alpha,\beta}$ of $A_{\alpha,\beta}$ by following procedures similar to that of $A_{\alpha,\beta}$. In fact, the Poisson derivations of $\mathcal{A}_{\alpha,\beta}$ are similar to their non-commutative counterparts. That is, every Poisson derivation of $\mathcal{A}_{\alpha,\beta}$ is inner (or hamiltonian) when α and β are non-zero, and the sum of an inner and a scalar Poisson derivation when either α or β is zero.

Notations and conventions

- K is a field with characteristic zero and $\mathbb{K}^* := \mathbb{K} \setminus \{0\}$.
- $q \in \mathbb{K}^*$ is not a root of unity.
- Let a, b ∈ N := {0, 1, 2, · · · }. Then, a ≤ i ≤ b means the set of natural numbers from a to b. We will used " < " instead of " ≤ " to indicate the exclusiveness of either a or b.
- $\mathbb{N}_{>a}$ denotes the set of all natural numbers greater than a.
- q[•] means an arbitrary integer power of q. This symbol will often be used whenever the power of q is of no interest.
- $|\triangle|$ denotes the cardinality of the set \triangle .

- ⟨Θ⟩_R means an ideal generated by the element Θ in the algebra R. Where no doubt arises, we will simply write ⟨Θ⟩.
- Z(R) denotes the center of the algebra R. If R is a Poisson algebra, then we will denote its Poisson center by Z_P(R).
- Spec(R) represents the set of all prime ideals of the algebra R. If R is a Poisson algebra, then we will denote its set of Poisson prime ideals by P.Spec(R).
- Fract(R) denotes the right ring of fractions of the ring R.
- DDA and PDDA represent *deleting derivations algorithm* and *Poisson deleting derivations algorithm* respectively.

Chapter 1

Preliminaries

In this chapter, we present the background materials needed for the subsequent chapters. The subtopics to be considered include root systems, quantized enveloping algebras, Ore and iterated Ore extensions, localization and ring of fractions, quantum affine space and quantum torus, and Cauchon's theory of deleting derivations algorithm.

1.1 Root systems

Most of the materials presented in this section can be found in [25] and [15].

Let a be a non-zero element in the euclidean vector space \mathbf{E} , and s_a be a reflection associated to a in \mathbf{E} . Recall that $s_a : \mathbf{E} \longrightarrow \mathbf{E}$ is defined by:

$$s_a(b) := b - 2\frac{(a,b)}{(a,a)}a,$$

for all $b \in \mathbf{E}$. Note, $(a, b) = ||a||||b||\cos(\theta)$ and $s_a s_a = s_a^2 = \mathrm{id}$.

1.1.1 Definition. A subset $\Phi \not\ni 0$ of **E** is a *root system* in **E** if the following axioms are satisfied:

(A1) Φ is finite and spans **E**.

(A2) If $\alpha \in \Phi$, then the only scalar multiples of α in Φ are α and $-\alpha$.

(A4) If $\alpha, \beta \in \Phi$; then $\frac{2(\alpha, \beta)}{(\alpha, \alpha)} \in \mathbb{Z}$.

Note, the elements of Φ are called *roots*. A subset Π of Φ is called a *base* for Φ if Π is a basis of **E**, and for all $\theta \in \Phi$, θ belongs to the span of Π over all non-positive or non-negative integers. When θ belongs to the span of Π over non-negative integers (resp. non-positive integers), we say that θ is a positive root (resp. a negative root). Let us denote the collection of all positive roots (resp. negative roots) by Φ^+ (resp. Φ^-). Then, $\Phi^+ = -\Phi^-$. The roots in Π are called *simple roots.* Again, Φ is *irreducible* if it cannot be written as a disjoint union of two proper subsets Φ_1 and Φ_2 such that $(lpha_1, lpha_2) = 0$ for all $\alpha_1 \in \Phi_1$ and $\alpha_2 \in \Phi_2$. Let Π be a base of Φ , it is well known that Φ is irreducible if and only if Π is irreducible (for example, see [25, Section 10.4]). Every root system Φ can be written as a disjoint union of some irreducible root systems. As a result, we will only focus on the irreducible root systems since they are the building blocks for root systems. If Φ is irreducible, then every root of Φ is either of the same length (usually referred to as simply laced) or of two lengths: short and long (usually referred to as non-simply laced). The short root is normally normalized to have a length of $\sqrt{2}$. The dimension of the euclidean vector space E (denoted by dim(E)) is called the *rank* of Φ (denoted by $\mathsf{rk}(\Phi)$). Since Π is a basis of \mathbf{E} , the cardinality of Π (denoted by $|\Pi|$) coincides with the dimension of **E**. As a result, we have the following equality: dim(**E**) = $|\Pi| = \mathsf{rk}(\Phi) = n$.

One important property of irreducible root systems is that, to every irreducible root system, one can associate one of the following complex simple Lie algebras: A_n $(n \ge 1)$, B_n $(n \ge 2)$, C_n $(n \ge 3)$, D_n $(n \ge 4)$, E_n (n = 6, 7, 8), F_4 and G_2 [25, Section 11.4]. Moreover, given any of these simple Lie algebras, one can also construct an irreducible root system associated to the simple Lie algebra [25, Section 12.1]. Note, each of A_n, B_n, C_n, D_n and E_n is the notation for a collection of a series of simple Lie algebras, and n is the rank of the associated Lie algebra or the root system. Of course, F_4 and G_2 are of rank 4 and 2 respectively. The roots of A_n, D_n, E_6, E_7 and E_8 are all simply laced, and the rest are non-simply laced. For the purpose of this thesis, we will only

discuss the root system for G_2 . We do this in line with the presentation in [25, Section 12.1]. Let **E** be a 2-dimensional subspace of \mathbb{R}^3 , and $\hat{\varepsilon}_1, \hat{\varepsilon}_2, \hat{\varepsilon}_3$ denote the standard unit vectors of \mathbb{R}^3 . Set $\alpha_1 := \hat{\varepsilon}_2 - \hat{\varepsilon}_1$ and $\alpha_2 = 2\hat{\varepsilon}_1 - \hat{\varepsilon}_2 - \hat{\varepsilon}_3$. One can verify that the set $\Phi = \pm \{\alpha_1, \alpha_2, \alpha_1 + \alpha_2, 2\alpha_1 + \alpha_2, 3\alpha_1 + \alpha_2, 3\alpha_1 + 2\alpha_2\}$ is a root system for G_2 in **E**. Moreover, one can also observe that the set $\{\alpha_1, \alpha_2\}$ is a base for Φ . Hence, α_1 and α_2 are the simple roots of G_2 . The root system of G_2 is non-simply laced since the simple roots α_1 and α_2 are of different lengths: short root (α_1) and long root (α_2) , which are of lengths $\sqrt{2}$ and $\sqrt{6}$ respectively. That is, $(\alpha_1, \alpha_1) = ||\alpha_1||^2 = 2$ and $(\alpha_2, \alpha_2) = ||\alpha_2||^2 = 6$. In total, G_2 has 12 roots of which 6 are short and the other 6 are long. Furthermore, it has 6 positive roots and 6 negative roots. The obtuse angle between the two simple roots α_1 and α_2 is $5\pi/6$. The root diagram of G_2 with only the positive roots is shown in the figure below.



Figure 1.1: G_2 root diagram.

In the rest of this section and beyond, \mathfrak{g} will denote a complex simple Lie algebra, and Φ will represent the root system of \mathfrak{g} . Fix \mathfrak{g} with rank n. We are going to discuss another concept called the *Weyl group* of \mathfrak{g} . Let Φ denote a root system of \mathfrak{g} , and Π denotes the set of simple roots of Φ . For every root $\alpha_i \in \Phi$, one can associate a reflection s_{α_i} . The collection of all these reflections s_{α_i} forms a group called the *Weyl group* of \mathfrak{g} . We denote this group by \mathcal{W} . Moreover, if $\alpha_i \in \Pi$, then the associated reflection s_{α_i} is called a simple reflection. The set of all the simple reflections generate \mathcal{W} . That is, $\mathscr{W} := \langle s_{\alpha_i} \mid \alpha_i \in \Pi \rangle.$

Let $\alpha_i, \alpha_j \in \Pi$ (note, *i* and *j* are non-zero natural numbers); and set

$$\eta_{ij} := \frac{2(\alpha_i, \alpha_j)}{(\alpha_i, \alpha_i)}.$$
(1.1.1)

It is well known that $\eta_{ij}\eta_{ji} \in \{0, 1, 2, 3\}$ (see [15, Lemma 11.4] for further details). Let $s_{\alpha_i}, s_{\alpha_j} \in \mathcal{W}$; where $\alpha_i, \alpha_j \in \Pi$, $i \neq j$. There exists $m_{ij} \in \mathbb{Z}$ such that $(s_{\alpha_i}s_{\alpha_j})^{m_{ij}} = s_{\alpha_i}^2 = 1$, where m_{ij} is the order of $s_{\alpha_i}s_{\alpha_j}$ in \mathcal{W} , and 1 is the identity element of \mathcal{W} . Again, for $\eta_{ij}\eta_{ji} = 0, 1, 2, 3$, we have that $m_{ij} = 2, 3, 4, 6$ respectively [7, Page 41].

Set $s_i := s_{\alpha_i}$. If $\Pi = \{\alpha_1, \cdots, \alpha_n\}$ is the set of simple roots of Φ , then

$$\mathscr{W} = \langle s_1, \cdots, s_n \mid s_i^2 = (s_i s_j)^{m_{ij}} = 1, \ i \neq j \rangle.$$

The relations $(s_i s_j)^{m_{ij}} = 1$ and $s_i^2 = 1$ imply that

$$s_i s_j s_i \cdots = s_j s_i s_j \cdots$$

Each side of the equality has exactly m_{ij} factors. The Weyl group \mathscr{W} acts on Φ as follows:

$$s_i(\alpha_j) = \alpha_j - \frac{2(\alpha_i, \alpha_j)}{(\alpha_i, \alpha_i)} \alpha_i = \alpha_j - \eta_{ij}\alpha_i,$$

where $s_i \in \mathcal{W}$ and $\alpha_j \in \Phi$. This action just permutes the elements of Φ as stated in axiom A3.

Denote the cardinality of the set of positive roots Φ^+ of Φ by N, and let $\omega \in \mathscr{W}$. Since ω is generated by s_i , one can write ω as a product of simple reflections. Let $r \in \mathbb{N}$ be minimum such that ω can be written as $\omega = s_{i_1}s_{i_2}\cdots s_{i_r}$. We call r the *length* of ω , and the notation $r = l(\omega)$ will often be used. It is well known that $l(\omega) \leq N$. Furthermore, there exists a unique element $\omega_0 \in \mathscr{W}$ such that $l(w_0) = N$. The element ω_0 is called the *longest element* of \mathscr{W} . One can recover the positive roots of Φ from any reduced decomposition of ω_0 . That is, if $\omega_0 = s_{i_1}s_{i_2}\cdots s_{i_N}$, then the positive roots of Φ are exactly β_1, \cdots, β_N defined by

$$\beta_r := s_{i_1} s_{i_2} \cdots s_{i_{r-1}}(\alpha_{i_r}) \quad (1 \le r \le N).$$

For example, given the set of simple roots $\{\alpha_1, \alpha_2\}$ of G_2 above, the associated set of simple reflections is $\{s_1, s_2\}$. One can easily confirm that $\eta_{12}\eta_{21} = 3$, hence the order of s_1s_2 in the Weyl group \mathscr{W} of G_2 is 6. Therefore, $\mathscr{W} = \langle s_1, s_2 | s_1^2 = s_2^2 = (s_1s_2)^6 = 1 \rangle$. Moreover, $|\mathscr{W}| = |\Phi| = 12$. Given that $(s_1s_2)^6 = 1$ and $s_1^2 = s_2^2 = 1$, it follows that $s_1s_2s_1s_2s_1s_2s_1s_2s_1s_2s_1s_2 = 1$. This implies that $s_1s_2s_1s_2s_1s_2 = s_2s_1s_2s_1s_2s_1$. This expression gives the longest and unique element ω_0 of \mathscr{W} . Thus, $\omega_0 = s_1s_2s_1s_2s_1s_2s_1s_2 = s_2s_1s_2s_1s_2s_1s_2$. Then, given the relation $s_i(\alpha_j) = \alpha_j - \frac{2(\alpha_i, \alpha_j)}{(\alpha_i, \alpha_i)}\alpha_i$, one can equally confirm that the positive roots of G_2 (see Figure 1.1) are as follows:

$$\beta_{1} = \alpha_{1} \qquad \qquad \beta_{4} = s_{1}s_{2}s_{1}(\alpha_{2}) = 2\alpha_{2} + 3\alpha_{1}$$

$$\beta_{2} = s_{1}(\alpha_{2}) = \alpha_{2} + 3\alpha_{1} \qquad \qquad \beta_{5} = s_{1}s_{2}s_{1}s_{2}(\alpha_{1}) = \alpha_{2} + \alpha_{1}$$

$$\beta_{3} = s_{1}s_{2}(\alpha_{1}) = \alpha_{2} + 2\alpha_{1} \qquad \qquad \beta_{6} = s_{1}s_{2}s_{1}s_{2}s_{1}(\alpha_{2}) = \alpha_{2}.$$

1.2 Quantized enveloping algebras

In this section, we present the defining relations and a class of automorphisms (called the *braid group* introduced by Lusztig [38]) of the quantized enveloping algebra $U_q(\mathfrak{g})$ of a finite dimensional complex simple Lie algebra \mathfrak{g} . Unless otherwise stated, we follow the presentations and conventions in [7, Chapter I.6]. For further details of the material presented here, [7, Chapter I.6] and [26, Chapter 8] will be helpful. Throughout this section, we assume that q is not a root of unity.

1.2.1 Definition. Let v be an indeterminate and $m, t \in \mathbb{N}$; we have the following definitions:

- (i). $[t]_v := \frac{v^t v^{-t}}{v v^{-1}} = v^{t-1} + v^{t-3} + \dots + v^{1-t}.$
- (ii). $[m]_v! := [m]_v[m-1]_v \cdots [1]_v$. By convention, $[0]_v! := 1$.
- (iii). for all integers $0 \le i \le m$, we have that:

$$\begin{bmatrix} m \\ i \end{bmatrix}_{v} := \frac{[m]_{v}!}{[i]_{v}![m-i]_{v}!} = \frac{(v^{m}-v^{-m})(v^{m-1}-v^{1-m})\cdots(v-v^{-1})}{(v^{i}-v^{-i})\cdots(v-v^{-1})(v^{m-i}-v^{i-m})\cdots(v-v^{-1})}$$

Now; $[t]_v$, $[m]_v!$ and $\begin{bmatrix} m \\ i \end{bmatrix}_v$ are all elements of the ring $\mathbb{Z}[v, v^{-1}]$. Particularly, $\begin{bmatrix} m \\ i \end{bmatrix}_v$ are called the *v*-binomial coefficients, and they are the *v*-analogue of the usual binomial coefficients. They can be evaluated at any given values of v. If v = q, with $q \in \mathbb{K}^*$, then we have the usual *q*-binomial coefficients.

Let \mathfrak{g} denote a finite dimensional complex simple Lie algebra of rank n. Recall from (1.1.1) that $\eta_{ij} = 2(\alpha_i, \alpha_j)/(\alpha_i, \alpha_i)$. The square matrix \mathcal{C} defined by $\mathcal{C} := (\eta_{ij})_{1 \le i,j \le n} \in M_n(\mathbb{Z})$ is called the *Cartan matrix* of \mathfrak{g} . For example, one can verify that the Cartan matrix of G_2 has the following entries: $\eta_{11} = \eta_{22} = 2$, $\eta_{12} = -3$, and $\eta_{21} = -1$. In fact, the diagonal entries of every Cartan matrix are 2 and the off-diagonal entries are non-positive integers.

Let $\Pi = \{\alpha_1, \cdots, \alpha_n\}$ represent the set of simple roots of g. For all $i \in \{1, \cdots, n\}$, set

$$q_i := q^{\frac{(\alpha_i, \alpha_i)}{2}}.$$

The quantized enveloping algebra $U_q(\mathfrak{g})$ of \mathfrak{g} is a \mathbb{K} -algebra generated by E_1, \dots, E_n , F_1, \dots, F_n and $K_1^{\pm 1}, \dots, K_n^{\pm 1}$ subject to the following relations:

$$K_i E_j = q_i^{\eta_{ij}} E_j K_i \qquad \qquad K_i F_j = q_i^{-\eta_{ij}} F_j K_i$$
$$E_i F_j = F_j E_i + \delta_{ij} \frac{K_i - K_i^{-1}}{q_i - q_i^{-1}} \qquad \qquad K_i K_j = K_j K_i$$

$$(\text{QSR1}) \qquad \sum_{l=0}^{1-\eta_{ij}} (-1)^l \begin{bmatrix} 1-\eta_{ij} \\ l \end{bmatrix}_{q_i} F_i^{1-\eta_{ij}-l} F_j F_i^l = 0 \quad i \neq j$$

$$(\text{QSR2}) \qquad \sum_{l=0}^{1-\eta_{ij}} (-1)^l \begin{bmatrix} 1-\eta_{ij} \\ l \end{bmatrix}_{q_i} E_i^{1-\eta_{ij}-l} E_j E_i^l = 0 \quad i \neq j.$$

The relations QSR1 and QSR2 are called the quantum Serre relations.

Recall that given the Lie algebra \mathfrak{g} , there is a triangular decomposition for \mathfrak{g} as $\mathfrak{g}^+ \oplus \mathfrak{h} \oplus \mathfrak{g}^-$. The associated decomposition for the enveloping algebra $U(\mathfrak{g})$ is of the form $U(\mathfrak{g}^+) \otimes U(\mathfrak{h}) \otimes U(\mathfrak{g}^-)$, where $U(\mathfrak{g}^+), U(\mathfrak{g}^-)$ and $U(\mathfrak{h})$ are called the positive nilpotent, negative nilpotent and Cartan subalgebras of $U(\mathfrak{g})$ respectively. Similarly, $U_q(\mathfrak{g})$ admits a triangular decomposition of the form $U_q^+(\mathfrak{g}) \otimes U^0(\mathfrak{h}) \otimes U_q^-(\mathfrak{g})$. One can understand $U_q^+(\mathfrak{g})$ and $U_q^-(\mathfrak{g})$ as the q-deformations of the subalgebras $U(\mathfrak{g}^+)$ and $U(\mathfrak{g}^-)$ respectively. The sets $\{E_1, \dots, E_n\}$, $\{F_1, \dots, F_n\}$ and $\{K_1^{\pm 1}, \dots, K_n^{\pm 1}\}$ generate the subalgebras $U_q^+(\mathfrak{g}), U_q^-(\mathfrak{g})$ and $U^0(\mathfrak{h})$ respectively. Furthermore, $U_q^-(\mathfrak{g})$ and $U_q^+(\mathfrak{g})$ satisfy the relations QSR1 and QSR2 respectively. The subalgebra $U^0(\mathfrak{h})$ is commutative. There is a unique automorphism of $U_q(\mathfrak{g})$ that maps E_i to F_i , F_i to E_i and K_i to K_i^{-1} (see [26, Lemma 4.6] or [7, Lemma 1.6.4]). Finally, $U_q^+(\mathfrak{g})$ and $U_q^-(\mathfrak{g})$ are isomorphic (for example, see [45, §2]).

We now present a subgroup of the automorphism group of $U_q(\mathfrak{g})$ called the *braid* group. The braid group was introduced by Lusztig [38].

Braid group. Recall that $\Pi = \{\alpha_1, \cdots, \alpha_n\}$ is the set of simple roots of g. Set $T_i := T_{\alpha_i}$, where $1 \le i \le n$. Given the Weyl group

$$\mathscr{W} = \langle s_1, \cdots, s_n \mid s_i^2 = (s_i s_j)^{m_{ij}} = 1 \Rightarrow s_i s_j s_i \cdots = s_j s_i s_j \cdots, \quad i \neq j \rangle$$

the associated braid group is given by

$$\mathscr{B}_{\mathscr{W}} = \langle T_1, \cdots, T_n \mid T_i T_j T_i \cdots = T_j T_i T_j \cdots, i \neq j \rangle.$$

Note, each side of the equality $T_iT_jT_i\cdots = T_jT_iT_j\cdots$ has exactly m_{ij} factors. The map $\varphi: \mathscr{B}_{\mathscr{W}} \longrightarrow \mathscr{W}$ defined by $\varphi(T_i) = s_i$ is an epimorphism. Note, $s_i^2 = 1$ in \mathscr{W} , however,

 T_i^2 is not necessarily 1 in $\mathscr{B}_{\mathscr{W}}$. In line with [26, Chapter 8], Lusztig defines an action of $\mathscr{B}_{\mathscr{W}}$ by automorphism on $U_q(\mathfrak{g})$ as follows:

$$T_i(E_i) = -F_i K_i \qquad T_i(E_j) = \sum_{\substack{r=0\\-\eta_{ij}}}^{-\eta_{ij}} (-1)^r q_i^{-r} E_i^{(-\eta_{ij}-r)} E_j E_i^{(r)} \qquad i \neq j$$

$$T_i(F_i) = -K_i^{-1}E_i \qquad T_i(F_j) = \sum_{r=0}^{\infty} (-1)^r q_i^r F_i^{(r)} F_j F_i^{(-\eta_{ij}-r)} \qquad i \neq j$$

$$T_i(K_\alpha) = K_{s_i(\alpha)}, \ \alpha \in Q,$$

where $E_i^{(r)} := \frac{E_i^r}{[r]_{q_i}!}$, $F_i^{(r)} := \frac{F_i^r}{[r]_{q_i}!}$ and Q (the root lattice) denotes the \mathbb{Z} - span of Π . Hence, $\alpha = z_1 \alpha_1 + \dots + z_n \alpha_n \in Q$, with $z_1, \dots, z_n \in \mathbb{Z}$, and $K_\alpha := K_1^{z_1} \dots K_n^{z_n}$. Note, one can also refer to [26, Chapter 8] for the actions of the inverse automorphisms T_i^{-1} on $U_q(\mathfrak{g})$.

From any reduced decomposition of ω_0 (discussed in Section 1.1), one can construct distinguished elements $E_{\beta_1}, \dots, E_{\beta_N}$ of $U_q(\mathfrak{g})$ as follows:

$$E_{\beta_r} = T_{i_1} \cdots T_{i_{r-1}}(E_{i_r}), \quad 1 \le r \le N.$$
(1.2.1)

The elements E_{β_r} depend on the reduced decomposition for ω_0 [32, §1.2]. Levendorskii and Soibelman [37] proved the result below (also, see [32, Theorem 1.1]).

1.2.2 Theorem. (1) The element E_{β_r} belongs to $U_q^+(\mathfrak{g})$ for all $r \in \{1, \cdots, N\}$.

- (2) If $\beta_r = \alpha_i$, then $E_{\beta_r} = E_i$.
- (3) The family of monomials $(E_{\beta_1}^{r_1}\cdots E_{\beta_N}^{r_N})_{r_1,\cdots,r_N\in\mathbb{N}}$ is a linear basis of $U_q^+(\mathfrak{g})$.
- (4) For all $1 \le i < j \le N$, we have:

$$E_{\beta_j} E_{\beta_i} - q^{-(\beta_i,\beta_j)} E_{\beta_i} E_{\beta_j} = \sum a_{r_{i+1},\cdots,r_{j-1}} E_{\beta_{i+1}}^{r_{i+1}} \cdots E_{\beta_{j-1}}^{r_{j-1}},$$

with $a_{r_{i+1},\cdots,r_{i-1}} \in \mathbb{K}$.

Technically, the basis in point (3) is called a PBW-basis of $U_q^+(\mathfrak{g})$, where PBW stands for Poincaré-Birkhoff-Witt. Also, as a corollary to this theorem, we have that the family of monomials $(F_{\beta_1}^{s_1}\cdots F_{\beta_N}^{s_N}K_{\alpha}E_{\beta_1}^{r_1}\cdots E_{\beta_N}^{r_N})_{s_i,r_i\in\mathbb{N},\alpha\in Q}$ form a PBW-basis of $U_q(\mathfrak{g})$ [7, §1.6.8].

1.2.3 Remark. Recall the definitions of ω and ω_0 in Section 1.1. It follows from [7, §I.6.8] that $U_q^+(\omega) := \mathbb{K} \langle E_{\beta_1}, \cdots, E_{\beta_r} \rangle$ is a subalgebra of $U_q^+(\mathfrak{g})$ for each $1 \leq r \leq N$, and $U_q^+(\omega_0) = U_q^+(\mathfrak{g})$.

1.3 Ore extension

Let A be a \mathbb{K} -algebra, σ be an automorphism of A and δ be a \mathbb{K} -linear map from A to A. Then,

- δ is a derivation if $\delta(xy) = x\delta(y) + \delta(x)y, \ \forall x, y \in A.$
- δ is a σ -derivation if $\delta(xy) = \sigma(x)\delta(y) + \delta(x)y, \ \forall x, y \in A.$

Observe that when $\sigma = id$, the σ -derivation and derivation coincide. One can easily verify that $\delta(1) = 0$. From [22, Chapter 2], we have the presentations in the following definition.

1.3.1 Definition. Let σ and δ denote an automorphism and a σ -derivation of a ring A respectively. Denote $R := A[x; \sigma, \delta]$ to mean the following:

- *R* is a ring containing the element *x*,
- A is a subring of R,
- R is a free left A-module with basis $(x^i)_{i \in \mathbb{N}}$,
- $xa = \sigma(a)x + \delta(a)$ for all $a \in A$.

The ring R is called an Ore extension of A or a skew polynomial ring over A. When $\sigma = id$, then $A[x; id, \delta]$ is simply written as $A[x; \delta]$ and when $\delta = 0$, then $A[x; \sigma, 0]$ is

simply written as $A[x;\sigma]$. The Ore extension described above can easily be iterated to obtain an *iterated Ore extension* of A, written as:

$$R = A[x_1; \sigma_1, \delta_1][x_2; \sigma_2, \delta_2] \cdots [x_n; \sigma_n, \delta_n],$$

where σ_i and δ_i are the automorphisms and σ_i -derivations of

 $A[x_1; \sigma_1, \delta_1][x_2; \sigma_2, \delta_2] \cdots [x_{i-1}; \sigma_{i-1}, \delta_{i-1}] \text{ respectively for all } i \in \{1, \cdots, n\}.$ Note, $A[x_0; \sigma_0, \delta_0] := A.$

- **1.3.2 Example.** (1) The quantum plane $\mathbb{K}_q[x, y]$, with xy = qyx, can be presented as an iterated Ore extension of the form $\mathbb{K}[y][x, \sigma]$, where σ is an automorphism of the polynomial ring $\mathbb{K}[y]$ defined by $\sigma(y) = qy$.
 - (2) The quantized enveloping algebra $U_q(\mathfrak{sl}_2)$ is generated by E, F and $K^{\pm 1}$; and satisfies the following relations:

$$KE = q^2 EK,$$
 $FK = q^2 KF,$
 $KK^{-1} = K^{-1}K = 1,$ $FE = EF + (K^{-1} - K)/(q^{-1} - q).$

An iterated Ore extension of $U_q(\mathfrak{sl}_2)$ can be written as:

$$U_q(\mathfrak{sl}_2) = \mathbb{K}[E][K^{\pm 1}; \sigma_1][F; \sigma_2.\delta_2],$$

where $\sigma_1(E) = q^2 E$, $\sigma_2(K) = q^2 K$, $\sigma_2(E) = E$, $\delta_2(K) = 0$ and $\delta_2(E) = (K^{-1} - K)/(q^{-1} - q)$. Note, σ_1 and σ_2 are automorphisms of the subalgebras $\mathbb{K}[E]$ and $\mathbb{K}[E][K^{\pm 1}; \sigma_1]$ respectively, and δ_2 is a σ_2 -derivation of $\mathbb{K}[E][K^{\pm 1}; \sigma_1]$.

One can refer to [22, Chapter 2] for more details on Ore and iterated Ore extensions, but we recall Hibert Basis Theorem below.

1.3.3 Theorem. [7, Lemma I.1.12 & Theorem I.1.13]. Let $R := A[x; \sigma, \delta]$ be a skew polynomial ring, where σ and δ are automorphism and σ -derivation of the ring A respectively.

- If A is left (right) noetherian then R is also left (right) noetherian [Hibert Basis Theorem].
- 2. If A is a domain then R is also a domain.
- **1.3.4 Remark.** $U_q^+(\mathfrak{g})$ can be presented as an iterated Ore extension of the form:

$$U_q^+(\mathfrak{g}) = \mathbb{K}[E_{\beta_1}][E_{\beta_2}; \sigma_2, \delta_2] \cdots [E_{\beta_N}; \sigma_N, \delta_N]$$

(see for instance [7, Section I.6.10]), where σ_i is an automorphism of

 $\mathbb{K}[E_{\beta_1}][E_{\beta_2}; \sigma_2, \delta_2] \cdots [E_{\beta_{i-1}}; \sigma_{i-1}, \delta_{i-1}]$ defined as $\sigma_i(E_{\beta_j}) = q^{-(\beta_j, \beta_i)}E_{\beta_j}$ and δ_i is a σ_i -derivation defined as $\delta_i(E_{\beta_j}) = E_{\beta_i}E_{\beta_j} - q^{-(\beta_j, \beta_i)}E_{\beta_j}E_{\beta_i}$ for all $1 \leq j < i \leq N$. As a result, $U_q^+(\mathfrak{g})$ is a noetherian domain since \mathbb{K} is a noetherian domain. From [32, §1.2], the group of invertible elements of $U_q^+(\mathfrak{g})$ is reduced to \mathbb{K}^* .

1.4 Localization and rings of fractions

The materials presented in this section are well known, nonetheless, we follow the presentations in [22, Chapters 6 and 10]. As a result, further details can be read from this reference.

Let S be a subset of a \mathbb{K} -algebra A such that $1 \in S$, then S is said to be a *multiplicative set* if it is closed under multiplication. The collection of all non-zero divisors (also known as regular elements) in A forms a multiplicative set. Given a multiplicative set S of A, if for each $s \in S$ and $a \in A$, there exists $t \in S$ and $b \in A$ such that sb = at (or $sA \cap aS \neq \emptyset$), then S is said to satisfy the *right Ore condition*. A multiplicative set S which satisfies the right Ore condition is called *right Ore set*. Symmetrically, one can also define the *left Ore condition* and *left Ore set*. A multiplicative set which satisfies both the right and left Ore condition is called an Ore set. In a commutative ring, every multiplicative set is an Ore set. A multiplicative set S is said to be *right reversible* if for $a \in A$ and $s \in S$ with sa = 0 then there exists $s' \in S$ such that as' = 0. A right

reversible right Ore set is called a *right denominator set*. A left denominator set is defined symmetrically. A *denominator set* is any right and left denominator set. Suppose that Sis a right Ore set. Then, the set $AS^{-1} := \{as^{-1} \mid s \in S \text{ and } a \in A\}$ is called the *right ring of fractions* or *right Ore localization*. Observe that every element of S is a unit in AS^{-1} and A is a subset of AS^{-1} . The *left ring of fractions* or *left Ore localization* is defined symmetrically.

- **1.4.1 Example.** 1. If A is a noetherian domain, then $S := A \setminus \{0\}$ satisfies the Ore condition, hence, $AS^{-1} = Fract(A)$ is a right ring of fractions of A.
 - 2. If A is a domain and $x \in A$ is normal (i.e. xA = Ax), then $S = \{\lambda x^i \mid \lambda \in \mathbb{K}^*, i \in \mathbb{N}\}$ is an Ore set. As a result, one can define $A[x^{\pm 1}, \sigma]$ as $A[x, \sigma]$ localized at S. That is, $A[x^{\pm 1}, \sigma] := A[x, \sigma]S^{-1}$.

1.5 Quantum affine space and quantum torus

Let $N \in \mathbb{N}$. Given a multiplicatively skew-symmetric matrix $\Lambda = (a_{ij}) \in M_N(\mathbb{K}^*)$, one can define a \mathbb{K} -algebra $\mathcal{O}_{\Lambda}(\mathbb{K}^N)$ associated to Λ as follows:

$$\mathcal{O}_{\Lambda}(\mathbb{K}^N) = \mathbb{K}_{\Lambda}[Y_1, \cdots, Y_N],$$

where $Y_jY_i = a_{ji}Y_iY_j$ for all $1 \le i, j \le N$. The algebra $\mathcal{O}_{\Lambda}(\mathbb{K}^N)$ is called a *quantum* affine space of rank N associated to Λ . The iterated Ore extension of $\mathcal{O}_{\Lambda}(\mathbb{K}^N)$ can be written as:

$$\mathcal{O}_{\Lambda}(\mathbb{K}^N) = \mathbb{K}[Y_1][Y_2; \sigma_2] \cdots [Y_N; \sigma_N],$$

where σ_j is an automorphism of $\mathbb{K}[Y_1][Y_2; \sigma_2] \cdots [Y_{j-1}; \sigma_{j-1}]$ defined by $\sigma_j(Y_i) = a_{ji}Y_i$ for all $1 \le i < j \le N$.

Since Y_1, \dots, Y_N are all normal elements of $\mathcal{O}_{\Lambda}(\mathbb{K}^N)$, the set $S := \{\lambda Y_1^{k_1} \dots Y_N^{k_N} \mid \lambda \in \mathbb{K}^* \text{ and } k_1, \dots, k_N \in \mathbb{N}\}$ is an Ore set of $\mathcal{O}_{\Lambda}(\mathbb{K}^N)$. One can therefore localize

 $\mathcal{O}_{\Lambda}(\mathbb{K}^N)$ at S as follows:

$$\mathcal{O}_{\Lambda}(\mathbb{K}^*)^N := \mathcal{O}_{\Lambda}(\mathbb{K}^N)S^{-1} = \mathbb{K}_{\Lambda}[Y_1^{\pm 1}, \cdots, Y_N^{\pm 1}]$$

Now, the localization $\mathcal{O}_{\Lambda}(\mathbb{K}^*)^N$ is called a *quantum torus* of rank N. Furthermore, $\mathcal{O}_{\Lambda}((\mathbb{K}^*)^N)$ can be written as an iterated Ore extension of the form:

$$\mathcal{O}_{\Lambda}((\mathbb{K}^*)^N) = \mathbb{K}[Y_1^{\pm 1}][Y_2^{\pm 1};\sigma_2]\cdots[Y_N^{\pm 1};\sigma_N],$$

where σ_j is an automorphism of $\mathbb{K}[Y_1^{\pm 1}][Y_2^{\pm 1}; \sigma_2] \cdots [Y_{j-1}^{\pm 1}; \sigma_{j-1}]$ defined by $\sigma_j(Y_i) = a_{ji}Y_i$ for all $1 \le i < j \le N$.

A basis of $\mathcal{O}_{\Lambda}((\mathbb{K}^*)^N)$ is given by the family $(Y_1^{k_1}\cdots Y_N^{k_N})_{k_1,\cdots,k_N\in\mathbb{Z}}$. If we restrict the powers k_1,\cdots,k_N to only \mathbb{N} , then $(Y_1^{k_1}\cdots Y_N^{k_N})$ becomes a basis of $\mathcal{O}_{\Lambda}(\mathbb{K}^N)$.

1.5.1 Remark. Given the finite dimensional complex simple Lie algebra \mathfrak{g} , recall from Section 1.1 that one can construct a set $\{\beta_1, \dots, \beta_N\}$ of positive roots of \mathfrak{g} . Set

$$\mu_{ij} := \begin{cases} (\beta_i, \beta_j) & \text{if } i < j \\\\ 0 & \text{if } i = j \\\\ -(\beta_j, \beta_i) & \text{if } i > j. \end{cases}$$
(1.5.1)

The matrix $\Lambda = (q^{\mu_{ij}}) \in M_N(\mathbb{K}^*)$ is a multiplicatively skew-symmetric matrix (i.e. $q^{\mu_{ij}}q^{\mu_{ji}} = q^{\mu_{ii}} = 1$ for all $1 \leq i, j \leq N$). As a result, one can define a quantum affine space or a quantum torus associated to Λ .

1.6 Prime spectrum

Let P be a proper ideal of a ring A and I_1, I_2 be ideals of A with $I_1I_2 \subseteq P$, the ideal P is called a *prime ideal* provided $P \supseteq I_1$ or $P \supseteq I_2$. A proper ideal P is said to be *completely prime* if $xy \in P$ implies that $x \in P$ or $y \in P$ for all $x, y \in A$. All completely prime ideals are prime ideals (see comments after [31, Theorem 12.5]), however, the converse is not always true. An ideal can be prime but not completely prime. For example, the zero ideal $\langle 0 \rangle$ of $M_2(\mathbb{Z})$ is prime but not completely prime.

Take

$$x = \begin{bmatrix} 2 & 2 \\ 2 & 2 \end{bmatrix} \text{ and } y = \begin{bmatrix} 2 & 2 \\ -2 & -2 \end{bmatrix}.$$

Clearly, $xy \in \langle 0 \rangle$ but $x \notin \langle 0 \rangle$ and $y \notin \langle 0 \rangle$. The collection of all the prime ideals of A is called the *prime spectrum* of A, denoted by Spec(A).

Let M be a right simple A-module. The set $\operatorname{ann}_A(M) := \{a \mid ma = 0, \forall m \in M\}$ is called the *right annihilator* of M. Now, $\operatorname{ann}_A(M)$ forms an ideal of A called the *right primitive ideal*. The left primitive ideal is defined symmetrically. Finally, an ideal of a ring A is *maximal* if it is not contained in any proper ideal of A. The set $\operatorname{Prim}(A)$ is the collection of all the (left/right) primitive ideals of A called the *primitive spectrum* of A, and the set $\operatorname{MaxSpec}(A)$ is the collection of all the maximal ideals of A called the *maximal spectrum* of A. We have the following set inclusions: $\operatorname{MaxSpec}(A) \subseteq \operatorname{Prim}(A) \subseteq \operatorname{Spec}(A)$ for any arbitrary ring A [22, Proposition 3.15].

Again, let A be a ring and $P_0, P_1, \dots, P_n \in \text{Spec}(A)$ such that $P_0 \subset P_1 \subset \dots \subset P_n$ is a chain of prime ideals of A. The number of strict inclusions n that we have in the chain is called the *length* of the chain. That is, n is the number of P_i 's in the chain minus 1. A chain is *saturated* if no prime ideal can be included in the chain. Furthermore, let $P \in \text{Spec}(A)$. The height of P denoted by ht(P) is the supremum of the lengths of all the chains of prime ideals contained in P. Finally, let P and Q be two distinct prime ideals of A such that $P \subset Q$. Then, A is said to be *catenary* if all saturated chains of prime ideals from P to Q have the same length for any fixed choice of P and Q. For example, $U_q^+(\mathfrak{g})$ is catenary [19, Theorem 4.8].

1.6.1 Remark. [42, Corollary of Theorem 3] If q is not a root of unity, then all prime ideals of $U_q^+(\mathfrak{g})$ are completely prime.

1.6.2 Rational torus action. In this section we define rational torus actions in line with the presentations in [7, Chapter II.2]. Let the group \mathcal{H} act on the \mathbb{K} -algebra A via \mathbb{K} -algebra automorphisms. Let x be a non-zero element of A. Then, x is said to be a \mathcal{H} -eigenvector if and only if $h \cdot x \in \mathbb{K}x$ for all $h \in \mathcal{H}$. If a torus \mathcal{H} acts on an algebra A such that the generators of A are simultaneous eigenvectors for all $h \in \mathcal{H}$, then the action of \mathcal{H} on A is said to be semisimple.

- **1.6.3 Example.** (1) The torus $\mathcal{H} = (\mathbb{K}^*)^2$ acts on the quantum plane $\mathbb{K}_q[x, y]$ by automorphisms as: $(\alpha, \beta) \cdot x^i y^j = \alpha^i \beta^j x^i y^j$ for all $(\alpha, \beta) \in \mathcal{H}$ and $i, j \ge 0$. This action is semisimple since each monomial $x^i y^j$ is an \mathcal{H} -eigenvector for all $i, j \ge 0$.
 - (2) In general, the torus $\mathcal{H} = (\mathbb{K}^*)^N$ acts on the quantum affine space $\mathbb{K}_{\Lambda}[Y_1, \cdots, Y_N]$ by automorphisms as follows:

$$(\alpha_1, \cdots, \alpha_N) \cdot Y_i = \alpha_i Y_i,$$

for all $i \in \mathbb{N}$ and $(\alpha_1, \dots, \alpha_N) \in \mathcal{H}$. This action extends uniquely to an action of the quantum torus, and it is semisimple.

Given an eigenvector, another terminology that obviously comes to mind is an eigenvalue. Therefore, given a \mathcal{H} -eigenvector x, the \mathcal{H} -eigenvalue of x is a group homomorphism $\lambda : \mathcal{H} \longrightarrow \mathbb{K}^*$ such that $h \cdot x = \lambda(h)x$ for all $h \in \mathcal{H}$. For instance, the group homomorphism $(\alpha, \beta) \longmapsto \alpha^i \beta^j$ is \mathcal{H} -eigenvalue of the \mathcal{H} -eigenvector $x^i y^j$ in Example 1.6.3. Given an \mathcal{H} -eigenvalue λ , the set $A_{\lambda} := \{x \in A \mid h \cdot x = \lambda(h)x, \forall h \in \mathcal{H}\}$ is called the \mathcal{H} -eigenspace of A provided it is non-zero.

The \mathcal{H} -eigenvalues λ are called *characters* of \mathcal{H} . If \mathcal{H} is an algebraic group over \mathbb{K} then a character λ of \mathcal{H} which is also a morphism of algebraic varieties is called a *rational character*. Suppose that \mathbb{K} is infinite, then it follows from [7, Theorem II.2.7] that an action of \mathcal{H} on A is *rational* if and only if it is a semisimple action and the corresponding \mathcal{H} -eigenvalues are all rational. One can easily verify that all the torus actions in Example 1.6.3 are rational actions. For the purpose of our studies, this discussion on rational torus

action is enough. However, one can read [7, Chapter II.2] for further details. Specifically, see [7, Definition II.2.6] for the general definition of rational torus action. We deduce the remark below from [4, §3.3].

1.6.4 Remark. The torus $\mathcal{H} = (\mathbb{K}^*)^n$ acts rationally on $U_q^+(\mathfrak{g})$ by automorphisms via:

$$(\alpha_1, \cdots, \alpha_n) \cdot E_i = \alpha_i E_i,$$

for all $1 \leq i \leq n$ and $(\alpha_1, \dots, \alpha_n) \in \mathcal{H}$. Recall; E_1, \dots, E_n are the standard generators of $U_q^+(\mathfrak{g})$, where n is the rank of \mathfrak{g} .

We now present the notion of \mathcal{H} -stratification by Goodearl and Letzter [21] which provides a partition of the prime spectrum of a given ring into a disjoint union of strata. More importantly, the \mathcal{H} -stratification also helps to study the primitive ideals of a ring. **1.6.5** \mathcal{H} -**Stratification.** Let A be a \mathbb{K} -algebra and I be any ideal of A. If $h \cdot I = I$ for all $h \in \mathcal{H}$, then I is said to be \mathcal{H} -invariant ideal of A. A prime \mathcal{H} -invariant ideal is called \mathcal{H} -invariant prime ideal (or \mathcal{H} -prime ideal for short). Also, $(I : \mathcal{H}) := \bigcap_{h \in \mathcal{H}} h \cdot I$ is the largest \mathcal{H} -invariant ideal contained in I. Let \mathcal{H} -Spec(A) represent the collection of all the \mathcal{H} -invariant prime ideals of A. For $J \in \mathcal{H}$ -Spec(A), the set Spec $_J(A) :=$ $\{P \in \text{Spec}(A) \mid (P : \mathcal{H}) = J\}$ is called the \mathcal{H} -stratum of Spec(A) associated to J. One can simply call it the J-stratum of Spec(A). The collection of all these J-strata forms a partition of Spec(A). That is,

$$\mathsf{Spec}(A) = \bigsqcup_{J \in \mathcal{H} - \mathsf{Spec}(A)} \mathsf{Spec}_J(A)$$

This partition is known as the \mathcal{H} -stratification of Spec(A). Note, given a non-zero algebra A, if the zero ideal and A are the only \mathcal{H} -invariant ideals of A, then A is said to be \mathcal{H} -simple.

For all J in $\mathcal{H}-\text{Spec}(A)$, we define $\text{Prim}_J(A) := \text{Spec}_J(A) \cap \text{Prim}(A)$. This gives a corresponding $\mathcal{H}-\text{stratification}$ of Prim(A). That is,

$$\mathsf{Prim}(A) = \bigsqcup_{J \in \mathcal{H} - \mathsf{Prim}(A)} \mathsf{Prim}_J(A),$$

where $\mathcal{H} - \mathsf{Prim}(A)$ is the collection of all \mathcal{H} -invariant primitive ideals of A.

This stratification theory applies to $U_q^+(\mathfrak{g}),$ and we have the following lemma.

1.6.6 Lemma. [7, Theorem II.8.4] Let $A = U_q^+(\mathfrak{g})$, the primitive ideals in $\text{Spec}_J(A)$ are just the maximal ideals in $\text{Spec}_J(A)$.

1.6.7 Remark. [24, §6,7] (also see [44, Theorem 3.8]) The poset of $\mathcal{H}-\text{Spec}(U_q^+(\mathfrak{g}))$ (ordered by inclusion) is isomorphic to the poset of the Weyl group (ordered by the Bruhat order) associated to \mathfrak{g} . That is, for any element of the Weyl group, one can always associate a unique element of $\mathcal{H}-\text{Spec}(U_q^+(\mathfrak{g}))$.

1.6.8 Example. Let $R = \mathbb{C}_q[x, y]$, with xy = qyx, and $\mathcal{H} = (\mathbb{C}^*)^2$. The ideals $\langle 0 \rangle$, $\langle x \rangle$, $\langle y \rangle$ and $\langle x, y \rangle$ are all \mathcal{H} -invariant prime ideals [7, Example II.2.3]. In fact, they are the only \mathcal{H} -invariant prime ideals we have in R. One can easily verify that $R/\langle x, y - \beta \rangle \cong \mathbb{C}$, where $\beta \in \mathbb{C}^*$. Since \mathbb{C} is a domain, $\langle x, y - \beta \rangle$ is a prime ideal. Similarly, $\langle x - \alpha, y \rangle$ with $\alpha \in \mathbb{C}^*$, is also a prime ideal. However, none of these two prime ideals is \mathcal{H} -invariant. From [7, Example II.2.3], we have the following strata in Spec(R) :

- Spec₍₀₎(R) = { $\langle 0 \rangle$ },
- $\operatorname{Spec}_{\langle x \rangle}(R) = \{ \langle x \rangle \} \cup \{ \langle x, y \beta \rangle \mid \beta \in \mathbb{C}^* \},\$
- Spec_{$\langle y \rangle$} $(R) = \{ \langle y \rangle \} \cup \{ \langle x \alpha, y \rangle \mid \alpha \in \mathbb{C}^* \}, \text{ and}$
- Spec $_{\langle x,y\rangle}(R) = \{\langle x,y\rangle\}.$

 $\mathsf{Consequently, } \mathsf{Spec}(R) = \mathsf{Spec}_{\langle 0 \rangle}(R) \cup \mathsf{Spec}_{\langle x \rangle}(R) \cup \mathsf{Spec}_{\langle y \rangle}(R) \cup \mathsf{Spec}_{\langle x, y \rangle}(R).$

For each $\alpha, \beta \in \mathbb{C}^*$, the poset of Spec(R) is shown in Figure 1.2. We also have that $\text{ht}(\langle 0 \rangle) = 0$, $\text{ht}(\langle x \rangle) = \text{ht}(\langle y \rangle) = 1$, and $\text{ht}(\langle x, y - \beta \rangle) = \text{ht}(\langle x - \alpha, y \rangle) = \text{ht}(\langle x, y \rangle) = 2$. Again; since $\langle 0 \rangle$, $\langle x, y - \beta \rangle$, $\langle x - \alpha, y \rangle$ and $\langle x, y \rangle$ are maximal in their respective strata for each $\alpha, \beta \in \mathbb{C}^*$; they are all primitive ideals. Although, $\langle 0 \rangle$ is maximal in its strata, it is obvious that it cannot be a maximal ideal in R, since it is strictly contained in other proper ideals of R. However; $\langle x, y - \beta \rangle$, $\langle x - \alpha, y \rangle$ and $\langle x, y \rangle$ are all maximal ideals in R.



Figure 1.2: Poset of Spec(R)

Following the notations in [4, §2.2], let W be the set of subsets of $I := \{1, \dots, N\}$. For each $w \in W$, set $K_w := \langle Y_i \mid i \in w \rangle$, where K_w is the two-sided ideal of the quantum affine space $\mathcal{O}_{\Lambda}(\mathbb{K}^N) = \mathbb{K}_{\Lambda}[Y_1, \dots, Y_N]$. From [20, Proposition 2.11] (also, see [4, Proposition 2.1]), we have the following proposition.

- **1.6.9 Proposition.** (1) The $\mathcal{H}-$ invariant prime ideals of $\mathcal{O}_{\Lambda}(\mathbb{K}^N)$ is the set $\{K_w \mid w \in W\}$. As a result, there are exactly $2^N \mathcal{H}-$ invariant prime ideals of $\mathcal{O}_{\Lambda}(\mathbb{K}^N)$.
- (2) $\operatorname{Spec}_{K_w}(\mathcal{O}_{\Lambda}(\mathbb{K}^N)) = \{P \in \operatorname{Spec}(\mathcal{O}_{\Lambda}(\mathbb{K}^N)) \mid P \cap \{Y_i \mid i \in I\} = \{Y_i \mid i \in w\}\}$ for all $w \in W$.

1.7 Deleting derivations algorithm (DDA)

In this section, we describe the notion of the deleting derivations algorithm (DDA) introduced by Cauchon [8]. We begin with the definition of quantum nilpotent algebras, which are also called Cauchon-Goodearl-Letzter (CGL) extensions. In line with [4, Definition 2], we provide a definition for CGL extension below.

1.7.1 Definition. Suppose that a ring A can be written as an iterated Ore extension as follows:

$$A = \mathbb{K}[X_1][X_2; \sigma_2, \delta_2] \cdots [X_N; \sigma_N, \delta_N],$$

with $N \in \mathbb{N}$. Then, A is said to be a quantum nilpotent algebra or a CGL extension if there exists a torus $\mathcal{H} = (\mathbb{K}^*)^m$ that acts rationally by \mathbb{K} -automorphism on A, and the following are satisfied:

- (a) X_1, \dots, X_N are \mathcal{H} -eigenvectors;
- (b) for all $2 \leq i \leq N$, there exists $h_i \in \mathcal{H}$ and $q_i \in \mathbb{K}^*$ (q_i is not a root of unity) such that $h_i \cdot X_i = q_i X_i$ and for all $1 \leq j < i$, there exists $\lambda_{ij} \in \mathbb{K}^*$ such that $h_i \cdot X_j = \lambda_{ij} X_j$;
- (c) the set $\{\lambda \in \mathbb{K}^* \mid \exists h \in \mathcal{H}, h \cdot X_1 = \lambda X_1\}$ is infinite;
- (d) for all $2 \le i \le N$, we have that δ_i is locally nilpotent, and
- (e) for all $1 \leq j < i \leq N$, there exists $\lambda_{ij} \in \mathbb{K}^*$ such that $\sigma_i(X_j) = \lambda_{ij}X_j$ (note, $\sigma_i = (h_i \cdot) \mid_{A_{i-1}}$, where $A_{i-1} := \mathbb{K}[X_1][X_2; \sigma_2, \delta_2] \cdots [X_{i-1}; \sigma_{i-1}, \delta_{i-1}]$).

Note, from the original definition, we have a condition that states that there exists $q_i \in \mathbb{K}^*$ (q_i is not a root of unity) such that $\sigma_i \circ \delta_i = q_i \delta_i \circ \sigma_i$. However, we did not include this condition in the above definition as it follows from the other conditions (see [23, Equation 3.1] for the necessary details). Suppose that A is a quantum nilpotent algebra, one can conclude from [21, Proposition 4.2] that all \mathcal{H} -invariant prime ideals of A are completely prime, and there are at most 2^N of these \mathcal{H} -invariant prime ideals.

Deleting derivations algorithm (DDA): let A be a quantum nilpotent algebra. One can use the theory of DDA constructed by Cauchon [8] (see also [4, §2.3]) to describe Spec(A). The algorithm relates Spec(A) to the prime spectrum of the quantum affine space obtained after sequentially 'deleting' all the skew-derivations $\delta_N, \dots, \delta_2$ from A. We now describe the DDA process below.

Let $j \in \{N + 1, \dots, 2\}$, the algorithm constructs a family $(X_{1,j}, \dots, X_{N,j})$ of elements of Fract(A) (the right ring of fractions of A) as follows. First, when j = N + 1, we set $(X_{1,N+1}, \dots, X_{N,N+1}) := (X_1, \dots, X_N)$. Second, for j < N + 1, suppose that the family $(X_{1,j+1}, \dots, X_{N,j+1})$ has already been constructed. Then, one can construct $X_{1,j}, \dots, X_{N,j}$ from $X_{1,j+1}, \dots, X_{N,j+1}$ using the relation below:

$$X_{i,j} := \begin{cases} X_{i,j+1} & \text{if } i \ge j \\ \sum_{k=0}^{+\infty} \frac{(1-q_j)^{-k}}{[k]_{q_j}!} \delta_j^k \circ \sigma_j^{-k} (X_{i,j+1}) X_{j,j+1}^{-k} & \text{if } i < j, \end{cases}$$
(1.7.1)

for all $i \in \{1, \dots, N\}$. Also, $[k]_{q_j}! = [0]_{q_j} \times \dots \times [k]_{q_j}$ with $[0]_{q_j} = 1$, and $[i]_{q_j} = 1 + q_j + \dots + q_j^{i-1}$ with $i \ge 1$. Note, from [8, Theorem 3.2.1], $X_{j,j+1} \ne 0$. Moreover, the summation is finite since δ_j are all locally nilpotent and $\delta_j \circ \sigma_j = q_j \sigma_j \circ \delta_j$.

For each $j \in \{2, \dots, N+1\}$, $A^{(j)}$ is a subalgebra of $\operatorname{Fract}(A)$ generated by $X_{1,j}, \dots, X_{N,j}$. That is, $A^{(j)} := \mathbb{K}\langle X_{1,j}, \dots, X_{N,j} \rangle$. Since $(X_{1,N+1}, \dots, X_{N,N+1}) = (X_1, \dots, X_N)$, we have that $A^{(N+1)} = A$. It follows from [8, Theorem 3.2.1] that

$$A^{(j)} \cong \mathbb{K}[X_1][X_2; \sigma_2, \delta_2] \cdots [X_{j-1}; \sigma_{j-1}, \delta_{j-1}][X_j; \tau_j] \cdots [X_N; \tau_N],$$

by an isomorphism that maps $X_{i,j}$ to X_i , and τ_j, \dots, τ_N are automorphisms defined by $\tau_l(X_i) = \lambda_{l,i}X_i$ for all $1 \le i < l \le N$. With a slight abuse of notation, one can identify τ_j, \dots, τ_N with $\sigma_j, \dots, \sigma_N$ respectively. With this isomorphism and the slight abuse of notation, we present $A^{(j)}$ as:

$$A^{(j)} = \mathbb{K}[X_{1,j}][X_{2,j}; \sigma_2, \delta_2] \cdots [X_{j-1,j}; \sigma_{j-1}, \delta_{j-1}][X_{j,j}; \sigma_j] \cdots [X_{N,j}; \sigma_N].$$

One can observe that for each $j \in \{2, \dots, N\}$, the derivations $\delta_j, \dots, \delta_N$ are all 'deleted' from $A^{(j)}$. For example, if j = 2, then $\delta_2, \dots, \delta_N$ will all be 'deleted' from $A^{(2)}$. As a result, $A^{(2)} = \mathbb{K}[T_1][T_2; \sigma_2] \cdots [T_N; \sigma_N]$, where $T_i := X_{i,2}$ for all $i \in \{1, \cdots, N\}$. Clearly, $\overline{A} := A^{(2)}$ is a quantum affine space. By Theorem 1.3.3, the algebra $A^{(j)}$ is a noetherian domain. Let $j \in \{2, \cdots, N\}$. The set $S_j := \{X_{j,j+1}^n \mid n \in \mathbb{N}\} = \{X_{j,j}^n \mid n \in \mathbb{N}\}$ is a multiplicative system of regular elements of $A^{(j)}$ and $A^{(j+1)}$ that satisfies the Ore condition in $A^{(j)}$ and $A^{(j+1)}$, and $A^{(j)}S_j^{-1} = A^{(j+1)}S_j^{-1}$ [8, Theorem 3.2.1].

Cauchon used DDA to relate Spec(A) to $\text{Spec}(\overline{A})$ by constructing an embedding $\psi_j : \text{Spec}(A^{(j+1)}) \hookrightarrow \text{Spec}(A^{(j)})$ for each $j \in \{2, \dots, N\}$. Suppose that $P \in \text{Spec}(A^{(j+1)})$ and $X_{j,j+1} \notin P$, then ψ_j is defined by

$$\psi_i(P) = PS_i^{-1} \cap A^{(j)} = Q,$$

with $Q \in \operatorname{Spec}(A^{(j)})$. The inverse map ψ_j^{-1} is also given by

$$\psi_j^{-1}(Q) = QS_j^{-1} \cap A^{(j+1)} = P,$$

provided $X_{j,j} \notin Q$.

The case where $X_{j,j+1} \in P$ is beyond the scope of this study, however, one can refer to [4, §2.3] and [8, §4.3] for the necessary details.

The map ψ_j is injective but not necessarily bijective. However, ψ_j induces a bijection from $\{P \in \text{Spec}(A^{(j+1)}) \mid P \cap S_j = \emptyset\}$ onto $\{Q \in \text{Spec}(A^{(j)}) \mid Q \cap S_j = \emptyset\}$. The so-called *canonical embedding* ψ : $\text{Spec}(A) \hookrightarrow \text{Spec}(\overline{A})$ is obtained by composing all the ψ_j . That is, $\psi := \psi_2 \circ \cdots \circ \psi_N$. This canonical embedding ψ helps to construct a partition of Spec(A) into a disjoint union of strata (known as the *canonical partition*) via the Cauchon diagrams (see [4, §2.3] for further details on this).

1.7.2 Remark. The algebra $U_q^+(\mathfrak{g})$ is a quantum nilpotent algebra [4, §3.1]. One can therefore apply the DDA to study the prime spectrum of $U_q^+(\mathfrak{g})$.
Chapter 2

Quest for height two maximal ideals of $U_q^+(G_2)$

The algebra $A = U_q^+(G_2)$ is a quantum nilpotent algebra, so one can apply the theory of deleting derivations algorithm (DDA) and \mathcal{H} -stratification theory to study its prime ideals. Note, from Remark 1.6.1, all prime ideals of A are completely prime. Since Ais a quantum nilpotent algebra, it follows from [18, Theorem 7.1] that Tauvel's height formula holds for A. That is, for any prime ideal P of A, we have that

$$\mathsf{GKdim}(A) = \mathsf{ht}(P) + \mathsf{GKdim}(A/P),$$

where GKdim(*) represents the Gelfand-Kirillov dimension of the algebra *, and ht(P) is the height of the prime ideal P. For more details on the Gelfand-Kirillov dimension of a ring, we recommend [30]. Since G_2 has 6 positive roots (Section 1.1), it follows from [45, §2.1] that GKdim(A) = 6. We are now interested in finding a family of prime ideals such that the quotients of A with those prime ideals are of GKdim 4 and simple. This implies that these prime ideals must be maximal, and using Tauvel's height formula, they must be of height 2. To achieve this result, we use \mathcal{H} -stratification theory to study some \mathcal{H} -strata of A. Note, the maximal ideals of height 2 can only belong to the \mathcal{H} -strata corresponding to an \mathcal{H} -invariant prime ideals of height at most 2, so we

identify these \mathcal{H} -prime ideals first. Of course, the only height 0 prime ideal is the zero ideal $\langle 0 \rangle$. In Section 2.3, we proved that the ideals $\langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$ are (completely) prime using Cauchon's theory of DDA, and in Lemma 2.4.2, we proved that they are of height 1. Moreover, studying the poset of \mathcal{H} -Spec(A) (as seen in Figure 2.1), no height 2 \mathcal{H} -prime ideal is maximal. As a result, the height 2 maximal ideals of A can only come from the \mathcal{H} -strata corresponding to $\langle 0 \rangle$, $\langle \Omega_1 \rangle$ or $\langle \Omega_2 \rangle$. In Subsection 2.4.3, we compute explicitly these \mathcal{H} -strata. This allows us to study the height 2 maximal ideals of A. We now begin by studying some properties of the algebra $U_q^+(G_2)$.

2.1 The algebra $U_a^+(G_2)$

The background of the materials presented here can be found in Section 1.2. Moreover, the positive part $U_q^+(G_2)$ of $U_q(G_2)$ is going to be of a major interest in this section and beyond. We will begin by finding the defining relations of $U_q(G_2)$, and then proceed to compute the defining relations of the subalgebra $U_q^+(G_2)$.

The algebra $U_q(G_2)$ is a \mathbb{C} -algebra generated by $F_1, F_2, K_1^{\pm 1}, K_2^{\pm 1}, E_1, E_2$ subject to the following relations:

$$K_{1}E_{1} = q^{2}E_{1}K_{1} \qquad K_{1}F_{1} = q^{-2}F_{1}K_{1} \qquad K_{1}K_{2} = K_{2}K_{1}$$

$$K_{1}E_{2} = q^{-3}E_{2}K_{1} \qquad K_{1}F_{2} = q^{3}F_{2}K_{1} \qquad E_{2}F_{2} = F_{2}E_{2} + \frac{K_{2} - K_{2}^{-1}}{q^{3} - q^{-3}}$$

$$K_{2}E_{1} = q^{-3}E_{1}K_{2} \qquad K_{2}F_{1} = q^{3}F_{1}K_{2} \qquad E_{1}F_{1} = F_{1}E_{1} + \frac{K_{1} - K_{1}^{-1}}{q - q^{-1}}$$

$$K_{2}E_{2} = q^{6}E_{2}K_{2} \qquad K_{2}F_{2} = q^{-6}F_{2}K_{2} \qquad E_{1}F_{2} = F_{2}E_{1}$$

 $E_2F_1 = F_1E_2,$

and the quantum Serre relations:

$$(S1) \quad E_1^4 E_2 - \begin{bmatrix} 4 \\ 1 \end{bmatrix}_q E_1^3 E_2 E_1 + \begin{bmatrix} 4 \\ 2 \end{bmatrix}_q E_1^2 E_2 E_1^2 - \begin{bmatrix} 4 \\ 1 \end{bmatrix}_q E_1 E_2 E_1^3 + E_2 E_1^4 = 0,$$

(S2)
$$E_2^2 E_1 - \begin{bmatrix} 2 \\ 1 \end{bmatrix}_{q^3} E_2 E_1 E_2 + E_1 E_2^2 = 0,$$

$$(S3) \quad F_1^4 F_2 - \begin{bmatrix} 4 \\ 1 \end{bmatrix}_q F_1^3 F_2 F_1 + \begin{bmatrix} 4 \\ 2 \end{bmatrix}_q F_1^2 F_2 F_1^2 - \begin{bmatrix} 4 \\ 1 \end{bmatrix}_q F_1 F_2 F_1^3 + F_2 F_1^4 = 0.$$

(S4)
$$F_2^2 F_1 - \begin{bmatrix} 2 \\ 1 \end{bmatrix}_{q^3} F_2 F_1 F_2 + F_1 F_2^2 = 0.$$

Now, $U_q^+(G_2)$ is a subalgebra of $U_q(G_2)$ generated by E_1 and E_2 subject to the quantum Serre relations (S1) and (S2). The actions of the automorphisms T_1 and T_2 of $U_q(G_2)$ on $U_q^+(G_2)$ are as follows:

$$T_{2}(E_{1}) = E_{2}E_{1} - q^{-3}E_{1}E_{2} \qquad T_{1}(E_{1}) = -F_{1}K_{1}$$
$$T_{1}(E_{2}) = \frac{E_{1}^{3}E_{2}}{[3]_{q}[2]_{q}} - q^{-1}\frac{E_{1}^{2}E_{2}E_{1}}{[2]_{q}} + q^{-2}\frac{E_{1}E_{2}E_{1}^{2}}{[2]_{q}} - q^{-3}\frac{E_{2}E_{1}^{3}}{[3]_{q}[2]_{q}} \qquad T_{2}(E_{2}) = -F_{2}K_{2}.$$

For the purpose of the computation of the relations of $U_q^+(G_2)$ in Appendix A.1, we include the following inverse automorphism actions: $T_1^{-1}(E_1) = -K_1^{-1}F_1$ and $T_2^{-1}(E_2) = -K_2^{-1}F_2$.

Now, recall the positive roots β_1, \dots, β_6 of G_2 in Section 1.1. From (1.2.1) and Theorem 1.2.2, the elements E_{β_r} (also called root vectors) of $U_q^+(G_2)$ are as follows:

$$E_{\beta_1} = E_1 \qquad E_{\beta_2} = T_1(E_2) \qquad E_{\beta_3} = T_1T_2(E_1)$$
$$E_{\beta_4} = T_1T_2T_1(E_2) \qquad E_{\beta_5} = T_1T_2T_1T_2(E_1) \qquad E_{\beta_6} = E_2.$$

With a slight abuse of notations, set $E_i := E_{\beta_i}$ for all $1 \le i \le 6$. Note, E_2 is no longer E_{β_6} as expected. With these notations, the defining relations of $U_q^+(G_2)$ (see Appendix A.1.1) are as follows:

$$E_{2}E_{1} = q^{-3}E_{1}E_{2}$$

$$E_{3}E_{2} = q^{-3}E_{2}E_{3}$$

$$E_{4}E_{2} = q^{-3}E_{2}E_{4} - \frac{q^{4} - 2q^{2} + 1}{q^{4} + q^{2} + 1}E_{3}^{3}$$

$$E_{4}E_{3} = q^{-3}E_{3}E_{4}$$

$$E_{5}E_{1} = qE_{1}E_{5} - (1 + q^{2})E_{3}$$

$$E_{5}E_{3} = q^{-1}E_{3}E_{5} - (q + q^{-1} + q^{-3})E_{4}$$

$$E_{5}E_{4} = q^{-3}E_{4}E_{5}$$

$$E_{6}E_{1} = q^{3}E_{1}E_{6} - q^{3}E_{5} \qquad E_{6}E_{2} = q^{3}E_{2}E_{6} + (q^{4} + q^{2} - 1)E_{4} + E_{6}E_{3} = E_{3}E_{6} + (1 - q^{2})E_{5}^{2} \qquad (q^{2} - q^{4})E_{3}E_{5} + (q^{4} - q^{2})E_{6}^{2} + (q$$

These relations have been confirmed with a mathematical software called GAP [10]. The GAP code can be found in Appendix A.1.2.

2.1.1 Remark. De Graaf [9, Appendix A] has also computed the relations of $U_q^+(G_2)$. If we set $E_1 := E_{\alpha}$, $E_2 := E_{3\alpha+\beta}$, $E_3 := E_{2\alpha+\beta}$, $E_4 := E_{3\alpha+2\beta}$, $E_5 := E_{\alpha+\beta}$ and $E_6 := E_{\beta}$, then our relations are the same as his relations. Of course, the only difference is that he used the notations α and β for the simple roots contrary to our notations α_1 and α_2 .

From Theorem 1.2.2, the set $\{E_1^{k_1}E_2^{k_2}E_3^{k_3}E_4^{k_4}E_5^{k_5}E_6^{k_6} \mid k_1, \cdots, k_6 \in \mathbb{N}\}$ forms a PBW-basis of $U_q^+(G_2)$. The iterated Ore extension of $U_q^+(G_2)$ is of the form:

$$U_q^+(G_2) = \mathbb{C}[E_1][E_2;\sigma_2][E_3;\sigma_3,\delta_3][E_4;\sigma_4,\delta_4][E_5;\sigma_5,\delta_5][E_6;\sigma_6,\delta_6];$$

where, σ_2 denotes the automorphism of $\mathbb{C}[E_1]$ defined by:

$$\sigma_2(E_1) = q^{-3}E_1,$$

 σ_3 denotes the automorphism of $\mathbb{C}[E_1][E_2;\sigma_2]$ defined by:

$$\sigma_3(E_1) = q^{-1}E_1 \qquad \sigma_3(E_2) = q^{-3}E_2,$$

 δ_3 denotes the σ_3 -derivation of $\mathbb{C}[E_1][E_2;\sigma_2]$ defined by:

$$\delta_3(E_1) = -(q + q^{-1} + q^{-3})E_2 \qquad \delta_3(E_2) = 0,$$

 σ_4 denotes the automorphism of $\mathbb{C}[E_1]\cdots [E_3;\sigma_3,\delta_3]$ defined by:

$$\sigma_4(E_1) = E_1 \qquad \sigma_4(E_2) = q^{-3}E_2 \qquad \sigma_4(E_3) = q^{-3}E_3,$$

 δ_4 denotes the σ_4 -derivation of $\mathbb{C}[E_1]\cdots [E_3;\sigma_3,\delta_3]$ defined by:

$$\delta_4(E_1) = (1 - q^2)E_3^2 \qquad \delta_4(E_2) = \frac{-q^4 + 2q^2 - 1}{q^4 + q^2 + 1}E_3^3 \qquad \delta_4(E_3) = 0$$

 σ_5 denotes the automorphism of $\mathbb{C}[E_1]\cdots [E_4;\sigma_4,\delta_4]$ defined by:

$$\sigma_5(E_1) = qE_1$$
 $\sigma_5(E_2) = E_2$ $\sigma_5(E_3) = q^{-1}E_3$ $\sigma_5(E_4) = q^{-3}E_4,$

 δ_5 denotes the σ_5 -derivation of $\mathbb{C}[E_1]\cdots [E_4;\sigma_4,\delta_4]$ defined by:

$$\delta_5(E_1) = -(1+q^2)E_3 \quad \delta_5(E_2) = (1-q^2)E_3^2 \quad \delta_5(E_3) = -(q+q^{-1}+q^{-3})E_4 \quad \delta_5(E_4) = 0,$$

 σ_6 denotes the automorphism of $\mathbb{C}[E_1]\cdots [E_5;\sigma_5,\delta_5]$ defined by:

$$\sigma_6(E_1) = q^3 E_1 \quad \sigma_6(E_2) = q^3 E_2 \quad \sigma_6(E_3) = E_3 \quad \sigma_6(E_4) = q^{-3} E_4 \quad \sigma_6(E_5) = q^{-3} E_5,$$

and δ_6 denotes the σ_6 -derivation of $\mathbb{C}[E_1]\cdots[E_5;\sigma_5,\delta_5]$ defined by:

$$\delta_6(E_1) = -q^3 E_5 \qquad \delta_6(E_2) = (q^2 - q^4) E_3 E_5 + (q^4 + q^2 - 1) E_4 \qquad \delta_6(E_3) = (1 - q^2) E_5^2$$
$$\delta_6(E_4) = \frac{-q^4 + 2q^2 - 1}{q^4 + q^2 + 1} E_5^3 \qquad \delta_6(E_5) = 0.$$

2.2 DDA and the center of $U_q^+(G_2)$

Henceforth, we set $A := U_q^+(G_2)$. We are now ready to describe the DDA of A. **2.2.1 DDA of** $U_q^+(G_2)$. The algebra A is a quantum nilpotent algebra (Remark 1.7.2), and therefore the theory of DDA studied in Section 1.7 applies to A. We construct the following elements of Fract(A) (computations have been omitted here, but can be found in Appendix A.2):

$$\begin{split} E_{1,6} &= E_1 + rE_5E_6^{-1} \\ E_{2,6} &= E_2 + tE_3E_5E_6^{-1} + uE_4E_6^{-1} + nE_5^3E_6^{-2} \\ E_{3,6} &= E_3 + sE_5^2E_6^{-1} \\ E_{4,6} &= E_4 + bE_5^3E_6^{-1} \\ E_{4,6} &= E_4 + bE_5^3E_6^{-1} \\ E_{1,5} &= E_{1,6} + hE_{3,6}E_{5,6}^{-1} + gE_{4,6}E_{5,6}^{-2} \\ E_{2,5} &= E_{2,6} + fE_{3,6}^2E_{5,6}^{-1} + pE_{3,6}E_{4,6}E_{5,6}^{-2} + eE_{4,6}^2E_{5,6}^{-3} \\ E_{3,5} &= E_{3,6} + aE_{4,6}E_{5,6}^{-1} \\ E_{1,4} &= E_{1,5} + sE_{3,5}^2E_{4,5}^{-1} \\ E_{2,4} &= E_{2,5} + bE_{3,5}^3E_{4,5}^{-1} \\ E_{1,3} &= E_{1,4} + aE_{2,4}E_{3,4}^{-1} \\ T_1 &:= E_{1,2} = E_{1,3} \\ T_2 &:= E_{2,2} = E_{2,3} = E_{2,4} \\ T_3 &:= E_{3,2} = E_{3,3} = E_{3,4} = E_{3,5} \\ T_4 &:= E_{4,2} = E_{4,3} = E_{4,4} = E_{4,5} = E_{4,6} \\ T_5 &:= E_{5,2} = E_{5,3} = E_{5,4} = E_{5,5} = E_{5,6} = E_5 \\ T_6 &:= E_{6,2} = E_{6,3} = E_{6,4} = E_{6,5} = E_{6,6} = E_6. \end{split}$$

The parameters a, b, e, f, g, h, n, p, r, s, t, u are all defined in Appendix C.

Again, from the theory of the DDA in Section 1.7, we have that for each $j \in \{2, \dots, 7\}$, the algebra $A^{(j)} := \mathbb{C}\langle E_{i,j} \mid i = 1, \dots, 6\rangle$ is the subalgebra of Fract(A). Since $(E_{1,7}, \dots, E_{6,7}) := (E_1, \dots, E_6)$, it follows that $A^{(7)} = A$.

2.2.2 Remark. Recall that $T_j = E_{j,j} = E_{j,j+1} \in A^{(j)} \cap A^{(j+1)}$. It follows from [8, Theorem 3.2.1] that the set $\Sigma_j := \{T_j^n \mid n \in \mathbb{N}\}$ is an Ore set in both $A^{(j)}$ and $A^{(j+1)}$ for each $1 \leq j \leq 6$, and that $A^{(j)}\Sigma_j^{-1} = A^{(j+1)}\Sigma_j^{-1}$.

2.2.3 The center of $U_q^+(G_2)$. Using (1.5.1), we have the skew-symmetric matrix M below:

$$M := \begin{bmatrix} 0 & (\beta_1, \beta_2) & \cdots & \cdots & (\beta_1, \beta_6) \\ -(\beta_1, \beta_2) & 0 & (\beta_2, \beta_3) & (\beta_2, \beta_6) \\ \vdots & \ddots & \ddots & \ddots & \vdots \\ \vdots & & \ddots & 0 & (\beta_5, \beta_6) \\ -(\beta_1, \beta_6) & \cdots & \cdots & -(\beta_5, \beta_6) & 0 \end{bmatrix}$$

	0	3	1	0	-1	-3
	-3	0	3	3	0	-3
	-1	-3	0	3	1	0
	0	-3	-3	0	3	3
	1	0	-1	-3	0	3
	3	3	0	-3	-3	0

Observe that $\overline{A} := A^{(2)} = \mathbb{C}_{q^M}[T_1, \cdots, T_6]$ is a quantum affine space. Set $\Omega_1 := T_1T_3T_5$ and $\Omega_2 := T_2T_4T_6$. One can verify that Ω_1 and Ω_2 are central elements of \overline{A} . That is, $\Omega_iT_j = T_j\Omega_i$ for all i = 1, 2, and $1 \le j \le 6$.

We now want to successively pull Ω_1 and Ω_2 from the quantum affine space \overline{A} into the algebra A using the data of the DDA of A discussed above. We only summarize the computations here, and provide a sketch of the computations in Appendix A.3.

$$\begin{aligned} \Omega_1 &:= T_1 T_3 T_5 \\ &= E_{1,4} E_{3,4} E_{5,4} + a E_{2,4} E_{5,4} \\ &= E_{1,5} E_{3,5} E_{5,5} + a E_{2,5} E_{5,5} \\ &= E_{1,6} E_{3,6} E_{5,6} + a E_{1,6} E_{4,6} + a E_{2,6} E_{5,6} + a' E_{3,6}^2 \\ &= E_1 E_3 E_5 + a E_1 E_4 + a E_2 E_5 + a' E_3^2, \end{aligned}$$

and

$$\Omega_2 := T_2 T_4 T_6$$

= $E_{2,5} E_{4,5} E_{6,5} + b E_{3,5}^3 E_{6,5}$
= $E_{2,6} E_{4,6} E_{6,6} + b E_{3,6}^3 E_{6,6}$
= $E_2 E_4 E_6 + b E_2 E_5^3 + b E_3^3 E_6 + b' E_3^2 E_5^2 + c' E_3 E_4 E_5 + d' E_4^2$

The parameters a, b, a', b', c', d' are all defined in Appendix C. Note, Ω_1 and Ω_2 are central elements of $A^{(j)}$ for each $2 \leq j \leq 7$, since $\operatorname{Fract}(A^{(j)}) = \operatorname{Fract}(\overline{A})$. From the commutation relations of A, one can observe that E_1 and E_6 are enough to generate A. Hence, it is sufficient to confirm that Ω_1 and Ω_2 are central elements of A by showing that they commute with only E_1 and E_6 . As a result, we have used GAP to confirm that $\Omega_1 E_i = E_i \Omega_1$ and $\Omega_2 E_i = E_i \Omega_2$ for each i = 1, 6. The GAP code can be found in Appendix A.3.1.

We now want to show that the center of A is a polynomial ring generated by Ω_1 and Ω_2 over \mathbb{C} . The following discussions will lead us to the proof.

Set $S_j := \{\lambda T_j^{i_j} T_{j+1}^{i_{j+1}} \cdots T_6^{i_6} \mid i_j, \cdots, i_6 \in \mathbb{N} \text{ and } \lambda \in \mathbb{C}^*\}$ for each $2 \leq j \leq 6$. One can observe that S_j is a multiplicative system of non-zero divisors of $A^{(j)} = \mathbb{C}\langle E_{i,j} \mid$ for all $i = 1, \cdots, 6\rangle$. Furthermore, the elements T_j, \cdots, T_6 are all normal in $A^{(j)}$. Hence, S_j is an Ore set in $A^{(j)}$. We can therefore localize $A^{(j)}$ at S_j as follows:

$$R_j := A^{(j)} S_j^{-1}.$$

Recall from Remark 2.2.2 that $\Sigma_j := \{T_j^n \mid n \in \mathbb{N}\}$ is an Ore set in both $A^{(j)}$ and $A^{(j+1)}$ for each $2 \leq j \leq 6$, and that

$$A^{(j)} \Sigma_j^{-1} = A^{(j+1)} \Sigma_j^{-1}.$$

For all $2 \le j \le 6$, we have that:

$$R_{j} = A^{(j)}S_{j}^{-1} = (A^{(j)}\Sigma_{j}^{-1})S_{1+j}^{-1} = (A^{(j+1)}\Sigma_{j}^{-1})S_{1+j}^{-1} = (A^{(j+1)}S_{j+1}^{-1})\Sigma_{j}^{-1} = R_{j+1}\Sigma_{j}^{-1}.$$
(2.2.1)

Note, $R_7 := A$.

Again, one can also observe that T_1 is normal in R_2 . As a result, the localization

$$R_1 := R_2[T_1^{-1}]$$

also holds in R_2 . In fact, R_1 is the quantum torus associated to the quantum affine space \overline{A} . As a result, $R_1 = \mathbb{C}_{q^M}[T_1^{\pm 1}, \cdots, T_6^{\pm 1}]$, where $T_iT_j = q^{\mu_{ij}}T_jT_i$ for all $1 \le i, j \le 6$ and $\mu_{ij} \in M$. Similar to [36, §31], we construct the following tower of algebras:

$$A = R_7 \subset R_6 = R_7 \Sigma_6^{-1} \subset R_5 = R_6 \Sigma_5^{-1} \subset R_4 = R_5 \Sigma_4^{-1}$$
$$\subset R_3 = R_4 \Sigma_3^{-1} \subset R_2 = R_3 \Sigma_2^{-1} \subset R_1.$$
 (2.2.2)

Note, the family $(E_{1,j}^{k_1}\cdots E_{6,j}^{k_6})$, where $k_i \in \mathbb{N}$ if i < j and $k_i \in \mathbb{Z}$ otherwise is a PBWbasis of R_j for all $1 \leq i, j \leq 7$. Therefore, the family $(T_1^{k_1}T_2^{k_2}T_3^{k_3}T_4^{k_4}T_5^{k_5}T_6^{k_6})_{k_1,\cdots,k_6 \in \mathbb{Z}}$ is a basis of R_1 .

2.2.4 Lemma. $Z(R_1) = \mathbb{C}[\Omega_1^{\pm 1}, \Omega_2^{\pm 1}].$

Proof. Obviously, $\mathbb{C}[\Omega_1^{\pm 1}, \Omega_2^{\pm 1}] \subseteq Z(R_1)$. For the reverse inclusion, let $y \in Z(R_1)$. Then, y can be written in terms of the basis of R_1 as:

$$y = \sum_{(i,\cdots,n)\in\mathbb{Z}^6} a_{(i,\cdots,n)} T_1^i T_2^j T_3^k T_4^l T_5^m T_6^n.$$

One can verify that $yT_1 = q^{-3j-k+m+3n}T_1y$. Since $y \in Z(R_1)$, it follows that

$$-3j - k + m + 3n = 0.$$

Similarly, $yT_2 = q^{3i-3k-3l+3n}T_2y$. Since $y \in Z(R_1)$, we have:

$$3i - 3k - 3l + 3n = 0.$$

Following the same pattern for T_3, T_4, T_5 and T_6 , one can confirm that

$$i + 3j - 3l - m = 0,$$

 $3j + 3k - 3m - 3n = 0,$
 $-i + k + 3l - 3n = 0,$
 $-3i - 3j + 3l + 3m = 0.$

Solving this system of six equations will reveal that i = k = m and j = l = n. One can therefore write

$$y = \sum_{(i,j)\in\mathbb{Z}^2} a_{(i,j)} T_1^i T_2^j T_3^i T_4^j T_5^j T_6^j = \sum_{(i,j)\in\mathbb{Z}^2} q^{\bullet} a_{(i,j)} T_1^i T_3^i T_5^j T_2^j T_4^j T_6^j = \sum_{(i,j)\in\mathbb{Z}^2} q^{\bullet} a_{(i,j)} \Omega_1^i \Omega_2^j.$$

This implies that $y \in \mathbb{C}[\Omega_1^{\pm 1}, \Omega_2^{\pm 1}]$ as expected.

2.2.5 Corollary. 1. $Z(R_3) = \mathbb{C}[\Omega_1, \Omega_2].$

2. $Z(\overline{A}) = \mathbb{C}[\Omega_1, \Omega_2].$

Proof. 1. Clearly, $\mathbb{C}[\Omega_1, \Omega_2] \subseteq Z(R_3)$. For the reverse inclusion, let $y \in Z(R_3)$. Then, y can be written in terms of the basis of R_3 (recall, $T_i = E_{i,3}$) as:

$$y = \sum_{(i,\cdots,n)\in\mathbb{N}^2\times\mathbb{Z}^4} a_{(i,\cdots,n)} T_1^i T_2^j T_3^k T_4^l T_5^m T_6^n.$$

Now; T_1, \dots, T_6 are all normal elements in R_3 . In fact, they satisfy the same commutation relations in R_1 . Hence, following procedures similar to the lemma above, one will arrive at the conclusion that i = k = m and j = l = n. Since $i, j \ge 0$; we have that $y = \sum_{(i,j)\in\mathbb{N}^2} q^{\bullet}a_{(i,j)}T_1^iT_3^iT_5^jT_2^jT_4^jT_6^j = \sum_{(i,j)\in\mathbb{N}^2} q^{\bullet}a_{(i,j)}\Omega_1^i\Omega_2^j$. This implies that

 $y \in \mathbb{C}[\Omega_1, \Omega_2]$ as expected.

- 2. Similar to (1).
- **2.2.6 Lemma.** $Z(A) = \mathbb{C}[\Omega_1, \Omega_2].$

Proof. Since R_i is a localization of R_{i+1} , it follows that $Z(R_{i+1}) \subseteq Z(R_i)$. From (2.2.2), we have that $Z(A) \subseteq Z(R_3)$. Observe that $\mathbb{C}[\Omega_1, \Omega_2] \subseteq Z(A) \subseteq Z(R_3) = \mathbb{C}[\Omega_1, \Omega_2]$.

2.2.7 Remark. Since $Z(A) = Z(R_3) = \mathbb{C}[\Omega_1, \Omega_2]$ and $Z(R_{i+1}) \subseteq Z(R_i)$, it follows from (2.2.2) that $Z(A) = Z(R_6) = Z(R_5) = Z(R_4) = Z(R_3) = \mathbb{C}[\Omega_1, \Omega_2]$. One can also deduce from Lemma 2.2.4 that $Z(R_2) = \mathbb{C}[\Omega_1, \Omega_2^{\pm 1}]$.

2.3 Proof of the completely primeness of $\langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$

Since all prime ideals of $A = U_q^+(G_2)$ are completely prime (Remark 1.6.1), it is sufficient to show that $\langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$ are completely prime by showing that they are prime ideals. Note, the data of the DDA of A and the expressions for Ω_1 and Ω_2 in the previous section will be helpful in the proofs of Lemmas 2.3.1 and 2.3.2. Recall, the notation $\langle \Theta \rangle_R$ means an ideal generated by the element Θ in any arbitrary ring R. Where no doubt arises, we will simply write $\langle \Theta \rangle$.

From Section 1.7 we know that there is a bijection between $\{P \in \operatorname{Spec}(A^{(j+1)}) \mid P \cap S_j = \emptyset\}$ and $\{Q \in \operatorname{Spec}(A^{(j)}) \mid Q \cap S_j = \emptyset\}$ via $P = QS_j^{-1} \cap A^{(j+1)}$. Note, $\langle T_1 \rangle$ and $\langle T_2 \rangle$ are prime ideals of the quantum affine space \overline{A} , since each of the factor algebras $\overline{A}/\langle T_1 \rangle$ and $\overline{A}/\langle T_2 \rangle$ is isomorphic to a quantum affine space of rank 5 which is well known to be a domain. From Section 1.7 we have that $\psi_j : \operatorname{Spec}(A^{(1+j)}) \hookrightarrow \operatorname{Spec}(A^{(j)})$ for $2 \leq j \leq 6$. Hence, $\psi := \psi_6 \circ \cdots \circ \psi_2 : \operatorname{Spec}(A) \hookrightarrow \operatorname{Spec}(\overline{A})$. Recall, $A^{(7)} = A$ and $A^{(2)} = \overline{A}$.

The following result shows that $\langle T_1 \rangle \in \text{Im}(\psi)$ (i.e. image of ψ) and that $\langle \Omega_1 \rangle$ is the completely prime ideal of A such that $\psi(\langle \Omega_1 \rangle) = \langle T_1 \rangle$.

2.3.1 Lemma. $\langle \Omega_1 \rangle \in \text{Spec}(A)$.

Proof. We will prove this result in several steps by showing that:

1.
$$\langle T_1 \rangle_{A^{(3)}} \in \text{Spec}(A^{(3)}).$$

2. $\langle E_{1,4}T_3 + aT_2 \rangle = \langle T_1 \rangle_{A^{(3)}} [T_3^{-1}] \cap A^{(4)}$, hence $Q_1 := \langle E_{1,4}T_3 + aT_2 \rangle \in \text{Spec}(A^{(4)})$
3. $\langle E_{1,5}T_3 + aE_{2,5} \rangle = Q_1 [T_4^{-1}] \cap A^{(5)}$, hence $Q_2 := \langle E_{1,5}T_3 + aE_{2,5} \rangle \in \text{Spec}(A^{(5)}).$
4. $\langle \Omega_1 \rangle_{A^{(6)}} = Q_2 [T_5^{-1}] \cap A^{(6)}$, hence $\langle \Omega_1 \rangle_{A^{(6)}} \in \text{Spec}(A^{(6)}).$

5. $\langle \Omega_1 \rangle_A = \langle \Omega_1 \rangle_{A^{(6)}} [T_6^{-1}] \cap A$, hence $\langle \Omega_1 \rangle_A \in \operatorname{Spec}(A)$.

We are now ready to prove the above claims.

1. One can easily verify that $A^{(3)}/\langle T_1 \rangle$ is isomorphic to a quantum affine space of rank 5, which is a domain, hence $\langle T_1 \rangle$ is a prime ideal in $A^{(3)}$.

2. Note, $T_1 = E_{1,4} + aT_2T_3^{-1}$. We want to show that $\langle E_{1,4}T_3 + aT_2 \rangle = \langle T_1 \rangle_{A^{(3)}}[T_3^{-1}] \cap$ $A^{(4)}$. Observe that $\langle E_{1,4}T_3 + aT_2 \rangle \subseteq \langle T_1 \rangle_{A^{(3)}} [T_3^{-1}] \cap A^{(4)}$. We established the reverse inclusion. Let $y \in \langle T_1 \rangle_{A^{(3)}}[T_3^{-1}] \cap A^{(4)}$. Then, $y \in \langle T_1 \rangle_{A^{(3)}}[T_3^{-1}]$. Therefore, there exists $i \in \mathbb{N}$ such that $yT_3^i \in \langle T_1 \rangle_{A^{(3)}}$. This implies that $yT_3^i = T_1v$, for some $v \in A^{(3)}$. Since $A^{(3)}[T_3^{-1}] = A^{(4)}[T_3^{-1}]$, there exists $j \in \mathbb{N}$ such that $vT_3^j = v'$, for some $v' \in A^{(4)}$. It follows that $yT_3^{i+j} = T_1vT_3^j = T_1v' = (E_{1,4} + aT_2T_3^{-1})v' = (E_{1,4}T_3 + aT_2)T_3^{-1}v'.$ The multiplicative system generated by T_3 satisfies the Ore condition in $A^{(4)}$, hence, there exists $k \in \mathbb{N}$ and $v'' \in A^{(4)}$ such that $T_3^{-1}v' = v''T_3^{-k}$. One can therefore write $yT_3^{i+j} = (E_{1,4}T_3 + aT_2)v''T_3^{-k}$. This implies that $yT_3^{\delta} = \Omega_1'v''$, where $\Omega_1' := E_{1,4}T_3 + aT_2$ and $\delta = i + j + k$. Set $S := \{s \in \mathbb{N} \mid \exists v'' \in A^{(4)} : yT_3^s = \Omega_1'v''\}$. Note, $S \neq i$ \emptyset , since $\delta \in S$. Let $s = s_0$ be the minimum element of S such that $yT_3^{s_0} = \Omega'_1v''$. We want to show that $s_0 = 0$. Remember, $\Omega'_1 T_5 = \Omega_1$ in $A^{(4)}$. Since Ω_1 is central in $A^{(4)}$, and T_5 is normal in $A^{(4)}$, we must have Ω'_1 to be a normal element in $A^{(4)}$, otherwise, there will be a contradiction. Therefore, there exists $w \in A^{(4)}$ such that $yT_3^{s_0} =$ $\Omega'_1 v'' = w \Omega'_1$. Now, $A^{(4)}$ can be viewed as a free left $\mathbb{C}\langle E_{1,4}, T_2, T_4, T_5, T_6 \rangle$ -module with basis $(T_3^{\xi})_{\xi \in \mathbb{N}}$. One can therefore write $y = \sum_{\xi=0}^n \alpha_{\xi} T_3^{\xi}$ and $w = \sum_{\xi=0}^n \beta_{\xi} T_3^{\xi}$, where $\alpha_{\xi}, \beta_{\xi} \in \mathbb{C}\langle E_{1,4}, T_2, T_4, T_5, T_6 \rangle$. This implies that $\sum_{\xi=0}^n \alpha_{\xi} T_3^{\xi+s_0} = \sum_{\xi=0}^n \beta_{\xi} T_3^{\xi} \Omega'_1 = \sum_{\xi=0}^n \beta_{\xi} T_3^{\xi} \Omega'_1$

$$\begin{split} \sum_{\xi=0}^n q^{\bullet} \beta_{\xi} \Omega_1' T_3^{\xi} \text{ (note, } T_3 \Omega_1' &= q^{-1} \Omega_1' T_3 \text{). Given that } \Omega_1' &= E_{1,4} T_3 + a T_2 \text{, we have that } \\ \sum_{\xi=0}^n \alpha_{\xi} T_3^{\xi+s_0} &= \sum_{\xi=0}^n q^{\bullet} \beta_{\xi} E_{1,4} T_3^{1+\xi} + \sum_{\xi=0}^n q^{\bullet} a \beta_{\xi} T_2 T_3^{\xi} \text{. Suppose that } s_0 > 0. \text{ Then,} \\ \text{identifying the constant coefficients, we have } q^{\bullet} a \beta_0 T_2 &= 0. \text{ As a result, } \beta_0 &= 0, \text{ since } \\ q^{\bullet} a T_2 \neq 0. \text{ Hence, } w \text{ can be written as } w &= \sum_{\xi=1}^n \beta_{\xi} T_3^{\xi} \text{. Returning to } y T_3^{s_0} &= w \Omega_1', \text{ we have that } y T_3^{s_0} &= \sum_{\xi=1}^n \beta_{\xi} T_3^{\xi} \Omega_1' &= \sum_{\xi=1}^n q^{\bullet} \beta_{\xi} \Omega_1' T_3^{\xi} &= \Omega_1' \sum_{\xi=1}^n q^{\bullet} \beta_{\xi}' T_3^{\xi} \text{. This implies that } \\ y T_3^{s_0-1} &= \Omega_1' w', \text{ where } w' &= \sum_{\xi=1}^n q^{\bullet} \beta_{\xi}' T_3^{\xi-1} \in A^{(4)}, \text{ with } \beta_{\xi}' \in \mathbb{C} \langle E_{1,4}, T_2, T_4, T_5, T_6 \rangle. \\ \text{Consequently, } s_0 - 1 \in S, \text{ a contradiction! Therefore, } s_0 &= 0 \text{ and } y = \Omega_1' v'' \in \langle \Omega_1' \rangle \\ &\leq (E_{1,4} T_3 + a T_2). \text{ Hence, } \langle T_1 \rangle_{A^{(3)}} [T_3^{-1}] \cap A^{(4)} \subseteq \langle E_{1,4} T_3 + a T_2 \rangle \text{ as desired.} \end{split}$$

3. We want to show that $\langle E_{1,5}T_3 + aE_{2,5} \rangle = \langle \Omega'_1 \rangle_{A^{(4)}} [T_4^{-1}] \cap A^{(5)}$. Note, $\Omega'_1 =$ $E_{1,4}T_3 + aT_2 = E_{1,5}T_3 + aE_{2,5}. \text{ Observe that } \langle E_{1,5}T_3 + aE_{2,5} \rangle \subseteq \langle \Omega_1' \rangle_{A^{(4)}}[T_4^{-1}] \cap A^{(5)}.$ We establish the reverse inclusion. Let $y \in \langle \Omega'_1 \rangle_{A^{(4)}}[T_4^{-1}] \cap A^{(5)}$. Then, $y \in \langle \Omega'_1 \rangle_{A^{(4)}}[T_4^{-1}]$. Therefore, there exists $i \in \mathbb{N}$ such that $yT_4^i \in \langle \Omega_1' \rangle_{A^{(4)}}$. This implies that $yT_4^i = \Omega_1' v$, for some $v \in A^{(4)}$. Since $A^{(4)}[T_4^{-1}] = A^{(5)}[T_4^{-1}]$, there exists $j \in \mathbb{N}$ such that $vT_4^j = v'$, for some $v' \in A^{(5)}$. It follows that $yT_4^{i+j} = \Omega'_1 vT_4^j = \Omega'_1 v'$. This implies that $yT_4^{\delta} = \Omega'_1 v'$, where $\delta = i + j$. Set $S := \{s \in \mathbb{N} \mid \exists v' \in A^{(5)} : yT_4^s = \Omega_1'v'\}$. Since $\delta \in S$, we have that $S \neq \emptyset$. Let $s = s_0$ be the minimum element of S such that $yT_4^{s_0} = \Omega'_1 v'$. We want to show that $s_0 = 0$. Note, $\Omega_1' T_5 = \Omega_1$ in $A^{(5)}$. Since Ω_1 is central in $A^{(5)}$, and T_5 is normal in $A^{(5)}$, we must have Ω'_1 as a normal element in $A^{(5)}$. Therefore, there exists some $v'' \in A^{(5)}$ such that $yT_4^{s_0} = \Omega_1'v' = v''\Omega_1'$. Now, $A^{(5)}$ can be viewed as a free left $\mathbb{C}\langle E_{1,5}, E_{2,5}, T_3, T_5, T_6 \rangle$ -module with basis $\left(T_4^{\xi}\right)_{\zeta \in \mathbb{N}}$. One can write $y = \sum_{\xi=0}^{n} \alpha_{\xi} T_4^{\xi}$ and $v'' = \sum_{\xi=0}^{n} \beta_{\xi} T_4^{\xi}$, where $\alpha_{\xi}, \beta_{\xi} \in \mathbb{C}\langle E_{1,5}, E_{2,5}, T_3, T_5, T_6 \rangle$. This implies that $\sum_{\xi=0}^n \alpha_\xi T_4^{\xi+s_0} = \sum_{\xi=0}^n \beta_\xi T_4^{\xi} \Omega_1' = \sum_{\xi=0}^n q^{\bullet} \beta_\xi \Omega_1' T_4^{\xi}$ (note, $T_4 \Omega_1' =$ $q^{-3}\Omega'_1T_4$). Suppose that $s_0 > 0$. Then, identifying the constant coefficients, we have that $q^{\bullet}\beta_0\Omega'_1 = 0$. Hence, $\beta_0 = 0$, since $q^{\bullet}\Omega'_1 \neq 0$. One can therefore write v'' as $v'' = \sum_{\xi=1}^n \beta_\xi T_4^\xi$. Returning to $yT_4^{s_0} = v''\Omega_1'$, we have that $yT_4^{s_0} = \sum_{\xi=1}^n \beta_\xi T_4^\xi \Omega_1' =$ $\sum_{\xi=1}^n q^{\bullet} \beta_{\xi} \Omega_1' T_4^{\xi} = \Omega_1' \sum_{\xi=1}^n q^{\bullet} \beta_{\xi}' T_4^{\xi}, \text{ where } \beta_{\xi}' \in \mathbb{C} \langle E_{1,5}, E_{2,5}, T_3, T_5, T_6 \rangle. \text{ This implies}$ that $yT_4^{s_0-1} = \Omega_1'w$, where $w = \sum_{\xi=1}^n q^{\bullet}\beta_{\xi}'T_4^{\xi-1} \in A^{(5)}$. Consequently, $s_0 - 1 \in S$, a contradiction! Therefore, $s_0 = 0$ and $y = \Omega'_1 v' \in \langle \Omega'_1 \rangle = \langle E_{1,5}T_3 + aE_{2,5} \rangle$. As a result,

 $\langle \Omega_1'\rangle_{A^{(4)}}[T_4^{-1}]\cap A^{(5)}\subseteq \langle E_{1,5}T_3+aE_{2,5}\rangle \text{ as desired}.$

4. Observe that $\Omega'_1 = E_{1,5}T_3 + aE_{2,5} = \Omega_1 T_5^{-1}$ in $A^{(6)}[T_5^{-1}]$. We want to show that $\langle \Omega_1' \rangle_{A^{(5)}} [T_5^{-1}] \cap A^{(6)} = \langle \Omega_1 \rangle_{A^{(6)}}.$ Obviously, $\langle \Omega_1 \rangle_{A^{(6)}} \subseteq \langle \Omega_1' \rangle_{A^{(5)}} [T_5^{-1}] \cap A^{(6)}.$ We establish the reverse inclusion. Let $y \in \langle \Omega'_1 \rangle_{A^{(5)}}[T_5^{-1}] \cap A^{(6)}$. This implies that $y \in \langle \Omega'_1 \rangle_{A^{(5)}}[T_5^{-1}]$. There exists $i \in \mathbb{N}$ such that $yT_5^i \in \langle \Omega_1 \rangle_{A^{(5)}}$. Hence, $yT_5^i = \Omega'_1 v$, for some $v \in A^{(5)}$. Furthermore, since $A^{(5)}[T_5^{-1}] = A^{(6)}[T_5^{-1}]$, there exists $j \in \mathbb{N}$ such that $vT_5^j = v'$, for some $v' \in A^{(6)}$. It follows from $yT_5^i = \Omega_1'v$ that $yT_5^{i+j} = \Omega_1'vT_5^j = \Omega_1'v' = \Omega_1T_5^{-1}v'$ (note, $\Omega'_1T_5 = \Omega_1$ in $A^{(6)}$). The multiplicative system generated by T_5 satisfies the Ore condition in $A^{(6)}$, hence, there exists $k \in \mathbb{N}$ and $v'' \in A^{(6)}$ such that $T_5^{-1}v' = v''T_5^{-k}$. One can therefore write $yT_5^{i+j} = \Omega_1 v''T_5^k$. Hence, $yT_5^{\delta} = \Omega_1 v''$, where $\delta = i+j+k$. Set S := $\{s \in \mathbb{N} \mid \exists v'' \in A^{(6)} : yT_5^s = \Omega_1 v''\}$. Since $\delta \in S$, we have that $S \neq \emptyset$. Let $s = s_0$ be the minimum element of S such that $yT_5^{s_0} = \Omega_1 v''$. We want to show that $s_0 = 0$. Now, $A^{(6)}$ can be viewed as a free $\mathbb{C}\langle E_{1,6}, E_{2,6}, E_{3,6}, T_4, T_6 \rangle$ -module with basis $\left(T_5^{\xi}\right)_{\xi \in \mathbb{N}}$. One can write $y = \sum_{\xi=0}^{n} \alpha_{\xi} T_{5}^{\xi}$ and $v'' = \sum_{\xi=0}^{n} \beta_{\xi} T_{5}^{\xi}$, where $\alpha_{\xi}, \beta_{\xi} \in \mathbb{C} \langle E_{1,6}, E_{2,6}, E_{3,6}, T_{4}, T_{6} \rangle$. With this, $yT_5^{s_0} = \Omega_1 v''$ implies that $\sum_{\xi=0}^n \alpha_\xi T_5^{\xi+s_0} = \sum_{\xi=0}^n \beta_\xi \Omega_1 T_5^{\xi}$. Write $\Omega_1 =$ $\gamma_1T_5 + \gamma_2$, where $\gamma_1 = E_{1,6}E_{3,6} + aE_{2,6}$ and $\gamma_2 = a'E_{3,6}^2 + aE_{1,6}E_{4,6}$. It follows that $\sum_{\xi=0}^n \alpha_\xi T_5^{\xi+s_0} = \sum_{\xi=0}^n \beta_\xi \gamma_1 T_5^{\xi+1} + \sum_{\xi=0}^n \beta_\xi \gamma_2 T_5^{\xi}$. Suppose that $s_0 > 0$. Then, identifying the constant coefficients, we have that $\beta_0\gamma_2 = 0$. Hence, $\beta_0 = 0$, since $\gamma_2 \neq 0$. One can therefore write $v'' = \sum_{\xi=1}^n \beta_\xi T_5^{\xi}$. Returning to $yT_5^{s_0} = \Omega_1 v''$, we have that $yT_5^{s_0} = \Omega_1 v''$ $\Omega_1 \sum_{\xi=1}^n \beta_\xi T_5^{\xi}$. This implies that $yT_5^{s_0-1} = \Omega_1 w$, where $w = \sum_{\xi=1}^n \beta_\xi T_5^{\xi-1} \in A^{(6)}$. As a result, $s_0 - 1 \in S$, a contradiction! Therefore, $s_0 = 0$ and $y = \Omega_1 v'' \in \langle \Omega_1 \rangle_{A^{(6)}}$. Consequently, $\langle \Omega'_1 \rangle_{A^{(5)}} [T_5^{-1}] \cap A^{(6)} \subseteq \langle \Omega_1 \rangle_{A^{(6)}}$ as desired.

5. We want to show that $\langle \Omega_1 \rangle_{A^{(6)}} [T_6^{-1}] \cap A = \langle \Omega_1 \rangle_A$.

Obviously, $\langle \Omega_1 \rangle_A \subseteq \langle \Omega_1 \rangle_{A^{(6)}} [T_6^{-1}] \cap A$. We establish the reverse inclusion. Let $y \in \langle \Omega_1 \rangle_{A^{(6)}} [T_6^{-1}] \cap A$. This implies that $y \in \langle \Omega_1 \rangle_{A^{(6)}} [T_6^{-1}]$. There exists $i \in \mathbb{N}$ such that $yT_6^i \in \langle \Omega_1 \rangle_{A^{(6)}}$. Hence, $yT_6^i = \Omega_1 v$, for some $v \in A^{(6)}$. Furthermore, since $A^{(6)}[T_6^{-1}] = A[T_6^{-1}]$, there exists $j \in \mathbb{N}$ such that $vT_6^j = v'$, for some $v' \in A$. It follows from $yT_6^i = \Omega_1 v$ that $yT_6^{i+j} = \Omega_1 vT_6^j = \Omega_1 v'$. Hence, $yT_6^\delta = \Omega_1 v'$, where $\delta = i + j$. Set

$$\begin{split} S &:= \{s \in \mathbb{N} \mid \exists v' \in A : yT_6^s = \Omega_1 v'\}. \text{ Note, } \delta \in S, \text{ hence } S \text{ is non-empty. Let } s = s_0 \\ \text{be the minimum element of } S \text{ such that } yT_6^{s_0} = \Omega_1 v'. \text{ We want to show that } s_0 = 0. \\ \text{Now, } A \text{ can be viewed as a free } \mathbb{C}\langle E_1, E_2, E_3, E_4, T_5 \rangle - \text{module with basis } \left(T_6^\xi\right)_{\xi \in \mathbb{N}}. \text{ One } \\ \text{can write } y = \sum_{\xi=0}^n \alpha_\xi T_6^\xi \text{ and } v' = \sum_{\xi=0}^n \beta_\xi T_6^\xi, \text{ where } \alpha_\xi, \beta_\xi \in \mathbb{C}\langle E_1, E_2, E_3, E_4, T_5 \rangle. \\ \text{With this, } yT_6^{s_0} = \Omega_1 v' \text{ implies that } \sum_{\xi=0}^n \alpha_\xi T_6^{\xi+s_0} = \sum_{\xi=0}^n \beta_\xi \Omega_1 T_6^\xi. \\ \text{Suppose that } s_0 > 0. \text{ Then, identifying constant coefficients, we have that } \beta_0\Omega_1 = 0. \\ \text{As a result, } \beta_0 = 0, \text{ since } \Omega_1 \neq 0. \\ \text{One can therefore write } v' = \sum_{\xi=1}^n \beta_\xi T_6^\xi. \\ \text{Returning to } yT_6^{s_0} = \Omega_1 \sum_{\xi=1}^n \beta_\xi T_6^\xi. \\ \text{This implies that } yT_6^{s_0-1} = \Omega_1 v', \\ \text{where } v'' = \sum_{\xi=1}^n \beta_\xi T_6^{\xi-1} \in A. \\ \text{Hence, } s_0 - 1 \in S, \text{ a contradiction! Therefore, } s_0 = 0 \\ \text{and } y = \Omega_1 v' \in \langle \Omega_1 \rangle_A. \\ \text{Consequently, } \langle \Omega_1 \rangle_{A^{(6)}} [T_6^{-1}] \cap A \subseteq \langle \Omega_1 \rangle_A \\ \text{as desired.} \\ \end{array}$$

The following result also shows that $\langle T_2 \rangle \in \text{Im}(\psi)$ and that $\langle \Omega_2 \rangle$ is the completely prime ideal of A such that $\psi(\langle \Omega_2 \rangle) = \langle T_2 \rangle$.

2.3.2 Lemma. $\langle \Omega_2 \rangle \in \text{Spec}(A)$.

Proof. Similar to the previous lemma, we also prove this result in several steps by showing that:

- 1. $\langle T_2 \rangle_{A^{(3)}} \in \text{Spec}(A^{(3)}).$
- 2. $\langle T_2 \rangle_{A^{(4)}} \in \text{Spec}(A^{(4)}).$
- 3. $\langle E_{2,5}T_4 + bT_3^3 \rangle = \langle T_2 \rangle_{A^{(4)}} [T_4^{-1}] \cap A^{(5)}$, hence $Q_1 := \langle E_{2,5}T_4 + bT_3^3 \rangle \in \operatorname{Spec}(A^{(5)})$.
- 4. $\langle E_{2,6}T_4 + bE_{3,6}^3 \rangle = Q_1[T_5^{-1}] \cap A^{(6)}$, hence $Q_2 := \langle E_{2,6}T_4 + bE_{3,6}^3 \rangle \in \text{Spec}(A^{(6)})$.
- 5. $\langle \Omega_2 \rangle = Q_2[T_6^{-1}] \cap A$, hence $\langle \Omega_2 \rangle \in \operatorname{Spec}(A)$.

We are now ready to prove the above claims.

1. The quotient algebra $A^{(3)}/\langle T_2 \rangle$ is isomorphic to a quantum affine space of rank 5, which is a domain, hence $\langle T_2 \rangle$ is a prime ideal in $A^{(3)}$.

2. Similar to (1).

3. Recall from the DDA of A that, $T_2 = E_{2,5} + bT_3^3T_4^{-1}$. We want to show that $\langle E_{2,5}T_4 + bT_3^3 \rangle = \langle T_2 \rangle_{A^{(4)}} [T_4^{-1}] \cap A^{(5)}$. Observe that $\langle E_{2,5}T_4 + bT_3^3 \rangle \subseteq \langle T_2 \rangle_{A^{(4)}} [T_4^{-1}] \cap A^{(5)}$.

We establish the reverse inclusion. Let $y \in \langle T_2 \rangle_{A^{(4)}}[T_4^{-1}] \cap A^{(5)}$. This implies that $y \in \langle T_2 \rangle_{A^{(4)}}[T_4^{-1}]$. Therefore, there exists $i \in \mathbb{N}$ such that $yT_4^i \in \langle T_2 \rangle_{A^{(4)}}$. This implies that $yT_4^i = T_2v$ for some $v \in A^{(4)}$. Since $A^{(4)}[T_4^{-1}] = A^{(5)}[T_4^{-1}]$, there exists $j \in \mathbb{N}$ such that $vT_4^j = v'$ for some $v' \in A^{(5)}$. It follows that $yT_4^{i+j} = T_2vT_4^j = v'$ $T_2v' = (E_{2,5} + bT_3^3T_4^{-1})v' = (E_{2,5}T_4 + bT_3^3)T_4^{-1}v'$. Note, the multiplicative set generated by T_4 satisfies the Ore condition in $A^{(5)}$, hence, there exist $k \in \mathbb{N}$ and $v'' \in A^{(5)}$ such that $T_4^{-1}v' = v''T_4^{-k}$. One can therefore write $yT_4^{i+j} = (E_{2,5}T_4 + bT_3^3)v''T_4^{-k}$. This implies that yT_4^δ = $\Omega_2'v''$, where Ω_2' := $E_{2,5}T_4 + bT_3^3$ and δ = i + j + k. Set $S := \{s \in \mathbb{N} \mid \exists v'' \in A^{(5)} : yT_4^s = \Omega'_2 v''\}. \text{ Since } \delta \in S, \text{ we have that } S \neq \emptyset.$ Let $s = s_0$ be the minimum element of S such that $yT_4^{s_0} = \Omega_2'v''$. Note, $\Omega_2'T_6 = \Omega_2$ in $A^{(5)}$. Given that Ω_2 is central in $A^{(5)}$, and T_6 is normal in $A^{(5)}$, we must have Ω'_2 to be a normal element in $A^{(5)}$. Therefore, there exists $w \in A^{(5)}$ such that $yT_4^{s_0}$ $\Omega'_2 v'' = w \Omega'_2$. Now, $A^{(5)}$ can be viewed as a free left $\mathbb{C}\langle E_{1,5}, E_{2,5}, T_3, T_5, T_6 \rangle$ -module with basis $\left(T_4^{\xi}\right)_{\xi \in \mathbb{N}}$. One can write $y = \sum_{\xi=0}^n \alpha_{\xi} T_4^{\xi}$ and $w = \sum_{\xi=0}^n \beta_{\xi} T_4^{\xi}$, where $\alpha_{\xi}, \beta_{\xi} \in \mathbb{C}\langle E_{1,5}, E_{2,5}, T_3, T_5, T_6 \rangle$. It follows from $yT_4^{s_0} = w\Omega'_2$ that $\sum_{\xi=0}^n \alpha_{\xi}T_4^{\xi+s_0} = w\Omega'_2$ $\sum_{\xi=0}^{n} \beta_{\xi} T_{4}^{\xi} \Omega_{2}' = \sum_{\xi=0}^{n} q^{\bullet} \beta_{\xi} \Omega_{2}' T_{4}^{\xi} \text{ (note, } T_{4} \Omega_{2}' = q^{-3} \Omega_{2}' T_{4} \text{). Since } \Omega_{2}' = E_{2,5} T_{4} + b T_{3}^{3},$ we have that $\sum_{\xi=0}^{n} \alpha_{\xi} T_{4}^{\xi+s_{0}} = \sum_{\xi=0}^{n} q^{\bullet} \beta_{\xi} E_{2,5} T_{4}^{1+\xi} + \sum_{\xi=0}^{n} q^{\bullet} b \beta_{\xi} T_{3}^{3} T_{4}^{\xi}$. Suppose that $s_0 > 0$. Then, identifying the constant coefficients, we have: $q^{\bullet}b\beta_0T_3^3 = 0$. Hence, $\beta_0 = 0$, since $q^{\bullet}bT_3^3 \neq 0$. One can therefore write $w = \sum_{\xi=1}^n \beta_{\xi}T_4^{\xi}$. Returning to $yT_4^{s_0} = w\Omega_2'$, we have that $yT_4^{s_0} = \sum_{\xi=1}^n \beta_\xi T_4^\xi \Omega_2' = \sum_{\xi=1}^n q^{\bullet} \beta_\xi \Omega_2' T_4^\xi = \Omega_2' \sum_{\xi=1}^n q^{\bullet} \beta_\xi' T_4^\xi$ for some $\beta'_{\xi} \in \mathbb{C}\langle E_{1,5}, E_{2,5}, T_3, T_5, T_6 \rangle$. This implies that $yT_4^{s_0-1} = \Omega'_2 w'$, where w' = w' $\sum_{\xi=1}^n q^{\bullet} \beta'_{\xi} T_4^{\xi-1} \in A^{(5)}$. Consequently, $s_0 - 1 \in S$, a contradiction! Therefore, $s_0 = 0$ and $y = \Omega'_2 v'' \in \langle \Omega'_2 \rangle = \langle E_{2,5}T_4 + bT_3^3 \rangle$. Hence, $\langle T_2 \rangle_{A^{(4)}} [T_4^{-1}] \cap A^{(5)} \subseteq \langle E_{2,5}T_4 + bT_3^3 \rangle$ as desired.

4. We want to show that $\langle E_{2,6}T_4 + bE_{3,6}^3 \rangle = \langle \Omega'_2 \rangle_{A^{(5)}}[T_5^{-1}] \cap A^{(6)}$. Observe that $\langle E_{2,6}T_4 + bE_{3,6}^3 \rangle \subseteq \langle \Omega'_2 \rangle_{A^{(5)}}[T_5^{-1}] \cap A^{(6)}$. We establish the reverse inclusion. Let $y \in \langle \Omega'_2 \rangle_{A^{(5)}}[T_5^{-1}] \cap A^{(6)}$. This implies that $y \in \langle \Omega'_2 \rangle_{A^{(5)}}[T_5^{-1}]$. Therefore, there exists $i \in \mathbb{N}$ such that $yT_5^i \in \langle \Omega'_2 \rangle_{A^{(5)}}$. This implies that $yT_5^i = \Omega'_2 v$, for some $v \in A^{(5)}$. Since

$$\begin{split} A^{(5)}[T_5^{-1}] &= A^{(6)}[T_5^{-1}], \text{ there exists } j \in \mathbb{N} \text{ such that } vT_5^j = v', \text{ for some } v' \in A^{(6)}. \text{ It} \\ \text{follows that } yT_5^{i+j} &= \Omega'_2 vT_5^j = \Omega'_2 v'. \text{ This implies that } yT_4^{\delta} &= \Omega'_2 v', \text{ where } \delta = i+j. \text{ Set} \\ S &:= \{s \in \mathbb{N} \mid \exists v' \in A^{(6)} : yT_5^s = \Omega'_2 v'\}. \text{ Since } \delta \in S, \text{ we have that } S \neq \emptyset. \text{ Let } s = s_0 \\ \text{be the minimum element of } S \text{ such that } yT_5^{s_0} = \Omega'_2 v'. \text{ Note, } \Omega'_2 T_6 = \Omega_2 \text{ in } A^{(6)}. \text{ Given} \\ \text{that } \Omega_2 \text{ is central in } A^{(6)}, \text{ and } T_6 \text{ is normal in } A^{(6)}, \text{ we must have } \Omega'_2 \text{ as a normal element} \\ \text{in } A^{(6)}. \text{ Therefore, there exists } v'' \in A^{(6)} \text{ such that } yT_5^{s_0} = \Omega'_2 v' = v''\Omega'_2. \text{ Now, } A^{(6)} \text{ can} \\ \text{be viewed as a free left } \mathbb{C}\langle E_{1,6}, E_{2,6}, E_{3,6}, T_4, T_6\rangle - \text{module with basis } \left(T_5^{\xi}\right)_{\xi \in \mathbb{N}}. \text{ One can} \\ \text{write } y = \sum_{\xi=0}^n \alpha_\xi T_5^{\xi} \text{ and } v'' = \sum_{\xi=0}^n \beta_\xi T_5^{\xi}, \text{ where } \alpha_\xi, \beta_\xi \in \mathbb{C}\langle E_{1,6}, E_{2,6}, E_{3,6}, E_{4,6}, T_6\rangle. \\ \text{It follows from } yT_5^{s_0} = v''\Omega_2' \text{ that } \sum_{\xi=0}^n \alpha_\xi T_5^{\xi+s_0} = \sum_{\xi=0}^n \beta_\xi T_5^{\xi}\Omega_2' = \sum_{\xi=0}^n q^\bullet \beta_\xi \Omega_2' T_5^{\xi} \\ (\text{note, } T_5\Omega_2' = q^{-3}\Omega_2'T_5). \text{ Suppose that } s_0 > 0. \text{ Then, identifying the constant coefficients, we have that } q^\bullet \beta_0 \Omega_2' = 0. \text{ Hence, } \beta_0 = 0, \text{ since } q^\bullet \Omega_2' \neq 0. \text{ One can therefore} \\ \text{write } v'' = \sum_{\xi=1}^n \beta_\xi T_5^{\xi}. \text{ Returning to } yT_5^{s_0} = v''\Omega_2', \text{ we have that } yT_5^{s_0} = \sum_{\xi=1}^n \beta_\xi T_5^{\xi}\Omega_2' = \sum_{\xi=1}^n \beta_\xi T_5^{\xi}\Omega_2' = \Omega_2' 2 \sum_{\xi=1}^n q^\bullet \beta_\xi T_5^{\xi}\Omega_1' = \Omega_2' 2 \sum_{\xi=1}^n q^\bullet \beta_\xi T_5^{\xi} . \text{ This implies that } yT_5^{s_0-1} = \Omega_2'w, \text{ where } w = \sum_{\xi=1}^n q^\bullet \beta_\xi T_5^{\xi}\Omega_1' \in A^{(6)}. \\ \text{Consequently, } s_0 - 1 \in S, \text{ a contradiction! Therefore, } s_0 = 0 \text{ and } y = \Omega_2'v' \in \langle \Omega_2' \rangle = \langle E_{2,6}T_4 + bE_{3,6}^{3} \rangle. \text{ Hence, } \langle \Omega_2' \rangle_{A^{(5)}} [T_5^{-1}] \cap A^{(6)} \subseteq \langle E_{2,6}T_4 + bE_{3,6}^{3} \rangle \text{ as desired.} \end{cases}$$

5. Note, $\Omega'_2 = E_{2,6}T_4 + bE_{3,6}^3 = \Omega_2 T_6^{-1}$ in $A[T_6^{-1}]$. We want to show that $\langle \Omega'_2 \rangle_{A^{(6)}}[T_6^{-1}] \cap A = \langle \Omega_2 \rangle_A$. Obviously, $\langle \Omega_2 \rangle_A \subseteq \langle \Omega'_2 \rangle_{A^{(6)}}[T_6^{-1}] \cap A$. We establish the reverse inclusion. Let $y \in \langle \Omega'_2 \rangle_{A^{(6)}}[T_6^{-1}] \cap A$. This implies that $y \in \langle \Omega'_2 \rangle_{A^{(6)}}[T_6^{-1}]$. There exists $i \in \mathbb{N}$ such that $yT_6^i \in \langle \Omega'_2 \rangle_{A^{(6)}}$. Hence, $yT_6^i = \Omega'_2 v$, for some $v \in A^{(6)}$. Furthermore, since $A^{(6)}[T_6^{-1}] = A[T_6^{-1}]$, there exist $j \in \mathbb{N}$ such that $vT_6^i = v'$, for some $v \in A^{(6)}$. Furthermore, since $A^{(6)}[T_6^{-1}] = A[T_6^{-1}]$, there exist $j \in \mathbb{N}$ such that $vT_6^i = v'$, for some $v' \in A$. It follows from $yT_6^i = \Omega'_2 v$ that $yT_6^{i+j} = \Omega'_2 vT_6^j = \Omega'_2 v' = \Omega_2 T_6^{-1} v'$. The multiplicative system generated by T_6 satisfies the Ore condition, hence, there exists $k \in \mathbb{N}$ and $v'' \in A$ such that $T_6^{-1}v' = v''T_6^k$. It follows that $yT_6^{i+j} = \Omega_2 T_6^{-1}v' = \Omega_2 v''T_6^{-k}$. Hence, $yT_6^\delta = \Omega_2 v''$, where $\delta = i + j + k$. Set $S := \{s \in \mathbb{N} \mid \exists v'' \in A : yT_6^s = \Omega_2 v''\}$. Since $\delta \in S$, we have that $S \neq \emptyset$. Let $s = s_0$ be the minimum element of S such that $yT_6^{i_6} = \Omega_2 v''$. Now, A can be viewed as a free $\mathbb{C}\langle E_1, E_2, E_3, E_4, T_5\rangle$ -module with basis $(T_6^{\xi})_{\xi \in \mathbb{N}}$. One can write $y = \sum_{\xi=0}^n \alpha_\xi T_6^\xi$ and $v'' = \sum_{\xi=0}^n \alpha_\xi T_6^\xi$, where $\alpha_\xi, \beta_\xi \in \mathbb{C}\langle E_1, E_2, E_3, E_4, T_5\rangle$. With this, $yT_6^{s_0} = \Omega_2 v''$ implies that $\sum_{\xi=0}^n \alpha_\xi T_6^{\xi+s_0} = \Omega_2 v''$.

 $\sum_{\xi=0}^{n} \beta_{\xi} \Omega_{2} T_{6}^{\xi}.$ From Subsection 2.2.3, one can write Ω_{2} as $\Omega_{2} = \gamma_{1} T_{6} + \gamma_{2}$, where $\gamma_{1} = E_{2} E_{4} + b E_{3}^{3}$ and $\gamma_{2} = b E_{2} T_{5}^{3} + b' E_{3}^{2} E_{5}^{2} + c' E_{3} E_{4} T_{5} + d' E_{4}^{2}.$ It follows that $\sum_{\xi=0}^{n} \alpha_{\xi} T_{6}^{\xi+s_{0}} = \sum_{\xi=0}^{n} \beta_{\xi} \gamma_{1} T_{6}^{\xi+1} + \sum_{\xi=0}^{n} \beta_{\xi} \gamma_{2} T_{6}^{\xi}.$ Suppose that $s_{0} > 0$. Then, identifying the constant coefficients, we have that $\beta_{0} \gamma_{2} = 0$. Hence, $\beta_{0} = 0$, since $\gamma_{2} \neq 0$. One can therefore write $v'' = \sum_{\xi=1}^{n} \beta_{\xi} T_{6}^{\xi}.$ Returning to $y T_{6}^{s_{0}} = \Omega_{2} v''$, we have that $y T_{6}^{s_{0}} = \Omega_{2} \sum_{\xi=1}^{n} \beta_{\xi} T_{6}^{\xi}.$ This implies that $y T_{6}^{s_{0}-1} = \Omega_{2} w$, where $w = \sum_{\xi=1}^{n} \beta_{\xi} T_{6}^{\xi-1} \in A$. Consequently, $s_{0} - 1 \in S$, a contradiction! Therefore, $s_{0} = 0$ and $y = \Omega_{2} v'' \in \langle \Omega_{2} \rangle_{A}.$ Hence, $\langle \Omega_{2}' \rangle_{A^{(6)}} [T_{6}^{-1}] \cap A \subseteq \langle \Omega_{2} \rangle_{A}$ as desired.

2.4 Height two maximal ideals of $U_q^+(G_2)$

This section focuses on studying the height two maximal ideals of $A = U_q^+(G_2)$. First and foremost, we will begin by describing the \mathcal{H} -invariant prime ideals of A, and then proceed to study the \mathcal{H} -strata corresponding to $\langle 0 \rangle$, $\langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$. We will conclude by describing the height two maximal ideals of A.

2.4.1 \mathcal{H} -**Spec**(A). Recall from Section 2.1 that $A = \mathbb{C}[E_1][E_2; \sigma_2][E_3; \sigma_3, \delta_3] \cdots [E_6; \sigma_6, \delta_6]$. The torus $\mathcal{H} := (\mathbb{C}^*)^2$ acts rationally on A via $(\alpha_1, \alpha_6) \cdot E_i = \alpha_i E_i$; with i = 1, 6; for all $(\alpha_1, \alpha_6) \in \mathcal{H}$. Using the defining relations of A, one can easily verify that

$$(\alpha_1, \alpha_6) \cdot E_2 = \alpha_1^3 \alpha_6 E_2 \qquad (\alpha_1, \alpha_6) \cdot E_3 = \alpha_1^2 \alpha_6 E_3$$
$$(\alpha_1, \alpha_6) \cdot E_4 = \alpha_1^3 \alpha_6^2 E_4 \qquad (\alpha_1, \alpha_6) \cdot E_5 = \alpha_1 \alpha_6 E_5.$$

We now proceed to describe the set $\mathcal{H}-\operatorname{Spec}(A)$. In Section 1.1, we studied the Weyl group \mathscr{W} of G_2 . It follows from Remark 1.6.7 that the poset of $\mathcal{H}-\operatorname{Spec}(A)$ is isomorphic to the poset of \mathscr{W} . For each $\varepsilon \in \mathscr{W}$, one can associate a unique element of $\mathcal{H}-\operatorname{Spec}(A)$, which we denote by $K(\varepsilon)$. As a result, $\mathcal{H}-\operatorname{Spec}(A) := \{K(\varepsilon) \mid \forall \varepsilon \in \mathscr{W}\}$. We have that $|\mathcal{H}-\operatorname{Spec}(A)| = |\mathscr{W}| = 12$. One can therefore partition $\operatorname{Spec}(A)$ into a disjoint union of strata as follows:

$$\operatorname{Spec}(A) = \bigsqcup_{\varepsilon \in \mathscr{W}} \operatorname{Spec}_{K(\varepsilon)}(A).$$

Moreover; for $\varepsilon, \varepsilon' \in \mathscr{W}$ with $\varepsilon \preceq \varepsilon'$, we have that $K(\varepsilon) \subset K(\varepsilon')$, where \preceq is the Bruhat order [44, Theorem 3.8]. From Section 1.1, we have that $\omega_0 := s_1 s_2 s_1 s_2 s_1 s_2 = s_2 s_1 s_2 s_1 s_2 s_1$ in \mathscr{W} . The posets of \mathscr{W} and $\mathcal{H}-\operatorname{Spec}(A)$ are seen in Figure 2.1.



Figure 2.1: Posets of \mathcal{W} and $\mathcal{H}-\operatorname{Spec}(A)$.

Recall from Subsection 2.2.3 that \bar{A} is a quantum affine space. The map $\psi := \psi_2 \circ \cdots \circ \psi_6$: Spec $(A) \hookrightarrow$ Spec (\bar{A}) is a canonical embedding. This canonical embedding restricts to the \mathcal{H} -invariant prime spectrum. That is, $\psi : \mathcal{H}$ -Spec $(A) \hookrightarrow \mathcal{H}$ -Spec (\bar{A}) . Let W be the set of all the subsets of $\{1, \cdots, 6\}$. For each $w \in W$, $K_w := \langle T_i \mid i \in w \rangle$ is a \mathcal{H} -invariant prime ideal of \bar{A} (Proposition 1.6.9). We therefore have that \mathcal{H} -Spec $(\bar{A}) := \{K_w \mid w \in W\}$. Observe that the set $\{\langle T_i \rangle \mid 1 \leq i \leq 6\}$ is the set of all the height one \mathcal{H} -invariant prime ideals of \bar{A} . Since ψ : Spec $(A) \hookrightarrow$ Spec (\bar{A}) , one can easily deduce from Figure 2.1 that there are only two height one \mathcal{H} -invariant prime ideals of A. From Lemmas 2.3.1 and 2.3.2, we have that $\langle T_1 \rangle = \psi(\langle \Omega_1 \rangle)$ and $\langle T_2 \rangle = \psi(\langle \Omega_2 \rangle)$. Since ψ preserves the height of a prime ideal, we have the following lemma.

- **2.4.2 Lemma.** 1. $\langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$ are the only height one \mathcal{H} -invariant prime ideals of A.
 - 2. Every non-zero \mathcal{H} -invariant prime ideal of A contains either $\langle \Omega_1 \rangle$ or $\langle \Omega_2 \rangle$ as a result of Figure 2.1.

2.4.3 \mathcal{H} -strata corresponding to $\langle 0 \rangle$, $\langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$. In this subsection, we aim to find the \mathcal{H} -strata corresponding to $\langle 0 \rangle$, $\langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$. We state and prove the results in Propositions 2.4.4, 2.4.5 and 2.4.6 using strategies similar to [33, Propositions 2.3 and 2.4]. Note, in this subsection, all ideals in $A = U_q^+(G_2)$ will simply be written as $\langle \Theta \rangle$, where $\Theta \in A$. However, if we want to refer to an ideal in any other algebra, say R, then that ideal will be written as $\langle \Theta \rangle_R$, where in this case, $\Theta \in R$.

2.4.4 Proposition. Let \mathcal{P} be the set of those unitary irreducible polynomials $P(\Omega_1, \Omega_2) \in \mathbb{C}[\Omega_1, \Omega_2]$ with $P(\Omega_1, \Omega_2) \neq \Omega_1$ and $P(\Omega_1, \Omega_2) \neq \Omega_2$. Then, $\text{Spec}_{\langle 0 \rangle}(A) = \{\langle 0 \rangle\} \cup \{\langle P(\Omega_1, \Omega_2) \rangle \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle \mid \alpha, \beta \in \mathbb{C}^*\}.$

Proof. We claim that $\operatorname{Spec}_{\langle 0 \rangle}(A) = \{Q \in \operatorname{Spec}(A) \mid \Omega_1, \Omega_2 \notin Q\}$. To establish this claim, let us assume that this is not the case. That is, suppose there exists $Q \in \operatorname{Spec}_{\langle 0 \rangle}(A)$ such that $\Omega_1, \Omega_2 \in Q$; then the product $\Omega_1\Omega_2$ which is an \mathcal{H} -eigenvector belongs to Q. Consequently, $\Omega_1\Omega_2 \in \bigcap_{h\in\mathcal{H}}h \cdot Q = \langle 0 \rangle$, a contradiction. Hence, we have shown that $\operatorname{Spec}_{\langle 0 \rangle}(A) \subseteq \{Q \in \operatorname{Spec}(A) \mid \Omega_1, \Omega_2 \notin Q\}$. Conversely, suppose that $Q \in \operatorname{Spec}(A)$ such that $\Omega_1, \Omega_2 \notin Q$, then $\bigcap_{h\in\mathcal{H}}h \cdot Q$ is an \mathcal{H} -invariant prime ideal of A, which contains neither Ω_1 nor Ω_2 . Obviously, the only possibility for $\bigcap_{h\in\mathcal{H}}h \cdot Q = \langle 0 \rangle$. Hence, $Q \in \operatorname{Spec}_{\langle 0 \rangle}(A)$. Therefore, $\{Q \in \operatorname{Spec}(A) \mid \Omega_1, \Omega_2 \notin Q\} \subseteq \operatorname{Spec}_{\langle 0 \rangle}(A)$. This confirms our claim.

Since $\Omega_1, \Omega_2 \in Z(A)$, we have that the set $\{\Omega_1^i \Omega_2^j \mid i, j \in \mathbb{N}\}$ is a right denominator set in the noetherian domain A. One can now localize A as $R := A[\Omega_1^{-1}, \Omega_2^{-1}]$. Let $Q \in$ $\operatorname{Spec}_{\langle 0 \rangle}(A)$, the map $\phi : Q \longrightarrow Q[\Omega_1^{-1}, \Omega_2^{-1}]$ is an increasing bijection from $\operatorname{Spec}_{\langle 0 \rangle}(A)$ onto $\operatorname{Spec}(R)$. Since Ω_1 and Ω_2 are \mathcal{H} -eigenvectors, and \mathcal{H} acts on A, we have that \mathcal{H} also acts on R. Let us verify that R is \mathcal{H} -simple before we describe Spec(R). Now, ϕ still induces a bijection between the set of those \mathcal{H} -invariant prime ideals of $\text{Spec}_{\langle 0 \rangle}(A)$ and the set of \mathcal{H} -invariant prime ideals of Spec(R). It is already known that the set of \mathcal{H} -invariant prime ideals of A that neither contains Ω_1 nor Ω_2 consists only of the zero ideal $\{\langle 0 \rangle\}$ (Lemma 2.4.2(2)). This implies that $\langle 0 \rangle_R$ is the only \mathcal{H} -invariant prime ideal of R. Every \mathcal{H} -invariant proper ideal of R is contained in an \mathcal{H} -invariant prime ideal of R[17, §3.1(v)]. Therefore, $\langle 0 \rangle_R$ is the only unique \mathcal{H} -invariant proper ideal of R. This confirms that R is \mathcal{H} -simple.

We proceed to describe Spec(R) and Spec₍₀₎(A). We deduce from [7, Exercise II.3.A] that the action of \mathcal{H} on R is rational. This rational action coupled with R being \mathcal{H} -simple implies that the extension and contraction maps provide a mutually inverse bijection between Spec(R) and Spec(Z(R)) [7, Corollary II.3.9]. From Lemma 2.2.6, $Z(A) = \mathbb{C}[\Omega_1, \Omega_2]$, and so $Z(R) = \mathbb{C}[\Omega_1^{\pm 1}, \Omega_2^{\pm 1}]$. Since \mathbb{C} is algebraically closed, we have that $\operatorname{Spec}(Z(R)) = \{\langle 0 \rangle_{Z(R)}\} \cup \{\langle P(\Omega_1, \Omega_2) \rangle_{Z(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_{Z(R)} \mid \alpha, \beta \in \mathbb{C}^*\}$. Since there is an inverse bijection between $\operatorname{Spec}(Z(R))$ and $\operatorname{Spec}(Z(R))$, and also R is \mathcal{H} -simple, one can recover $\operatorname{Spec}(R)$ from $\operatorname{Spec}(Z(R))$ as follows: $\operatorname{Spec}(R) = \{\langle 0 \rangle_R\} \cup \{\langle P(\Omega_1, \Omega_2) \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid \alpha, \beta \in \mathbb{C}^*\}$. It follows that $\operatorname{Spec}_{(0)}(A) = \{\langle 0 \rangle_R \cap A\} \cup \{\langle P(\Omega_1, \Omega_2) \rangle_R \cap A \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid \alpha, \beta \in \mathbb{C}^*\}$.

Undoubtedly, $\langle 0 \rangle_R \cap A = \langle 0 \rangle$. We now have to show that $\langle P(\Omega_1, \Omega_2) \rangle_R \cap A = \langle P(\Omega_1, \Omega_2) \rangle$, $\forall P(\Omega_1, \Omega_2) \in \mathcal{P}$, and $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \cap A = \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$, $\forall \alpha, \beta \in \mathbb{C}^*$ to complete the proof.

Fix $P(\Omega_1, \Omega_2) \in \mathcal{P}$. Observe that $\langle P(\Omega_1, \Omega_2) \rangle \subseteq \langle P(\Omega_1, \Omega_2) \rangle_R \cap A$. To show the reverse inclusion, let $y \in \langle P(\Omega_1, \Omega_2) \rangle_R \cap A$. This implies that $y = dP(\Omega_1, \Omega_2)$, where $d \in R$, since $y \in \langle P(\Omega_1, \Omega_2) \rangle_R$. Also, $d \in R$ implies that there exist $i, j \in \mathbb{N}$ such that $d = a\Omega_1^{-i}\Omega_2^{-j}$, where $a \in A$. Therefore, $y = a\Omega_1^{-i}\Omega_2^{-j}P(\Omega_1, \Omega_2)$, which implies that $y\Omega_1^i\Omega_2^j = aP(\Omega_1, \Omega_2)$. Choose $(i, j) \in \mathbb{N}^2$ minimal (in the lexicographic order on \mathbb{N}^2) such that the equality holds. Without loss of generality, suppose that i > 0, then $aP(\Omega_1, \Omega_2) \in \langle \Omega_1 \rangle$. Given that $\langle \Omega_1 \rangle$ is a completely prime ideal, it implies that $a \in \langle \Omega_1 \rangle$ or $P(\Omega_1, \Omega_2) \in \langle \Omega_1 \rangle$. Since $P(\Omega_1, \Omega_2) \in \mathcal{P}$, it implies that $P(\Omega_1, \Omega_2) \notin \langle \Omega_1 \rangle$, hence $a \in \langle \Omega_1 \rangle$. This further implies that $a = t\Omega_1$, where $t \in A$. Returning to $y\Omega_1^i\Omega_2^j = aP(\Omega_1, \Omega_2)$, we have that $y\Omega_1^i\Omega_2^j = t\Omega_1P(\Omega_1, \Omega_2)$. Therefore, $y\Omega_1^{i-1}\Omega_2^j = tP(\Omega_1, \Omega_2)$. This clearly contradicts the minimality of (i, j), hence (i, j) = (0, 0), and $y = aP(\Omega_1, \Omega_2) \in \langle P(\Omega_1, \Omega_2) \rangle$. Consequently, $\langle P(\Omega_1, \Omega_2) \rangle_R \cap A = \langle P(\Omega_1, \Omega_2) \rangle$ for all $P(\Omega_1, \Omega_2) \in \mathcal{P}$ as desired.

Similarly, we show that $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \cap A = \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle; \forall \alpha, \beta \in \mathbb{C}^*$. Fix $\alpha, \beta \in \mathbb{C}^*$. Observe that $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle \subseteq \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \cap A$. We establish the reverse inclusion. Let $y \in \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \cap A$. Since $y \in \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R$, there exist $i, j \in \mathbb{N}$ such that $y\Omega_1^i\Omega_2^j = m(\Omega_1 - \alpha) + n(\Omega_2 - \beta)$, where $m, n \in A$. Choose $(i, j) \in \mathbb{N}^2$ minimal (in the lexicographic order on \mathbb{N}^2) such that the equality holds. Without loss of generality, suppose that i > 0 and let $f : A \longrightarrow A/\langle \Omega_2 - \beta \rangle$ be a canonical surjection. We have that $f(y)f(\Omega_1)^if(\Omega_2)^j = f(m)f(\Omega_1 - \alpha)$. It follows that $f(m)f(\Omega_1 - \alpha) \in \langle f(\Omega_1) \rangle$. Observe that $f(\Omega_1 - \alpha) \notin \langle f(\Omega_1) \rangle$, hence $f(m) \in \langle f(\Omega_1) \rangle$. Therefore, $\exists \lambda \in A$ such that $f(m) = f(\lambda)f(\Omega_1)$. Consequently, $f(y)f(\Omega_1)^if(\Omega_2)^j = f(\lambda)f(\Omega_1)f(\Omega_1 - \alpha)$. Since $f(\Omega_1) \neq 0$, it implies that $f(y)f(\Omega_1)^{i-1}f(\Omega_2)^j = f(\lambda)f(\Omega_1 - \alpha)$. Therefore, $y\Omega_1^{i-1}\Omega_2^j = \lambda(\Omega_1 - \alpha) + \lambda'(\Omega_2 - \beta)$ for some $\lambda' \in A$. This contradicts the minimality of (i, j). Hence, (i, j) = (0, 0) and so $y = m(\Omega_1 - \alpha) + n(\Omega_2 - \beta) \in \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$. In conclusion, $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \cap A = \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle, \forall \alpha, \beta \in \mathbb{C}^*$.

2.4.5 Proposition. Spec $_{\langle \Omega_1 \rangle}(A) = \{ \langle \Omega_1 \rangle \} \cup \{ \langle \Omega_1, \Omega_2 - \beta \rangle \mid \beta \in \mathbb{C}^* \}.$

Proof. We claim that $\operatorname{Spec}_{\langle\Omega_1\rangle}(A) = \{Q \in \operatorname{Spec}(A) \mid \Omega_1 \in Q \text{ and } \Omega_2 \notin Q\}$. To establish this claim, let us assume that this is not the case. That is, suppose there exists $Q \in \operatorname{Spec}_{\langle\Omega_1\rangle}(A)$ such that $\Omega_2 \in Q$. Since Ω_2 is an \mathcal{H} -eigenvector, $\Omega_2 \in \bigcap_{h \in \mathcal{H}} h \cdot Q = \langle\Omega_1\rangle$, a contradiction. Hence, we have shown that $\operatorname{Spec}_{\langle\Omega_1\rangle}(A) \subseteq \{Q \in \operatorname{Spec}(A) \mid \Omega_1 \in Q \text{ and } \Omega_2 \notin Q\}$. Conversely, suppose that $Q \in \operatorname{Spec}(A)$ and $\Omega_1 \in Q$ but $\Omega_2 \notin Q$. Then $\bigcap_{h \in \mathcal{H}} h \cdot Q$ is an \mathcal{H} -invariant prime ideal of A, which contains Ω_1 but does not contain

 Ω_2 . The only possibility for $\bigcap_{h \in \mathcal{H}} h \cdot Q$ is the ideal $\langle \Omega_1 \rangle$. Hence, $Q \in \operatorname{Spec}_{\langle \Omega_1 \rangle}(A)$. Therefore, $\{Q \in \operatorname{Spec}(A) \mid \Omega_1 \in Q \text{ and } \Omega_2 \notin Q\} \subseteq \operatorname{Spec}_{\langle \Omega_1 \rangle}(A)$. This confirms our claim.

Let $\Lambda : A \longrightarrow A/\langle \Omega_1 \rangle$ be a canonical surjection. Since A is a noetherian domain and $\langle \Omega_1 \rangle$ is a completely prime ideal, one can also deduce that $A/\langle \Omega_1 \rangle$ is a noetherian domain. Given that $\Omega_2 \in Z(A)$, we have that $\Lambda(\Omega_2) \neq 0 \in Z(A/\langle \Omega_1 \rangle)$. Therefore, the set $\{\Lambda(\Omega_2)^i \mid i \in \mathbb{N}\}$ is a right denominator set in $A/\langle \Omega_1 \rangle$. One can therefore localize $A/\langle \Omega_1 \rangle$ at this denominator set as $R := \frac{A}{\langle \Omega_1 \rangle} [\Lambda(\Omega_2)^{-1}]$. Given that $Q \in \operatorname{Spec}_{\langle \Omega_1 \rangle}(A)$, the map $\phi : Q \longrightarrow \frac{Q}{\langle \Omega_1 \rangle} [\Lambda(\Omega_2)^{-1}]$ is an increasing bijection from $\operatorname{Spec}_{\langle \Omega_1 \rangle}(A)$ onto $\operatorname{Spec}(R)$.

Since $\langle \Omega_1 \rangle$ is an \mathcal{H} -invariant prime ideal, $\Lambda(\Omega_2)$ is an \mathcal{H} -eigenvector, and \mathcal{H} acts on A, we have that \mathcal{H} also acts on R. Let us verify that R is \mathcal{H} -simple before we describe $\operatorname{Spec}(R)$. Now, ϕ still induces a bijection between the set of those \mathcal{H} -invariant prime ideals of $\operatorname{Spec}_{\langle\Omega_1\rangle}(A)$ and the set of \mathcal{H} -invariant prime ideals of $\operatorname{Spec}(R)$. The set of \mathcal{H} -invariant prime ideals of A that contains Ω_1 but does not contain Ω_2 is $\{\langle\Omega_1\rangle\}$, implying that $\langle 0 \rangle_R$ is the only \mathcal{H} -invariant prime ideal of R (note, $\Omega_1 = 0$ in R). Every \mathcal{H} -invariant proper ideal of R is contained in an \mathcal{H} -invariant prime ideal of R[17, §3.1(v)]. Therefore, $\langle 0 \rangle_R$ is the only unique \mathcal{H} -invariant proper ideal of R. This confirms that R is \mathcal{H} -simple.

We proceed to describe the Spec(R) and Spec_(Ω_1)(A). We deduce from [7, Exercise II.3.A] that the action of \mathcal{H} on R is rational. This rational action coupled with R being \mathcal{H} -simple implies that the extension and contraction maps provide a mutually inverse bijection between Spec(R) and Spec(Z(R)) [7, Corollary II.3.9]. One can deduce that $Z(A/\langle\Omega_1\rangle) = \mathbb{C}[\Lambda(\Omega_2)]$, and so $Z(R) = \mathbb{C}[\Lambda(\Omega_2)^{\pm 1}]$. Since \mathbb{C} is algebraically closed, we have that Spec(Z(R)) = { $\langle 0 \rangle_{Z(R)}$ } \cup { $\langle \Lambda(\Omega_2) - \beta \rangle_{Z(R)} | \beta \in \mathbb{C}^*$ }. We now recover Spec(R) from Spec(Z(R)) as follows: Spec(R) = { $\langle 0 \rangle_R$ } \cup { $\langle \Lambda(\Omega_2) - \beta \rangle_R | \beta \in \mathbb{C}^*$ }, since there is a mutually inverse bijection between Spec(R) and Spec(Z(R)), and also R is \mathcal{H} -simple. Moreover, since ϕ is an increasing bijection from Spec_(Ω_1)(A) onto $\operatorname{Spec}(R), \text{ we have that } \operatorname{Spec}_{\langle \Omega_1 \rangle}(A) = \{ \phi^{-1}(\langle 0 \rangle_R) \} \cup \{ \phi^{-1}(\langle \Lambda(\Omega_2) - \beta \rangle_R) \mid \beta \in \mathbb{C}^* \}.$

Naturally, $\phi^{-1}(\langle 0 \rangle_R) = \langle \Omega_1 \rangle$. We now show that $\phi^{-1}(\langle \Lambda(\Omega_2) - \beta \rangle_R) = \langle \Omega_1, \Omega_2 - \beta \rangle$. Fix $\beta \in \mathbb{C}^*$. Observe that $\langle \Omega_1, \Omega_2 - \beta \rangle \subseteq \phi^{-1}(\langle \Lambda(\Omega_2) - \beta \rangle_R)$. To show the reverse inclusion, let $x \in \phi^{-1}(\langle \Lambda(\Omega_2) - \beta \rangle_R)$. Then $\Lambda(x) \in \langle \Lambda(\Omega_2) - \beta \rangle_R$. This implies that $\Lambda(x) = d(\Lambda(\Omega_2) - \beta)$, where $d \in R$. Also, $d \in R$ implies that there exists $t \ge 0$ and $a \in A/\langle \Omega_1 \rangle$ such that $d = a\Lambda(\Omega_2)^{-t}$. As a result, $\Lambda(x) = d(\Lambda(\Omega_2) - \beta) = a\Lambda(\Omega_2)^{-t}(\Lambda(\Omega_2) - \beta)$. Hence, $\Lambda(x)\Lambda(\Omega_2)^t = a(\Lambda(\Omega_2) - \beta)$. Choose t minimum such that the equality holds. If t > 0, then $\Lambda(x)\Lambda(\Omega_2)^t = a\Lambda(\Omega_2) - a\beta$. One can therefore write $a = \frac{1}{\beta}\Lambda(\Omega_2)(a - \Lambda(x)\Lambda(\Omega_2)^{t-1})$. Set $b := \frac{1}{\beta}(a - \Lambda(x)\Lambda(\Omega_2)^{t-1})$, then $a = b\Lambda(\Omega_2)$. This implies that $\Lambda(x)\Lambda(\Omega_2)^t = a(\Lambda(\Omega_2) - \beta) = b\Lambda(\Omega_2)(\Lambda(\Omega_2) - \beta)$ which further implies that $\Lambda(x)\Lambda(\Omega_2)^{t-1} = b(\Lambda(\Omega_2) - \beta)$. This contradicts the minimality of t, and so t = 0. Therefore, $\Lambda(x) = a(\Lambda(\Omega_2) - \beta) \in \langle \Lambda(\Omega_2) - \beta \rangle_R$.

2.4.6 Proposition. Spec_{$(\Omega_2)}(A) = { (\Omega_2) } \cup { (\Omega_1 - \alpha, \Omega_2) | \alpha \in \mathbb{C}^* }.$ </sub>

Proof. We claim that $\operatorname{Spec}_{\langle\Omega_2\rangle}(A) = \{Q \in \operatorname{Spec}(A) \mid \Omega_1 \notin Q \text{ and } \Omega_2 \in Q\}$. To establish this claim, let us assume that this is not the case. That is, suppose there exists $Q \in \operatorname{Spec}_{\langle\Omega_2\rangle}(A)$ such that $\Omega_1 \in Q$. Since Ω_1 is an \mathcal{H} -eigenvector, $\Omega_1 \in \bigcap_{h \in \mathcal{H}} h \cdot Q = \langle\Omega_2\rangle$, a contradiction. Hence, we have shown that $\operatorname{Spec}_{\langle\Omega_2\rangle}(A) \subseteq \{Q \in \operatorname{Spec}(A) \mid \Omega_1 \notin Q \text{ and } \Omega_2 \in Q\}$. Conversely, suppose that $Q \in \operatorname{Spec}(A)$ and $\Omega_2 \in Q$ but $\Omega_1 \notin Q$, then $\bigcap_{h \in \mathcal{H}} h \cdot Q$ is an \mathcal{H} -invariant prime ideal of A, which contains Ω_2 but does not contain Ω_1 . The only possibility for $\bigcap_{h \in \mathcal{H}} h \cdot Q$ is the ideal $\langle\Omega_2\rangle$. Hence, $Q \in \operatorname{Spec}_{\langle\Omega_2\rangle}(A)$. Therefore, $\{Q \in \operatorname{Spec}(A) \mid \Omega_1 \notin Q \text{ and } \Omega_2 \in Q\} \subseteq \operatorname{Spec}_{\langle\Omega_2\rangle}(A)$. This confirms our claim.

Let $\Lambda : A \longrightarrow A/\langle \Omega_2 \rangle$ be a canonical surjection. Since A is a noetherian domain and $\langle \Omega_2 \rangle$ is a completely prime ideal, one can also deduce that $A/\langle \Omega_2 \rangle$ is a noetherian domain. Given that $\Omega_1 \in Z(A)$, we have $\Lambda(\Omega_1) \in Z(A/\langle \Omega_2 \rangle)$. It follows that the set $\{\Lambda(\Omega_1)^i \mid i \in \mathbb{N}\}$ is a right denominator set in $A/\langle \Omega_2 \rangle$. One can therefore localize $A/\langle \Omega_2 \rangle$ as $R := \frac{A}{\langle \Omega_2 \rangle} [\Lambda(\Omega_1)^{-1}]$. Given $Q \in \operatorname{Spec}_{\langle \Omega_2 \rangle}(A)$, we have a map $\phi : Q \longrightarrow$ $\frac{Q}{\langle \Omega_2 \rangle} [\Lambda(\Omega_1)^{-1}] \text{ which is an increasing bijection from } \operatorname{Spec}_{\langle \Omega_2 \rangle}(A) \text{ onto } \operatorname{Spec}(R).$

Similar to the proof of Proposition 2.4.5, one can show that R is \mathcal{H} -simple and that the extension and contraction maps provide a mutually inverse bijection between $\operatorname{Spec}(R)$ and $\operatorname{Spec}(Z(R))$ [7, Corollary II.3.9]. We also have that $Z(A/\langle \Omega_2 \rangle) = \mathbb{C}[\Lambda(\Omega_1)]$, and so $Z(R) = \mathbb{C}[\Lambda(\Omega_1)^{\pm 1}]$. Since \mathbb{C} is algebraically closed, we have that $\operatorname{Spec}(Z(R)) =$ $\{\langle 0 \rangle_{Z(R)}\} \cup \{\langle \Lambda(\Omega_1) - \alpha \rangle_{Z(R)} \mid \alpha \in \mathbb{C}^*\}$. One can recover $\operatorname{Spec}(R)$ from $\operatorname{Spec}(Z(R))$ as follows: $\operatorname{Spec}(R) = \{\langle 0 \rangle_R\} \cup \{\langle \Lambda(\Omega_1) - \alpha \rangle_R \mid \alpha \in \mathbb{C}^*\}$, since there is a mutual inverse bijection between $\operatorname{Spec}(R)$ and $\operatorname{Spec}(Z(R))$, and also R is \mathcal{H} -simple. Moreover, since ϕ is an increasing bijection between $\operatorname{Spec}_{\langle \Omega_2 \rangle}(A)$ and $\operatorname{Spec}(R)$, we have that $\operatorname{Spec}_{\langle \Omega_2 \rangle}(A) =$ $\{\phi^{-1}(\langle 0 \rangle_R)\} \cup \{\phi^{-1}(\langle \Lambda(\Omega_1) - \alpha \rangle_R) \mid \alpha \in \mathbb{C}^*\}$.

Naturally, $\phi^{-1}(\langle 0 \rangle_R) = \langle \Omega_2 \rangle$. We now show that $\phi^{-1}(\langle \Lambda(\Omega_1) - \alpha \rangle_R) = \langle \Omega_1 - \alpha, \Omega_2 \rangle$. Fix $\alpha \in \mathbb{C}^*$. Observe that $\langle \Omega_1 - \alpha, \Omega_2 \rangle \subseteq \phi^{-1}(\langle \Lambda(\Omega_1) - \alpha \rangle_R)$. To show the reverse inclusion, let $x \in \phi^{-1}(\langle \Lambda(\Omega_1) - \alpha \rangle_R)$. Then, $\Lambda(x) \in \langle \Lambda(\Omega_1) - \alpha \rangle_R$. Let $d \in R$, then there exists $t \geq 0$ and $a \in A/\langle \Omega_2 \rangle$ such that $d = a\Lambda(\Omega_1)^{-t}$. Given that $\Lambda(x) \in \langle \Lambda(\Omega_1) - \alpha \rangle_R$, it follows that $\Lambda(x) = d(\Lambda(\Omega_1) - \alpha) = a\Lambda(\Omega_1)^{-t}(\Lambda(\Omega_1) - \alpha)$ which further implies that $\Lambda(x)\Lambda(\Omega_1)^t = a(\Lambda(\Omega_1) - \alpha)$. Choose t minimum such that the equality holds. Suppose that t > 0, then $\Lambda(x)\Lambda(\Omega_1)^t = a\Lambda(\Omega_1) - a\alpha$. We have that $a = \frac{1}{\alpha}\Lambda(\Omega_1)(a - \Lambda(x)\Lambda(\Omega_1)^{t-1})$. Set $b := \frac{1}{\alpha}(a - \Lambda(x)\Lambda(\Omega_1)^{t-1})$. It follows that $a = b\Lambda(\Omega_1)$, with $b \in A/\langle \Omega_2 \rangle$. This implies that $\Lambda(x)\Lambda(\Omega_1)^t = a(\Lambda(\Omega_1) - \alpha) = b\Lambda(\Omega_1)(\Lambda(\Omega_1) - \alpha)$, hence $\Lambda(x)\Lambda(\Omega_1)^{t-1} = b(\Lambda(\Omega_1) - \alpha)$. This contradicts the minimality of t, and so t = 0. Therefore, $\Lambda(x) = a(\Lambda(\Omega_1) - \alpha)$ belongs to the ideal of $A/\langle \Omega_2 \rangle$ generated by $\Lambda(\Omega_1) - \alpha$.

2.4.7 Corollary. Let $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}$. The prime ideal $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$ of A is primitive.

Proof. Since $\operatorname{Prim}_J(A)$ are just the maximal prime ideals in $\operatorname{Spec}_J(A)$ (Lemma 1.6.6), it follows that $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$ is a primitive ideal of A for each $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}$.

2.4.8 Proposition. Let $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}$. The prime ideal $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$ of A is maximal.

Proof. Let $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0,0)\}$. Suppose that there exists a maximal ideal I of A such that $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle \subsetneq I \varsubsetneq A$. Let J be the \mathcal{H} -invariant prime ideal in A such that $I \in \operatorname{Spec}_J(A)$. By Propositions 2.4.4, 2.4.5 and 2.4.6, J cannot be $\langle 0 \rangle$, $\langle \Omega_1 \rangle$ or $\langle \Omega_2 \rangle$, since either of these will lead to a contradiction. Every non-zero \mathcal{H} -invariant prime ideal contains Ω_1 only or Ω_2 only or both. Since $J \neq \langle \Omega_1 \rangle, \langle \Omega_2 \rangle$, it implies that J contains both Ω_1 and Ω_2 . Moreover, since $J \subseteq I$, it implies that $\Omega_1, \Omega_2 \in I$. Given that $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle \subset I$, we have that $\Omega_1 - \alpha, \Omega_2 - \beta \in I$. It follows that $\alpha, \beta \in I$, hence I = A, a contradiction! This confirms that $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$ is a maximal ideal in A.

2.4.9 Remark. Since the algebra A is catenary [19, Theorem 4.8], one can prove that the height of $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$ is 2 for all $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}$ by constructing chains of prime ideals of $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$. Nonetheless, we use a different approach to prove the result in Section 3.1. We do this because the discussion in that section (i.e. Section 3.1) will be useful in the subsequent chapters.

Chapter 3

Simple quotients of $U_q^+(G_2)$ and their relation to the second Weyl algebra

Now that we have found some family of maximal ideals of $A = U_q^+(G_2)$, we are going to study the main algebra of interest in this thesis, namely, the corresponding simple quotients. In view of Dixmier's theorem, we consider these simple quotients as deformations of (a quadratic extension of) the second Weyl algebra $A_2(\mathbb{C})$, and so we compare their properties with some known properties of the Weyl algebras. Recall from Proposition 2.4.8 that $\Omega_1 - \alpha$ and $\Omega_2 - \beta$, where $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0,0)\}$, generate a maximal ideal of A. As a result, the corresponding quotient

$$A_{\alpha,\beta} := \frac{A}{\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle}$$

is a simple noetherian domain. Denote the canonical images of E_i in $A_{\alpha,\beta}$ by $e_i := E_i + \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$ for all $1 \le i \le 6$. The algebra $A_{\alpha,\beta}$ satisfies the following relations:

$$e_{2}e_{1} = q^{-3}e_{1}e_{2} \qquad e_{3}e_{1} = q^{-1}e_{1}e_{3} - (q + q^{-1} + q^{-3})e_{2}$$

$$e_{3}e_{2} = q^{-3}e_{2}e_{3} \qquad e_{4}e_{1} = e_{1}e_{4} + (1 - q^{2})e_{3}^{2}$$

$$e_{4}e_{2} = q^{-3}e_{2}e_{4} - \frac{q^{4} - 2q^{2} + 1}{q^{4} + q^{2} + 1}e_{3}^{3} \qquad e_{4}e_{3} = q^{-3}e_{3}e_{4}$$

$$e_{5}e_{1} = qe_{1}e_{5} - (1 + q^{2})e_{3} \qquad e_{5}e_{2} = e_{2}e_{5} + (1 - q^{2})e_{3}^{2}$$

$$e_{5}e_{3} = q^{-1}e_{3}e_{5} - (q + q^{-1} + q^{-3})e_{4} \qquad e_{5}e_{4} = q^{-3}e_{4}e_{5}$$

$$e_{6}e_{1} = q^{3}e_{1}e_{6} - q^{3}e_{5} \qquad e_{6}e_{2} = q^{3}e_{2}e_{6} + (q^{4} + q^{2} - 1)e_{4} + (q^{2} - q^{4})e_{3}e_{5}$$

$$e_{6}e_{3} = e_{3}e_{6} + (1 - q^{2})e_{5}^{2} \qquad e_{6}e_{4} = q^{-3}e_{4}e_{6} - \frac{q^{4} - 2q^{2} + 1}{q^{4} + q^{2} + 1}e_{5}^{3}$$

$$e_{6}e_{5} = q^{-3}e_{5}e_{6},$$

 and

$$e_1e_3e_5 + ae_1e_4 + ae_2e_5 + a'e_3^2 = \alpha, \qquad (3.0.1)$$

$$e_2e_4e_6 + be_2e_5^3 + be_3^3e_6 + b'e_3^2e_5^2 + c'e_3e_4e_5 + d'e_4^2 = \beta.$$
 (3.0.2)

The following additional relations of $A_{\alpha,\beta}$ in the lemma below will be very helpful in this chapter, particularly, in Section 3.2. Note, we put constant coefficients of monomials in a square bracket [] in order to distinguish them from monomials easily. These constants are defined in Appendix C.

3.0.1 Lemma.

(1)
$$e_3^2 = c_1 \alpha + [c_2]e_2e_5 + [c_2]e_1e_4 + [c_3]e_1e_3e_5.$$

$$(2) \ e_4^2 = b_1\beta + [b_2]e_2e_4e_6 + [b_3]e_2e_5^3 + [b_4\alpha]e_3e_6 + [b_5]e_2e_3e_5e_6 + [b_6]e_1e_3e_4e_6 + [b_7\alpha]e_1e_5e_6 + [b_8]e_1e_2e_5^2e_6 + [b_9]e_1^2e_4e_5e_6 + [b_{10}]e_1^2e_3e_5^2e_6 + [b_{11}\alpha]e_5^2 + [b_{12}]e_1e_3e_5^3 + [b_{13}]e_3e_4e_5 + [b_{14}]e_1e_4e_5^2.$$

$$(3) \ e_3^2 e_4 = [c_1 \alpha] e_4 + [q^{-3} c_2] e_2 e_4 e_5 + [c_2 b_4 \alpha] e_1 e_3 e_6 + [b_{15}] e_1 e_3 e_4 e_5 + [\beta b_1 c_2] e_1 \\ + [c_2 b_3] e_1 e_2 e_5^3 + [c_2 b_5] e_1 e_2 e_3 e_5 e_6 + [c_2 b_6] e_1^2 e_3 e_4 e_6 + [c_2 b_7 \alpha] e_1^2 e_5 e_6 \\ + [c_2 b_{11} \alpha] e_1 e_5^2 + [c_2 b_{12}] e_1^2 e_3 e_5^3 + [c_2 b_8] e_1^2 e_2 e_5^2 e_6 + [c_2 b_9] e_1^3 e_4 e_5 e_6 \\ + [c_2 b_{10}] e_1^3 e_3 e_5^2 e_6 + [c_2 b_{14}] e_1^2 e_4 e_5^2 + [c_2 b_2] e_1 e_2 e_4 e_6.$$

$$(4) \ e_{3}e_{4}^{2} = [\beta b_{1}]e_{3} + [k_{1}]e_{2}e_{3}e_{4}e_{6} + [k_{2}]e_{2}e_{3}e_{5}^{3} + [k_{3}\alpha^{2}]e_{6} + [k_{4}\alpha]e_{2}e_{5}e_{6} + [k_{5}\alpha]e_{1}e_{4}e_{6} \\ + [k_{6}\alpha]e_{1}e_{3}e_{5}e_{6} + [k_{7}]e_{2}^{2}e_{5}^{2}e_{6} + [k_{8}\beta]e_{1}e_{5} + [k_{9}]e_{1}e_{2}e_{4}e_{5}e_{6} + [k_{10}]e_{1}^{3}e_{2}e_{5}^{2}e_{6}^{2} \\ + [k_{11}\alpha]e_{4}e_{5} + [k_{12}\alpha]e_{1}^{3}e_{5}e_{6}^{2} + [k_{13}]e_{1}^{2}e_{3}e_{4}e_{5}e_{6} + [k_{14}\beta]e_{1}^{2}e_{6} + [b_{11}\alpha]e_{3}e_{5}^{2} \\ + [k_{15}]e_{1}^{2}e_{2}e_{4}e_{6}^{2} + [k_{16}]e_{1}^{2}e_{2}e_{5}^{3}e_{6} + [k_{17}\alpha]e_{1}^{2}e_{3}e_{6}^{2} + [k_{18}]e_{1}^{2}e_{2}e_{3}e_{5}e_{6}^{2} + [k_{19}]e_{1}^{3}e_{3}e_{4}e_{6}^{2} \\ + [k_{20}]e_{1}^{4}e_{4}e_{5}e_{6}^{2} + [k_{21}\alpha]e_{1}^{2}e_{5}^{2}e_{6} + [k_{22}]e_{1}^{4}e_{3}e_{5}^{2}e_{6}^{2} + [k_{23}]e_{1}^{3}e_{3}e_{5}^{3}e_{6} + [k_{24}]e_{1}^{3}e_{4}e_{5}^{2}e_{6} \\ + [k_{25}]e_{1}e_{5}^{3} + [k_{26}]e_{1}^{2}e_{4}e_{5}^{3} + [k_{27}]e_{1}^{2}e_{3}e_{5}^{4} + [k_{28}]e_{2}e_{4}e_{5}^{2} + [k_{29}]e_{1}e_{3}e_{4}e_{5}^{2} \\ + [k_{30}]e_{1}e_{2}e_{5}^{4} + [k_{31}]e_{1}e_{2}e_{3}e_{5}^{2}e_{6}.$$

The rest of the chapter is organized as follows. In Section 3.1, we prove that the GKdim of $A_{\alpha,\beta}$ is 4 and consequently prove that the height of the maximal ideal $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$ is 2 as expected. Section 3.2 focuses on describing a linear basis of $A_{\alpha,\beta}$. Ultimately, we show in Section 3.3 that at appropriate choices of α and β , the algebra $A_{\alpha,\beta}$ is a quadratic extension of the second Weyl algebra $A_2(\mathbb{C})$ at q = 1. In the next chapter, we study the derivations of $A_{\alpha,\beta}$.

3.1 Gelfand-Kirillov dimension of $A_{\alpha,\beta}$

Let $\alpha, \beta \neq 0$. Recall from Section 2.2 that $R_1 = \mathbb{C}_{q^M}[T_1^{\pm 1}, \cdots, T_6^{\pm 1}]$ is the quantum torus associated to the quantum affine space $\bar{A} = A^{(2)}$. Also, $\Omega_1 = T_1 T_3 T_5$ and $\Omega_2 = T_2 T_4 T_6$ in \bar{A} . It follows from [8, Theorem 5.4.1] that there exists an Ore set $S_{\alpha,\beta}$ in $A_{\alpha,\beta}$ such that $A_{\alpha,\beta}S_{\alpha,\beta}^{-1} \cong R_1/\langle T_1 T_3 T_5 - \alpha, T_2 T_4 T_6 - \beta \rangle$. Now, set

$$\mathscr{A}_{\alpha,\beta} := \frac{R_1}{\langle T_1 T_3 T_5 - \alpha, T_2 T_4 T_6 - \beta \rangle}.$$

Let $t_i := T_i + \langle T_1 T_3 T_5 - \alpha, T_2 T_4 T_6 - \beta \rangle$ denote the canonical images of the generators T_i of R_1 in $\mathscr{A}_{\alpha,\beta}$. The algebra $\mathscr{A}_{\alpha,\beta}$ is a quantum torus generated by $t_1^{\pm 1}, \cdots, t_6^{\pm 1}$ subject to the following relations:

$$t_i t_j = q^{\mu_{ij}} t_j t_i \qquad t_1 = \alpha t_5^{-1} t_3^{-1} \qquad t_2 = \beta t_6^{-1} t_4^{-1}$$

for all $1 \leq i, j \leq 6$ and $\mu_{ij} \in M$ (the skew-symmetric matrix in Section 2.2). Observe that $\mathscr{A}_{\alpha,\beta} \cong \mathbb{C}_{q^N}[t_3^{\pm 1}, t_4^{\pm 1}, t_5^{\pm 1}, t_6^{\pm 1}]$, where the skew-symmetric matrix N can easily be deduced from M as follows:

$$N := \begin{bmatrix} 0 & 3 & 1 & 0 \\ -3 & 0 & 3 & 3 \\ -1 & -3 & 0 & 3 \\ 0 & -3 & -3 & 0 \end{bmatrix}.$$

Secondly, suppose that $\alpha = 0$ and $\beta \neq 0$.

Then, $A_{0,\beta}S_{0,\beta}^{-1} \cong \mathscr{A}_{0,\beta} = R_1/\langle T_1, T_2T_4T_6 - \beta \rangle$. The algebra $\mathscr{A}_{0,\beta}$ is generated by $t_2^{\pm 1}, \cdots, t_6^{\pm 1}$ subject to the relations

$$t_i t_j = q^{\mu_{ij}} t_j t_i \qquad t_1 = 0 \qquad t_2 = \beta t_6^{-1} t_4^{-1},$$

for all $1 \leq i, j \leq 6$ and $\mu_{ij} \in M$. We also have that $\mathscr{A}_{0,\beta} \cong \mathbb{C}_{q^N}[t_3^{\pm 1}, t_4^{\pm 1}, t_5^{\pm 1}, t_6^{\pm 1}]$. Finally, when $\alpha \neq 0$ and $\beta = 0$, then one can also verify that $\mathscr{A}_{\alpha,0} \cong \mathbb{C}_{q^N}[t_3^{\pm 1}, t_4^{\pm 1}, t_5^{\pm 1}, t_6^{\pm 1}]$. As a result, we have that $A_{\alpha,\beta}S_{\alpha,\beta}^{-1} \cong \mathscr{A}_{\alpha,\beta} \cong \mathbb{C}_{q^N}[t_3^{\pm 1}, t_4^{\pm 1}, t_5^{\pm 1}, t_6^{\pm 1}]$. With a slight abuse of notation, we write $A_{\alpha,\beta}S_{\alpha,\beta}^{-1} = \mathscr{A}_{\alpha,\beta} = \mathbb{C}_{q^N}[t_3^{\pm 1}, t_4^{\pm 1}, t_5^{\pm 1}, t_6^{\pm 1}]$ for all $(\alpha, \beta) \in$ $\mathbb{C}^2 \setminus \{(0,0)\}$. It follows from [18, Theorem 6.3] that $\operatorname{GKdim}(A_{\alpha,\beta}) = \operatorname{GKdim}(A_{\alpha,\beta}S_{\alpha,\beta}^{-1}) =$ $\operatorname{GKdim}(\mathscr{A}_{\alpha,\beta}) = 4$. Since Tauvel's height formula holds in $A = U_q^+(G_2)$ (Chapter 2), we have that $\operatorname{GKdim}(A) = \operatorname{ht}(\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle) + \operatorname{GKdim}(A_{\alpha,\beta})$. We already know from Chapter 2 that the $\mathsf{GKdim}(A) = 6$. Therefore, $\mathsf{ht}(\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle) = 2$ for all $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}.$

3.2 Linear basis for $A_{\alpha,\beta}$

Set $A_{\beta} := A/\langle \Omega_2 - \beta \rangle$, where $\beta \in \mathbb{C}$. Now, denote the canonical images of E_i by $\widehat{e}_i := E_i + \langle \Omega_2 - \beta \rangle$ in A_{β} . Clearly, $A_{\alpha,\beta} \cong A_{\beta}/\langle \widehat{\Omega}_1 - \alpha \rangle$. As a result, one can identify $A_{\alpha,\beta}$ with $A_{\beta}/\langle \widehat{\Omega}_1 - \alpha \rangle$. Moreover, the algebra A_{β} satisfies the relations of $A = U_q^+(G_2)$ and

$$\hat{e}_2\hat{e}_4\hat{e}_6 + b\hat{e}_2\hat{e}_5^3 + b\hat{e}_3^3\hat{e}_6 + b'\hat{e}_3^2\hat{e}_5^2 + c'\hat{e}_3\hat{e}_4\hat{e}_5 + d'\hat{e}_4^2 = \beta.$$
(3.2.1)

From Propositions 2.4.4 and 2.4.6, one can conclude that $\langle \Omega_2 - \beta \rangle$ is a completely prime ideal (since it is a prime ideal) of A for all $\beta \in \mathbb{C}$. Hence, similar to $A_{\alpha,\beta}$, the algebra A_{β} is a noetherian domain.

We are now going to find a linear basis for $A_{\alpha,\beta}$, where $(\alpha,\beta) \in \mathbb{C}^2 \setminus \{(0,0)\}$. Since $A_{\alpha,\beta}$ is identified with $A_\beta/\langle \widehat{\Omega}_1 - \alpha \rangle$, we will first and foremost find a basis for A_β , and then proceed to find a basis for $A_{\alpha,\beta}$. Note, the relations in Lemma A.1.4 are also valid in A_β and $A_{\alpha,\beta}$, and are going to be very useful in this section.

3.2.1 Proposition. The set $\mathfrak{S} = \{\widehat{e_1}^i \widehat{e_2}^j \widehat{e_3}^k \widehat{e_4}^{\xi} \widehat{e_5}^l \widehat{e_6}^m \mid i, j, k, l, m \in \mathbb{N} \text{ and } \xi = 0, 1\}$ is a \mathbb{C} -basis of A_{β} .

Proof. Since the family $(\prod_{s=1}^{6} E_s^{i_s})_{i_s \in \mathbb{N}}$ is a PBW-basis of A over \mathbb{C} , it follows that the family $(\prod_{s=1}^{6} \hat{e_s}^{i_s})_{i_s \in \mathbb{N}}$ is a spanning set of A_β over \mathbb{C} . We want to show that \mathfrak{S} spans A_β . We do this by showing that $\prod_{s=1}^{6} \hat{e_s}^{i_s}$ can be written as a finite linear combination of the elements of \mathfrak{S} for all $i_1, \dots, i_6 \in \mathbb{N}$ by an induction on i_4 . The result is obvious when $i_4 = 0$ or 1. For $i_4 \geq 1$, assume that

$$\Pi_{s=1}^6 \widehat{e_s}^{i_s} = \sum_{(\xi,\underline{v})\in I} a_{(\xi,\underline{v})} \widehat{e_1}^i \widehat{e_2}^j \widehat{e_3}^k \widehat{e_4}^\xi \widehat{e_5}^l \widehat{e_6}^m,$$

where $\underline{v} := (i, j, k, l, m) \in \mathbb{N}^5$ and $a_{(\xi,\underline{v})}$ are all complex numbers. Note, I is a finite subset of $\{0, 1\} \times \mathbb{N}^5$. It follows from the commutation relations of A_β (see Lemma A.1.4) that

$$\widehat{e_1}^{i_1} \widehat{e_2}^{i_2} \widehat{e_3}^{i_3} \widehat{e_4}^{i_4+1} \widehat{e_5}^{i_5} \widehat{e_6}^{i_6} = q^{\bullet} \prod_{s=1}^6 \widehat{e_s}^{i_s} \widehat{e_4} - q^{\bullet} d_1[i_6] \widehat{e_1}^{i_1} \widehat{e_2}^{i_2} \widehat{e_3}^{i_3} \widehat{e_4}^{i_4} \widehat{e_5}^{i_5+3} \widehat{e_6}^{i_6-1}.$$

From the inductive hypothesis, $\widehat{e_1}^{i_1} \widehat{e_2}^{i_2} \widehat{e_3}^{i_3} \widehat{e_4}^{i_4} \widehat{e_5}^{i_5+3} \widehat{e_6}^{i_6-1} \in \text{Span}(\mathfrak{S})$. Hence, we proceed to show that $\prod_{s=1}^6 \widehat{e_s}^{i_s} \widehat{e_4}$ is also in the span of \mathfrak{S} . From the inductive hypothesis, we have

$$\Pi_{s=1}^6 \widehat{e_s}^{i_s} \widehat{e_4} = \sum_{(\xi,\underline{v})\in I} a_{(\xi,\underline{v})} \widehat{e_1}^i \widehat{e_2}^j \widehat{e_3}^k \widehat{e_4}^\xi \widehat{e_5}^l \widehat{e_6}^m \widehat{e_4}.$$

Using the commutation relations in Lemma A.1.4, we have that

$$\Pi_{s=1}^{6} \widehat{e_s}^{i_s} \widehat{e_4} = \sum_{(\xi,\underline{v})\in I} q^{\bullet} a_{(\xi,\underline{v})} \widehat{e_1}^i \widehat{e_2}^j \widehat{e_3}^k \widehat{e_4}^{\xi+1} \widehat{e_5}^l \widehat{e_6}^m \\ + \sum_{(\xi,\underline{v})\in I} q^{\bullet} d_1[m] a_{(\xi,\underline{v})} \widehat{e_1}^i \widehat{e_2}^j \widehat{e_3}^k \widehat{e_4}^{\xi} \widehat{e_5}^{l+3} \widehat{e_6}^{m-1}$$

All the terms in the above expression belong to the span of \mathfrak{S} except $\hat{e_1}^i \hat{e_2}^j \hat{e_3}^k \hat{e_4}^2 \hat{e_5}^l \hat{e_6}^m$. From (3.2.1), we have that

$$\hat{e_4}^2 = \beta_0 \hat{e_2} \hat{e_4} \hat{e_6} + b\beta_0 \hat{e_2} \hat{e_5}^3 + b\beta_0 \hat{e_3}^3 \hat{e_6} + b'\beta_0 \hat{e_3}^2 \hat{e_5}^2 + c'\beta_0 \hat{e_3} \hat{e_4} \hat{e_5} - \beta\beta_0, \qquad (3.2.2)$$

where $\beta_0 = -1/d'$. Substituting (3.2.2) into $\hat{e_1}^i \hat{e_2}^j \hat{e_3}^k \hat{e_4}^2 \hat{e_5}^l \hat{e_6}^m$, one can easily verify that $\hat{e_1}^i \hat{e_2}^j \hat{e_3}^k \hat{e_4}^2 \hat{e_5}^l \hat{e_6}^m \in \text{Span}(\mathfrak{S})$. Therefore, $\hat{e_1}^{i_1} \hat{e_2}^{i_2} \hat{e_3}^{i_3} \hat{e_4}^{i_4+1} \hat{e_5}^{i_5} \hat{e_6}^{i_6}$ can be written as a finite linear combination of the elements of \mathfrak{S} over \mathbb{C} for all $i_1, \dots, i_6 \in \mathbb{N}$. By the principle of mathematical induction, \mathfrak{S} is a spanning set of A_β over \mathbb{C} .

We further show that \mathfrak{S} is a linearly independent set. Suppose that

$$\sum_{(\xi,\underline{v})\in I} a_{(\xi,\underline{v})} \widehat{e_1}^i \widehat{e_2}^j \widehat{e_3}^k \widehat{e_4}^\xi \widehat{e_5}^l \widehat{e_6}^m = 0.$$

Since $A_{\beta} = A/\langle \Omega_2 - \beta \rangle$, it implies that

$$\sum_{(\xi,\underline{v})\in I} a_{(\xi,\underline{v})} E_1^i E_2^j E_3^k E_4^{\xi} E_5^l E_6^m = (\Omega_2 - \beta)\nu,$$

with $\nu \in A$. Write $\nu = \sum_{(i,\dots,n)\in J} b_{(i,\dots,n)} E_1^i E_2^j E_3^k E_4^l E_5^m E_6^n$, where J is a finite subset of \mathbb{N}^6 and $b_{(i,\dots,n)}$ are all complex numbers. It follows that

$$\sum_{(\xi,\underline{v})\in I} a_{(\xi,\underline{v})} E_1^i E_2^j E_3^k E_4^{\xi} E_5^l E_6^m = \sum_{(i,\dots,n)\in J} b_{(i,\dots,n)} E_1^i E_2^j E_3^k (\Omega_2 - \beta) E_4^l E_5^m E_6^n.$$
(3.2.3)

Before we continue the proof, the following needs to be noted.

♣ Let (i', j', k', l', m', n'), $(i, j, k, l, m, n) \in \mathbb{N}^6$. We say that $(i, j, k, l, m, n) <_4$ (i', j', k', l', m', n') if [l < l'] or [l = l' and i < i'] or [l = l', i = i' and j < j'] or [l = l', i = i', j = j' and k < k'] or [l = l', i = i', j = j', k = k' and m < m']or [l = l', i = i', j = j', k = k', m = m' and $n \le n']$. Note, the purpose of the square bracket [] is to differentiate the options.

From Section 2.2, we have that $\Omega_2 = E_2 E_4 E_6 + b E_2 E_5^3 + b E_3^3 E_6 + b' E_3^2 E_5^2 + c' E_3 E_4 E_5 + d' E_4^2$ in $A = U_q^+(G_2)$. Now,

$$\begin{split} \sum_{(\xi,\underline{v})\in I} a_{(\xi,\underline{v})} E_1^i E_2^j E_3^k E_4^{\xi} E_5^l E_6^m &= \sum_{(i,\cdots,n)\in J} b_{(i,\cdots,n)} E_1^i E_2^j E_3^k (\Omega_2 - \beta) E_4^l E_5^m E_6^n \\ &= \sum_{(i,\cdots,n)\in J} d' b_{(i,\cdots,m)} E_1^i E_2^j E_3^k E_4^{l+2} E_5^m E_6^n + \mathsf{LT}_{<_4}, \end{split}$$

where $LT_{<_4}$ contains lower order terms with respect to $<_4$ (as in \clubsuit). Moreover, $LT_{<_4}$ vanishes when $b_{(i,\dots,n)} = 0$ for all $(i,\dots,n) \in J$ (one can easily confirm this by fully expanding the right hand side of (3.2.3)).

Now, suppose that there exists $(i, j, k, l, m, n) \in J$ such that $b_{(i,j,k,l,m,n)} \neq 0$. Let (i', j', k', l', m', n') be the greatest element of J with respect to $<_4$ (defined in \clubsuit above) such that $b_{(i',j',k',l',m',n')} \neq 0$. Note, the family $(E_1^i E_2^j E_3^k E_4^l E_5^m E_6^n)_{(i,\cdots,n) \in \mathbb{N}^6}$ is a basis of A. Since $LT_{<_4}$ contains lower order terms, identifying the coefficients of $E_1^{i'}E_2^{j'}E_3^{k'}E_4^{l'+2}E_5^{m'}E_6^{n'}$ in the above equality, we have that $d'b_{(i',\cdots,n')} = 0$. Since $b_{(i',j',k',l',m',n')} \neq 0$, it follows that $d' = q^{12}/(q^6 - 1) = 0$, a contradiction (see Appendix C for the expression of d'). As a result, $b_{(i,j,k,l,m,n)} = 0$ for all $(i, j, k, l, m, n) \in J$. Therefore, $\sum_{(\xi,\underline{v})\in I} a_{(\xi,\underline{v})}E_1^iE_2^jE_3^kE_4^\xi E_5^lE_6^m = 0$. Since $(E_1^iE_2^jE_3^kE_4^lE_5^mE_6^n)_{(i,\cdots,n)\in\mathbb{N}^6}$ is a basis of A, it follows that $a_{(\xi,\underline{v})} = 0$ for all $(\xi,\underline{v}) \in I$. In conclusion, \mathfrak{S} is a linearly independent set and hence forms a basis of A_β as desired.

3.2.2 Proposition. Let $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}$. The set $\mathcal{B} = \{e_1^i e_2^j e_3^{\epsilon_1} e_4^{\epsilon_2} e_5^k e_6^l \mid i, j, k, l \in \mathbb{N} \text{ and } \epsilon_1, \epsilon_2 \in \{0, 1\}\}$ is a \mathbb{C} -basis of $A_{\alpha, \beta}$.

Proof. Since the set $\mathfrak{S} = \{\widehat{e_1}^{i_1} \widehat{e_2}^{i_2} \widehat{e_3}^{i_3} \widehat{e_4}^{\xi} \widehat{e_5}^{i_5} \widehat{e_6}^{i_6} \mid i_1, i_2, i_3, i_5, i_6 \in \mathbb{N} \text{ and } \xi = 0, 1\}$ is a \mathbb{C} -basis of A_{β} (Proposition 3.2.1), and $A_{\alpha,\beta}$ is identified with $A_{\beta}/\langle \widehat{\Omega}_1 - \alpha \rangle$, it follows that $(e_1^{i_1} e_2^{i_2} e_3^{i_3} e_4^{\xi} e_5^{i_5} e_6^{i_6})_{i_1, \dots, i_6 \in \mathbb{N}}$, with $\xi \in \{0, 1\}$, is a spanning set of $A_{\alpha,\beta}$ over \mathbb{C} . We want to show that \mathcal{B} spans $A_{\alpha,\beta}$ by showing that $e_1^{i_1} e_2^{i_2} e_3^{i_3} e_4^{\xi} e_5^{i_5} e_6^{i_6}$ can be written as a finite linear combination of the elements of \mathcal{B} over \mathbb{C} for all $i_1, i_2, i_3, i_5, i_6 \in \mathbb{N}$ and $\xi = 0, 1$. By Proposition 3.2.1, it is sufficient to do this by an induction on i_3 . The result is obvious when $i_3 = 0$ or 1. For $i_3 \geq 1$, suppose that

$$e_1^{i_1} e_2^{i_2} e_3^{i_3} e_4^{\xi} e_5^{i_5} e_6^{i_6} = \sum_{(\epsilon_1, \epsilon_2, \underline{\upsilon}) \in I} a_{(\epsilon_1, \epsilon_2, \underline{\upsilon})} e_1^i e_2^j e_3^{\epsilon_1} e_4^{\epsilon_2} e_5^k e_6^l,$$

where $\underline{v} = (i, j, k, l) \in \mathbb{N}^4$, I is a finite subset of $\{0, 1\}^2 \times \mathbb{N}^4$, and $(a_{(\epsilon_1, \epsilon_2, \underline{v})})_{(\epsilon_1, \epsilon_2, \underline{v}) \in I}$ is a family of complex numbers. Using the commutation relations in $A_{\alpha,\beta}$ (Lemma A.1.4), we have that:

$$e_1^{i_1}e_2^{i_2}e_3^{i_3+1}e_4^{\xi}e_5^{i_5}e_6^{i_6} = q^{\bullet}e_3e_1^{i_1}e_2^{i_2}e_3^{i_3}e_4^{\xi}e_5^{i_5}e_6^{i_6} - q^{\bullet}d_2[i_1]e_1^{i_1-1}e_2^{1+i_2}e_3^{i_3}e_4^{\xi}e_5^{i_5}e_6^{i_6}$$

From the inductive hypothesis, $e_1^{i_1-1}e_2^{1+i_2}e_3^{i_3}e_4^{\xi}e_5^{i_5}e_6^{i_6} \in \text{Span}(\mathcal{B})$ for all $i_1 > 0$ (note: $d_2[0] = 0$). As a result, we proceed to show that $e_3e_1^{i_1}e_2^{i_2}e_3^{i_3}e_4^{\xi}e_5^{i_5}e_6^{i_6}$ is also in the span of \mathcal{B} . It follows from the inductive hypothesis that

$$e_{3}e_{1}^{i_{1}}e_{2}^{i_{2}}e_{3}^{i_{3}}e_{4}^{\xi}e_{5}^{i_{5}}e_{6}^{i_{6}} = \sum_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon})\in I} a_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon})}e_{3}e_{1}^{i}e_{2}^{j}e_{3}^{\epsilon_{1}}e_{4}^{\epsilon_{2}}e_{5}^{i}e_{6}^{l}$$
$$= \sum_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon})\in I} a_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon})}e_{1}^{i}e_{2}^{j}e_{3}^{\epsilon_{1}+1}e_{4}^{\epsilon_{2}}e_{5}^{k}e_{6}^{l}$$
$$+ \sum_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon})\in I} d_{2}[i]a_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon})}e_{1}^{i-1}e_{2}^{1+j}e_{3}^{\epsilon_{1}}e_{4}^{\epsilon_{2}}e_{5}^{k}e_{6}^{l}.$$

Clearly, the monomial $e_1^{i-1}e_2^{1+j}e_3^{\epsilon_1}e_4^{\epsilon_2}e_5^ke_6^l$ belongs to the span of \mathcal{B} for all $(\epsilon_1, \epsilon_2, \underline{v}) \in I$ (with i > 0). Again, the monomial $e_1^i e_2^j e_3^{\epsilon_1+1} e_4^{\epsilon_2} e_5^k e_6^l$ belongs to the span of \mathcal{B} for all $(\epsilon_1, \epsilon_2, \underline{v}) \in I$; with $(\epsilon_1, \epsilon_2) = (0, 0), (0, 1)$. For $(\epsilon_1, \epsilon_2) = (1, 0), (1, 1)$; we must show that $e_1^i e_2^j e_3^2 e_5^k e_6^l$ and $e_1^i e_2^j e_3^2 e_4 e_5^k e_6^l$ belong to the span of \mathcal{B} . From Lemma 3.0.1, one can write $e_1^i e_2^j e_3^2 e_5^k e_6^l$ and $e_1^i e_2^j e_3^2 e_4 e_5^k e_6^l$ as finite linear combinations of the elements of \mathcal{B} over \mathbb{C} . Hence, $e_1^i e_2^j e_3^2 e_5^k e_6^l$ and $e_1^i e_2^j e_3^2 e_4 e_5^k e_6^l$ belong to the span of \mathcal{B} for all $(\epsilon_1, \epsilon_2, \underline{v}) \in I$; with $(\epsilon_1, \epsilon_2) = (1, 0), (1, 1)$. We have therefore established that $e_3 e_1^{i_1} e_2^{i_2} e_3^{i_3} e_4^k e_5^{i_6} e_6^i \in$ Span (\mathcal{B}) . Consequently, each $e_1^{i_1} e_2^{i_2} e_3^{i_3+1} e_4^k e_5^{i_5} e_6^{i_6}$ belongs to the span of \mathcal{B} . By the principle of mathematical induction, \mathcal{B} is a spanning set of $A_{\alpha,\beta}$ over \mathbb{C} as expected.

Finally, we show that $\mathcal B$ is a linearly independent set. Suppose that

$$\sum_{(\epsilon_1,\epsilon_2,\underline{\upsilon})\in I} a_{(\epsilon_1,\epsilon_2,\underline{\upsilon})} e_1^i e_2^j e_3^{\epsilon_1} e_4^{\epsilon_2} e_5^k e_6^l = 0.$$

In A_{β} , we have that

$$\sum_{(\epsilon_1,\epsilon_2,\underline{\nu})\in I} a_{(\epsilon_1,\epsilon_2,\underline{\nu})} \widehat{e_1}^i \widehat{e_2}^j \widehat{e_3}^{\epsilon_1} \widehat{e_4}^{\epsilon_2} \widehat{e_5}^k \widehat{e_6}^l = (\widehat{\Omega}_1 - \alpha)\nu,$$

with $\nu \in A_{\beta}$. One can write ν in terms of the basis \mathfrak{S} of A_{β} (Proposition 3.2.1) as:

$$\nu = \sum_{\underline{w}\in J_1} b_{\underline{w}} \widehat{e_1}^i \widehat{e_2}^j \widehat{e_3}^k \widehat{e_4} \widehat{e_5}^m \widehat{e_6}^n + \sum_{\underline{w}\in J_2} c_{\underline{w}} \widehat{e_1}^i \widehat{e_2}^j \widehat{e_3}^k \widehat{e_5}^l \widehat{e_6}^m$$

where $b_{\underline{w}} \text{ and } c_{\underline{w}}$ are all complex numbers and w:=(i,j,k,l,m). Hence,

$$\sum_{(\epsilon_1,\epsilon_2,\underline{v})\in I} a_{(\epsilon_1,\epsilon_2,\underline{v})} \widehat{e_1}^i \widehat{e_2}^j \widehat{e_3}^{\epsilon_1} \widehat{e_4}^{\epsilon_2} \widehat{e_5}^k \widehat{e_6}^l = \sum_{\underline{w}\in J_1} b_{\underline{w}} \widehat{e_1}^i \widehat{e_2}^j \widehat{e_3}^k (\widehat{\Omega}_1 - \alpha) \widehat{e_4} \widehat{e_5}^l \widehat{e_6}^m + \sum_{\underline{w}\in J_2} c_{\underline{w}} \widehat{e_1}^i \widehat{e_2}^j \widehat{e_3}^k (\widehat{\Omega}_1 - \alpha) \widehat{e_5}^l \widehat{e_6}^m.$$

Note, $\widehat{\Omega}_1 = \widehat{e_1}\widehat{e_3}\widehat{e_5} + a\widehat{e_1}\widehat{e_4} + a\widehat{e_2}\widehat{e_5} + a'\widehat{e_3}^2$. Using (3.2.2) and the relation $e_3^k e_1 = q^{-k}e_1e_3^k + d_2[k]e_2e_3^{k-1}$ (see Lemma A.1.4), one can verify that the above equality can be written in terms of the basis of A_β (Propositions 3.2.1) as:

$$\sum_{(\epsilon_{1},\epsilon_{2},\underline{v})\in I} a_{(\epsilon_{1},\epsilon_{2},\underline{v})} \widehat{e_{1}}^{i} \widehat{e_{2}}^{j} \widehat{e_{3}}^{\epsilon_{1}} \widehat{e_{4}}^{\epsilon_{2}} \widehat{e_{5}}^{k} \widehat{e_{6}}^{l} = \sum_{\underline{w}\in J_{1}} b_{\underline{w}} a' \widehat{e_{1}}^{i} \widehat{e_{2}}^{j} \widehat{e_{3}}^{k+2} \widehat{e_{4}} \widehat{e_{5}}^{l} \widehat{e_{6}}^{m} + \sum_{\underline{w}\in J_{1}} q^{\bullet} b_{\underline{w}} ab \beta_{0} \widehat{e_{1}}^{i+1} \widehat{e_{2}}^{j} \widehat{e_{3}}^{k+3} \widehat{e_{5}}^{l} \widehat{e_{6}}^{m+1} + \sum_{\underline{w}\in J_{2}} q^{\bullet} c_{\underline{w}} a \widehat{e_{1}}^{i+1} \widehat{e_{2}}^{j} \widehat{e_{3}}^{k+2} \widehat{e_{5}}^{l} \widehat{e_{6}}^{m} + \sum_{\underline{w}\in J_{2}} c_{\underline{w}} a' \widehat{e_{1}}^{i} \widehat{e_{2}}^{j} \widehat{e_{3}}^{k+2} \widehat{e_{5}}^{l} \widehat{e_{6}}^{m} + \Upsilon, \quad (3.2.4)$$

where Υ is defined on the next page.
$$\begin{split} &\Upsilon = \sum_{\underline{w} \in J_1} q^{\bullet} b_{\underline{w}} \hat{e_1}^{i+1} \hat{e_2}^j \hat{e_3}^{k+1} \hat{e_4} \hat{e_5}^{l+1} \hat{e_6}^m + \sum_{\underline{w} \in J_1} q^{\bullet} b_{\underline{w}} d_2[k] \hat{e_1}^{i} \hat{e_2}^{1+j} \hat{e_3}^{k} \hat{e_4} \hat{e_5}^{l+1} \hat{e_6}^m \\ &- \sum_{\underline{w} \in J_1} q^{\bullet} b_{\underline{w}} a\beta\beta_0 \hat{e_1}^{i+1} \hat{e_2}^{j} \hat{e_3}^{k} \hat{e_5}^{l} \hat{e_6}^m + \sum_{\underline{w} \in J_1} q^{\bullet} b_{\underline{w}} a\beta_0 \hat{e_1}^{i+1} \hat{e_2}^{1+j} \hat{e_3}^{k} \hat{e_4} \hat{e_5}^{l+2} \hat{e_6}^m \\ &+ \sum_{\underline{w} \in J_1} q^{\bullet} b_{\underline{w}} ab\beta_0 \hat{e_1}^{i+1} \hat{e_2}^{j+1} \hat{e_3}^{k} \hat{e_5}^{l+3} \hat{e_6}^m + \sum_{\underline{w} \in J_1} q^{\bullet} b_{\underline{w}} a\beta\beta_0 \hat{e_1}^{i+1} \hat{e_2}^{j} \hat{e_3}^{k+2} \hat{e_5}^{l+2} \hat{e_6}^m \\ &+ \sum_{\underline{w} \in J_1} q^{\bullet} b_{\underline{w}} ac' \beta_0 \hat{e_1}^{i+1} \hat{e_2}^{j} \hat{e_3}^{k+1} \hat{e_4} \hat{e_5}^{l+1} \hat{e_6}^m - \sum_{\underline{w} \in J_1} q^{\bullet} b_{\underline{w}} a\beta\beta_0 d_2[k] \hat{e_1}^{i} \hat{e_2}^{1+j} \hat{e_3}^{k-1} \hat{e_5}^{l} \hat{e_6}^m \\ &+ \sum_{\underline{w} \in J_1} q^{\bullet} b_{\underline{w}} a\beta_0 d_2[k] \hat{e_1}^{i} \hat{e_2}^{j+2} \hat{e_3}^{k-1} \hat{e_4} \hat{e_5}^{l} \hat{e_6}^{m+1} + \sum_{\underline{w} \in J_1} q^{\bullet} b_{\underline{w}} a\beta\beta_0 d_2[k] \hat{e_1}^{i} \hat{e_2}^{j+2} \hat{e_3}^{k-1} \hat{e_5}^{l+3} \hat{e_6}^m \\ &+ \sum_{\underline{w} \in J_1} q^{\bullet} b_{\underline{w}} a\beta_0 d_2[k] \hat{e_1}^{i+1} \hat{e_2}^{j} \hat{e_3}^{k+2} \hat{e_5}^{l} \hat{e_6}^{m+1} + \sum_{\underline{w} \in J_1} q^{\bullet} b_{\underline{w}} a\beta\beta_0 d_2[k] \hat{e_1}^{i} \hat{e_2}^{j+1} \hat{e_3}^{k+1} \hat{e_5}^{l+2} \hat{e_6}^m \\ &+ \sum_{\underline{w} \in J_1} q^{\bullet} b_{\underline{w}} a\beta\beta_0 d_2[k] \hat{e_1}^{i+1} \hat{e_2}^{j} \hat{e_3}^{k+2} \hat{e_5}^{l} \hat{e_6}^{m+1} + \sum_{\underline{w} \in J_1} q^{\bullet} b_{\underline{w}} a\beta\beta_0 d_2[k] \hat{e_1}^{i} \hat{e_2}^{j+1} \hat{e_3}^{k+1} \hat{e_5}^{l+2} \hat{e_6}^m \\ &+ \sum_{\underline{w} \in J_1} q^{\bullet} b_{\underline{w}} ac^{\prime} \beta_0 d_2[k] \hat{e_1}^{i} \hat{e_2}^{j+1} \hat{e_3}^{k} \hat{e_4} \hat{e_5}^{l+1} \hat{e_6}^m + \sum_{\underline{w} \in J_1} q^{\bullet} b_{\underline{w}} a\hat{e_1}^{i} \hat{e_2}^{j+1} \hat{e_3}^{k} \hat{e_4} \hat{e_5}^{l+1} \hat{e_6}^m \\ &- \sum_{\underline{w} \in J_2} q^{\bullet} c_{\underline{w}} d_2[k] \hat{e_1}^{i} \hat{e_2}^{j+1} \hat{e_3}^{k} \hat{e_5}^{l+1} \hat{e_6}^m + \sum_{\underline{w} \in J_2} q^{\bullet} c_{\underline{w}} ad_2[k] \hat{e_1}^{i} \hat{e_2}^{j+1} \hat{e_3}^{k-1} \hat{e_4} \hat{e_5}^{l} \hat{e_6}^m \\ &+ \sum_{\underline{w} \in J_2} q^{\bullet} c_{\underline{w}} d_2[k] \hat{e_1}^{i} \hat{e_2}^{j+1} \hat{e_3}^{k} \hat{e_5}^{l+1} \hat{e_6}^m + \sum_{\underline{w} \in J_2} q^{\bullet} c_{\underline{w}} ad_2[k] \hat{e_1}^{i} \hat{e_2}^{j+1} \hat{e_3}^{k$$

Before we continue the proof, the following point needs to be noted.

▲ Let $(\vartheta_1, \vartheta_2, \vartheta_3, \vartheta_5, \vartheta_6)$, $(\varsigma_1, \varsigma_2, \varsigma_3, \varsigma_5, \varsigma_6) \in \mathbb{N}^5$. We say that $(\varsigma_1, \varsigma_2, \varsigma_3, \varsigma_5, \varsigma_6) <_3$ $(\vartheta_1, \vartheta_2, \vartheta_3, \vartheta_5, \vartheta_6)$ if $[\vartheta_3 > \varsigma_3]$ or $[\vartheta_3 = \varsigma_3$ and $\vartheta_1 > \varsigma_1]$ or $[\vartheta_3 = \varsigma_3, \ \vartheta_1 = \varsigma_1$ and $\vartheta_2 > \varsigma_2]$ or $[\vartheta_3 = \varsigma_3, \ \vartheta_1 = \varsigma_1, \ \vartheta_2 = \varsigma_2$ and $\vartheta_5 > \varsigma_5]$ or $[\vartheta_3 = \varsigma_3, \ \vartheta_1 = \varsigma_1, \ \vartheta_2 = \varsigma_2$, $\vartheta_5 = \varsigma_5$ and $\vartheta_6 \ge \varsigma_6]$.

Observe that Υ contains lower order terms with respect to $<_3$ (defined in \clubsuit above) in each monomial type (note, there are two different types of monomials in the basis of A_β : one with \hat{e}_4 and the other without \hat{e}_4). Now, suppose that there exists $(i, j, k, l, m) \in J_1$ and $(i, j, k, l, m) \in J_2$ such that $b_{(i,j,k,l,m)} \neq 0$ and $c_{(i,j,k,l,m)} \neq 0$. Let $(v_1, v_2, v_3, v_5, v_6)$ and $(w_1, w_2, w_3, w_5, w_6)$ be the greatest elements of J_1 and J_2 respectively with re-

spect to $<_3$ such that $b_{(v_1,v_2,v_3,v_5,v_6)}$ and $c_{(w_1,w_2,w_3,w_5,w_6)}$ are non-zero. Since \mathfrak{S} is a linear basis for $A_eta,$ and Υ contains lower order terms with respect to $<_3$, we have the following: if $w_3 - v_3 < 2$, then identifying the coefficients of $\hat{e_1}^{v_1} \hat{e_2}^{v_2} \hat{e_3}^{v_3+2} \hat{e_4} \hat{e_5}^{v_5} \hat{e_6}^{v_6}$ in (3.2.4), we have that $a'b_{(v_1,v_2,v_3,v_5,v_6)} = 0$. But $b_{(v_1,v_2,v_3,v_5,v_6)} \neq 0$, hence a' = $q^6/(q^2-1) = 0$, a contradiction (see Appendix C for the expression of a'). Finally, if $w_3 - v_3 \ge 2$, then identifying the coefficient of $\widehat{e_1}^{w_1} \widehat{e_2}^{w_2} \widehat{e_3}^{w_3+2} \widehat{e_5}^{w_5} \widehat{e_6}^{w_6}$, we have that $a'c_{(w_1,w_2,w_3,w_5,w_6)} = 0$. But $c_{(w_1,w_2,w_3,w_5,w_6)} \neq 0$, hence a' = 0, a contradiction! This implies that either all $b_{(i,j,k,l,m)}$ or all $c_{(i,j,k,l,m)}$ are zero. Without loss of generality, suppose that there exists $(i, j, k, l, m) \in J_2$ such that $c_{(i, j, k, l, m)}$ is not zero. Then, $b_{(i,j,k,l,m)}$ are all zero. Let $(w_1, w_2, w_3, w_5, w_6)$ be the greatest element of J_2 such that $c_{(w_1,w_2,w_3,w_5,w_6)} \neq 0$. Identifying the coefficients of $\hat{e_1}^{w_1} \hat{e_2}^{w_2} \hat{e_3}^{w_3+2} \hat{e_5}^{w_5} \hat{e_6}^{w_6}$ in the above equality, we have that $a'c_{(w_1,w_2,w_3,w_5,w_6)}=0.$ Since $c_{(w_1,w_2,w_3,w_5,w_6)}
eq 0$, it follows that a' = 0, a contradiction! We can therefore conclude that $b_{(i,j,k,l,m)}$ and $c_{(i,j,k,l,m)}$ are all zero. Consequently, $\sum_{(\epsilon_1,\epsilon_2,\underline{v})\in I} a_{(\epsilon_1,\epsilon_2,\underline{v})} \widehat{e_1}^i \widehat{e_2}^j \widehat{e_3}^{\epsilon_1} \widehat{e_4}^{\epsilon_2} \widehat{e_5}^k \widehat{e_6}^l = 0.$ Since $(\widehat{e_1}^{i_1}\widehat{e_2}^{i_2}\widehat{e_3}^{i_3}\widehat{e_4}^{\xi}\widehat{e_5}^{i_5}\widehat{e_6}^{i_6})_{(\xi,i_1,\cdots,i_6)\in\{0,1\}\times\mathbb{N}^5} \text{ is a basis of } A_{\beta}\text{, it implies that } a_{(\epsilon_1,\epsilon_2,\underline{v})} = 0 \text{ for } a_{(\epsilon_1,$ all $(\epsilon_1, \epsilon_2, \underline{v}) \in I$. Therefore, \mathcal{B} is a linearly independent set.

3.2.3 Corollary. Let $\underline{v} = (i, j, k, l) \in \mathbb{N}^2 \times \mathbb{Z}^2$, I represent a finite subset of $\{0, 1\} \times \mathbb{N}^2 \times \mathbb{Z}^2$ and $(a_{(\epsilon_1, \epsilon_2, \underline{v})})_{(\epsilon_1, \epsilon_2, \underline{v}) \in I}$ be a family of complex numbers. If

$$\sum_{(\epsilon_1,\epsilon_2,\underline{v})\in I} a_{(\epsilon_1,\epsilon_2,\underline{v})} e_1^i e_2^j e_3^{\epsilon_1} e_4^{\epsilon_2} t_5^k t_6^l = 0,$$

then $a_{(\epsilon_1,\epsilon_2,\underline{v})} = 0$ for all $(\epsilon_1,\epsilon_2,\underline{v}) \in I$.

Proof. From Proposition 3.2.2, the result is obvious when $k, l \ge 0$. When k (resp. l) is negative, then one can multiply the above equality enough times (on the right) by t_5 (resp. t_6) to kill all the negative powers and then apply Proposition 3.2.2 to complete the proof.

3.2.4 Remark. Given the basis of $A_{\alpha,\beta}$, we have computed the group of units of $A_{\alpha,\beta}$, however, we do not include the details in this thesis due to the voluminous computations

involved. We only summarize our findings below. Set

$$h_1 := e_3 e_5 + a e_4$$
 and $h_2 := (q^{-3} - q^{-9})e_2 e_4 - (q^4 - 2q^2 + 1)/(q^4 + q^2 + 1)e_3^3$.

3.2.5 Theorem. Let $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}$ and $\mathcal{U}(A_{\alpha, \beta})$ denote the group of units of $A_{\alpha, \beta}$. We have that:

$$\mathcal{U}(A_{\alpha,\beta}) = \begin{cases} \{\lambda h_1^i \mid \lambda \in \mathbb{C}^*, \ i \in \mathbb{Z}\} & \text{if } \alpha = 0\\ \{\lambda h_2^i \mid \lambda \in \mathbb{C}^*, \ i \in \mathbb{Z}\} & \text{if } \beta = 0\\ \mathbb{C}^* & \text{otherwise.} \end{cases}$$

3.3 $A_{\alpha,\beta}$ as a *q*-deformation of a quadratic extension of $A_2(\mathbb{C})$

We are now ready to establish that $A_{\alpha,\beta}$ is a q-deformation of a quadratic extension of $A_2(\mathbb{C})$. Recall that $A_2(\mathbb{C})$ is generated by x_1, x_2, y_1 and y_2 subject to the relations:

$y_1y_2 = y_2y_1$	$x_2y_1 = y_1x_2$	$x_1 x_2 = x_2 x_1$	$x_1 y_1 - y_1 x_1 = 1$
$y_1y_2 = y_2y_1$	$x_1y_2 = y_2x_1$	$x_2y_1 = y_1x_2$	$x_2y_2 - y_2x_2 = 1.$

Given the relations of $A_{\alpha,\beta}$ at the onset of this chapter, we have that $A_{1,\frac{1}{9(q^6-1)}}$ satisfies the following relations:

$$e_{2}e_{1} = q^{-3}e_{1}e_{2} \qquad e_{3}e_{1} = q^{-1}e_{1}e_{3} - (q+q^{-1}+q^{-3})e_{2}$$

$$e_{3}e_{2} = q^{-3}e_{2}e_{3} \qquad e_{4}e_{1} = e_{1}e_{4} + (1-q^{2})e_{3}^{2}$$

$$e_{4}e_{2} = q^{-3}e_{2}e_{4} - \frac{q^{4}-2q^{2}+1}{q^{4}+q^{2}+1}e_{3}^{3} \qquad e_{4}e_{3} = q^{-3}e_{3}e_{4}$$

$$e_{5}e_{1} = qe_{1}e_{5} - (1+q^{2})e_{3} \qquad e_{5}e_{2} = e_{2}e_{5} + (1-q^{2})e_{3}^{2}$$

$$e_{5}e_{3} = q^{-1}e_{3}e_{5} - (q+q^{-1}+q^{-3})e_{4} \qquad e_{5}e_{4} = q^{-3}e_{4}e_{5}$$

$$e_{6}e_{1} = q^{3}e_{1}e_{6} - q^{3}e_{5} \qquad e_{6}e_{2} = q^{3}e_{2}e_{6} + (q^{4} + q^{2} - 1)e_{4} + (q^{2} - q^{4})e_{3}e_{5}$$

$$e_{6}e_{3} = e_{3}e_{6} + (1 - q^{2})e_{5}^{2} \qquad e_{6}e_{4} = q^{-3}e_{4}e_{6} - \frac{q^{4} - 2q^{2} + 1}{q^{4} + q^{2} + 1}e_{5}^{3}$$

$$e_{6}e_{5} = q^{-3}e_{5}e_{6},$$

and

$$(q^{-2}-1)e_1e_3e_5 + (q^2+1+q^{-2})e_1e_4 + (q^2+1+q^{-2})e_2e_5 - q^4e_3^2 = q^{-2}-1,$$

$$\begin{aligned} (q^{6}-1)e_{2}e_{4}e_{6} + \frac{2q^{-1}-q^{-3}-q}{q^{4}+q^{2}+1}e_{2}e_{5}^{3} + \frac{2q^{-1}-q^{-3}-q}{q^{4}+q^{2}+1}e_{3}^{3}e_{6} \\ + \frac{(q^{6}-1)(q^{13}-q^{11})}{(q^{4}+q^{2}+1)^{2}}e_{3}^{2}e_{5}^{2} - \frac{q^{9}(q^{6}-1)}{q^{4}+q^{2}+1}e_{3}e_{4}e_{5} + q^{12}e_{4}^{2} = \frac{1}{9}. \end{aligned}$$

Note, we have made substitutions for a, a', b, b', c' and d' (see Appendix C).

Set $C := \mathbb{C}[z^{\pm 1}]$. One can define a $C[(z^4 + z^2 + 1)^{-1}]$ -algebra A_z generated by e_1, \cdots, e_6 subject to the following relations:

$$\begin{split} e_{2}e_{1} &= z^{-3}e_{1}e_{2} & e_{3}e_{1} &= z^{-1}e_{1}e_{3} - (z + z^{-1} + z^{-3})e_{2} \\ e_{3}e_{2} &= z^{-3}e_{2}e_{3} & e_{4}e_{1} &= e_{1}e_{4} + (1 - z^{2})e_{3}^{2} \\ e_{4}e_{2} &= z^{-3}e_{2}e_{4} - \frac{z^{4} - 2z^{2} + 1}{z^{4} + z^{2} + 1}e_{3}^{3} & e_{4}e_{3} &= z^{-3}e_{3}e_{4} \\ e_{5}e_{1} &= ze_{1}e_{5} - (1 + z^{2})e_{3} & e_{5}e_{2} &= e_{2}e_{5} + (1 - z^{2})e_{3}^{2} \\ e_{5}e_{3} &= z^{-1}e_{3}e_{5} - (z + z^{-1} + z^{-3})e_{4} & e_{5}e_{4} &= z^{-3}e_{4}e_{5} \\ e_{6}e_{1} &= z^{3}e_{1}e_{6} - z^{3}e_{5} & e_{6}e_{2} &= z^{3}e_{2}e_{6} + (z^{4} + z^{2} - 1)e_{4} + (z^{2} - z^{4})e_{3}e_{5} \\ e_{6}e_{3} &= e_{3}e_{6} + (1 - z^{2})e_{5}^{2} & e_{6}e_{4} &= z^{-3}e_{4}e_{6} - \frac{z^{4} - 2z^{2} + 1}{z^{4} + z^{2} + 1}e_{5}^{3} \\ e_{6}e_{5} &= z^{-3}e_{5}e_{6}, \end{split}$$

$$(z^{-2}-1)e_1e_3e_5 + (z^2+1+z^{-2})e_1e_4 + (z^2+1+z^{-2})e_2e_5 - z^4e_3^2 = z^{-2}-1$$
, and

$$(z^{6}-1)e_{2}e_{4}e_{6} + \frac{2z^{-1}-z^{-3}-z}{z^{4}+z^{2}+1}e_{2}e_{5}^{3} + \frac{2z^{-1}-z^{-3}-z}{z^{4}+z^{2}+1}e_{3}^{3}e_{6} + \frac{(z^{6}-1)(z^{13}-z^{11})}{(z^{4}+z^{2}+1)^{2}}e_{3}^{2}e_{5}^{2} - \frac{z^{9}(z^{6}-1)}{z^{4}+z^{2}+1}e_{3}e_{4}e_{5} + z^{12}e_{4}^{2} = \frac{1}{9}$$

Observe that A_1 (i.e. when z = 1) satisfies the following relations:

$$e_{2}e_{1} = e_{1}e_{2} \qquad e_{3}e_{1} = e_{1}e_{3} - 3e_{2} \qquad e_{3}e_{2} = e_{2}e_{3}$$

$$e_{4}e_{1} = e_{1}e_{4} \qquad e_{4}e_{2} = e_{2}e_{4} \qquad e_{4}e_{3} = e_{3}e_{4}$$

$$e_{5}e_{1} = e_{1}e_{5} - 2e_{3} \qquad e_{5}e_{2} = e_{2}e_{5} \qquad e_{5}e_{3} = e_{3}e_{5} - 3e_{4}$$

$$e_{5}e_{4} = e_{4}e_{5} \qquad e_{6}e_{1} = e_{1}e_{6} - e_{5} \qquad e_{6}e_{2} = e_{2}e_{6} + e_{4}$$

$$e_{6}e_{3} = e_{3}e_{6} \qquad e_{6}e_{4} = e_{4}e_{6} \qquad e_{6}e_{5} = e_{5}e_{6}$$

$$e_{4}^{2} - 1/9 = 0 \qquad e_{3}^{2} - 3e_{1}e_{4} - 3e_{2}e_{5} = 0.$$

3.3.1 Lemma. $e_4 \in Z(A_1)$ and it is also invertible.

Proof. Since $e_4e_i = e_ie_4$ for all $1 \le i \le 6$, we have that $e_4 \in Z(A_1)$. Again, from $e_4^2 - 1/9 = 0$, we have that $e_4(9e_4) = (9e_4)e_4 = 1$. Hence e_4 is invertible with $e_4^{-1} = 9e_4$.

Given that $e_4^{-1} = 9e_4$ and $e_4 \in Z(A_1)$, it follows from the relation $e_3^2 - 3e_1e_4 - 3e_2e_5 = 0$ that $e_1 = 3e_3^2e_4 - 9e_2e_4e_5$. Therefore, A_1 can be generated by only e_2, \dots, e_6 . All these generators commute except

$$e_6e_2 = e_2e_6 + e_4$$
 and $e_5e_3 = e_3e_5 - 3e_4$.

Since e_4 is invertible, one can also verify that $9e_2e_4$, $3e_3e_4$, e_4 , e_5 and e_6 generate A_1 .

Let R be an algebra generated by f_2, f_3, f_4, f_5, f_6 subject to the following defining relations:

$$f_3f_2 = f_2f_3 \qquad f_4f_2 = f_2f_4 \qquad f_4f_3 = f_3f_4$$

$$f_5f_2 = f_2f_5 \qquad f_5f_4 = f_4f_5 \qquad f_6f_3 = f_3f_6$$

$$f_6f_4 = f_4f_6 \qquad f_6f_5 = f_5f_6 \qquad f_4^2 = 1/9$$

$$f_6f_2 = f_2f_6 + 1 \qquad f_5f_3 = f_3f_5 - 1.$$

3.3.2 Proposition. $R \cong A_1$.

Proof. One can define a homomorphism $\phi: R \longrightarrow A_1$ such that

$$\phi(f_2) = 9e_2e_4$$
 $\phi(f_3) = 3e_3e_4$ $\phi(f_4) = e_4$ $\phi(f_5) = e_5$ $\phi(f_6) = e_6$

Recall, $e_4^2 = 1/9$. One can confirm that ϕ is indeed a homomorphism by verifying that it is compatible with the relations of R. We check this on the relation $f_6f_2 - f_2f_6 = 1$ and $f_3f_5 - f_5f_3 = 1$, and leave the remaining ones for the reader to verify. We do that as follows: $\phi(f_6)\phi(f_2) - \phi(f_2)\phi(f_6) = 9e_6e_2e_4 - 9e_2e_4e_6 = 9e_4(e_6e_2 - e_2e_6) =$ $9e_4^2 = 9(1/9) = 1$ as expected. Also, $\phi(f_3)\phi(f_5) - \phi(f_5)\phi(f_3) = 3e_3e_4e_5 - 3e_5e_3e_4 =$ $3e_4(e_3e_5 - e_5e_3) = 3e_4(3e_4) = 9e_4^2 = 9(1/9) = 1$.

Conversely, one can also define a homomorphism $\varphi: A_1 \longrightarrow R$ such that

$$\varphi(e_1) = 3f_3^2 f_4 - f_2 f_5$$
 $\varphi(e_2) = f_2 f_4$ $\varphi(e_3) = 3f_3 f_4$
 $\varphi(e_4) = f_4$ $\varphi(e_5) = f_5$ $\varphi(e_6) = f_6.$

We check this on the relation $e_3^2 - 3e_1e_4 - 3e_2e_5 = 0$, and leave the remaining ones for the reader to verify. We do that as follows: $\varphi(e_3)^2 - 3\varphi(e_1)\varphi(e_4) - 3\varphi(e_2)\varphi(e_5) = (3f_3f_4)^2 - 3(3f_3^2f_4 - f_2f_5)f_4 - 3f_2f_4f_5 = 9f_3^2f_4^2 - 9f_3^2f_4^2 + 3f_2f_4f_5 - 3f_2f_4f_5 = 0$ as expected.

Now, $\phi \circ \varphi = id_{A_1}$ (one can check this on the generators of A_1) and $\varphi \circ \phi = id_R$ (one can also check this on the generators of R) (note, id_* means an identity map of the algebra *). This confirms that $R \cong A_1$.

The corollary below can easily be deduced from the above proposition.

3.3.3 Corollary. Set $\mathbb{F} := \mathbb{C}[f_4]/\langle f_4^2 - 1/9 \rangle$, we have that $R \cong A_2(\mathbb{F})$, where $A_2(\mathbb{F})$ is the second Weyl algebra over the ring \mathbb{F} .

3.3.4 Remark. Observe that the subalgebra B of R generated by f_2, f_3, f_5, f_6 is isomorphic to $A_2(\mathbb{C})$ and $R \cong B[f_4] \cong A_2(\mathbb{C})[f_4]$. Therefore, R is a quadratic extension of $A_2(\mathbb{C})$. Note, $A_{1,\frac{1}{9(q^6-1)}}$ is a q-deformation of $A_1 \cong R \cong A_2(\mathbb{F}) \cong A_2(\mathbb{C})[f_4]$.



3.3.5 Remark. Given the algebra $A_{\alpha,\beta}$, the construction in this section was done for $\alpha = 1$ and $\beta = \frac{1}{9(q^6 - 1)}$. However, any $\alpha \in \mathbb{R}$ and β of the form $\beta = \frac{\varsigma}{(q^6 - 1)}$, where $\varsigma \in \mathbb{R}^*$, will also work.

Chapter 4

Derivations of the simple quotients of $U_q^+(G_2)$

In this chapter, we compute the derivations of the algebra $A_{\alpha,\beta}$. In finding the derivations of $A_{\alpha,\beta}$, we use the theory of deleting derivations algorithm by Cauchon [8] and localization theory to embed $A_{\alpha,\beta}$ into a suitable quantum torus. This is because every derivation of the quantum torus, through the work of Osborn and Passman [40], is known to be the sum of an inner derivation and a central/scalar derivation. Since $A_{\alpha,\beta}$ can be embedded into a quantum torus, we extend every derivation of $A_{\alpha,\beta}$ to a derivation of the quantum torus, and then restrict the derivations of the quantum torus back to a derivation of $A_{\alpha,\beta}$. We conclude that every derivation of $A_{\alpha,\beta}$ is inner when α and β are non-zero. However, when either α or β is zero, we conclude that every derivation of $A_{\alpha,\beta}$ is the sum of an inner and a scalar derivation. In fact, the first Hochschild cohomology group of $A_{\alpha,\beta}$ is of dimension 0 when α and β are non-zero and 1 when either α or β is zero.

4.1 Preliminaries

Let $2 \leq j \leq 7$ and $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}$. Set

$$A_{\alpha,\beta}^{(j)} := \frac{A^{(j)}}{\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle},$$

where $A^{(j)}$ is defined in Section 2.2 and, Ω_1 and Ω_2 are the generators of the center of $A^{(j)}$ (Subsection 2.2.3). Recall that $A^{(7)} = A = U_q^+(G_2)$ (Section 2.2). It follows that $A_{\alpha,\beta}^{(7)} = A_{\alpha,\beta}$. For each $2 \leq j \leq 7$, denote the canonical images of the generators $E_{i,j}$ of $A^{(j)}$ in $A_{\alpha,\beta}^{(j)}$ by $e_{i,j}$ for all $1 \leq i \leq 6$. Since the data of the DDA of $A_{\alpha,\beta}$ is going to be crucial in this section, we present them below (note, we deduce them from that of $U_q^+(G_2)$ in Section 2.2):

$$e_{1,6} = e_1 + re_5e_6^{-1}$$

$$e_{2,6} = e_2 + te_3e_5e_6^{-1} + ue_4e_6^{-1} + ne_5^3e_6^{-2}$$

$$e_{3,6} = e_3 + se_5^2e_6^{-1}$$

$$e_{4,6} = e_4 + be_5^3e_6^{-1}$$

$$e_{1,5} = e_{1,6} + he_{3,6}e_{5,6}^{-1} + ge_{4,6}e_{5,6}^{-2}$$

$$e_{2,5} = e_{2,6} + fe_{3,6}^2e_{5,6}^{-1} + pe_{3,6}e_{4,6}e_{5,6}^{-2} + ee_{4,6}^2e_{5,6}^{-3}$$

$$e_{1,4} = e_{1,5} + se_{3,5}^2e_{4,5}^{-1}$$

$$e_{1,3} = e_{1,4} + ae_{2,4}e_{3,4}^{-1}$$

$$t_1 := e_{1,2} = e_{1,3}$$

$$t_2 := e_{2,2} = e_{2,3} = e_{2,4}$$

$$t_3 := e_{3,2} = e_{3,3} = e_{3,4} = e_{3,5}$$

$$t_4 := e_{4,2} = e_{4,3} = e_{4,4} = e_{4,5} = e_{4,6}$$

$$t_5 := e_{5,2} = e_{5,3} = e_{5,4} = e_{5,5} = e_{5,6} = e_{5}$$

$$t_6 := e_{6,2} = e_{6,3} = e_{6,4} = e_{6,5} = e_{6,6} = e_{6}.$$

Note, t_i is also the canonical image of T_i in $A_{\alpha,\beta}^{(2)}$ for each $1 \le i \le 6$. For each $3 \le j \le 6$, define $S_j := \left\{ \lambda t_j^{i_j} t_{j+1}^{i_{j+1}} \cdots t_6^{i_6} \mid i_j, \cdots, i_6 \in \mathbb{N} \text{ and } \lambda \in \mathbb{C}^* \right\}$. One can observe that S_j is a multiplicative system of non-zero divisors (or regular elements) of $A_{\alpha,\beta}^{(j)}$. Furthermore;

 t_j, \dots, t_6 are all normal elements of $A_{\alpha,\beta}^{(j)}$. It follows from Section 1.7 that S_j is an Ore set in $A_{\alpha,\beta}^{(j)}$. As a result, one can localize $A_{\alpha,\beta}^{(j)}$ at S_j as follows:

$$R_j := A_{\alpha,\beta}^{(j)} S_j^{-1}.$$

Let $3 \leq j \leq 6$, and set $\Sigma_j := \{t_j^k \mid k \in \mathbb{N}\}$. By [8, Theorem 3.2.1], Σ_j is an Ore set in both $A_{\alpha,\beta}^{(j)}$ and $A_{\alpha,\beta}^{(j+1)}$. Consequently,

$$A_{\alpha,\beta}^{(j)}\Sigma_j^{-1} = A_{\alpha,\beta}^{(j+1)}\Sigma_j^{-1}.$$

Similar to (2.2.1), we have that

$$R_j = R_{j+1} \Sigma_j^{-1}, \tag{4.1.1}$$

for all $2 \le j \le 6$. By convention, $R_7 := A_{\alpha,\beta}$. We also construct the following tower of algebras in a manner similar to (2.2.2):

$$R_7 = A_{\alpha,\beta} \subset R_6 = R_7 \Sigma_6^{-1} \subset R_5 = R_6 \Sigma_5^{-1} \subset R_4 = R_5 \Sigma_4^{-1} \subset R_3.$$
(4.1.2)

Note, $R_3 = A_{\alpha,\beta}^{(3)} S_3^{-1} = R_4 \Sigma_3^{-1}$ is the quantum torus $\mathscr{A}_{\alpha,\beta} = \mathbb{C}_{q^N}[t_3^{\pm 1}, t_4^{\pm 1}, t_5^{\pm 1}, t_6^{\pm 1}]$ studied in Section 3.1.

Linear bases for R_3, R_4 and R_5 .

Let $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0,0)\}$. We aim to find a basis of R_j for each j = 3, 4, 5. Since $R_3 = \mathscr{A}_{\alpha,\beta}$, the set $\{t_3^i t_4^j t_5^k t_6^l \mid i, j, k, l \in \mathbb{Z}\}$ is a \mathbb{C} -basis of R_3 .

For simplicity, we set

$$f_{1} := e_{1,4} F_{1} := E_{1,4} Z_{1} := E_{1,5} Z_{2} := e_{2,5} Z_{2} := E_{2,5}.$$

Basis for R_4 . Observe that

$$A_{\alpha,\beta}^{(4)} = \frac{A^{(4)}}{\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle}$$

where $\Omega_1 = F_1T_3T_5 + aT_2T_5$ and $\Omega_2 = T_2T_4T_6$ in $A^{(4)}$ (Subsection 2.2.3). Recall from Section 3.2 that finding a basis for the algebra A_β served as a good ground for finding a basis for $A_{\alpha,\beta}$. In a similar manner, to find a basis for R_4 , we will first and foremost find a basis for the algebra

$$A_{\beta}^{(4)}S_4^{-1} = \frac{A^{(4)}S_4^{-1}}{\langle \Omega_2 - \beta \rangle} = \frac{A^{(4)}S_4^{-1}}{\langle T_2T_4T_6 - \beta \rangle},$$

where $\beta \in \mathbb{C}^*$. We will denote the canonical images of $E_{i,4}$ (resp. T_i) in $A_{\beta}^{(4)}$ by $\widehat{e_{i,4}}$ (resp. $\widehat{t_i}$) for all $1 \leq i \leq 6$. Observe that $\widehat{t_2} = \beta \widehat{t_6}^{-1} \widehat{t_4}^{-1}$ in $A_{\beta}^{(4)} S_4^{-1}$. Note, when $\beta = 0$, then one can easily deduce that $A_{\beta}^{(4)} S_4^{-1} = A^{(4)} S_4^{-1} / \langle T_2 \rangle$, hence, $\widehat{t_2} = 0$.

4.1.1 Proposition. The set $\mathfrak{S}_4 = \left\{ \widehat{f_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} \mid (i_1, i_3, i_4, i_5, i_6) \in \mathbb{N}^2 \times \mathbb{Z}^3 \right\}$ is a \mathbb{C} -basis of $A_{\beta}^{(4)} S_4^{-1}$, where $\beta \in \mathbb{C}$.

Proof. We begin by showing that \mathfrak{S}_4 is a spanning set for $A_{\beta}^{(4)}S_4^{-1}$. It is sufficient to do this by showing that $\widehat{f_1}^{k_1}\widehat{t_2}^{k_2}\widehat{t_3}^{k_3}\widehat{t_4}^{k_4}\widehat{t_5}^{k_5}\widehat{t_6}^{k_6}$ can be written as a finite linear combination of the elements of \mathfrak{S}_4 for all $(k_1, \cdots, k_6) \in \mathbb{N}^3 \times \mathbb{Z}^3$. We do this by an induction on k_2 . The result is clear when $k_2 = 0$. Assume that the statement is true for $k_2 \ge 0$. That is,

$$\widehat{f}_1^{\ k_1} \widehat{t}_2^{\ k_2} \widehat{t}_3^{\ k_3} \widehat{t}_4^{\ k_4} \widehat{t}_5^{\ k_5} \widehat{t}_6^{\ k_6} = \sum_{\underline{i} \in I} a_{\underline{i}} \widehat{f}_1^{\ i_1} \widehat{t}_3^{\ i_3} \widehat{t}_4^{\ i_4} \widehat{t}_5^{\ i_5} \widehat{t}_6^{\ i_6},$$

where $\underline{i} = (i_1, i_3, i_4, i_5, i_6) \in I \subset \mathbb{N}^2 \times \mathbb{Z}^3$ and $a_{\underline{i}} \in \mathbb{C}$. It follows that

$$\begin{split} \widehat{f_1}^{k_1} \widehat{t_2}^{k_2+1} \widehat{t_3}^{k_3} \widehat{t_4}^{k_4} \widehat{t_5}^{k_5} \widehat{t_6}^{k_6} &= q^{\bullet} \left(\widehat{f_1}^{k_1} \widehat{t_2}^{k_2} \widehat{t_3}^{k_3} \widehat{t_4}^{k_4} \widehat{t_5}^{k_5} \widehat{t_6}^{k_6} \right) \widehat{t_2} \\ &= \sum_{\underline{i} \in I} q^{\bullet} a_{\underline{i}} \widehat{f_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} \widehat{t_2} \\ &= \sum_{\underline{i} \in I} q^{\bullet} \beta a_{\underline{i}} \widehat{f_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4-1} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6-1} \end{split}$$

 $\left(\text{note, } \widehat{t_2} = \beta \widehat{t_6}^{-1} \widehat{t_4}^{-1}\right)$. By the principle of mathematical induction, \mathfrak{S}_4 is a spanning set for $A_{\beta}^{(4)} S_4^{-1}$ for all $i_2 \in \mathbb{N}$. We now prove that \mathfrak{S}_4 is a linearly independent set. Suppose that

$$\sum_{\underline{i}\in I} a_{\underline{i}} \widehat{f_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} = 0.$$

This implies that

$$\sum_{\underline{i}\in I} a_{\underline{i}} F_1^{i_1} T_3^{i_3} T_4^{i_4} T_5^{i_5} T_6^{i_6} = (\Omega_2 - \beta)\nu_4$$

for some $\nu \in A^{(4)}S_4^{-1}$. Write $\nu = \sum_{\underline{j} \in J} b_{\underline{j}}F_1^{i_1}T_2^{i_2}T_3^{i_3}T_4^{i_4}T_5^{i_5}T_6^{i_6}$, where $\underline{j} = (i_1, i_2, i_3, i_4, i_5, i_6) \in J \subset \mathbb{N}^3 \times \mathbb{Z}^3$ and $b_{\underline{j}}$ is a family of complex numbers. Given that $\Omega_2 = T_2T_4T_6$, it follows from the above equality that

$$\sum_{\underline{i}\in I} a_{\underline{i}} F_1^{i_1} T_3^{i_3} T_4^{i_4} T_5^{i_5} T_6^{i_6} = \sum_{\underline{j}\in J} q^{\bullet} b_{\underline{j}} F_1^{i_1} T_2^{i_2+1} T_3^{i_3} T_4^{i_4+1} T_5^{i_5} T_6^{i_6+1} - \sum_{\underline{j}\in J} \beta b_{\underline{j}} F_1^{i_1} T_2^{i_2} T_3^{i_3} T_4^{i_4} T_5^{i_5} T_6^{i_6}.$$

Suppose that there exists $(i_1, \cdots, i_6) \in J$ such that $b_{(i_1, \cdots, i_6)} \neq 0$. Let $(w_1, \cdots, w_6) \in J$ be the greatest element of J with respect to $<_2$ ¹ such that $b_{(w_1, \cdots, w_6)} \neq 0$. Note, $(F_1^{i_1}T_2^{i_2}T_3^{i_3}T_4^{i_4}T_5^{i_5}T_6^{i_6})_{(i_1, \cdots, i_6) \in J}$ is a basis of $A^{(4)}S_4^{-1}$. Identifying the coefficients of $F_1^{w_1}T_2^{w_2+1}T_3^{w_3}T_4^{w_4+1}T_5^{w_5}T_6^{w_6+1}$, we have that $b_{(w_1, \cdots, w_6)} = 0$. This is a contradiction to our assumption, hence $b_{(i_1, \cdots, i_6)} = 0$ for all $(i_1, \cdots, i_6) \in J$. This implies that

$$\sum_{\underline{i}\in I} a_{\underline{i}} F_1^{i_1} T_3^{i_3} T_4^{i_4} T_5^{i_5} T_6^{i_6} = 0.$$

 $[\]begin{array}{c} {}^{1}(i_{1},i_{2},i_{3},i_{4},i_{5},i_{6}) <_{2}(w_{1},w_{2},w_{3},w_{4},w_{5},w_{6}) \text{ if } [w_{2}>i_{2}] \text{ or } [w_{2}=i_{2},\ w_{1}>i_{1}] \text{ or } [w_{2}=i_{2},\ w_{1}=i_{1},\ w_{3}>i_{3}] \text{ or } \cdots \text{ or } [w_{l}=i_{l},\ w_{6}\geq t_{6},\ l=2,1,3,4,5] \text{ for all } (i_{1},\cdots,i_{6})\in J. \end{array}$

Consequently, $a_{\underline{i}} = 0$ for all $\underline{i} \in I$. Therefore, \mathfrak{S}_4 is a linearly independent set.

In $R_4 = A_{\alpha,\beta}^{(4)}S_4^{-1}$, we have the following two relations: $f_1t_3t_5 + at_2t_5 = \alpha$ and $t_2t_4t_6 = \beta$. This implies that $f_1t_3 = \alpha t_5^{-1} - at_2$ and $t_2 = \beta t_6^{-1}t_4^{-1}$. Putting these two relations together, we have that

$$f_1 t_3 = \alpha t_5^{-1} - a\beta t_6^{-1} t_4^{-1}. \tag{4.1.3}$$

Note, we will usually identify R_4 with $A_{\beta}^{(4)}S_4^{-1}/\langle \widehat{\Omega}_1 - \alpha \rangle$.

4.1.2 Proposition. The set $\mathcal{B}_4 = \left\{ f_1^{i_1} t_4^{i_4} t_5^{i_5} t_6^{i_6}, t_3^{i_3} t_4^{i_4} t_5^{i_5} t_6^{i_6} \mid i_1, i_3 \in \mathbb{N} \text{ and } i_4, i_5, i_6 \in \mathbb{Z} \right\}$ is a \mathbb{C} -basis of R_4 .

Proof. Since $(\widehat{f}_1^{k_1}\widehat{t}_3^{k_3}\widehat{t}_4^{k_4}\widehat{t}_5^{k_5}\widehat{t}_6^{k_6})_{(k_1,k_3,\cdots,k_6)\in\mathbb{N}^2\times\mathbb{Z}^3}$ is a basis of $A_\beta^{(4)}S_4^{-1}$ (Proposition 4.1.1), the set $(f_1^{k_1}t_3^{k_3}t_4^{k_4}t_5^{k_5}t_6^{k_6})_{(k_1,k_3,\cdots,k_6)\in\mathbb{N}^2\times\mathbb{Z}^3}$ spans R_4 . We show that \mathcal{B}_4 is a spanning set of R_4 by showing that $f_1^{k_1}t_3^{k_3}t_4^{k_4}t_5^{k_5}t_6^{k_6}$ can be written as a finite linear combination of the elements of \mathcal{B}_4 for all $(k_1,k_3,\cdots,k_6)\in\mathbb{N}^2\times\mathbb{Z}^3$. By Proposition 4.1.1, it is sufficient to do this by induction on k_1 . The result is clear when $k_1 = 0$. Assume that the statement is true for $k_1 \ge 0$. That is,

$$f_1^{k_1} t_3^{k_3} t_4^{k_4} t_5^{k_5} t_6^{k_6} = \sum_{\underline{i} \in I_1} a_{\underline{i}} f_1^{i_1} t_4^{i_4} t_5^{i_5} t_6^{i_6} + \sum_{\underline{j} \in I_2} b_{\underline{j}} t_3^{i_3} t_4^{i_4} t_5^{i_5} t_6^{i_6},$$

where $\underline{i} = (i_1, i_4, i_5, i_6) \in I_1 \subset \mathbb{N} \times \mathbb{Z}^3$ and $\underline{j} = (i_3, i_4, i_5, i_6) \in I_2 \subset \mathbb{N} \times \mathbb{Z}^3$. Note, $a_{\underline{i}}$ and b_j are all complex numbers. It follows that

$$f_1^{k_1+1}t_3^{k_3}t_4^{k_4}t_5^{k_5}t_6^{k_6} = f_1\left(f_1^{k_1}t_3^{k_3}t_4^{k_4}t_5^{k_5}t_6^{k_6}\right) = \sum_{\underline{i}\in I_1}a_{\underline{i}}f_1^{i_1+1}t_4^{i_4}t_5^{i_5}t_6^{i_6} + \sum_{\underline{j}\in I_2}b_{\underline{j}}f_1t_3^{i_3}t_4^{i_4}t_5^{i_5}t_6^{i_6}.$$

Clearly, the monomial $f_1^{i_1+1}t_4^{i_4}t_5^{i_5}t_6^{i_6} \in \text{Span}(\mathcal{B}_4)$. We have to also show that $f_1t_3^{i_3}t_4^{i_4}t_5^{i_5}t_6^{i_6} \in \text{Span}(\mathcal{B}_4)$ for all $i_3 \in \mathbb{N}$ and $i_4, i_5, i_6 \in \mathbb{Z}$. This can easily be achieved by an induction on i_3 , and the use of the relation $f_1t_3 = \alpha t_5^{-1} - \alpha\beta t_6^{-1}t_4^{-1}$. Therefore, by the principle of mathematical induction, \mathcal{B}_4 is a spanning set of R_4 over \mathbb{C} .

We prove that \mathcal{B}_4 is a linearly independent set. Suppose that

$$\sum_{\underline{i}\in I_1} a_{\underline{i}} f_1^{i_1} t_4^{i_4} t_5^{i_5} t_6^{i_6} + \sum_{\underline{j}\in I_2} b_{\underline{j}} t_3^{i_3} t_4^{i_4} t_5^{i_5} t_6^{i_6} = 0.$$

It follows that there exists $\nu \in A_{\beta}^{(4)}S_4^{-1}$ such that

$$\sum_{\underline{i}\in I_1} a_{\underline{i}} \widehat{f_1}^{i_1} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} + \sum_{\underline{j}\in I_2} b_{\underline{j}} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} = \left(\widehat{\Omega}_1 - \alpha\right) \nu.$$

Write $\nu = \sum_{\underline{l}\in J} c_{\underline{l}} \widehat{f_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6}$, where $\underline{l} = (i_1, i_3, i_4, i_5, i_6) \in J \subset \mathbb{N}^2 \times \mathbb{Z}^3$ and $c_{\underline{l}} \in \mathbb{C}$. Note, $\widehat{t_2} = \beta \widehat{t_6}^{-1} \widehat{t_4}^{-1}$. We have that $\widehat{\Omega}_1 = \widehat{f_1} \widehat{t_3} \widehat{t_5} + a \widehat{t_2} \widehat{t_5} = \widehat{f_1} \widehat{t_3} \widehat{t_5} + a \beta \widehat{t_6}^{-1} \widehat{t_4}^{-1} \widehat{t_5}$. Therefore,

$$\begin{split} \sum_{\underline{i}\in I_1} a_{\underline{i}} \widehat{f_1}^{i_1} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} + \sum_{\underline{j}\in I_2} b_{\underline{j}} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} &= \sum_{\underline{l}\in J} q^{\bullet} c_{\underline{l}} \widehat{f_1}^{i_1+1} \widehat{t_3}^{i_3+1} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5+1} \widehat{t_6}^{i_6} \\ &+ \sum_{\underline{l}\in J} q^{\bullet} \beta a c_{\underline{l}} \widehat{f_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4-1} \widehat{t_5}^{i_5+1} \widehat{t_6}^{i_6-1} \\ &- \sum_{\underline{l}\in J} \alpha c_{\underline{l}} \widehat{f_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6}. \end{split}$$

Suppose that there exists $(i_1, i_3, i_4, i_5, i_6) \in J$ such that $c_{(i_1, i_3, i_4, i_5, i_6)} \neq 0$.

Let $(w_1, w_3, w_4, w_5, w_6) \in J$ be the greatest element (in the lexicographic order on $\mathbb{N}^2 \times \mathbb{Z}^3$) of J such that $c_{(w_1, w_3, w_4, w_5, w_6)} \neq 0$. Since $\left(\widehat{f_1}^{k_1} \widehat{t_3}^{k_3} \widehat{t_4}^{k_4} \widehat{t_5}^{k_5} \widehat{t_6}^{k_6}\right)_{(k_1, k_3, \cdots, k_6) \in \mathbb{N}^2 \times \mathbb{Z}^3}$ is a basis of $A^{(4)} S_4^{-1}$, it implies that the coefficients of $\widehat{f_1}^{w_1+1} \widehat{t_3}^{w_3+1} \widehat{t_4}^{w_4} \widehat{t_5}^{w_5+1} \widehat{t_6}^{w_6}$ in the above equality can be identified as: $q^{\bullet} c_{(w_1, w_3, w_4, w_5, w_6)} = 0$. Hence, $c_{(w_1, w_3, w_4, w_5, w_6)} = 0$, a contradiction! Therefore, $c_{(i_1, i_3, i_4, i_5, i_6)} = 0$ for all $(i_1, i_3, i_4, i_5, i_6) \in J$. This further implies that

$$\sum_{\underline{i}\in I_1} a_{\underline{i}} \widehat{f_1}^{i_1} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} + \sum_{\underline{j}\in I_2} b_{\underline{j}} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} = 0.$$

Consequently, $a_{\underline{i}}$ and b_j are all zero. In conclusion, \mathcal{B}_4 is a linearly independent set.

Basis for
$$R_5$$
. We will identify R_5 with $A_{\alpha}^{(5)}S_5^{-1}/\langle \widehat{\Omega}_2 - \beta \rangle$, where $A_{\alpha}^{(5)}S_5^{-1} = \frac{A^{(5)}S_5^{-1}}{\langle \Omega_1 - \alpha \rangle}$.

Note, the canonical images of $E_{i,j}$ (resp. T_i) in $A_{\alpha}^{(5)}$ will be denoted by $\widehat{e_{i,j}}$ (resp. $\widehat{t_i}$). We now find a basis for $A_{\alpha}^{(5)}S_5^{-1}$. Recall from Subsection 2.2.3 that $\Omega_1 = Z_1T_3T_5 + aZ_2T_5$ and $\Omega_2 = Z_2T_4T_6 + bT_3^3T_6$ in $A^{(5)}$ (remember, $Z_1 := E_{1,5}$ and $Z_2 := E_{2,5}$). Since $z_2t_4t_6 + bt_3^3t_6 = \beta$ and $\widehat{z_1}\widehat{t_3}\widehat{t_5} + a\widehat{z_2}\widehat{t_5} = \alpha$ in R_5 and $A_{\alpha}^{(5)}S_5^{-1}$ respectively, we have the relation $\widehat{z_2} = \frac{1}{a} \left(\alpha \widehat{t_5}^{-1} - \widehat{z_1}\widehat{t_3} \right)$ in $A_{\alpha}^{(5)}S_5^{-1}$ and, in R_5 , we have the following two relations:

$$z_2 = \frac{1}{a} \left(\alpha t_5^{-1} - z_1 t_3 \right). \tag{4.1.4}$$

$$t_3^3 = \frac{1}{b} \left(\beta t_6^{-1} - z_2 t_4\right) = \frac{\beta}{b} t_6^{-1} - \frac{q^3 \alpha}{ab} t_4 t_5^{-1} + \frac{1}{ab} z_1 t_3 t_4.$$
(4.1.5)

4.1.3 Proposition. The set $\mathfrak{S}_5 = \left\{ \widehat{z_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} \mid (i_1, i_3, \cdots, i_6) \in \mathbb{N}^3 \times \mathbb{Z}^2 \right\}$ is a \mathbb{C} -basis of $A_{\alpha}^{(5)} S_5^{-1}$, where $\alpha \in \mathbb{C}$.

Proof. The proof is similar to that of Proposition 4.1.1. Hence, we will provide the proof without many details. Assume that the statement is true for $k_2 \ge 0$. That is,

$$\widehat{z_1}^{k_1} \widehat{z_2}^{k_2} \widehat{t_3}^{k_3} \widehat{t_4}^{k_4} \widehat{t_5}^{k_5} \widehat{t_6}^{k_6} = \sum_{\underline{i} \in I} a_{\underline{i}} \widehat{z_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6},$$

where $\underline{i} = (i_1, i_3, i_4, i_5, i_6) \in I \subset \mathbb{N}^3 \times \mathbb{Z}^2$ and $a_{\underline{i}} \in \mathbb{C}$. Given that $\widehat{z}_2 = \frac{1}{a} \left(\alpha \widehat{t_5}^{-1} - \widehat{z_1} \widehat{t_3} \right)$, it follows from the inductive hypothesis that

$$\begin{split} \widehat{z_1}^{k_1} \widehat{z_2}^{k_2+1} \widehat{t_3}^{k_3} \widehat{t_4}^{k_4} \widehat{t_5}^{k_5} \widehat{t_6}^{k_6} =& q^{\bullet} \widehat{z_2} \left(\widehat{z_1}^{k_1} \widehat{z_2}^{k_2} \widehat{t_3}^{k_3} \widehat{t_4}^{k_4} \widehat{t_5}^{k_5} \widehat{t_6}^{k_6} \right) \\ &= \sum_{\underline{i} \in I} q^{\bullet} a_{\underline{i}} \widehat{z_1}^{i_1} \widehat{z_2} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} \\ &= \sum_{\underline{i} \in I} \frac{q^{\bullet} \alpha}{a} a_{\underline{i}} \widehat{z_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5-1} \widehat{t_6}^{i_6} \\ &- \sum_{\underline{i} \in I} \frac{q^{\bullet} \alpha}{a} a_{\underline{i}} \widehat{z_1}^{i_1+1} \widehat{t_3}^{i_3+1} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5-1} \widehat{t_6}^{i_6} \end{split}$$

where $(k_1, \cdots, k_6) \in \mathbb{N}^4 \times \mathbb{Z}^2$. Hence, \mathfrak{S}_5 is a spanning set of $A_{\alpha}^{(5)}S_5^{-1}$.

Secondly, suppose that

$$\sum_{\underline{i}\in I} a_{\underline{i}} \widehat{z_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} = 0.$$

Then,

$$\sum_{\underline{i}\in I} a_{\underline{i}} Z_1^{i_1} T_3^{i_3} T_4^{i_4} T_5^{i_5} T_6^{i_6} = (\Omega_1 - \alpha)\nu,$$

with $\nu \in A^{(5)}S_5^{-1}$. Write $\nu = \sum_{\underline{j}\in J} b_{\underline{j}}Z_1^{i_1}Z_2^{i_2}T_3^{i_3}T_4^{i_4}T_5^{i_5}T_6^{i_6}$, where $\underline{j} = (i_1, i_3, i_4, i_5, i_6) \in J \subset \mathbb{N}^4 \times \mathbb{Z}^2$ and $b_{\underline{j}}$ is a family of complex numbers. Given that $\Omega_1 = Z_1T_3T_5 + aZ_2T_5$, it follows that

$$\begin{split} \sum_{\underline{i}\in I} a_{\underline{i}} Z_1^{i_1} T_3^{i_3} T_4^{i_4} T_5^{i_5} T_6^{i_6} &= \sum_{\underline{j}\in J} q^{\bullet} b_{\underline{j}} Z_1^{i_1+1} Z_2^{i_2} T_3^{i_3+1} T_4^{i_4} T_5^{i_5+1} T_6^{i_6} \\ &- \sum_{\underline{j}\in J} \alpha b_{\underline{j}} Z_1^{i_1} Z_2^{i_2} T_3^{i_3} T_4^{i_4} T_5^{i_5} T_6^{i_6} \\ &+ \sum_{\underline{j}\in J} q^{\bullet} a b_{\underline{j}} Z_1^{i_1} Z_2^{i_2+1} T_3^{i_3} T_4^{i_4} T_5^{i_5+1} T_6^{i_6} \end{split}$$

Suppose that there exists $(i_1, i_2, i_3, i_4, i_5, i_6) \in J$ such that $b_{(i_1, i_2, i_3, i_4, i_5, i_6)} \neq 0$. Let $(w_1, w_2, w_3, w_4, w_5, w_6) \in J$ be the greatest element (in the lexicographic order on $\mathbb{N}^4 \times \mathbb{Z}^2$) of J such that $b_{(w_1, w_2, w_3, w_4, w_5, w_6)} \neq 0$. Since the family of monomials $(Z_1^{i_1} Z_2^{i_2} T_3^{i_3} T_4^{i_4} T_5^{i_5} T_6^{i_6})_{(i_1, \cdots, k_6) \in \mathbb{N}^4 \times \mathbb{Z}^2}$ is a basis of $A^{(5)} S_5^{-1}$, the coefficients of $Z_1^{w_1+1} Z_2^{w_2} T_3^{w_3+1} T_4^{w_4} T_5^{w_5+1} T_6^{w_6}$ in the above equality can be identified as: $q^{\bullet} b_{(w_1, w_2, w_3, w_4, w_5, w_6)} = 0$. This implies that $b_{(w_1, w_2, w_3, w_4, w_5, w_6)} = 0$, a contradiction! Hence, $b_{(i_1, \cdots, i_6)} = 0$ for all $(i_1, \cdots, i_6) \in J$. Consequently, $a_{(i_1, i_3, i_4, i_5, i_6)} = 0$ for all $(i_1, i_3, i_4, i_5, i_6) \in I$. Therefore, \mathfrak{S}_5 is a linearly independent set.

4.1.4 Proposition. The set $\mathcal{B}_5 = \left\{ z_1^{i_1} t_3^{\xi} t_4^{i_4} t_5^{i_5} t_6^{i_6} \mid (\xi, i_1, i_4, i_5, i_6) \in \{0, 1, 2\} \times \mathbb{N}^2 \times \mathbb{Z}^2 \right\}$ is a \mathbb{C} -basis of R_5 .

Proof. Since R_5 is identified with $A_{\alpha}^{(5)}S_5^{-1}/\langle \widehat{\Omega}_2 - \beta \rangle$ and \mathfrak{S}_5 is a basis for $A_{\alpha}^{(5)}S_5^{-1}$

(Proposition 4.1.3), we show that the spanning set $z_1^{k_1}t_3^{k_3}t_4^{k_4}t_5^{k_5}t_6^{k_6}$ of R_5 can be written as a finite linear combination of the elements of \mathcal{B}_5 for all $(k_1, k_3, \dots, k_6) \in \mathbb{N}^3 \times \mathbb{Z}^2$. By Proposition 4.1.3, it is sufficient to do this by an induction on k_3 . The result is obvious when $k_3 = 0, 1$ or 2. For $k_3 \ge 2$, suppose that

$$z_1^{k_1} t_3^{k_3} t_4^{k_4} t_5^{k_5} t_6^{k_6} = \sum_{(\xi,\underline{i})\in I} a_{(\xi,\underline{i})} z_1^{i_1} t_3^{\xi} t_4^{i_4} t_5^{i_5} t_6^{i_6},$$

where I is a finite subset of $\{0, 1, 2\} \times \mathbb{N}^2 \times \mathbb{Z}^2$ and $a_{(\xi,\underline{i})}$ are all complex numbers. It follows that

$$z_1^{k_1}t_3^{k_3+1}t_4^{k_4}t_5^{k_5}t_6^{k_6} = q^{\bullet}\left(z_1^{k_1}t_3^{k_3}t_4^{k_5}t_5^{k_6}\right)t_3 = \sum_{(\xi,\underline{i})\in I}q^{\bullet}a_{(\xi,\underline{i})}z_1^{i_1}t_3^{\xi+1}t_4^{i_4}t_5^{i_5}t_6^{i_6}$$

Now, $z_1^{i_1}t_3^{\xi+1}t_4^{i_4}t_5^{i_5}t_6^{i_6} \in \text{Span}(\mathcal{B}_5)$ when $\xi = 0, 1$. For $\xi = 2$, one can easily verify that $z_1^{i_1}t_3^3t_4^{i_4}t_5^{i_5}t_6^{i_6} \in \text{Span}(\mathcal{B}_5)$ by using the relation (4.1.5). Therefore, by the principle of mathematical induction, \mathcal{B}_5 spans R_5 .

We now prove that \mathcal{B}_5 is a linearly independent set. Suppose that

$$\sum_{(\xi,\underline{i})\in I} a_{(\xi,\underline{i})} z_1^{i_1} t_3^{\xi} t_4^{i_4} t_5^{i_5} t_6^{i_6} = 0.$$

Since R_5 is identified with $A_{\alpha}^{(5)}S_5^{-1}/\langle\widehat{\Omega}_2-\beta\rangle,$ we have that

$$\sum_{(\underline{\xi},\underline{i})\in I} a_{(\underline{\xi},\underline{i})} \widehat{z_1}^{i_1} \widehat{t_3}^{\xi} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} = \langle \widehat{\Omega}_2 - \beta \rangle \nu,$$

where $\nu \in A_{\alpha}^{(5)}S_5^{-1}$. Write $\nu = \sum_{\underline{j}\in J} b_{\underline{j}}\widehat{z_1}^{i_1}\widehat{t_3}^{i_3}\widehat{t_4}^{i_4}\widehat{t_5}^{i_5}\widehat{t_6}^{i_6}$, with $\underline{j} = (i_1, i_3, i_4, i_5, i_6) \in J \subset \mathbb{N}^3 \times \mathbb{Z}^2$ and $b_{\underline{j}} \in \mathbb{C}$. Given that $\Omega_2 = Z_2 T_4 T_6 + b T_3^3 T_6$ in $A^{(5)}$ and the relation (4.1.4), one can deduce that

$$\widehat{\Omega}_2 = \widehat{z_2}\widehat{t_4}\widehat{t_6} + b\widehat{t_3}^3\widehat{t_6} = \frac{q^3\alpha}{a}\widehat{t_4}\widehat{t_5}^{-1}\widehat{t_6} - \frac{1}{a}\widehat{z_1}\widehat{t_3}\widehat{t_4}\widehat{t_6} + b\widehat{t_3}^3\widehat{t_6}.$$

Therefore,

$$\begin{split} \sum_{(\xi,\underline{i})\in I} a_{(\xi,\underline{i})} \widehat{z_1}^{i_1} \widehat{t_3}^{\xi} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} &= \sum_{\underline{j}\in J} \frac{q^{\bullet}\alpha}{a} b_{\underline{j}} \widehat{z_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4+1} \widehat{t_5}^{i_5-1} \widehat{t_6}^{i_6+1} \\ &+ \sum_{\underline{j}\in J} q^{\bullet} b b_{\underline{j}} \widehat{z_1}^{i_1} \widehat{t_3}^{i_3+3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6+1} \\ &- \sum_{\underline{j}\in J} \frac{q^{\bullet}}{a} b_{\underline{j}} \widehat{z_1}^{i_1+1} \widehat{t_3}^{i_3+1} \widehat{t_4}^{i_4+1} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6+1} \\ &- \sum_{\underline{j}\in J} \beta b_{\underline{j}} \widehat{z_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6}. \end{split}$$

Suppose that there exists $(i_1, i_3, i_4, i_5, i_6) \in J$ such that $b_{(i_1, i_3, i_4, i_5, i_6)} \neq 0$. Let $(w_1, w_3, w_4, w_5, w_6) \in J$ be the greatest element (in the lexicographic order on $\mathbb{N}^3 \times \mathbb{Z}^2$) of J such that $b_{(w_1, w_3, w_4, w_5, w_6)} \neq 0$. Since $\left(\widehat{z_1}^{k_1} \widehat{t_3}^{k_3} \widehat{t_4}^{k_4} \widehat{t_5}^{k_5} \widehat{t_6}^{k_6}\right)_{(k_1, k_3, \cdots, k_6) \in \mathbb{N}^3 \times \mathbb{Z}^2}$ is a basis of $A_{\alpha}^{(5)} S_5^{-1}$, one can identify the coefficients of $\widehat{z_1}^{w_1+1} \widehat{t_3}^{w_3+1} \widehat{t_4}^{w_4+1} \widehat{t_5}^{w_5} \widehat{t_6}^{w_6+1}$ in the above equality as: $\frac{q^{\bullet}}{a} b_{(w_1, w_3, w_4, w_5, w_6)} = 0$. Hence, $b_{(w_1, w_3, w_4, w_5, w_6)} = 0$, a contradiction! Therefore, $b_{(i_1, i_3, i_4, i_5, i_6)} = 0$ for all $(i_1, i_3, i_4, i_5, i_6) \in J$. Consequently,

$$\sum_{(\xi,\underline{i})\in I} a_{(\xi,\underline{i})} \widehat{z_1}^{i_1} \widehat{t_3}^{\xi} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} = 0.$$

It follows that $a_{(\xi,\underline{i})} = 0$ for all $(\xi,\underline{i}) \in I$. As a result, \mathcal{B}_5 is a linearly independent set.

4.1.5 Corollary. Let I be a finite subset of $\{0, 1, 2\} \times \mathbb{N} \times \mathbb{Z}^3$ and $(a_{(\xi,\underline{i})})_{\underline{i} \in I}$ be a family of complex numbers. If

$$\sum_{\xi,\underline{i})\in I} a_{(\xi,\underline{i})} z_1^{i_1} t_3^{\xi} t_4^{i_4} t_5^{i_5} t_6^{i_6} = 0,$$

(

then $a_{(\xi,\underline{i})} = 0$ for all $(\xi,\underline{i}) \in I$.

Proof. When $i_4 \ge 0$, then the result is obvious from the result of Proposition 4.1.4. For $i_4 < 0$, multiply both sides of the equality (on the right) enough times by t_4 to kill all the negative powers of t_4 , and then apply Proposition 4.1.4 to complete the proof.

4.1.6 Remark. We were not successful in finding a basis for R_6 . However, this has no

effect on our main result in this chapter. Since $R_7 = A_{\alpha,\beta}$, we already have a basis for R_7 (Proposition 3.2.2).

4.1.7 Lemma. Let $Z(R_i)$ denote the center of R_i , then $Z(R_i) = \mathbb{C}$ for each $3 \le i \le 7$. *Proof.* One can easily verify that $Z(R_3) = \mathbb{C}$. Note, $R_7 = A_{\alpha,\beta}$. Since R_i is a localization of R_{i+1} (see (4.1.1)), we have that $\mathbb{C} \subseteq Z(R_7) \subseteq Z(R_6) \subseteq Z(R_5) \subseteq Z(R_4) \subseteq Z(R_3) =$ \mathbb{C} . Therefore, $Z(R_7) = Z(R_6) = Z(R_5) = Z(R_4) = Z(R_3) = \mathbb{C}$.

4.1.8 Remark. Recall the notations:

$f_1 := e_{1,4}$	$F_1 := E_{1,4}$
$z_1 := e_{1,5}$	$Z_1 := E_{1,5}$
$z_2 := e_{2,5}$	$Z_2 := E_{2,5}.$

Let $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}$. Given the above notations, we present the following selected data of the DDA of $A_{\alpha,\beta}$, listed at the beginning of this section, in a manner that would be very useful in Subsections 4.2.1 and 4.2.5. They are as follows:

$$f_{1} = t_{1} - at_{2}t_{3}^{-1} \qquad e_{3,6} = t_{3} - at_{4}t_{5}^{-1}$$

$$z_{1} = f_{1} - st_{3}^{2}t_{4}^{-1} \qquad e_{1} = e_{1,6} - rt_{5}t_{6}^{-1}$$

$$z_{2} = t_{2} - bt_{3}^{3}t_{4}^{-1} \qquad e_{3} = e_{3,6} - st_{5}^{2}t_{6}^{-1}$$

$$e_{1,6} = z_{1} - he_{3,6}t_{5}^{-1} - gt_{4}t_{5}^{-2} \qquad e_{4} = t_{4} - bt_{5}^{3}t_{6}^{-1}.$$

4.2 Derivations of $A_{\alpha,\beta}$

We are now going to study the derivations of $A_{\alpha,\beta}$. We will begin with the case where both α and β are non-zero, and then proceed to look at the case where either α or β is zero.

4.2.1 Derivations of $A_{\alpha,\beta}$ ($\alpha, \beta \neq 0$). Throughout this subsection, we assume that α and β are non-zero. Let $\text{Der}(A_{\alpha,\beta})$ denote the \mathbb{C} -derivations of $A_{\alpha,\beta}$ and $D \in \text{Der}(A_{\alpha,\beta})$.

From the relation (4.1.1), D extends uniquely to a derivation of each of the series of algebras in (4.1.2) via localization. Therefore, D extends to a derivation of the quantum torus $R_3 = \mathbb{C}_{q^N}[t_3^{\pm 1}, t_4^{\pm 1}, t_5^{\pm 1}, t_6^{\pm 1}]$. It follows from [40, Corollary 2.3] that D can uniquely be written as:

$$D = \mathsf{ad}_x + \delta$$

where $x \in R_3$, and δ is a scalar derivation of R_3 defined as $\delta(t_i) = \lambda_i t_i$ for each i = 3, 4, 5, 6. Note, $\lambda_i \in Z(R_3) = \mathbb{C}$. Also, ad_x is an inner derivation of R_3 defined as $ad_x(L) = xL - Lx$ for all $L \in R_3$.

We aim to describe D as a derivation of $A_{\alpha,\beta}$. We do this in several steps. We first describe D as a derivation of R_4 .

4.2.2 Lemma. 1. $x \in R_4$.

2.
$$\lambda_5 = \lambda_4 + \lambda_6$$
, $\delta(f_1) = -(\lambda_3 + \lambda_5)f_1$ and $\delta(t_2) = -\lambda_5 t_2$.

3. Set $\lambda_1 := -(\lambda_3 + \lambda_5)$ and $\lambda_2 := -\lambda_5$. Then, $D(e_{\kappa,4}) = \operatorname{ad}_x(e_{\kappa,4}) + \lambda_{\kappa}e_{\kappa,4}$ for all $\kappa \in \{1, \dots, 6\}$.

Proof. 1. Set $Q_q := \mathbb{C}_{q^{\mu}}[t_4^{\pm 1}, t_5^{\pm 1}, t_6^{\pm 1}]$, where μ is some skew-symmetric matrix in $M_3(\mathbb{Z})$. Observe that Q_q is a subalgebra of both R_3 and R_4 with central element

$$z := t_4 t_5^{-1} t_6.$$

Furthermore, since R_3 is a quantum torus, we can present it as a free left Q_q -module with basis $(t_3^s)_{s\in\mathbb{Z}}$. With this presentation, $x \in R_3$ can be written as

$$x = \sum_{s \in \mathbb{Z}} y_s t_3^s,$$

where $y_s \in \mathcal{Q}_q$. Set

$$x_+ := \sum_{s \geq 0} y_s t_3^s$$
 and $x_- := \sum_{s < 0} y_s t_3^s$.

Clearly, $x = x_+ + x_-$. Obviously, $x_+ \in R_4$, hence we aim to also show that x_- belongs

to R_4 by following a pattern similar to [29, Proposition 7.1.2]. As D is a derivation of R_4 , we have that $D(z^j) \in R_4$ for all $j \in \mathbb{N}_{\geq 1}$. Now $D(z^j) = \operatorname{ad}_x(z^j) + \delta(z^j) =$ $\operatorname{ad}_{x_+}(z^j) + \operatorname{ad}_{x_-}(z^j) + \delta(z^j)$. Observe that $\operatorname{ad}_{x_+}(z^j) \in R_4$; since $x_+, z^j \in R_4$. Also, $\delta(z) = \delta(t_4 t_5^{-1} t_6) = (\lambda_4 - \lambda_5 + \lambda_6) t_4 t_5^{-1} t_6 = (\lambda_4 - \lambda_5 + \lambda_6) z$, where $\lambda_4, \lambda_5, \lambda_6 \in \mathbb{C}$. It follows that $\delta(z^j) = j(\lambda_4 - \lambda_5 + \lambda_6) z^j \in R_4$. We can therefore conclude that each $\operatorname{ad}_{x_-}(z^j)$ belongs to R_4 since $D(z^j), \operatorname{ad}_{x_+}(z^j), \delta(z^j) \in R_4$. We have:

$$\mathsf{ad}_{x_{-}}(z^{j}) = x_{-}z^{j} - z^{j}x_{-} = \sum_{s=-1}^{-n} y_{s}t_{3}^{s}z^{j} - \sum_{s=-1}^{-n} y_{s}z^{j}t_{3}^{s}.$$

One can verify that $zt_3 = q^{-2}t_3z$. Therefore,

$$\mathsf{ad}_{x_{-}}(z^{j}) = \sum_{s=-1}^{-n} (1 - q^{-2js}) y_{s} t_{3}^{s} z^{j}, \text{ hence, } \mathsf{ad}_{x_{-}}(z^{j}) z^{-j} = \sum_{s=-1}^{-n} (1 - q^{-2js}) y_{s} t_{3}^{s} z^{j}$$

Set $\nu_j := \operatorname{ad}_{x_-}(z^j) z^{-j} \in R_4$. It follows that

$$\nu_j = \sum_{s=-1}^{-n} (1 - q^{-2js}) y_s t_3^s,$$

for each $j \in \{1, \dots, n\}$. One can therefore write the above equality as a matrix equation as follows:

$$\begin{bmatrix} (1-q^2) & (1-q^4) & (1-q^6) & \cdots & (1-q^{2n}) \\ (1-q^4) & (1-q^8) & (1-q^{12}) & \cdots & (1-q^{4n}) \\ (1-q^6) & (1-q^{12}) & (1-q^{18}) & \cdots & (1-q^{6n}) \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ (1-q^{2n}) & (1-q^{4n}) & (1-q^{6n}) & \cdots & (1-q^{2n^2}) \end{bmatrix} \begin{bmatrix} y_{-1}t_3^{-1} \\ y_{-2}t_3^{-2} \\ y_{-3}t_3^{-3} \\ \vdots \\ y_{-n}t_3^{-n} \end{bmatrix} = \begin{bmatrix} \nu_1 \\ \nu_2 \\ \nu_3 \\ \vdots \\ \nu_n \end{bmatrix}$$

We already know that each ν_j belongs to R_4 . We want to show that $y_s t_3^s$ also belongs to R_4 for each $s \in \{-1, \dots, -n\}$. To establish this, it is sufficient to show that the coefficient matrix of the above matrix equation is invertible. Let U represent this matrix. Thus,

$$U = \begin{bmatrix} (1-q^2) & (1-q^4) & (1-q^6) & \cdots & (1-q^{2n}) \\ (1-q^4) & (1-q^8) & (1-q^{12}) & \cdots & (1-q^{4n}) \\ (1-q^6) & (1-q^{12}) & (1-q^{18}) & \cdots & (1-q^{6n}) \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ (1-q^{2n}) & (1-q^{4n}) & (1-q^{6n}) & \cdots & (1-q^{2n^2}) \end{bmatrix}.$$

Apply row operations: $-r_{n-1} + r_n \rightarrow r_n, \cdots, -r_2 + r_3 \rightarrow r_3, -r_1 + r_2 \rightarrow r_2$ to U to obtain:

$$U' = \begin{bmatrix} l_1 & l_2 & l_3 & \cdots & l_n \\ q^2 l_1 & q^4 l_2 & q^6 l_3 & \cdots & q^{2n} l_n \\ q^4 l_1 & q^8 l_2 & q^{12} l_3 & \cdots & q^{4n} l_n \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ q^{2(n-1)} l_1 & q^{4(n-1)} l_2 & q^{6(n-1)} l_3 & \cdots & q^{2n(n-1)} l_n \end{bmatrix}$$

where $l_i := 1 - q^{2i}$; $i \in \{1, 2, \dots, n\}$. Clearly, U' is similar to a Vandermonde matrix (since the terms in each column form a geometric sequence) which is well known to be invertible. This further implies that U is invertible. So each $y_s t_3^s$ is a linear combination of the $\nu_j \in R_4$. As a result, $y_s t_3^s \in R_4$ for all $s \in \{-1, \dots, -n\}$. Consequently, $x_- = \sum_{s=-1}^{-n} y_s t_3^s \in R_4$ as desired.

2. Recall that $\delta(t_{\kappa}) = \lambda_{\kappa}t_{\kappa}$ for all $\kappa \in \{3, 4, 5, 6\}$ and $\lambda_{\kappa} \in \mathbb{C}$. From Remark 4.1.8, we have that $f_1 = t_1 - at_2t_3^{-1}$. Recall from Section 3.1 that $t_1 = \alpha t_5^{-1}t_3^{-1}$ and $t_2^{-1} = \beta t_6^{-1}t_4^{-1}$ in $R_3 = \mathscr{A}_{\alpha,\beta}$. As a result, $f_1 = \alpha t_5^{-1}t_3^{-1} - a\beta t_6^{-1}t_4^{-1}t_3^{-1}$. Hence,

$$\delta(f_1) = -(\lambda_5 + \lambda_3)\alpha t_5^{-1} t_3^{-1} + (\lambda_6 + \lambda_4 + \lambda_3)a\beta t_6^{-1} t_4^{-1} t_3^{-1}.$$
(4.2.1)

From Proposition 4.1.2, the set $\mathcal{B}_4 = \left\{ f_1^{i_1} t_4^{i_4} t_5^{i_5} t_6^{i_6}, t_3^{i_3} t_4^{i_4} t_5^{i_5} t_6^{i_6} \mid i_1, i_3 \in \mathbb{N} \text{ and } i_4, i_5, i_6 \in \mathbb{Z} \right\}$ is a \mathbb{C} -basis of R_4 . Since t_4, t_5 and t_6 q-commute with f_1 and t_3 , one can also write $\delta(f_1) \in R_4$ in terms of \mathcal{B}_4 as follows:

$$\delta(f_1) = \sum_{r>0} a_r f_1^r + \sum_{s\ge 0} b_s t_3^s, \tag{4.2.2}$$

where a_r and b_s belong to $\mathcal{Q}_q = \mathbb{C}_{q^{\mu}}[t_4^{\pm 1}, t_5^{\pm 1}, t_6^{\pm 1}].$

$$f_{1}^{r} = (\alpha t_{5}^{-1} t_{3}^{-1} - a\beta t_{6}^{-1} t_{4}^{-1} t_{3}^{-1})^{r} = \sum_{i=0}^{r} {r \choose i}_{q^{\bullet}} (\alpha t_{5}^{-1} t_{3}^{-1})^{i} (-a\beta t_{6}^{-1} t_{4}^{-1} t_{3}^{-1})^{r-i}$$
$$= \sum_{i=0}^{r} {r \choose i}_{q^{\bullet}} \alpha^{i} (-a\beta)^{r-i} q^{\frac{1}{2}i(i-1) + \frac{3}{2}(r-i)(r-i-1) + 3i(i-r)} t_{5}^{-i} (t_{6}^{-1} t_{4}^{-1})^{r-i} t_{3}^{-r}$$
$$= c_{r} t_{3}^{-r}, \qquad (4.2.3)$$

where

$$c_{r} = \sum_{i=0}^{r} {\binom{r}{i}}_{q^{\bullet}} q^{\frac{1}{2}i(i-1) + \frac{3}{2}(r-i)(r-i-1) + 3i(i-r)} \alpha^{i} (-a\beta)^{r-i} t_{5}^{-i} (t_{6}^{-1} t_{4}^{-1})^{r-i} \in \mathcal{Q}_{q} \setminus \{0\}.$$

$$(4.2.4)$$

Substitute (4.2.3) into (4.2.2) to obtain;

$$\delta(f_1) = \sum_{r>0} a_r c_r t_3^{-r} + \sum_{s\geq 0} b_s t_3^s.$$
(4.2.5)

One can rewrite (4.2.1) as

$$\delta(f_1) = dt_3^{-1},\tag{4.2.6}$$

where $d = -(\lambda_5 + \lambda_3)\alpha t_5^{-1} + (\lambda_6 + \lambda_4 + \lambda_3)a\beta t_6^{-1}t_4^{-1} \in Q_q$. Comparing (4.2.5) to (4.2.6) shows that $b_s = 0$ for all $s \ge 0$, and $a_r c_r = 0$ for all $r \ne 1$. Therefore $\delta(f_1) = a_1 c_1 t_3^{-1}$. Moreover, from (4.2.4), $c_1 = -a\beta t_6^{-1} t_4^{-1} + \alpha t_5^{-1}$. Hence,

$$\delta(f_1) = a_1 c_1 t_3^{-1} = a_1 (-a\beta t_6^{-1} t_4^{-1} + \alpha t_5^{-1}) t_3^{-1} = a_1 \alpha t_5^{-1} t_3^{-1} - a_1 a\beta t_6^{-1} t_4^{-1} t_3^{-1}.$$
 (4.2.7)

Comparing (4.2.7) to (4.2.1) reveals that $a_1 = -(\lambda_5 + \lambda_3) = -(\lambda_6 + \lambda_4 + \lambda_3)$. Conse-

quently, $\lambda_5 = \lambda_6 + \lambda_4$. Hence, $\delta(f_1) = -(\lambda_5 + \lambda_3)\alpha t_5^{-1} t_3^{-1} + (\lambda_5 + \lambda_3)a\beta t_6^{-1} t_4^{-1} t_3^{-1} = -(\lambda_5 + \lambda_3)f_1$. Finally, since $t_2 = \beta t_6^{-1} t_4^{-1}$ in R_4 , it follows that $\delta(t_2) = -(\lambda_6 + \lambda_4)\beta t_6^{-1} t_4^{-1} = -(\lambda_6 + \lambda_4)t_2 = -\lambda_5 t_2$.

3. Set $\lambda_1 := -(\lambda_3 + \lambda_5)$ and $\lambda_2 := -\lambda_5$. it follows from points (1) and (2) that $D(e_{\kappa,4}) = \operatorname{ad}_x(e_{\kappa,4}) + \delta(e_{\kappa,4}) = \operatorname{ad}_x(e_{\kappa,4}) + \lambda_{\kappa}e_{\kappa,4}$ for all $\kappa \in \{1, \dots, 6\}$. In conclusion, $D = \operatorname{ad}_x + \delta$, with $x \in R_4$.

We proceed to describe D as a derivation of R_5 .

4.2.3 Lemma. 1. $x \in R_5$.

2.
$$\lambda_4 = 3\lambda_3 + \lambda_5$$
, $\lambda_6 = -3\lambda_3$, $\delta(z_1) = -(\lambda_3 + \lambda_5)z_1$ and $\delta(z_2) = -\lambda_5 z_2$

3. Set $\lambda_1 := -(\lambda_3 + \lambda_5)$, $\lambda_2 := -\lambda_5$ and $\lambda_6 := -3\lambda_3$. Then, $D(e_{\kappa,5}) = \operatorname{ad}_x(e_{\kappa,5}) + \lambda_{\kappa}e_{\kappa,5}$ for all $\kappa \in \{1, \dots, 6\}$.

Proof. In this proof, we denote $\underline{v} := (i, j, k, l) \in \mathbb{N} \times \mathbb{Z}^3$.

1. We already know that $x \in R_4 = R_5[t_4^{-1}]$. Given the basis \mathcal{B}_5 of R_5 (Proposition 4.1.4), x can be written as $x = \sum_{(\xi,\underline{v})\in I} a_{(\xi,\underline{v})} z_1^i t_3^{\xi} t_4^j t_5^k t_6^l$, where I is a finite subset of $\{0, 1, 2\} \times \mathbb{N} \times \mathbb{Z}^3$ and $a_{(\xi,\underline{v})}$ are complex numbers. Write $x = x_- + x_+$, where

$$x_{+} = \sum_{\substack{(\xi,\underline{\upsilon}) \in I \\ j \ge 0}} a_{(\xi,\underline{\upsilon})} z_{1}^{i} t_{3}^{\xi} t_{4}^{j} t_{5}^{k} t_{6}^{l} \text{ and } x_{-} = \sum_{\substack{(\xi,\underline{\upsilon}) \in I \\ j < 0}} a_{(\xi,\underline{\upsilon})} z_{1}^{i} t_{3}^{\xi} t_{4}^{j} t_{5}^{k} t_{6}^{l}$$

Suppose that there exists a minimum $j_0 < 0$ such that $a_{(\xi,i,j_0,k,l)} \neq 0$ for some $(\xi, i, j_0, k, l) \in I$ and $a_{(\xi,i,j,k,l)} = 0$ for all $(\xi, i, j_0, k, l) \in I$ with $j < j_0$. Given this assumption, write

$$x_{-} = \sum_{\substack{(\xi,\underline{\upsilon})\in I\\j_0 \le j \le -1}} a_{(\xi,\underline{\upsilon})} z_1^i t_3^\xi t_4^j t_5^k t_6^j.$$

Now, $D(t_6) = ad_{x_+}(t_6) + ad_{x_-}(t_6) + \delta(t_6) \in R_5$. This implies that $ad_{x_-}(t_6) \in R_5$, since $ad_{x_+}(t_6) + \delta(t_6) = ad_{x_+}(t_6) + \lambda_6 t_6 \in R_5$. We aim to show that $x_- = 0$. Since t_6 is normal

in R_5 , one can easily verify that

$$\mathsf{ad}_{x_{-}}(t_{6}) = \sum_{\substack{(\xi,\underline{\upsilon})\in I\\j_{0}\leq j\leq -1}} \left(q^{3(i-j-k)} - 1\right) a_{(\xi,\underline{\upsilon})} z_{1}^{i} t_{3}^{\xi} t_{4}^{j} t_{5}^{k} t_{6}^{l+1}.$$

Set $\underline{w} := (i, j, k, l) \in \mathbb{N}^2 \times \mathbb{Z}^2$. One can equally write $\operatorname{ad}_{x_-}(t_6) \in R_5$ in terms of the basis \mathcal{B}_5 of R_5 (Proposition 4.1.4) as:

$$\mathsf{ad}_{x_-}(t_6) = \sum_{(\underline{\xi},\underline{w})\in J} b_{(\underline{\xi},\underline{w})} z_1^i t_3^{\underline{\xi}} t_4^j t_5^k t_6^j,$$

where J is a finite subset of $\{0, 1, 2\} \times \mathbb{N}^2 \times \mathbb{Z}^2$ and $b_{(\xi,\underline{w})}$ are all complex numbers. It follows that

$$\sum_{\substack{(\xi,\underline{\upsilon})\in I\\j_0\leq j\leq -1}} \left(q^{3(i-j-k)} - 1\right) a_{(\xi,\underline{\upsilon})} z_1^i t_3^{\xi} t_4^j t_5^k t_6^{l+1} = \sum_{(\xi,\underline{w})\in J} b_{(\xi,\underline{w})} z_1^i t_3^{\xi} t_4^j t_5^k t_6^l$$

As \mathcal{B}_5 is a basis for R_5 , we deduce from Corollary 4.1.5 that $\left(z_1^i t_3^{\xi} t_4^j t_5^k t_6^l\right)_{(i \in \mathbb{N}; j, k, l \in \mathbb{Z}; \xi \in \{0, 1, 2\})}$ is a basis for $R_5[t_4^{-1}]$. Now, at $j = j_0$, denote $\underline{v} = (i, j, k, l)$ by $\underline{v}_0 := (i, j_0, k, l)$. Since $\underline{v}_0 \in \mathbb{N} \times \mathbb{Z}^3$ (with $j_0 < 0$) and $\underline{w} = (i, j, k, l) \in \mathbb{N}^2 \times \mathbb{Z}^2$ (with $j \ge 0$), it follows from the above equality that, at \underline{v}_0 , we must have

$$\left(q^{3(i-j_0-k)}-1\right)a_{(\xi,\underline{v}_0)}=0.$$

From our initial assumption, the coefficients $a_{(\xi,\underline{v}_0)}$ are all not zero, therefore $q^{3(i-j_0-k)} - 1 = 0$. This implies that

$$k = i - j_0,$$
 (4.2.8)

for some $(\xi, \underline{v}_0) \in I$.

In a similar manner, $D(t_3) = \mathsf{ad}_{x_+}(t_3) + \mathsf{ad}_{x_-}(t_3) + \delta(t_3) \in R_5$. This implies that

 $\mathsf{ad}_{x_-}(t_3) \in R_5$, since $\mathsf{ad}_{x_+}(t_3) + \delta(t_3) = \mathsf{ad}_{x_+}(t_3) + \lambda_3 t_3 \in R_5$. We have that

$$\mathsf{ad}_{x_{-}}(t_{3}) = \sum_{\substack{(\xi,\underline{\upsilon})\in I\\j_{0}\leq j\leq -1}} a_{(\xi,\underline{\upsilon})} z_{1}^{i} t_{3}^{\xi} t_{4}^{j} t_{5}^{k} t_{6}^{l} t_{3} - \sum_{\substack{(\xi,\underline{\upsilon})\in I\\j_{0}\leq j\leq -1}} a_{(\xi,\underline{\upsilon})} t_{3} z_{1}^{i} t_{3}^{\xi} t_{4}^{j} t_{5}^{k} t_{6}^{l}$$

One can deduce from Lemma A.1.4(3a) that

$$t_3 z_1^i = q^{-i} z_1^i t_3 + d_2[i] z_1^{i-1} z_2,$$

where $d_2[i] = q^{1-i}d_2[1]\left(\frac{1-q^{-2i}}{1-q^{-2}}\right)$, $d_2[1] = -(q+q^{-1}+q^{-3})$ and $d_2[0] = 0$. Therefore, the above expression for $ad_{x_-}(t_3)$ can be expressed as:

$$\begin{aligned} \mathsf{ad}_{x_{-}}(t_{3}) &= \sum_{\substack{(0,\underline{\upsilon})\in I\\ j_{0}\leq j\leq -1}} f[i,j,k] a_{(0,\underline{\upsilon})} z_{1}^{i} t_{3} t_{4}^{j} t_{5}^{k} t_{6}^{l} + \sum_{\substack{(1,\underline{\upsilon})\in I\\ j_{0}\leq j\leq -1}} f[i,j,k] a_{(1,\underline{\upsilon})} z_{1}^{i} t_{3}^{2} t_{4}^{j} t_{5}^{k} t_{6}^{l} + \\ &+ \sum_{\substack{(2,\underline{\upsilon})\in I\\ j_{0}\leq j\leq -1}} f[i,j,k] a_{(2,\underline{\upsilon})} z_{1}^{i} t_{3}^{3} t_{4}^{j} t_{5}^{k} t_{6}^{l} - \sum_{\substack{(\xi,\underline{\upsilon})\in I\\ j_{0}\leq j\leq -1}} a_{(\xi,\underline{\upsilon})} d_{2}[i] z_{1}^{i-1} z_{2} t_{3}^{\xi} t_{4}^{j} t_{5}^{k} t_{6}^{l}, \end{aligned}$$

where $f[i, j, k] := q^{-(k+3j)} - q^{-i}$. Recall from (4.1.4) and (4.1.5) that

where a and b are non-zero scalars (Appendix C). Using these two expressions, one can write $ad_{x_{-}}(t_{3})$ in terms of the basis of R_{5} as:

$$\begin{aligned} \mathsf{ad}_{x_{-}}(t_{3}) = &\mathcal{K} + \sum_{(0,\underline{\upsilon}_{0})\in I} g[i,j_{0},k]a_{(0,\underline{\upsilon}_{0})}z_{1}^{i}t_{3}t_{4}^{j_{0}}t_{5}^{k}t_{6}^{l} + \sum_{(1,\underline{\upsilon}_{0})\in I} g[i,j_{0},k]a_{(1,\underline{\upsilon}_{0})}z_{1}^{i}t_{3}^{2}t_{4}^{j_{0}}t_{5}^{k}t_{6}^{l} \\ &+ \sum_{(2,\underline{\upsilon}_{0})\in I} \frac{q^{\bullet}\beta}{b}a_{(2,\underline{\upsilon}_{0})}g[i,j_{0},k]z_{1}^{i}t_{4}^{j_{0}}t_{5}^{k}t_{6}^{l-1} - \sum_{(\xi,\underline{\upsilon}_{0})\in I} \frac{q^{\bullet}\alpha}{a}d_{2}[i]a_{(\xi,\underline{\upsilon}_{0})}z_{1}^{i-1}t_{3}^{\xi}t_{4}^{j_{0}}t_{5}^{k-1}t_{6}^{l} \\ &= \sum 1/b\left(q^{\bullet}\beta g[i,j_{0},k]a_{(2,i,j_{0},k,l+1)} + (q^{\bullet}\alpha bd_{2}[i+1]/a)a_{(0,i+1,j_{0},k+1,l)}\right)z_{1}^{i}t_{4}^{j_{0}}t_{5}^{k}t_{6}^{l} \\ &+ \sum \left(g[i,j_{0},k]a_{(0,i,j_{0},k,l)} + (q^{\bullet}\alpha d_{2}[i+1]/a)a_{(1,i+1,j_{0},k+1,l)}\right)z_{1}^{i}t_{3}^{j_{0}}t_{5}^{k}t_{6}^{l} \\ &+ \sum \left(g[i,j_{0},k]a_{(1,i,j_{0},k,l)} + (q^{\bullet}\alpha d_{2}[i+1]/a)a_{(2,i+1,j_{0},k+1,l)}\right)z_{1}^{i}t_{3}^{2}t_{4}^{j_{0}}t_{5}^{k}t_{6}^{l} + \mathcal{K}, \end{aligned}$$

$$(4.2.9)$$

where $g[i, j_0, k] := q^{-(k+3j_0)} - q^{-i} + d_2[i]/a$ and

$$\mathcal{K} \in \mathsf{Span}\left(\mathcal{B}_5 \setminus \{z_1^i t_3^\xi t_4^{j_0} t_5^k t_6^l \mid (\xi, i, j_0, k, l) \in \{0, 1, 2\} \times \mathbb{N} \times \mathbb{Z}^3\}\right).$$

One can also write $\operatorname{ad}_{x_{-}}(t_{3}) \in R_{5}$ in terms of the basis \mathcal{B}_{5} of R_{5} (Proposition 4.1.4) as:

$$\mathsf{ad}_{x_{-}}(t_{3}) = \sum_{(\xi,\underline{w})\in J} b_{(\xi,\underline{w})} z_{1}^{i} t_{3}^{\xi} t_{4}^{j} t_{5}^{k} t_{6}^{l},$$
(4.2.10)

where J is a finite subset of $\{0, 1, 2\} \times \mathbb{N}^2 \times \mathbb{Z}^2$, and $b_{(\xi,\underline{w})} \in \mathbb{C}$. Recall: $\underline{w} = (i, j, k, l) \in \mathbb{N}^2 \times \mathbb{Z}^2$. Now, (4.2.9) and (4.2.10) imply that

$$\begin{split} \sum_{(\xi,\underline{w})\in J} b_{(\xi,\underline{w})} z_1^i t_3^{\xi} t_4^j t_5^k t_6^l \\ &= \sum 1/b \left(q^{\bullet} \beta g[i, j_0, k] a_{(2,i,j_0,k,l+1)} + (q^{\bullet} \alpha b d_2[i+1]/a) a_{(0,i+1,j_0,k+1,l)} \right) z_1^i t_4^{j_0} t_5^k t_6^l \\ &+ \sum \left(g[i, j_0, k] a_{(0,i,j_0,k,l)} + (q^{\bullet} \alpha d_2[i+1]/a) a_{(1,i+1,j_0,k+1,l)} \right) z_1^i t_3 t_4^{j_0} t_5^k t_6^l \\ &+ \sum \left(g[i, j_0, k] a_{(1,i,j_0,k,l)} + (q^{\bullet} \alpha d_2[i+1]/a) a_{(2,i+1,j_0,k+1,l)} \right) z_1^i t_3^2 t_4^{j_0} t_5^k t_6^l + \mathcal{K}. \end{split}$$

We have already established that $(z_1^i t_3^{\xi} t_4^j t_5^k t_6^l)_{(i \in \mathbb{N}; j, k, l \in \mathbb{Z}; \xi \in \{0, 1, 2\})}$ is a basis for $R_5[t_4^{-1}]$. Given that $\underline{v}_0 = (i, j_0, k, l) \in \mathbb{N} \times \mathbb{Z}^3$ (with $j_0 < 0$) and $\underline{w} = (i, j, k, l) \in \mathbb{N}^2 \times \mathbb{Z}^2$ (with $j \ge 0$), it follows that

$$q^{\bullet}\beta g[i, j_0, k]a_{(2,i,j_0,k,l+1)} + (q^{\bullet}\alpha bd_2[i+1]/a)a_{(0,i+1,j_0,k+1,l)} = 0.$$
(4.2.11)

$$g[i, j_0, k]a_{(0,i,j_0,k,l)} + (q^{\bullet}\alpha d_2[i+1]/a)a_{(1,i+1,j_0,k+1,l)} = 0.$$
(4.2.12)

$$g[i, j_0, k]a_{(1,i,j_0,k,l)} + (q^{\bullet}\alpha d_2[i+1]/a)a_{(2,i+1,j_0,k+1,l)} = 0.$$
(4.2.13)

Suppose that there exists $(\xi, i, j_0, k, l) \in I$ such that $g[i, j_0, k] = 0$. Then,

$$g[i, j_0, k] = q^{-(k+3j_0)} - q^{-i} + d_2[i]/a = 0.$$

Note, $d_2[i] = d_2[1]q^{1-i}\left(\frac{1-q^{-2i}}{1-q^{-2}}\right)$, where $d_2[1] = -(q+q^{-1}+q^{-3})$ and $d_2[0] = 0$. Again, recall from Appendix C that $a = (q^2+1+q^{-2})/(q^{-2}-1) = \frac{qd_2[1]}{1-q^{-2}}$. Given these expressions for $d_2[i]$ and a, we have that

$$g[i, j_0, k] = q^{-(k+3j_0)} - q^{-i} + d_2[i]/a = q^{-3j_0-k} - q^{-3i} = 0.$$

Since q is not a root of unity, we get

$$k = 3(i - j_0). \tag{4.2.14}$$

Comparing (4.2.14) to (4.2.8) shows that $i - j_0 = 0$ which implies that $i = j_0 < 0$, a contradiction (note, $i \ge 0$). Therefore, $g[i, j_0, k] \ne 0$ for all $(\xi, i, j, k, l) \in I$.

Now, observe that if there exists $\xi \in \{0, 1, 2\}$ such that $a_{(\xi,i,j_0,k,l)} = 0$ for all $(i, j_0, k, l) \in \mathbb{N} \times \mathbb{Z}^3$, then one can easily deduce from equations (4.2.11), (4.2.12) and (4.2.13) that $a_{(\xi,i,j_0,k,l)} = 0$ for all $(\xi, i, j_0, k, l) \in I$. This will contradict our initial assumption. Therefore, there exists some $(i, j_0, k, l) \in \mathbb{N} \times \mathbb{Z}^3$ such that $a_{(\xi,i,j_0,k,l)} \neq 0$ for each $\xi \in \{0, 1, 2\}$. Without loss of generality, let (u, j_0, v, w) be the greatest element in the lexicographic order on $\mathbb{N} \times \mathbb{Z}^3$ such that $a_{(0,u,j_0,v,w)} \neq 0$ and $a_{(0,i,j_0,k,l)} = 0$ for all i > u.

From (4.2.12), at $(i, j_0, k, l) = (u, j_0, v, w)$, we have:

$$g[u, j_0, v]a_{(0,u,j_0,v,w)} + (q^{\bullet}\alpha d_2[u+1]/a)a_{(1,u+1,j_0,v+1,w)} = 0.$$

From (4.2.13), at $(i, j_0, k, l) = (u + 1, j_0, v + 1, w)$, we have:

$$g[u+1, j_0, v+1]a_{(1,u+1,j_0,v+1,w)} + (q^{\bullet}\alpha d_2[u+2]/a)a_{(2,u+2,j_0,v+2,w)} = 0.$$

Finally, from (4.2.11), at $(i, j_0, k, l) = (u + 2, j_0, v + 2, w - 1)$, we have:

$$q^{\bullet}\beta g[u+2, j_0, v+2]a_{(2,u+2,j_0,v+2,w)} + (q^{\bullet}\alpha bd_2[u+3]/a)a_{(0,u+3,j_0,v+3,w-1)} = 0.$$

Note: $a, b, \alpha, \beta, q^{\bullet} \neq 0$; $g[i, j_0, k] \neq 0$ for all $(\xi, i, j_0, k, l) \in I$; and $d_2[i] \neq 0$ for i > 0. Since u + 3 > u, it follows from the above list of equations (starting from the last one) that

$$a_{(0,u+3,j_0,v+3,w-1)} = 0 \Rightarrow a_{(2,u+2,j_0,v+2,w)} = 0 \Rightarrow a_{(1,u+1,j_0,v+1,w)} = 0 \Rightarrow a_{(0,u,j_0,v,w)} = 0,$$

a contradiction! Hence, $a_{(0,i,j_0,k,l)} = 0$ for all $(i, j_0, k, l) \in \mathbb{N} \times \mathbb{Z}^3$. From (4.2.11), (4.2.12) and (4.2.13), one can easily conclude that $a_{(\xi,i,j_0,k,l)} = 0$ for all $(\xi, i, j_0, k, l) \in I$. This contradicts our initial assumption, hence $x_- = 0$. Consequently, $x = x_+ \in R_5$ as desired.

2. From Remark 4.1.8, we have $z_2 = t_2 - bt_3^3 t_4^{-1}$. Since $\delta(t_{\kappa}) = \lambda_{\kappa} t_{\kappa}$, $\kappa \in \{2, \dots, 6\}$, with $\lambda_2 := -\lambda_5$ (see Lemma 4.2.2), it follows that

$$\delta(z_2) = -\lambda_5 t_2 - b(3\lambda_3 - \lambda_4) t_3^3 t_4^{-1} = -\lambda_5 z_2 - b(3\lambda_3 - \lambda_4 + \lambda_5) t_3^3 t_4^{-1}.$$

Furthermore,

$$D(z_2) = \mathsf{ad}_x(z_2) + \delta(z_2) = \mathsf{ad}_x(z_2) - \lambda_5 z_2 - b(3\lambda_3 - \lambda_4 + \lambda_5) t_3^3 t_4^{-1} \in R_5.$$

Hence $b(3\lambda_3 - \lambda_4 + \lambda_5)t_3^3t_4^{-1} \in R_5$, since $\operatorname{ad}_x(z_2) - \lambda_5 z_2 \in R_5$. This implies that

 $b(3\lambda_3 - \lambda_4 + \lambda_5)t_3^3 \in R_5t_4$ (note, from Appendix C, $b \neq 0$). Set $w := 3\lambda_3 - \lambda_4 + \lambda_5$. Suppose that $w \neq 0$. From (4.1.5), we have:

$$t_3^3 = \frac{\beta}{b}t_6^{-1} - \frac{q^3\alpha}{ab}t_4t_5^{-1} + \frac{1}{ab}z_1t_3t_4.$$

It follows that

$$wbt_3^3 = w\beta t_6^{-1} - \frac{q^3w\alpha}{a}t_4t_5^{-1} + \frac{w}{a}z_1t_3t_4 \in R_5t_4.$$

Since t_3^3 , $t_4t_5^{-1}$ and $z_1t_3t_4$ are all elements of R_5t_4 , it implies that $t_6^{-1} \in R_5t_4$. Hence, $1 \in R_5t_4t_6$. Using the basis \mathcal{B}_5 of R_5 (Proposition 4.1.4), this leads to contradiction. Therefore, w = 0. That is, $3\lambda_3 - \lambda_4 + \lambda_5 = 0$, and so $\lambda_4 = 3\lambda_3 + \lambda_5$. This further implies that $\delta(z_2) = -\lambda_5 z_2$ as desired.

Again, from Lemma 4.2.2, we have that $\delta(f_1) = -(\lambda_3 + \lambda_5)f_1$. Recall from Remark 4.1.8 that $z_1 = f_1 - st_3^2t_4^{-1}$. It follows that

$$\delta(z_1) = -(\lambda_3 + \lambda_5)f_1 - s(2\lambda_3 - \lambda_4)t_3^2t_4^{-1} = -(\lambda_3 + \lambda_5)z_1 - s(3\lambda_3 - \lambda_4 + \lambda_5)t_3^2t_4^{-1}$$
$$= -(\lambda_3 + \lambda_5)z_1 - s(3\lambda_3 - (3\lambda_3 + \lambda_5) + \lambda_5)t_3^2t_4^{-1} = -(\lambda_3 + \lambda_5)z_1.$$

Finally, we know that $\delta(t_6) = \lambda_6 t_6$. This implies that $\delta(t_6^{-1}) = -\lambda_6 t_6^{-1}$. From (4.1.5), we have that

$$t_3^3 = \frac{\beta}{b}t_6^{-1} - \frac{q^3\alpha}{ab}t_4t_5^{-1} + \frac{1}{ab}z_1t_3t_4,$$

where a and b are non-zero scalars (Appendix C). This implies that

$$t_6^{-1} = \frac{b}{\beta} t_3^3 + \frac{q^3 \alpha}{a\beta} t_4 t_5^{-1} - \frac{1}{a\beta} z_1 t_3 t_4.$$

Given that $\delta(z_1) = -(\lambda_3 + \lambda_5)z_1$, $\delta(t_3) = \lambda_3 t_3$, $\delta(t_4) = (3\lambda_3 + \lambda_5)t_4$ and $\delta(t_5) = \lambda_5 t_5$, applying δ to the above relation gives

$$-\lambda_6 t_6^{-1} = 3\lambda_3 \left(\frac{b}{\beta} t_3^3 + \frac{q^3 \alpha}{a\beta} t_4 t_5^{-1} - \frac{1}{a\beta} z_1 t_3 t_4 \right).$$

It follows that $\lambda_6 = -3\lambda_3$ as desired.

3. Set $\lambda_1 := -(\lambda_3 + \lambda_5)$ and $\lambda_2 := -\lambda_5$. It follows from points (1) and (2) that $D(e_{\kappa,5}) = \operatorname{ad}_x(e_{\kappa,5}) + \delta(e_{\kappa,5}) = \operatorname{ad}_x(e_{\kappa,5}) + \lambda_{\kappa}e_{\kappa,5}$ for all $\kappa \in \{1, \dots, 6\}$. In conclusion, $D = \operatorname{ad}_x + \delta$ with $x \in R_5$.

We are now ready to describe D as a derivation of $A_{\alpha,\beta}$.

4.2.4 Lemma. 1. $x \in A_{\alpha,\beta}$.

- 2. $\delta(e_{\kappa}) = 0$ for all $\kappa \in \{1, \cdots, 6\}$.
- 3. $D = \operatorname{ad}_x$.

Proof. In this proof, we denote $\underline{v} := (i, j, k, l) \in \mathbb{N}^2 \times \mathbb{Z}^2$. Also, recall from the DDA of $A_{\alpha,\beta}$ at the beginning of this section that $t_5 = e_5$ and $t_6 = e_6$.

1. Given the basis \mathcal{B} of $A_{\alpha,\beta}$ (Proposition 3.2.2), one can write $x \in R_5 = A_{\alpha,\beta}[t_5^{-1}, t_6^{-1}]$ as:

$$x = \sum_{(\epsilon_1, \epsilon_2, \underline{\upsilon}) \in I} a_{(\epsilon_1, \epsilon_2, \underline{\upsilon})} e_1^i e_2^j e_3^{\epsilon_1} e_4^{\epsilon_2} t_5^k t_6^l,$$

where I is a finite subset of $\{0,1\}^2 \times \mathbb{N}^2 \times \mathbb{Z}^2$, and $a_{(\epsilon_1,\epsilon_2,\underline{v})}$ are complex numbers. Write $x = x_- + x_+$, where

$$x_{+} = \sum_{\substack{(\epsilon_1, \epsilon_2, \underline{v}) \in I \\ k, \ l \ge 0}} a_{(\epsilon_1, \epsilon_2, \underline{v})} e_1^i e_2^j e_3^{\epsilon_1} e_4^{\epsilon_2} t_5^k t_6^l,$$

and

$$x_{-} = \sum_{\substack{(\epsilon_1, \epsilon_2, \underline{\upsilon}) \in I \\ k < 0 \text{ or } l < 0}} a_{(\epsilon_1, \epsilon_2, \underline{\upsilon})} e_1^i e_2^j e_3^{\epsilon_1} e_4^{\epsilon_2} t_5^k t_6^l.$$

Suppose that there exists a minimum negative integer k_0 or l_0 such that $a_{(\epsilon_1,\epsilon_2,i,j,k_0,l)} \neq 0$ or $a_{(\epsilon_1,\epsilon_2,i,j,k,l_0)} \neq 0$ for some $(\epsilon_1, \epsilon_2, i, j, k_0, l), (\epsilon_1, \epsilon_2, i, j, k, l_0) \in I$, and $a_{(\epsilon_1,\epsilon_2,i,j,k,l)} = 0$ whenever $k < k_0$ or $l < l_0$. Write

$$x_- = \sum_{\substack{(\epsilon_1, \epsilon_2, \underline{\upsilon}) \in I \\ k_0 \leq k \leq -1 \text{ or } l_0 \leq l \leq -1}} a_{(\epsilon_1, \epsilon_2, \underline{\upsilon})} e_1^i e_2^j e_3^{\epsilon_1} e_4^{\epsilon_2} t_5^k t_6^l.$$

Now, $D(e_3) = \operatorname{ad}_{x_+}(e_3) + \operatorname{ad}_{x_-}(e_3) + \delta(e_3) \in A_{\alpha,\beta}$. From Remark 4.1.8, we have that $e_3 = e_{3,6} - st_5^2t_6^{-1}$ and $e_{3,6} = t_3 - at_4t_5^{-1}$. Putting these two together gives

$$e_3 = t_3 - at_4 t_5^{-1} - st_5^2 t_6^{-1}.$$

Again, from Remark 4.1.8, we also have that $t_4 = e_4 + bt_5^3 t_6^{-1}$. Note, $\delta(t_{\kappa}) = \lambda_{\kappa} t_{\kappa}$, $\kappa \in \{3, 4, 5, 6\}$. Now,

$$\delta(e_3) = \lambda_3 t_3 - a(\lambda_4 - \lambda_5) t_4 t_5^{-1} - s(2\lambda_5 - \lambda_6) t_5^2 t_6^{-1}$$

$$= \lambda_3 (e_{3,6} + a t_4 t_5^{-1}) + a(\lambda_5 - \lambda_4) t_4 t_5^{-1} + s(\lambda_6 - 2\lambda_5) t_5^2 t_6^{-1}$$

$$= \lambda_3 e_{3,6} + a(\lambda_3 - \lambda_4 + \lambda_5) t_4 t_5^{-1} + s(\lambda_6 - 2\lambda_5) t_5^2 t_6^{-1}$$

$$= \lambda_3 (e_3 + s t_5^2 t_6^{-1}) + a(\lambda_3 - \lambda_4 + \lambda_5) (e_4 + b t_5^3 t_6^{-1}) t_5^{-1} + s(\lambda_6 - 2\lambda_5) t_5^2 t_6^{-1}$$

$$= \lambda_3 e_3 + \alpha_1 e_4 t_5^{-1} + \alpha_2 t_5^2 t_6^{-1}, \qquad (4.2.15)$$

where $\alpha_1 = a(\lambda_3 - \lambda_4 + \lambda_5)$ and $\alpha_2 = s(\lambda_3 - 2\lambda_5 + \lambda_6) + q^{-3}ab(\lambda_3 - \lambda_4 + \lambda_5)$. Therefore, $D(e_3) = ad_{x_+}(e_3) + ad_{x_-}(e_3) + \lambda_3e_3 + \alpha_1e_4t_5^{-1} + \alpha_2t_5^2t_6^{-1} \in A_{\alpha,\beta}$. It follows that $D(e_3)t_5t_6 = ad_{x_+}(e_3)t_5t_6 + ad_{x_-}(e_3)t_5t_6 + \lambda_3e_3t_5t_6 + \alpha_1e_4t_6 + q^3\alpha_2t_5^3 \in A_{\alpha,\beta}$. Hence, $ad_{x_-}(e_3)t_5t_6 \in A_{\alpha,\beta}$, since $ad_{x_+}(e_3)t_5t_6 + \lambda_3e_3t_5t_6 + \alpha_1e_4t_6 + q^3\alpha_2t_5^3 \in A_{\alpha,\beta}$.

Now,

$$\mathsf{ad}_{x_{-}}(e_{3}) = \sum_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon})\in I} a_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon})} e_{1}^{i} e_{2}^{j} e_{3}^{\epsilon_{1}} e_{4}^{\epsilon_{2}} t_{5}^{k} t_{6}^{l} e_{3} - \sum_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon})\in I} a_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon})} e_{3} e_{1}^{i} e_{2}^{j} e_{3}^{\epsilon_{1}} e_{4}^{\epsilon_{2}} t_{5}^{k} t_{6}^{l}.$$
(4.2.16)

Using Lemma A.1.4, we have the following:

$$t_5^k t_6^l e_3 = q^{-k} e_3 t_5^k t_6^l + d_2[k] e_4 t_5^{k-1} t_6^l + d_3[l] t_5^{k+2} t_6^{l-1},$$
(4.2.17)

$$e_3 e_1^i e_2^j = q^{-i-3j} e_1^i e_2^j e_3 + d_2[i] e_1^{i-1} e_2^{j+1}, (4.2.18)$$

(note: $d_2[i], d_2[k]$ and $d_3[l]$ are defined in Lemma A.1.4). Substitute (4.2.17) and (4.2.18)

into (4.2.16), simplify and multiply (on the right) by t_5t_6 to obtain

$$\begin{aligned} \mathsf{ad}_{x_{-}}(e_{3})t_{5}t_{6} &= \\ &\sum_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon})\in I} a_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon})} \left(g[i,j,\epsilon_{2},l]e_{1}^{i}e_{2}^{j}e_{3}^{\epsilon_{1}+1}e_{4}^{\epsilon_{2}}t_{5}^{k+1}t_{6}^{l+1} + q^{-3l}d_{2}[k]e_{1}^{i}e_{2}^{j}e_{3}^{\epsilon_{1}}e_{4}^{\epsilon_{2}+1}t_{5}^{k}t_{6}^{l+1} \\ &+ q^{-3(l-1)}d_{3}[l]e_{1}^{i}e_{2}^{j}e_{3}^{\epsilon_{1}}e_{4}^{\epsilon_{2}}t_{5}^{k+3}t_{6}^{l} - q^{-3l}d_{2}[i]e_{1}^{i-1}e_{2}^{j+1}e_{3}^{\epsilon_{1}}e_{4}^{\epsilon_{2}}t_{5}^{k+1}t_{6}^{l+1}\right), \end{aligned}$$

$$(4.2.19)$$

where $g[i, j, \epsilon_2, l] := q^{-k-3\epsilon_2-3l} - q^{-i-3j-3l}$.

Assume that there exists l < 0 such that $a_{(\epsilon_1,\epsilon_2,\underline{v})} \neq 0$. It follows from our initial assumption that $a_{(\epsilon_1,\epsilon_2,i,j,k,l_0)} \neq 0$. Now, at $l = l_0$, denote $\underline{v} = (i, j, k, l)$ by $\underline{v}_0 := (i, j, k, l_0)$. From (4.2.19), we have that

$$\mathsf{ad}_{x_{-}}(e_{3})t_{5}t_{6} = \sum_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon}_{0})\in I} q^{-3(l_{0}-1)}a_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon}_{0})}d_{3}[l_{0}]e_{1}^{i}e_{2}^{j}e_{3}^{\epsilon_{1}}e_{4}^{\epsilon_{2}}t_{5}^{k+3}t_{6}^{l_{0}} + \mathcal{J}_{1},$$

where $\mathcal{J}_1 \in \text{Span}\left(\mathcal{B} \setminus \{e_1^i e_2^j e_3^{\epsilon_1} e_4^{\epsilon_2} t_5^k t_6^{l_0} \mid \epsilon_1, \epsilon_2 \in \{0, 1\}, k \in \mathbb{Z} \text{ and } i, j \in \mathbb{N}\}\right)$.

Set $\underline{w} := (i, j, k, l) \in \mathbb{N}^4$. One can also write $\operatorname{ad}_{x_-}(e_3)t_5t_6 \in A_{\alpha,\beta}$ in terms of the basis \mathcal{B} of $A_{\alpha,\beta}$ (Proposition 3.2.2) as:

$$\mathsf{ad}_{x_{-}}(e_3)t_5t_6 = \sum_{(\epsilon_1,\epsilon_2,\underline{w})\in J} b_{(\epsilon_1,\epsilon_2,\underline{w})} e_1^i e_2^j e_3^{\epsilon_1} e_4^{\epsilon_2} t_5^k t_6^l,$$
(4.2.20)

where J is a finite subset of $\{0,1\}^2 \times \mathbb{N}^4$, and $b_{(\epsilon_1,\epsilon_2,\underline{w})} \in \mathbb{C}$. It follows that

$$\sum_{(\epsilon_1,\epsilon_2,\underline{w})\in J} b_{(\epsilon_1,\epsilon_2,\underline{w})} e_1^i e_2^j e_3^{\epsilon_1} e_4^{\epsilon_2} t_5^k t_6^l = \sum_{(\epsilon_1,\epsilon_2,\underline{v}_0)\in I} q^{-3(l_0-1)} a_{(\epsilon_1,\epsilon_2,\underline{v}_0)} d_3[l_0] e_1^i e_2^j e_3^{\epsilon_1} e_4^{\epsilon_2} t_5^{k+3} t_6^{l_0} + \mathcal{J}_1.$$

Since \mathcal{B} is a basis for $A_{\alpha,\beta}$, we deduce from Corollary 3.2.3 that

 $(e_1^i e_2^j e_3^{\epsilon_1} e_4^{\epsilon_2} t_5^k t_6^l)_{((\epsilon_1, \epsilon_2, \underline{v}) \in \{0, 1\}^2 \times \mathbb{N}^2 \times \mathbb{Z}^2)} \text{ is also a basis for } A_{\alpha, \beta}[t_5^{-1}, t_6^{-1}]. \text{ Since } \underline{v}_0 = (i, j, k, l_0) \in \mathbb{N}^2 \times \mathbb{Z}^2 \text{ (with } l_0 < 0) \text{ and } \underline{w} = (i, j, k, l) \in \mathbb{N}^4 \text{ (with } l \ge 0) \text{ in the above equality, we must have}$

$$q^{-3(l_0-1)}a_{(\epsilon_1,\epsilon_2,\underline{\upsilon}_0)}d_3[l_0] = 0.$$

Given that $q^{-3(l_0-1)}d_3[l_0] \neq 0$, it follows that $a_{(\epsilon_1,\epsilon_2,\underline{v}_0)} = a_{(\epsilon_1,\epsilon_2,i,j,k,l_0)}$ are all zero. This is a contradiction. Therefore, $l \geq 0$ (i.e. there is no negative exponent for t_6).

Since $l \ge 0$, it follows from our initial assumption that there exists $k = k_0 < 0$ such that $a_{(\epsilon_1, \epsilon_2, i, j, k_0, l)} \ne 0$. The rest of the proof will show that this assumption cannot also hold. Set $\underline{v}_0 := (i, j, k_0, l) \in \mathbb{N}^2 \times \mathbb{Z} \times \mathbb{N}$. From (4.2.19), we have that

$$\mathsf{ad}_{x_{-}}(e_{3})t_{5}t_{6} = \sum_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon}_{0})\in I} q^{-3l} a_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon}_{0})} d_{2}[k_{0}] e_{1}^{i} e_{2}^{j} e_{3}^{\epsilon_{1}} e_{4}^{\epsilon_{2}+1} t_{5}^{k_{0}} t_{6}^{l+1} + V,$$

where $V \in \mathcal{J}_2 := \text{Span}\left(\mathcal{B} \setminus \{e_1^i e_2^j e_3^{\epsilon_1} e_4^{\epsilon_2} t_5^{k_0} t_6^l \mid \epsilon_1, \epsilon_2 \in \{0, 1\} \text{ and } i, j, l \in \mathbb{N}\}\right)$. It follows that:

$$\begin{aligned} \mathsf{ad}_{x_{-}}(e_{3})t_{5}t_{6} &= \\ & \sum_{(0,0,\underline{\upsilon}_{0})\in I} q^{-3l}a_{(0,0,\underline{\upsilon}_{0})}d_{2}[k_{0}]e_{1}^{i}e_{2}^{j}e_{4}t_{5}^{k_{0}}t_{6}^{l+1} + \sum_{(1,0,\underline{\upsilon}_{0})\in I} a_{(1,0,\underline{\upsilon}_{0})}d_{2}[k_{0}]e_{1}^{i}e_{2}^{j}e_{3}e_{4}t_{5}^{k_{0}}t_{6}^{l+1} \\ &+ \sum_{(0,1,\underline{\upsilon}_{0})\in I} q^{-3l}a_{(0,1,\underline{\upsilon}_{0})}d_{2}[k_{0}]e_{1}^{i}e_{2}^{j}e_{4}^{2}t_{5}^{k_{0}}t_{6}^{l+1} \sum_{(1,1,\underline{\upsilon}_{0})\in I} a_{(1,1,\underline{\upsilon}_{0})}d_{2}[k_{0}]e_{1}^{i}e_{2}^{j}e_{3}e_{4}^{2}t_{5}^{k_{0}}t_{6}^{l+1} + V. \end{aligned}$$

$$(4.2.21)$$

Write the relations in Lemma 3.0.1(2),(4) as:

$$e_4^2 = b_1\beta + b_2e_2e_4e_6 + b_4\alpha e_3e_6 + b_6e_1e_3e_4e_6 + L_1, \qquad (4.2.22)$$

$$e_{3}e_{4}^{2} = \beta b_{1}e_{3} + k_{1}e_{2}e_{3}e_{4}e_{6} + k_{3}\alpha^{2}e_{6} + k_{5}\alpha e_{1}e_{4}e_{6} + k_{14}\beta e_{1}^{2}e_{6} + k_{15}e_{1}^{2}e_{2}e_{4}e_{6}^{2} + k_{17}\alpha e_{1}^{2}e_{3}e_{6}^{2} + k_{19}e_{1}^{3}e_{3}e_{4}e_{6}^{2} + L_{2},$$
(4.2.23)

where L_1 and L_2 are some elements of the left ideal $A_{\alpha,\beta}t_5 \subseteq \mathcal{J}_2$. Substitute (4.2.22) and (4.2.23) into (4.2.21), and simplify to obtain:

$$\begin{aligned} \mathsf{ad}_{x_{-}}(e_{3})t_{5}t_{6} &= \sum [\lambda_{1,1}\beta a_{(0,1,i,j,k_{0},l-1)} + \lambda_{1,2}\alpha^{2}a_{(1,1,i,j,k_{0},l-2)} \\ &+ \lambda_{1,3}\beta a_{(1,1,i-2,j,k_{0},l-2)}]e_{1}^{i}e_{2}^{j}t_{5}^{k_{0}}t_{6}^{l} \\ &+ \sum [\lambda_{2,1}\alpha a_{(0,1,i,j,k_{0},l-2)} + \lambda_{2,2}\beta a_{(1,1,i,j,k_{0},l-1)} \\ &+ \lambda_{2,3}\alpha a_{(1,1,i-2,j,k_{0},l-3)}]e_{1}^{i}e_{2}^{j}e_{3}t_{5}^{k_{0}}t_{6}^{l} \\ &+ \sum [\lambda_{3,1}a_{(0,1,i,j-1,k_{0},l-2)} + \lambda_{3,2}\alpha a_{(1,1,i-1,j,k_{0},l-2)} \\ &+ \lambda_{3,3}a_{(1,1,i-2,j-1,k_{0},l-3)} + \lambda_{3,4}a_{(0,0,i,j,k_{0},l-1)}]e_{1}^{i}e_{2}^{j}e_{4}t_{5}^{k_{0}}t_{6}^{l} \\ &+ \sum [\lambda_{4,1}a_{(0,1,i-1,j,k_{0},l-2)} + \lambda_{4,2}a_{(1,1,i,j-1,k_{0},l-2)} \\ &+ \lambda_{4,3}a_{(1,1,i-3,j,k_{0},l-3)} + \lambda_{4,4}a_{(1,0,i,j,k_{0},l-1)}]e_{1}^{i}e_{2}^{j}e_{3}e_{4}t_{5}^{k_{0}}t_{6}^{l} + V', \end{aligned}$$

$$(4.2.24)$$

where $V' \in \mathcal{J}_2$. Also, $\lambda_{s,t} := \lambda_{s,t}(j, k_0, l)$ are some families of complex numbers which are non-zero for all $s, t \in \{1, 2, 3, 4\}$ and $j, l \in \mathbb{N}$, except $\lambda_{1,4}$ and $\lambda_{2,4}$ which are assumed to be zero since they do not exist in the above expression. Note, although each $\lambda_{s,t}$ depends on j, k_0, l , we have not made this dependency explicit in the above expression since the minimum requirement we need to complete the proof is for all the $\lambda_{s,t}$ existing in the above expression to be non-zero, which we already have. Observe that (4.2.24) and (4.2.20) are equal, hence,

$$\begin{split} \sum_{(\epsilon_1,\epsilon_2,\underline{w})\in J} b_{(\epsilon_1,\epsilon_2,\underline{w})} e_1^i e_2^j e_3^{\epsilon_1} e_4^{\epsilon_2} t_5^k t_6^l &= \sum [\lambda_{1,1}\beta a_{(0,1,i,j,k_0,l-1)} + \lambda_{1,2}\alpha^2 a_{(1,1,i,j,k_0,l-2)} \\ &+ \lambda_{1,3}\beta a_{(1,1,i-2,j,k_0,l-2)}] e_1^i e_2^j t_5^{k_0} t_6^l \\ &+ \sum [\lambda_{2,1}\alpha a_{(0,1,i,j,k_0,l-2)} + \lambda_{2,2}\beta a_{(1,1,i,j,k_0,l-1)} \\ &+ \lambda_{2,3}\alpha a_{(1,1,i-2,j,k_0,l-3)}] e_1^i e_2^j e_3 t_5^{k_0} t_6^l \\ &+ \sum [\lambda_{3,1}a_{(0,1,i,j-1,k_0,l-2)} + \lambda_{3,2}\alpha a_{(1,1,i-1,j,k_0,l-2)} \\ &+ \lambda_{3,3}a_{(1,1,i-2,j-1,k_0,l-3)} + \lambda_{3,4}a_{(0,0,i,j,k_0,l-1)}] e_1^i e_2^j e_4 t_5^{k_0} t_6^l \\ &+ \sum [\lambda_{4,1}a_{(0,1,i-1,j,k_0,l-2)} + \lambda_{4,2}a_{(1,1,i,j-1,k_0,l-2)} \\ &+ \lambda_{4,3}a_{(1,1,i-3,j,k_0,l-3)} + \lambda_{4,4}a_{(1,0,i,j,k_0,l-1)}] e_1^i e_2^j e_3 e_4 t_5^{k_0} t_6^l + V'. \end{split}$$

We have previously established that $(e_1^i e_2^j e_3^{\epsilon_1} e_4^{\epsilon_2} t_5^k t_6^l)_{((\epsilon_1, \epsilon_2, \underline{v}) \in \{0,1\}^2 \times \mathbb{Z}^2 \times \mathbb{Z}^2)}$ is a basis for $A_{\alpha,\beta}[t_5^{-1}, t_6^{-1}]$ (note, in this part of the proof $l \ge 0$). Since $\underline{v}_0 = (i, j, k_0, l) \in \mathbb{N}^2 \times \mathbb{Z} \times \mathbb{N}$ (with $k_0 < 0$) and $\underline{w} = (i, j, k, l) \in \mathbb{N}^4$ (with $k \ge 0$) in the above equality, it follows that

$$\lambda_{1,1}\beta a_{(0,1,i,j,k_0,l-1)} + \lambda_{1,2}\alpha^2 a_{(1,1,i,j,k_0,l-2)} + \lambda_{1,3}\beta a_{(1,1,i-2,j,k_0,l-2)} = 0, \qquad (4.2.25)$$

$$\lambda_{2,1}\alpha a_{(0,1,i,j,k_0,l-2)} + \lambda_{2,2}\beta a_{(1,1,i,j,k_0,l-1)} + \lambda_{2,3}\alpha a_{(1,1,i-2,j,k_0,l-3)} = 0,$$
(4.2.26)

$$\lambda_{3,1}a_{(0,1,i,j-1,k_0,l-2)} + \lambda_{3,2}\alpha a_{(1,1,i-1,j,k_0,l-2)} + \lambda_{3,3}a_{(1,1,i-2,j-1,k_0,l-3)} + \lambda_{3,4}a_{(0,0,i,j,k_0,l-1)} = 0,$$
(4.2.27)

$$\lambda_{4,1}a_{(0,1,i-1,j,k_0,l-2)} + \lambda_{4,2}a_{(1,1,i,j-1,k_0,l-2)} + \lambda_{4,3}a_{(1,1,i-3,j,k_0,l-3)} + \lambda_{4,4}a_{(1,0,i,j,k_0,l-1)} = 0.$$
(4.2.28)

From (4.2.25) and (4.2.26), one can easily deduce that

$$a_{(0,1,i,j,k_0,l)} = -\frac{\alpha^2 \lambda_{1,2}}{\beta \lambda_{1,1}} a_{(1,1,i,j,k_0,l-1)} - \frac{\lambda_{1,3}}{\lambda_{1,1}} a_{(1,1,i-2,j,k_0,l-1)},$$
(4.2.29)

$$a_{(1,1,i,j,k_0,l)} = -\frac{\alpha\lambda_{2,1}}{\beta\lambda_{2,2}}a_{(0,1,i,j,k_0,l-1)} - \frac{\alpha\lambda_{2,3}}{\beta\lambda_{2,2}}a_{(1,1,i-2,j,k_0,l-2)}.$$
(4.2.30)
Note, $a_{(\epsilon_1, \epsilon_2, i, j, k_0, l)} := 0$ whenever i < 0 or j < 0 or l < 0 for all $\epsilon_1, \epsilon_2 \in \{0, 1\}$.

Claim. The coefficients $a_{(0,1,i,j,k_0,l)}$ and $a_{(1,1,i,j,k_0,l)}$ are all zero for all $l \ge 0$. We now justify the claim by an induction on l. From (4.2.29) and (4.2.30), the result is obviously true when l = 0. For $l \ge 0$, assume that $a_{(0,1,i,j,k_0,l)} = a_{(1,1,i,j,k_0,l)} = 0$. Then, it follows from (4.2.29) and (4.2.30) that

$$a_{(0,1,i,j,k_0,l+1)} = -\frac{\alpha^2 \lambda_{1,2}}{\beta \lambda_{1,1}} a_{(1,1,i,j,k_0,l)} - \frac{\lambda_{1,3}}{\lambda_{1,1}} a_{(1,1,i-2,j,k_0,l)},$$

$$a_{(1,1,i,j,k_0,l+1)} = -\frac{\alpha\lambda_{2,1}}{\beta\lambda_{2,2}}a_{(0,1,i,j,k_0,l)} - \frac{\alpha\lambda_{2,3}}{\beta\lambda_{2,2}}a_{(1,1,i-2,j,k_0,l-1)}$$

From the inductive hypothesis, $a_{(1,1,i,j,k_0,l)} = a_{(1,1,i-2,j,k_0,l)} = a_{(0,1,i,j,k_0,l)} = a_{(1,1,i-2,j,k_0,l-1)}$ = 0. Hence, $a_{(1,1,i,j,k_0,l+1)} = a_{(0,1,i,j,k_0,l+1)} = 0$. By the principle of mathematical induction, $a_{(0,1,i,j,k_0,l)} = a_{(1,1,i,j,k_0,l)} = 0$ for all $l \ge 0$ as desired. Given that the families $a_{(0,1,i,j,k_0,l)}$ and $a_{(1,1,i,j,k_0,l)}$ are all zero, it follows from (4.2.27) and (4.2.28) that $a_{(0,0,i,j,k_0,l)}$ and $a_{(1,0,i,j,k_0,l)}$ are also zero for all $(i, j, k_0, l) \in \mathbb{N}^2 \times \mathbb{Z} \times \mathbb{N}$. Since $a_{(\epsilon_1,\epsilon_2,i,j,k_0,l)}$ are all zero, it contradicts our assumption. Hence, $x_- = 0$. Consequently, $x = x_+ \in A_{\alpha,\beta}$ as desired.

2. From Remark 4.1.8, we have $e_4 = t_4 - bt_5^3t_6^{-1}$. Again, from Lemma 4.2.3, we have that $\lambda_4 = 3\lambda_3 + \lambda_5$ and $\lambda_6 = -3\lambda_3$. Therefore,

$$\delta(e_4) = \lambda_4 t_4 - b(3\lambda_5 - \lambda_6) t_5^3 t_6^{-1}$$

= $(3\lambda_3 + \lambda_5) e_{4,6} - 3b(\lambda_3 + \lambda_5) t_5^3 t_6^{-1}$
= $(3\lambda_3 + \lambda_5) (e_4 + bt_5^3 t_6^{-1}) - 3b(\lambda_3 + \lambda_5) t_5^3 t_6^{-1}$
= $(3\lambda_3 + \lambda_5) e_4 - 2b\lambda_5 t_5^3 t_6^{-1}$.

Moreover, $D(e_4) = \operatorname{ad}_x(e_4) + \delta(e_4) = \operatorname{ad}_x(e_4) + (3\lambda_3 + \lambda_5)e_4 - 2b\lambda_5t_5^3t_6^{-1} \in A_{\alpha,\beta}$. It follows that $b\lambda_5t_5^3t_6^{-1} \in A_{\alpha,\beta}$, since $\operatorname{ad}_x(e_4) + (3\lambda_3 + \lambda_5)e_4 \in A_{\alpha,\beta}$. Consequently, $b\lambda_5t_5^3 \in A_{\alpha,\beta}t_6$. Since $b \neq 0$ (Appendix C), we must have $\lambda_5 = 0$, otherwise, there will be a contradiction using the basis of $A_{\alpha,\beta}$ (Proposition 3.2.2). Therefore, $\delta(e_4) =$ $3\lambda_3 e_4$ and $\delta(t_5) = 0$. We already know from Lemma 4.2.3 that $\delta(t_6) = -3\lambda_3 t_6$. From (4.2.15), we have that $\delta(e_3) = \lambda_3 e_3 + a(\lambda_3 - \lambda_4 + \lambda_5)e_4t_5^{-1} + [s(\lambda_3 - 2\lambda_5 + \lambda_6) + q^{-3}ab(\lambda_3 - \lambda_4 + \lambda_5)]t_5^2t_6^{-1}$. Given that $\lambda_4 = 3\lambda_3$, $\lambda_5 = 0$ and $\lambda_6 = -3\lambda_3$, we have that $\delta(e_3) = \lambda_3 e_3 - 2a\lambda_3 e_4 t_5^{-1}$ (note, from Appendix C, one can confirm that $q^{-3}ab + s = 0$). Now, $D(e_3) = ad_x(e_3) + \delta(e_3) = ad_x(e_3) + \lambda_3 e_3 - 2a\lambda_3 e_4 t_5^{-1} \in A_{\alpha,\beta}$. Observe that $ad_x(e_3) + \lambda_3 e_3 \in A_{\alpha,\beta}$. Hence, $2a\lambda_3 e_4 t_5^{-1} \in A_{\alpha,\beta}$, and so $2a\lambda_3 e_4 \in A_{\alpha,\beta} t_5$. Since $a \neq 0$, it implies that $\lambda_3 = 0$, otherwise, there will be a contradiction using the basis of $A_{\alpha,\beta}$. We now have that $\delta(e_3) = \delta(e_4) = \delta(e_5) = \delta(e_6) = 0$. We finish the proof by showing that $\delta(e_1) = \delta(e_2) = 0$. Recall from (3.0.2) that

$$e_2e_4e_6 + be_2e_5^3 + be_3^3e_6 + b'e_3^2e_5^2 + c'e_3e_4e_5 + d'e_4^2 = \beta.$$

Apply δ to this relation to obtain $\delta(e_2)e_4e_6 + b\delta(e_2)e_5^3 = 0$. This implies that $\delta(e_2)(e_4e_6 + be_5^3) = 0$. Since $e_4e_6 + be_5^3 \neq 0$, it follows that $\delta(e_2) = 0$. Similarly, from (3.0.1), we have that

$$e_1e_3e_5 + ae_1e_4 + ae_2e_5 + a'e_3^2 = \alpha.$$

Apply δ to this relation to obtain $\delta(e_1)(e_3e_5 + ae_4) = 0$. Since $e_3e_5 + ae_4 \neq 0$, we have that $\delta(e_1) = 0$. In conclusion, $\delta(e_{\kappa}) = 0$ for all $\kappa \in \{1, \dots, 6\}$.

3. As a result of (1) and (2), we have that $D(e_{\kappa}) = \operatorname{ad}_{x}(e_{\kappa})$. Therefore, $D = \operatorname{ad}_{x}$ as desired.

4.2.5 Derivations of $A_{\alpha,0}$ and $A_{0,\beta}$. In this subsection, we explore the derivations of $A_{\alpha,\beta}$ when either α or β is zero (but not both). We are going to follow the same pattern used to compute the derivations of $A_{\alpha,\beta}$ ($\alpha, \beta \neq 0$) in the previous subsection. Of course, results that can easily be obtained from the previous subsection are not going to be repeated here. In this case, the appropriate references shall be made. We will begin with the derivations of $A_{\alpha,0}$.

Derivations of $A_{\alpha,0}$. Let $\text{Der}(A_{\alpha,0})$ denote the \mathbb{C} -derivations of $A_{\alpha,0}$ and $D \in \text{Der}(A_{\alpha,0})$. We already know from the previous subsection that D extends uniquely to

a derivation of each of the series of algebras in (4.1.2) via localization. Therefore, D extends to a derivation of the quantum torus $R_3 = \mathbb{C}_{q^N}[t_3^{\pm 1}, t_4^{\pm 1}, t_5^{\pm 1}, t_6^{\pm 1}]$. It follows from [40, Corollary 2.3] that D can uniquely be written as:

$$D = \operatorname{ad}_x + \delta,$$

where $x \in R_3$, and δ is a scalar derivation of R_3 defined as $\delta(t_i) = \lambda_i t_i$ for each i = 3, 4, 5, 6. Note, $\lambda_i \in Z(R_3) = \mathbb{C}$ (Lemma 4.1.7). Also, ad_x is an inner derivation of R_3 defined as $ad_x(L) = xL - Lx$ for all $L \in R_3$.

We will describe D as a derivation of $A_{\alpha,0}$. We first describe D as a derivation of R_4 .

- **4.2.6 Lemma.** 1. $x \in R_4$.
 - 2. $\delta(f_1) = -(\lambda_3 + \lambda_5)f_1$ and $\delta(t_2) = 0$.
 - 3. Set $\lambda_1 := -(\lambda_3 + \lambda_5)$ and $\lambda_2 := 0$. Then, $D(e_{\kappa,4}) = \operatorname{ad}_x(e_{\kappa,4}) + \lambda_{\kappa}e_{\kappa,4}$ for all $\kappa \in \{1, \dots, 6\}$.

Proof. 1. Similar to that of Lemma 4.2.2(1).

2. From Section 3.1, we have that $t_1 = \alpha t_5^{-1} t_3^{-1}$ and $t_2 = \beta t_6^{-1} t_4^{-1}$ in $R_3 = \mathscr{A}_{\alpha,\beta}$. Clearly, $t_2 = 0$ since $\beta = 0$. From Remark 4.1.8, we have that $f_1 = t_1 - a t_2 t_3^{-1}$. This simplifies to $f_1 = t_1 = \alpha t_5^{-1} t_3^{-1}$. It follows that $\delta(f_1) = -(\lambda_3 + \lambda_5) f_1$. Obviously, $\delta(t_2) = 0$.

3. As a result of (1) and (2), $D(e_{\kappa,4}) = \operatorname{ad}_x(e_{\kappa,4}) + \lambda_{\kappa}e_{\kappa,4}$ for all $\kappa \in \{1, \dots, 6\}$.

We proceed to also describe D as a derivation of R_5 .

4.2.7 Lemma. 1. $x \in R_5$.

- 2. $\lambda_5 = \lambda_4 3\lambda_3$, $\delta(z_1) = -(\lambda_3 + \lambda_5)z_1$ and $\delta(z_2) = -\lambda_5 z_2$.
- 3. Set $\lambda_1 := -(\lambda_3 + \lambda_5)$ and $\lambda_2 := -\lambda_5$. Then, $D(e_{\kappa,5}) = \operatorname{ad}_x(e_{\kappa,5}) + \lambda_{\kappa}e_{\kappa,5}$ for all $\kappa \in \{1, \dots, 6\}$.

Proof. 1. Similar to that of Lemma 4.2.3(1).

2. Recall that $\delta(t_i) = \lambda_i t_i$, i = 3, 4, 5, 6, and $\delta(f_1) = -(\lambda_3 + \lambda_5)f_1$. From Remark 4.1.8, we have that $z_1 = f_1 - st_3^2t_4^{-1}$. One can confirm that $\delta(z_1) = -(\lambda_3 + \lambda_5)z_1 + s(\lambda_4 - 3\lambda_3 - \lambda_5)t_3^2t_4^{-1}$. Now, $D(z_1) = ad_x + \delta(z_1) = ad_x - (\lambda_3 + \lambda_5)z_1 + s(\lambda_4 - 3\lambda_3 - \lambda_5)t_3^2t_4^{-1} \in R_5$. Since $ad_x - (\lambda_3 + \lambda_5)z_1 \in R_5$, we have that $s(\lambda_4 - 3\lambda_3 - \lambda_5)t_3^2t_4^{-1} \in R_5$. Consequently, $s(\lambda_4 - 3\lambda_3 - \lambda_5)t_3^2 \in R_5t_4$. Since $s \neq 0$ (Appendix C), it follows that $\lambda_4 - 3\lambda_3 - \lambda_5 = 0$, and so $\lambda_5 = \lambda_4 - 3\lambda_3$. Otherwise, there will be a contradiction using the basis of R_5 (Proposition 4.1.4). We now have $\delta(z_1) = -(\lambda_3 + \lambda_5)z_1$. From the previous lemma, we have that $t_2 = 0$. As a result, the relation $z_2 = t_2 - bt_3^3t_4^{-1}$ in Remark 4.1.8 becomes $z_2 = -bt_3^3t_4^{-1}$. It follows that $\delta(z_2) = (3\lambda_3 - \lambda_4)z_2 = -\lambda_5z_2$.

3. Set $\lambda_1 := -(\lambda_3 + \lambda_5)$ and $\lambda_2 := -\lambda_5$. It follows from points (1) and (2) that $D(e_{\kappa,5}) = \operatorname{ad}_x(e_{\kappa,5}) + \lambda_{\kappa}e_{\kappa,5}$ for all $\kappa \in \{1, \cdots, 6\}$.

We now describe D as a derivation of $A_{\alpha,0}$.

4.2.8 Lemma. 1. $x \in A_{\alpha,0}$.

- 2. $\lambda_6 = 2\lambda_5$, $\delta(e_1) = -\lambda_5 e_1$, $\delta(e_2) = -\lambda_5 e_2$, $\delta(e_3) = 0$ and $\delta(e_4) = \lambda_5 e_4$.
- 3. Set $\lambda_1 := -\lambda_5$, $\lambda_2 := -\lambda_5$, $\lambda_3 := 0$ and $\lambda_4 := \lambda_5$. Then, $D(e_{\kappa}) = \operatorname{ad}_x(e_{\kappa}) + \lambda_{\kappa}e_{\kappa}$ for all $\kappa \in \{1, \dots, 6\}$.

Proof. 1. Similar to the proof of Lemma 4.2.4(1).

2. From Lemma 4.2.7, we have $\lambda_5 = \lambda_4 - 3\lambda_3$, which implies that $\lambda_4 = 3\lambda_3 + \lambda_5$. Furthermore, from Remark 4.1.8, we have $e_4 = t_4 - bt_5^3t_6^{-1}$. One can confirm that $\delta(e_4) = (3\lambda_3 + \lambda_5)e_4 + b(3\lambda_3 - 2\lambda_5 + \lambda_6)t_5^3t_6^{-1}$. Now, $D(e_4) = ad_x(e_4) + \delta(e_4) = ad_x(e_4) + (3\lambda_3 + \lambda_5)e_4 + b(3\lambda_3 - 2\lambda_5 + \lambda_6)t_5^3t_6^{-1} \in A_{\alpha,0}$. Since $ad_x(e_4) + (3\lambda_3 + \lambda_5)e_4 \in A_{\alpha,0}$, it follows that $b(3\lambda_3 - 2\lambda_5 + \lambda_6)t_5^3t_6^{-1} \in A_{\alpha,0}$. This implies that $b(3\lambda_3 - 2\lambda_5 + \lambda_6)t_5^3 \in A_{\alpha,0}t_6$. Note, $b \neq 0$ (Appendix C). As a result, $3\lambda_3 - 2\lambda_5 + \lambda_6 = 0$, otherwise, we will have a contradiction using the basis of $A_{\alpha,0}$ (Proposition 3.2.2). Consequently, $\lambda_6 = 2\lambda_5 - 3\lambda_3$ and $\delta(e_4) = (3\lambda_3 + \lambda_5)e_4$. The relation $\delta(e_3) = \lambda_3 e_3 + a(\lambda_3 - \lambda_4 + \lambda_5)e_4t_5^{-1} + [s(\lambda_3 - 2\lambda_5 + \lambda_6) + q^{-3}ab(\lambda_3 - \lambda_4 + \lambda_5)]t_5^2t_6^{-1}$ (see (4.2.15)) is also valid in $A_{\alpha,0}$. Given that $\lambda_4 = 3\lambda_3 + \lambda_5$ and $\lambda_6 = 2\lambda_5 - 3\lambda_3$, one can easily verify that $\delta(e_3) = \lambda_3 e_3 - 2a\lambda_3 e_4 t_5^{-1}$ (note, from Appendix C, we have that $q^{-3}ab + s = 0$). Now, $D(e_3) = ad_x(e_3) + \delta(e_3) = ad_x(e_3) + \lambda_3 e_3 - 2a\lambda_3 e_4 t_5^{-1} \in A_{\alpha,0}$. Since $ad_x(e_3) + \lambda_3 e_3 \in A_{\alpha,0}$, we have that $2a\lambda_3 e_4 t_5^{-1} \in A_{\alpha,0}$, which implies that $2a\lambda_3 e_4 \in A_{\alpha,0}t_5$. Hence, $\lambda_3 = 0$, otherwise, we will have a contradiction using the basis of $A_{\alpha,0}$ (note, from Appendix C, $a \neq 0$). Since $\lambda_3 = 0$, we have: $\lambda_6 = 2\lambda_5$, $\delta(e_4) = \lambda_5 e_4$ and $\delta(e_3) = 0$. From Remark 4.1.8, we have: $e_1 = e_{1,6} - rt_5 t_6^{-1}$, $e_{1,6} = z_1 - he_{3,6} t_5^{-1} - gt_4 t_5^{-2}$ and $e_{3,6} = t_3 - at_4 t_5^{-1}$. Putting these three relations together gives

$$e_1 = z_1 - rt_5t_6^{-1} - ht_3t_5^{-1} + (ah - g)t_4t_5^{-2}.$$
(4.2.31)

From the previous lemma, we have that $\delta(z_1) = -(\lambda_3 + \lambda_5)z_1$. Given that $\lambda_6 = 2\lambda_5$, $\lambda_4 = \lambda_5$, $\lambda_3 = 0$ and $\delta(t_5) = \lambda_5 t_5$, one can verify that $\delta(e_1) = -\lambda_5 e_1$. Finally, from the commutation relations of $A_{\alpha,0}$ in Chapter 3, we have that $e_3e_1 = q^{-1}e_1e_3 - (q + q^{-1} + q^{-3})e_2$. Since $\delta(e_1) = -\lambda_5 e_1$ and $\delta(e_3) = 0$, applying δ to this relation gives $\delta(e_2) = -\lambda_5 e_2$.

3. Set $\lambda_1 := -\lambda_5$, $\lambda_2 := -\lambda_5$, $\lambda_3 := 0$ and $\lambda_4 := \lambda_5$, it follows from points (1) and (2) that $D(e_{\kappa}) = \operatorname{ad}_x(e_{\kappa}) + \lambda_{\kappa}e_{\kappa}$ for all $\kappa \in \{1, \dots, 6\}$.

Derivations of $A_{0,\beta}$. Every derivation D of $A_{0,\beta}$ extends uniquely to a derivation of each of the series of algebras in (4.1.2). Therefore, D extends to a derivation of the quantum torus $R_3 = \mathbb{C}_{q^N}[t_3^{\pm 1}, t_4^{\pm 1}, t_5^{\pm 1}, t_6^{\pm 1}]$. It follows from [40, Corollary 2.3] that D can uniquely be written as

$$D = \mathsf{ad}_x + \delta,$$

where $x \in R_3$, and δ is a scalar derivation of R_3 defined as $\delta(t_i) = \lambda_i t_i$ for each i = 3, 4, 5, 6. Note, $\lambda_i \in Z(R_3) = \mathbb{C}$ (Lemma 4.1.7). Also, ad_x is an inner derivation of R_3 defined as $ad_x(L) = xL - Lx$ for all $L \in R_3$.

We want to describe D as a derivation of $A_{0,\beta}$. We begin by describing D as a

derivation of R_4 .

4.2.9 Lemma. 1. $x \in R_4$.

- 2. $\delta(f_1) = -(\lambda_3 + \lambda_4 + \lambda_6)f_1$ and $\delta(t_2) = -(\lambda_6 + \lambda_4)t_2$.
- 3. Set $\lambda_1 := -(\lambda_3 + \lambda_4 + \lambda_6)$ and $\lambda_2 := -(\lambda_6 + \lambda_4)$. Then, $D(e_{\kappa,4}) = \operatorname{ad}_x(e_{\kappa,4}) + \lambda_{\kappa}e_{\kappa,4}$ for all $\kappa \in \{1, \dots, 6\}$.

Proof. 1. Similar to that of Lemma 4.2.2(1).

2. Recall from Section 3.1 that $t_1 = 0$ and $t_2 = \beta t_6^{-1} t_4^{-1}$ in $R_3 = \mathscr{A}_{0,\beta}$. Observe that $\delta(t_2) = -(\lambda_4 + \lambda_6)t_2$. From Remark 4.1.8, the relation $f_1 = t_1 - at_2t_3^{-1}$ becomes $f_1 = -at_2t_3^{-1} = -a\beta t_6^{-1}t_4^{-1}t_3^{-1}$. It follows that $\delta(f_1) = -(\lambda_3 + \lambda_4 + \lambda_6)f_1$.

3. Set $\lambda_1 := -(\lambda_3 + \lambda_4 + \lambda_6)$ and $\lambda_2 := -(\lambda_6 + \lambda_4)$, it follows from points (1) and (2) that $D(e_{\kappa,4}) = \operatorname{ad}_x(e_{\kappa,4}) + \lambda_{\kappa}e_{\kappa,4}$ for all $\kappa \in \{1, \dots, 6\}$.

We proceed to describe D as a derivation of R_5 .

4.2.10 Lemma. 1. $x \in R_5$.

- 2. $\lambda_6 = -3\lambda_3$, $\delta(z_1) = (2\lambda_3 \lambda_4)z_1$ and $\delta(z_2) = (3\lambda_3 \lambda_4)z_2$.
- 3. Set $\lambda_1 := (2\lambda_3 \lambda_4)$ and $\lambda_2 := 3\lambda_3 \lambda_4$. Then, $D(e_{\kappa,5}) = \operatorname{ad}_x(e_{\kappa,5}) + \lambda_{\kappa}e_{\kappa,5}$ for all $\kappa \in \{1, \dots, 6\}$.

Proof. 1. Similar to the proof of Lemma 4.2.3(1).

2. Recall that $\delta(t_i) = \lambda_i t_i$, i = 3, 4, 5, 6, and $\delta(t_2) = -(\lambda_4 + \lambda_6)t_2$. From Remark 4.1.8, we have that $z_2 = t_2 - bt_3^3 t_4^{-1}$. This implies that $\delta(z_2) = -(\lambda_4 + \lambda_6)z_2 - b(\lambda_6 + 3\lambda_3)t_3^3 t_4^{-1}$. Now, $D(z_2) = ad_x(z_2) + \delta(z_2) = ad_x(z_2) - (\lambda_4 + \lambda_6)z_2 - b(\lambda_6 + 3\lambda_3)t_3^3 t_4^{-1} \in R_5$. Since $ad_x(z_2) - (\lambda_4 + \lambda_6)z_2 \in R_5$, we have that $b(\lambda_6 + 3\lambda_3)t_3^3 t_4^{-1} \in R_5$. This implies that $b(\lambda_6 + 3\lambda_3)t_3^3 \in R_5 t_4$. Since $\alpha = 0$, from (4.1.5), we have that $t_3^3 = \frac{\beta}{b}t_6^{-1} + \frac{1}{ab}z_1t_3t_4$. Therefore, $b(\lambda_6 + 3\lambda_3)t_3^3 = (\lambda_6 + 3\lambda_3)\left(\beta t_6^{-1} + \frac{1}{a}z_1t_3t_4\right) = \beta(\lambda_6 + 3\lambda_3)t_6^{-1} + \frac{1}{a}(\lambda_6 + 3\lambda_3)z_1t_3t_4 \in R_5t_4$. Observe that $z_1t_3t_4 \in R_5t_4$. This implies that $\beta(\lambda_6 + 3\lambda_3)t_6^{-1} = R_5t_5$. Hence, $\beta(\lambda_6 + 3\lambda_3) \in R_5t_5t_6$. Since $\beta \neq 0$, we must have: $\lambda_6 + 3\lambda_3 = 0$, and so $\lambda_6 = \beta(\lambda_6 + 3\lambda_3) \in R_5t_5t_6$.

 $-3\lambda_3$. Otherwise, there will be a contradiction using the basis of R_5 (Proposition 4.1.4). As a result, $\delta(z_2) = -(\lambda_4 + \lambda_6)z_2 = (3\lambda_3 - \lambda_4)z_2$.

Again, from Remark 4.1.8, we have that $z_1 = f_1 - st_3^2 t_4^{-1}$. From the previous lemma, we also have $\delta(f_1) = -(\lambda_3 + \lambda_4 + \lambda_6)f_1$. Given that $\lambda_6 = -3\lambda_3$, one can easily verify that $\delta(z_1) = (2\lambda_3 - \lambda_4)z_1$.

3. Set $\lambda_1 := (2\lambda_3 - \lambda_4)$ and $\lambda_2 := 3\lambda_3 - \lambda_4$, it follows from points (1) and (2) that $D(e_{\kappa,5}) = \operatorname{ad}_x(e_{\kappa,5}) + \lambda_{\kappa}e_{\kappa,5}$ for all $\kappa \in \{1, \dots, 6\}$.

We now describe D as a derivation of $A_{0,\beta}$.

4.2.11 Lemma. 1. $x \in A_{0,\beta}$.

- 2. $\lambda_6 = 3\lambda_5, \ \delta(e_1) = -2\lambda_5 e_1, \ \delta(e_2) = -3\lambda_5 e_2, \ \delta(e_3) = -\lambda_5 e_3 \text{ and } \delta(e_4) = 0.$
- 3. Set $\lambda_1 := -2\lambda_5$, $\lambda_2 := -3\lambda_5$, $\lambda_3 := -\lambda_5$ and $\lambda_4 := 0$. Then, $D(e_{\kappa}) = \operatorname{ad}_x(e_{\kappa}) + \lambda_{\kappa}e_{\kappa}$ for all $\kappa \in \{1, \dots, 6\}$.

Proof. 1. Similar to the proof of Lemma 4.2.4(1).

2. From Remark 4.1.8, we have $e_4 = t_4 - bt_5^3 t_6^{-1}$. One can confirm that $\delta(e_4) = \lambda_4 e_4 + b(\lambda_4 - 3\lambda_5 - 3\lambda_3)t_5^3 t_6^{-1}$. Now, $D(e_4) = ad_x(e_4) + \delta(e_4) = ad_x(e_4) + \lambda_4 e_4 + b(\lambda_4 - 3\lambda_5 - 3\lambda_3)t_5^3 t_6^{-1} \in A_{0,\beta}$. Since $ad_x(e_4) + \lambda_4 e_4 \in A_{0,\beta}$, it follows that $b(\lambda_4 - 3\lambda_5 - 3\lambda_3)t_5^3 t_6^{-1} \in A_{0,\beta}$. This implies that $b(\lambda_4 - 3\lambda_5 - 3\lambda_3)t_5^3 \in A_{0,\beta}t_6$. Note, from Appendix C, $b \neq 0$. Therefore, $\lambda_4 - 3\lambda_5 - 3\lambda_3 = 0$, otherwise, there will be a contradiction using the basis of $A_{0,\beta}$ (Proposition 3.2.2). Hence, $\lambda_4 = 3(\lambda_3 + \lambda_5)$, and $\delta(e_4) = 3(\lambda_3 + \lambda_5)e_4$. The relation $\delta(e_3) = \lambda_3 e_3 + a(\lambda_3 - \lambda_4 + \lambda_5)e_4 t_5^{-1} + [s(\lambda_3 - 2\lambda_5 + \lambda_6) + q^{-3}ab(\lambda_3 - \lambda_4 + \lambda_5)]t_5^2 t_6^{-1}$ (see (4.2.15)) is also valid in $A_{0,\beta}$. From Lemma 4.2.10, we have that $\lambda_6 = -3\lambda_3$. Given that $\lambda_4 = 3(\lambda_3 + \lambda_5)$ and $\lambda_6 = -3\lambda_3$, one can confirm that $\delta(e_3) = \lambda_3 e_3 - 2a(\lambda_3 + \lambda_5)e_4 t_5^{-1}$ (note, from Appendix C, $q^{-3}ab + s = 0$). Now, $D(e_3) = ad_x(e_3) + \delta(e_3) = ad_x(e_3) + \lambda_5 e_4 t_5^{-1} \in A_{0,\beta}$. Since $ad_x(e_3) + \lambda_3 e_3 \in A_{0,\beta}$, it follows that $2a(\lambda_3 + \lambda_5)e_4 t_5^{-1} \in A_{0,\beta}$. This implies that $2a(\lambda_3 + \lambda_5)e_4 \in A_{0,\beta}t_5$. We must have $\lambda_3 + \lambda_5 = 0$, and so $\lambda_3 = -\lambda_5$ (note, $a \neq 0$). Otherwise, there will be a contradiction

using the basis of $A_{0,\beta}$ (Proposition 3.2.2). As a result, $\delta(e_3) = -\lambda_5 e_3$, $\delta(e_4) = 0$ and $\lambda_6 = 3\lambda_5$.

From (4.2.31), we have $e_1 = z_1 - rt_5t_6^{-1} - ht_3t_5^{-1} + (ah - g)t_4t_5^{-2}$. Recall from Lemma 4.2.10 that $\delta(z_1) = (2\lambda_3 - \lambda_4)z_1$. Given that $\lambda_6 = 3\lambda_5$, $\lambda_4 = 0$, $\lambda_3 = -\lambda_5$ and $\delta(t_5) = \lambda_5 t_5$, one can verify that $\delta(e_1) = -2\lambda_5 e_1$. From the commutation relations of $A_{0,\beta}$ in Chapter 3, we have that $e_3e_1 = q^{-1}e_1e_3 - (q+q^{-1}+q^{-3})e_2$. Since $\delta(e_1) = -2\lambda_5 e_1$ and $\delta(e_3) = -\lambda_5 e_3$, applying δ to this relation gives $\delta(e_2) = -3\lambda_5 e_2$.

In the next lemma, we prove that the first Hochschild cohomology groups of $A_{0,\beta}$ and $A_{\alpha,0}$ are non-trivial.

4.2.12 Lemma. Let $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}$ and, θ and $\tilde{\theta}$ be linear maps of $A_{\alpha,0}$ and $A_{0,\beta}$ respectively defined by

$$\theta(e_1) = -e_1, \ \ \theta(e_2) = -e_2, \ \ \theta(e_3) = 0, \ \ \theta(e_4) = e_4, \ \ \theta(e_5) = e_5, \ \ \theta(e_6) = 2e_6,$$

and

$$\tilde{\theta}(e_1) = -2e_1, \quad \tilde{\theta}(e_2) = -3e_2, \quad \tilde{\theta}(e_3) = -e_3, \quad \tilde{\theta}(e_4) = 0, \quad \tilde{\theta}(e_5) = e_5, \quad \tilde{\theta}(e_6) = 3e_6.$$

Then, θ and $\tilde{\theta}$ are \mathbb{C} -derivations of $A_{\alpha,0}$ and $A_{0,\beta}$ respectively.

Proof. Given the algebra relations of $A_{\alpha,\beta}$ in Chapter 3, we verify that θ satisfies the relations of $A_{\alpha,\beta}$ when $\alpha \neq 0$ and $\beta = 0$, and $\tilde{\theta}$ satisfies the relations of $A_{\alpha,\beta}$ when $\alpha = 0$ and $\beta \neq 0$. We will verify this for only one of the relations, and leave the remaining ones for the reader to check. From (3.0.1), we have that

$$e_1e_3e_5 + ae_1e_4 + ae_2e_5 + a'e_3^2 = \alpha$$

in $A_{\alpha,0}$.

Now,

$$\theta(e_1)e_3e_5 + e_1\theta(e_3)e_5 + e_1e_3\theta(e_5) + a\theta(e_1)e_4 + ae_1\theta(e_4) + a\theta(e_2)e_5 + ae_2\theta(e_5) + 2a'\theta(e_3)e_3$$

= 0

as expected. In $A_{0,\beta}$, we have that

$$e_1e_3e_5 + ae_1e_4 + ae_2e_5 + a'e_3^2 = 0.$$

Observe that

$$\begin{split} \tilde{\theta}(e_1)e_3e_5 + e_1\tilde{\theta}(e_3)e_5 + e_1e_3\tilde{\theta}(e_5) + a\tilde{\theta}(e_1)e_4 + ae_1\tilde{\theta}(e_4) + a\tilde{\theta}(e_2)e_5 + ae_2\tilde{\theta}(e_5) + 2a'\tilde{\theta}(e_3)e_3 \\ &= -2(e_1e_3e_5 + ae_1e_4 + ae_2e_5 + a'e_3^2) = 0, \end{split}$$

as expected.

4.2.13 Remark. From Lemma 4.1.7, $Z(A_{\alpha,\beta}) = \mathbb{C}$. Let $(\alpha,\beta) \in \mathbb{C}^2 \setminus \{(0,0)\}$ and Der $(A_{\alpha,\beta})$ represent the \mathbb{C} -derivations of $A_{\alpha,\beta}$. Then, the first Hochschild cohomology group of $A_{\alpha,\beta}$ (denoted by $HH^1(A_{\alpha,\beta})$) defined by

$$HH^{1}(A_{\alpha,\beta}) := \frac{\mathsf{Der}(A_{\alpha,\beta})}{\mathsf{Inn}\mathsf{Der}(A_{\alpha,\beta})}$$

is a free module over $Z(A_{\alpha,\beta}) = \mathbb{C}$, where $\text{InnDer}(A_{\alpha,\beta}) := \{ \text{ad}_x \mid x \in A_{\alpha,\beta} \}$ is the set of inner derivations of $A_{\alpha,\beta}$.

We summarize our main results in this chapter in the theorem below.

4.2.14 Theorem. Given that $A_{\alpha,\beta} = U_q^+(G_2)/\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$, with $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0,0)\}$, we have the following results:

- 1. if $\alpha, \beta \neq 0$; then every derivation D of $A_{\alpha,\beta}$ can uniquely be written as $D = \mathsf{ad}_x$, where $x \in A_{\alpha,\beta}$.
- 2. if $\alpha \neq 0$ and $\beta = 0$, then every derivation D of $A_{\alpha,0}$ can uniquely be written as $D = ad_x + \lambda \theta$, where $\lambda \in \mathbb{C}$ and $x \in A_{\alpha,0}$.
- 3. if $\alpha = 0$ and $\beta \neq 0$, then every derivation D of $A_{0,\beta}$ can uniquely be written as $D = ad_x + \lambda \tilde{\theta}$, where $\lambda \in \mathbb{C}$ and $x \in A_{0,\beta}$.
- 4. $HH^1(A_{\alpha,0}) = \mathbb{C}[\theta]$ and $HH^1(A_{0,\beta}) = \mathbb{C}[\tilde{\theta}]$, where $[\theta]$ and $[\tilde{\theta}]$ respectively denote the classes of θ and $\tilde{\theta}$ modulo the space of inner derivations.
- 5. if $\alpha, \beta \neq 0$; then $HH^1(A_{\alpha,\beta}) = \{[0]\}$, where [0] denotes the class of 0 modulo the space of inner derivations.

Proof. Points (1), (2) and (3) are as a result of Lemmas 4.2.4, 4.2.8 and 4.2.11 respectively. Point (4) is a consequence of Lemma 4.2.12, and (5) is a consequence of (1).

Chapter 5

Semiclassical limit of the simple quotients of $U_q^+(G_2)$

In this chapter, we study a semiclassical limit of $U_q^+(G_2)$, and its simple quotients. Unless stated explicitly, we do not transfer notations used in the previous chapters to this and the subsequent chapter. Given a non-commutative algebra A, one can obtain a Poisson algebra \mathcal{A} from A through a process called *semiclassical limit*. Conversely, given a Poisson algebra \mathcal{A} , one can obtain a non-commutative algebra A from \mathcal{A} through a process called *quantization*. We briefly explain these concepts in line with the presentations in [14] in the next section. Since $U_q^+(G_2)$ is a non-commutative algebra, we study its semiclassical limit in Section 5.2. Following strategies similar to that of $U_q^+(G_2)$ in Chapter 2, we study the Poisson maximal ideals (of height 2) of the semiclassical limit of $U_q^+(G_2)$ using \mathcal{H} -stratification by Goodearl [16], and Poisson deleting derivations algorithm by Launois and Lecoutre [34]. We finally conclude in Section 5.3 that the simple quotients of the semiclassical limit of $U_q^+(G_2)$ are the semiclassical limits of the non-commutative algebra $A_{\alpha,\beta}$. Most of the results in this and the subsequent chapters are analogues to their non-commutative counterparts.

We begin with some preliminaries on Poisson algebras.

5.1 Preliminaries

This section studies some preliminary materials such as semiclassical limit, Poisson prime spectrum and Poisson deleting derivations algorithm. We begin with the following definitions and examples.

5.1.1 Definitions and examples. A Poisson algebra \mathcal{A} is a commutative algebra over a field \mathbb{K} endowed with a skew-symmetric \mathbb{K} -bilinear map $\{-,-\} : \mathcal{A} \times \mathcal{A} \longrightarrow \mathcal{A}$ which satisfies the following properties:

- $\{x, yz\} = \{x, y\}z + y\{x, z\}$ for all $x, y, z \in \mathcal{A}$ (Leibniz rule).
- $\{x, \{y, z\}\} + \{y, \{z, x\}\} + \{z, \{x, y\}\} = 0$ for all $x, y, z \in \mathcal{A}$ (Jacobi identity).

Note, the skew-symmetric \mathbb{K} -bilinear map $\{-,-\}$ is called the *Poisson bracket*. Every Poisson algebra is also a Lie algebra since they both satisfy the Jacobi identity. Since the Poisson bracket also satisfies the Leibniz rule, it is well known that for each $x \in \mathcal{A}$ the \mathbb{K} -linear map ham_x := $\{x, -\} : \mathcal{A} \longrightarrow \mathcal{A}$ is a derivation known as the *Hamiltonian* derivation associated to x. A Poisson ideal I of \mathcal{A} is any ideal such that $\{x, u\} \in I$ for all $x \in \mathcal{A}$ and $u \in I$. The set $Z_P(\mathcal{A}) := \{a \in \mathcal{A} \mid \{a, x\} = 0, \forall x \in \mathcal{A}\}$ is a Poisson subalgebra of \mathcal{A} called the *Poisson center* of \mathcal{A} .

5.1.2 Remark. If \mathcal{A} is a Poisson algebra and $\{x_1, \dots, x_n\}$ is a generating set for \mathcal{A} (as an algebra), then it is enough to define a Poisson bracket $\{-, -\}$ on \mathcal{A} by defining it on only the generating set.

5.1.3 Example. The following are some examples of Poisson algebras.

- 1. The polynomial ring $\mathbb{C}[x, y]$ is a Poisson algebra via $\{x, y\} = a$, where $a \in \mathbb{C}$. When a = 0, then we have a trivial Poisson structure on $\mathbb{C}[x, y]$. In fact, any commutative algebra can be endowed with the trivial Poisson structure to form a Poisson algebra.
- 2. Let $\mathcal{R} = \mathbb{K}[y_1, \cdots, y_N]$ be a polynomial algebra over the field \mathbb{K} , with $\{y_i, y_j\} = \mu_{ij}y_iy_j, i, j \in \mathbb{N}_{\geq 1}$, where $M = (\mu_{ij})$ is an $N \times N$ skew-symmetric matrix over

the field \mathbb{K} . Then, \mathcal{R} is a Poisson algebra known as the *Poisson affine space*. Note, for all $f, g \in \mathcal{R}$, we have that:

$$\{f,g\} = \sum_{i,j=1}^{n} \mu_{ij} y_i y_j \frac{\partial f}{\partial y_i} \frac{\partial g}{\partial y_j}$$

The set $S := \{\lambda y_1^{i_1} \cdots y_N^{i_N} \mid i_1, \cdots, i_N \in \mathbb{N} \text{ and } \lambda \in \mathbb{C}^*\}$ is a multiplicative set of \mathcal{R} . Hence, one can localize \mathcal{R} at S as $\mathcal{R}S^{-1} = \mathbb{K}[y_1^{\pm 1}, \cdots, y_N^{\pm 1}]$. Note, the Poisson bracket of \mathcal{R} extends uniquely to a Poisson bracket of $\mathcal{R}S^{-1}$. Therefore, $\mathcal{R}S^{-1}$ is a Poisson algebra known as the *Poisson torus* associated to the Poisson affine space \mathcal{R} .

In general, let $S \ni 1$ be a multiplicative set of a Poisson algebra \mathcal{A} . Then, the localization $\mathcal{A}S^{-1}$ admits a Poisson bracket extended uniquely from \mathcal{A} as follows:

$$\{xs^{-1}, yt^{-1}\} = \{x, y\}s^{-1}t^{-1} - \{x, t\}ys^{-1}t^{-2} - \{s, y\}xs^{-2}t^{-1} + \{s, t\}xys^{-2}t^{-2},$$

for all $x, y \in \mathcal{A}$ and all $s, t \in S$.

In a special case where A is a domain, then the Poisson structure of A extends uniquely to its field of fractions.

Set R := O(M₂(C)) = C [^{a b}_{c d}]. Then, R is a Poisson algebra called 2 × 2 Poisson matrix algebra with Poisson bracket defined as follows:

$$\{a,b\} = ab, \ \{a,c\} = ac, \ \{b,c\} = 0, \ \{b,d\} = bd, \ \{c,d\} = cd, \ \{a,d\} = 2bc.$$

4. Take any arbitrary Poisson algebra A and a Poission ideal I of A. Then, it is well known that the quotient algebra A/I is a Poisson algebra with an induced Poisson bracket defined as {x̄, ȳ} = {x, y}, where x̄ := x + I and ȳ := y + I, with x, y ∈ A.

5.1.4 Semiclassical limit. As already stated, given a non-commutative algebra A, one can obtain a Poisson algebra A from A through a process called *semiclassical limit*.

Conversely, given a Poisson algebra \mathcal{A} , one can obtain a non-commutative algebra A from \mathcal{A} through a process called *quantization*. That is, one can either view A as a quantization of \mathcal{A} or \mathcal{A} as the semiclassical limit of A under certain conditions, which we present in the next paragraph. Unless stated otherwise, we present ideas similar to [14, §1.1.3].

Let R be a commutative principal ideal domain containing the field \mathbb{K} and hR be a maximal ideal of R for a fix $h \in R$. According to [7, Chapter III.5], in practice, R will always be either a polynomial or Laurent polynomial ring over \mathbb{K} in one variable, and h will be a linear polynomial. Take an algebra A (which is not necessarily commutative torsion-free R-algebra) such that the factor algebra $\mathcal{A} := A/hA$ is commutative. Let $u, v \in A$. Then, $\bar{u} := u + hA$ and $\bar{v} := v + hA$ are the canonical images of u and v in \mathcal{A} respectively. Since $\bar{u}\bar{v} = \bar{v}\bar{u}$, we have that $[u, v] := uv - vu \in hA$. There exists a unique element $\gamma(u, v)$ of A such that $[u, v] = h\gamma(u, v)$. It follows that

$$\{\bar{u},\bar{v}\} := \gamma(u,v) + hA = \frac{[u,v]}{h} + hA$$

defines a Poisson bracket on \mathcal{A} . One can prove that the above definition defines a Poisson bracket on \mathcal{A} by showing that it is well-defined, the Jacobi identity and the Leibniz rule hold. The details of the proof can be found in [14, §1.1.3]. Given the above presentation, A is said to be a *quantization* of \mathcal{A} , and \mathcal{A} is termed as the *semiclassical limit* of A. Let $\lambda \in \mathbb{K}$, then the algebra $\mathcal{A}_{\lambda} := A/(h - \lambda)A$ is a *deformation* of the Poisson algebra $\mathcal{A} = \mathcal{A}_0$ provided the central element $h - \lambda$ is not invertible in A.

5.1.5 Example. 1. From Remark 1.5.1, we can define a quantum affine space of the form $\mathcal{O}_{\Lambda}(\mathbb{K}^N) = \mathbb{K}_{\Lambda}[Y_1, \cdots, Y_N]$, with $Y_jY_i = q^{\mu_{ji}}Y_iY_j$, where $\Lambda = (q^{\mu_{ij}})$ is a skew-symmetric matrix defined in Remark 1.5.1. Set $R := \mathbb{K}[z^{\pm 1}]\langle Y_1, \cdots, Y_N | Y_jY_i = z^{\mu_{ji}}Y_iY_j$ for all $1 \le i, j \le N \rangle$. The element z-1 is central and not invertible in R. Hence, $\mathcal{R} := R/(z-1)R = \mathbb{K}[y_1, \cdots, y_N]$, with $y_i := Y_i + (z-1)R$, $1 \le i \le N$. Now, $y_iy_j = y_jy_i$ for all $1 \le i, j \le N$, hence, \mathcal{R} is a commutative algebra.

A Poisson structure is defined on \mathcal{R} as follows:

$$\{y_j, y_i\} = \frac{[Y_j, Y_i]}{z - 1} + (z - 1)R = \frac{(z^{\mu_{ji}} - 1)Y_iY_j}{z - 1} + (z - 1)R = \mu_{ji}y_iy_j,$$

for all $1 \leq i, j \leq n$. Therefore, \mathcal{R} is a Poisson affine space and it is the semiclassical limit of the algebra R. Moreover, $R/(z-q)R = \mathcal{O}_{\Lambda}(\mathbb{K}^N)$ is a deformation of the Poisson affine space \mathcal{R} . Since a Poisson bracket on a Poisson affine space extends uniquely to a Poission bracket of the associated Poisson torus, one can also conclude that the Poisson torus $\mathbb{K}[y_1^{\pm 1}, \cdots, y_N^{\pm 1}]$ is the semiclassical limit of the quantum torus $\mathbb{K}_{\Lambda}[Y_1^{\pm 1}, \cdots, Y_N^{\pm 1}]$.

2. Recall that the 2×2 quantum matrices $\mathcal{O}_q(M_2(\mathbb{C})) = \mathbb{C} \begin{bmatrix} a & b \\ c & d \end{bmatrix}$ satisfies the relation:

ab = qba, ac = qca, bc = cb, bd = qdb, cd = qdc, $ad - da = (q - q^{-1})bc$

(see [7, Example 1.1.7]). Now, the 2×2 Poisson matrix algebra in Example 5.1.3(3) is the semiclassical limit of the non-commutative algebra $\mathcal{O}_q(M_2(\mathbb{C}))$.

5.1.6 Poisson prime spectrum. In Section 1.6 we discussed the prime spectrum Spec(A) of an algebra A. In this section, we discuss the Poisson analogue called the Poisson prime spectrum. Let P be a proper Poisson ideal of a Poisson algebra A and I_1 , I_2 be Poisson ideals of A such that $P \supseteq I_1I_2$, the ideal P is called a *Poisson prime ideal* provided $P \supseteq I_1$ or $P \supseteq I_2$. It is well known that a Poisson ideal which is also a prime ideal is a Poisson prime ideal, however, the converse is not always true, except the case where A is noetherian (see [16, Lemma 1.1]). The collection of all Poisson prime ideals of A is called the *Poisson prime spectrum* of A, denoted by P.Spec(A). The largest Poisson prime ideal of A. The collection of all these primitive ideals is also called *Poisson primitive spectrum* of A, denoted by P.Prim(A).

Let A be a \mathbb{K} -algebra. Recall from Section 1.6 that the prime spectrum Spec(A) of A endowed with a suitable torus action can be partitioned into a disjoint union of

strata, a partition known as the \mathcal{H} -Stratification of Spec(A). In a similar manner, P.Spec(A) can also be partitioned into a disjoint union of strata in the presence of a suitable torus. We discuss this partition in the next paragraph. In [16, §3], Goodearl described this process for a commutative noetherian differential \mathbb{K} -algebras and, in [35, §2.2], Launois and Lecoutre described the process for noetherian Poisson algebras via the Cauchon diagrams. Since we have already described this process for any arbitrary \mathbb{K} -algebra in Section 1.6, we will only provide a summary for the Poisson version.

Let \mathcal{A} be a noetherian Poisson algebra and \mathcal{H} be an algebraic torus acting rationally on \mathcal{A} by Poisson automorphism (an automorphism that preserves the Poisson bracket). A Poisson prime ideal P is \mathcal{H} -invariant if $h \cdot P = P$ for all $h \in \mathcal{H}$. The set

$$\mathsf{P}.\mathsf{Spec}_J(\mathcal{A}) := \{ P \in \mathsf{P}.\mathsf{Spec}(\mathcal{A}) \mid (P : \mathcal{H}) = J \}$$

is called the J-stratum of P.Spec(\mathcal{A}). Note, $J := \bigcap_{h \in \mathcal{H}} h \cdot P$ is the largest Poisson \mathcal{H} -invariant prime ideal contained in P. The \mathcal{H} -strata P.Spec_J(\mathcal{A}) partitioned P.Spec(\mathcal{A}) into a disjoint union of strata. Hence,

$$\mathsf{P}.\mathsf{Spec}(\mathcal{A}) = \bigsqcup_{J \in \mathcal{H} - \mathsf{P}.\mathsf{Spec}(\mathcal{A})} \mathsf{P}.\mathsf{Spec}_J(A),$$

where $\mathcal{H} - P.Spec(\mathcal{A})$ is the collection of all the Poisson \mathcal{H} -invariant prime ideals of \mathcal{A} . This process is called \mathcal{H} -stratification of $P.Spec(\mathcal{A})$. In a similar manner, a \mathcal{H} -stratification of $P.Prim(\mathcal{A})$ is obtained as follows:

$$\mathsf{P}.\mathsf{Prim}(\mathcal{A}) = \bigsqcup_{J \in \mathcal{H} - \mathsf{P}.\mathsf{Prim}(\mathcal{A})} \mathsf{P}.\mathsf{Prim}_J(\mathcal{A}),$$

where $P.Prim_J(\mathcal{A}) = P.Spec_J(\mathcal{A}) \cap P.Prim(\mathcal{A})$ and $\mathcal{H} - P.Prim(\mathcal{A})$ is the collection of all the Poisson \mathcal{H} -invariant primitive ideals of \mathcal{A} .

5.1.7 Proposition. [16, Theorem 4.3] Let $P \in P.Spec_J(\mathcal{A})$, P is Poisson primitive if and only if P is maximal in $P.Spec_J(\mathcal{A})$.

5.1.8 Poisson deleting derivation algorithm (PDDA). We have already studied Cauchon's theory of deleting derivation algorithm for non-commutative algebras in Section **1.7.** In [35], Launois and Lecoutre studied the Poisson version of the deleting derivation algorithm called the *Poisson deleting derivation algorithm (PDDA)* for some class of Poisson algebras with a base field of characteristic zero and prime characteristic. In this section, we will only discuss the algorithm for the characteristic zero case. We begin with the following theorem.

5.1.9 Theorem. [39, Theorem 1.1] Let \mathcal{A} be a Poisson \mathbb{K} -algebra and $\sigma, \delta : \mathcal{A} \to \mathcal{A}$ be \mathbb{K} -linear maps. Then, the polynomial algebra $\mathcal{R} = \mathcal{A}[X]$ endowed with a Poisson bracket (i.e. $\{X, a\} = \sigma(a)X + \delta(a)$ for all $a \in \mathcal{A}$) extending from the Poisson bracket of \mathcal{A} is a Poisson algebra if and only if:

1. σ is a Poisson derivation of A. That is, σ is a \mathbb{K} -derivation of A with

$$\sigma(\{a,b\}) = \{\sigma(a),b\} + \{a,\sigma(b)\} \text{ for all } a,b \in \mathcal{A}.$$

2. δ is a Poisson σ -derivation of A. That is, δ is a \mathbb{K} -derivation of A with

$$\delta(\{a,b\}) = \{\delta(a),b\} + \{a,\delta(b)\} + \sigma(a)\delta(b) - \delta(a)\sigma(b) \text{ for all } a,b \in \mathcal{A}.$$

Given a Poisson algebra $\mathcal{R} = \mathcal{A}[X]$ satisfying the above conditions, one can simply write \mathcal{R} as $\mathcal{R} = \mathcal{A}[X, \sigma, \delta]_P$. This is called the *Poisson-Ore extension* of \mathcal{A} . If $\delta = 0$, then $\mathcal{A}[X, \sigma, 0]_P$ is simply written as $\mathcal{A}[X, \sigma]_P$. Furthermore, \mathcal{R} is called an *iterated Poisson-Ore extension over* \mathcal{A} if there exists a set $\{\sigma_i \mid i = 1, \dots, N\}$ of Poisson derivations and a set $\{\delta_i \mid i = 1, \dots, N\}$ of Poisson σ_i - derivations such that

$$\mathcal{R} = \mathcal{A}[X_1; \sigma_1, \delta_1]_p \cdots [X_n; \sigma_n, \delta_n]_P.$$

5.1.10 Example. In Example 5.1.3(3), we have that the matrix Poisson algebra

 $\mathcal{R} = \mathbb{C} \left[\begin{smallmatrix} a & b \\ c & d \end{smallmatrix} \right]$ satisfies the relations:

$$\{a,b\} = ab, \ \{a,c\} = ac, \ \{b,c\} = 0, \ \{b,d\} = bd, \ \{c,d\} = cd, \ \{a,d\} = 2bc.$$

Now, set $X_1 := a$, $X_2 := b$, $X_3 := c$ and $X_4 := d$. Then, \mathcal{R} can be written as an iterated Poisson-Ore extension as:

$$\mathcal{R} = \mathbb{C}[X_1][X_2;\sigma_2]_P[X_3;\sigma_3]_P[X_4;\sigma_4,\delta_4]_P,$$

where $\sigma_2(X_1) = X_1$, $\sigma_3(X_1) = X_1$, $\sigma_3(X_2) = 0$, $\sigma_4(X_1) = 0$, $\sigma_4(X_2) = X_2$, $\sigma_4(X_3) = X_3$, and $\delta_4(X_1) = 2X_2X_3$, $\delta_4(X_2) = \delta_4(X_3) = 0$. Note, σ_j is a Poisson derivation and δ_j is a Poisson σ_j -derivation of $\mathbb{C}[X_1][X_2;\sigma_2]_P \cdots [X_{j-1};\sigma_{j-1},\delta_{j-1}]_P$ for each $2 \le j \le 4$ (note, $\delta_2 = \delta_3 = \delta_4 = 0$).

We are now going to discuss the PDDA process.

Let \mathbb{K} be a field with characteristic zero, and $\mathcal{A} = \mathbb{K}[X_1][X_2; \sigma_2, \delta_2]_p \cdots [X_N; \sigma_N, \delta_N]_P$ be an iterated Poisson polynomial algebra over \mathbb{K} . Suppose that \mathcal{A} satisfies the conditions in the hypothesis below.

- **5.1.11 Hypothesis.** (H1) For all $1 \le j < i \le N$, there exists $\mu_{ij} \in \mathbb{K}$, with $\mu_{ji} := -\mu_{ij}$, such that $\sigma_i(X_j) = \mu_{ij}X_j$.
- (H2) The derivation δ_i is locally nilpotent and $\delta_i \alpha \alpha \delta_i = \eta_i \delta_i$ for some non-zero scalar η_i for all $2 \le i \le N$.

Then, the PDDA can be used to study the Poisson prime spectrum of \mathcal{A} . Let $j \in \{N+1, \dots, 2\}$. The algorithm (PDDA) constructs a family $(X_{1,j}, \dots, X_{N,j})$ of elements of Fract(\mathcal{A}) as follows. For j = N + 1, set $(X_{1,N+1}, \dots, X_{N,N+1}) := (X_1, \dots, X_N)$. Now, suppose that the family $(X_{1,j+1}, \dots, X_{N,j+1})$ has already been constructed. Then, for j < N + 1, construct $X_{1,j}, \dots, X_{N,j}$ from $X_{1,j+1}, \dots, X_{N,j+1}$ using the relation

$$X_{i,j} := \begin{cases} X_{i,j+1} & \text{if } i \ge j \\ \sum_{k=0}^{+\infty} \frac{1}{\eta_j^k k!} \delta_j^k(X_{i,j+1}) X_{j,j+1}^{-k} & \text{if } i < j, \end{cases}$$

for all $i \in \{1, \dots, N\}$. Since δ_i is locally nilpotent, the summation is finite.

For each $j \in \{2, \dots, N+1\}$, the algebra $\mathcal{A}^{(j)}$ represents the subalgebra of $\text{Fract}(\mathcal{A})$ generated by all the $X_{i,j}$. That is, $\mathcal{A}^{(j)} = \mathbb{K}[X_{1,j}, \dots, X_{N,j}]$. Note, $\mathcal{A}^{(N+1)} = \mathcal{A}$. It follows from [35, Proposition 1.11] that

$$\mathcal{A}^{(j)} \cong \mathbb{K}[X_1][X_2; \sigma_2, \delta_2]_P \cdots [X_{j-1}; \sigma_{j-1}, \delta_{j-1}]_P [X_j; \tau_j]_P \cdots [X_N; \tau_N]_P,$$

by an isomorphism that maps $X_{i,j}$ to X_i , and τ_j, \dots, τ_N denote the Poisson derivations defined by $\tau_l(X_i) = \mu_{li}X_i$ for all $1 \le i < l \le N$. With a slight abuse of notation, one can identify τ_j, \dots, τ_N with $\sigma_j, \dots, \sigma_N$ respectively.

Moreover, the set $S_j := \{X_{j,j+1}^n \mid n \in \mathbb{N}\} = \{X_{j,j}^n \mid n \in \mathbb{N}\}$ is a multiplicative system of regular elements of $\mathcal{A}^{(j)}$ and $\mathcal{A}^{(j+1)}$, and $\mathcal{A}^{(j)}S_j^{-1} = \mathcal{A}^{(j+1)}S_j^{-1}$ [35, Proposition 1.11]. Launois and Lecoutre [35] used the PDDA to relate P.Spec(\mathcal{A}) to P.Spec($\overline{\mathcal{A}}$), where $\overline{\mathcal{A}} := \mathcal{A}^{(2)}$, by constructing an embedding $\psi_j : \text{P.Spec}(\mathcal{A}^{(j+1)}) \hookrightarrow \text{P.Spec}(\mathcal{A}^{(j)})$ for each $j \in \{2, \dots, N\}$. Suppose that $P \in \text{P.Spec}(\mathcal{A}^{(j+1)})$ and $X_{j,j+1} \notin P$, then ψ_j is defined by

$$\psi_j(P) = PS_j^{-1} \cap \mathcal{A}^{(j)} = Q_j$$

with $Q \in \mathsf{P}.\mathsf{Spec}(\mathcal{A}^{(j)})$. In the case where $X_{j,j} \notin Q$, then the inverse map ψ_j^{-1} is also given by

$$\psi_j^{-1}(Q) = QS_j^{-1} \cap \mathcal{A}^{(j+1)} = P.$$

The map ψ_j is injective but not necessarily bijective, however, ψ_j induces a bijection from $\{P \in \mathsf{P.Spec}(\mathcal{A}^{(j+1)}) \mid P \cap S_j = \emptyset\}$ onto $\{Q \in \mathsf{P.Spec}(\mathcal{A}^{(j)}) \mid Q \cap S_j = \emptyset\}$ [35, Section 2.1]. The so-called *canonical embedding* ψ : $\mathsf{P.Spec}(\mathcal{A}) \hookrightarrow \mathsf{P.Spec}(\bar{\mathcal{A}})$ is obtained by composing all the ψ_j . That is, $\psi := \psi_2 \circ \cdots \circ \psi_N$. This canonical embedding ψ helps to

construct a partition of P.Spec(\mathcal{A}) into a disjoint union of strata known as the *canonical* partition via the Cauchon diagrams. See [35] for further details on this, and how the map ψ_j is defined when $X_{j,j+1} \in P$ (we omit this case).

5.2 Semiclassical limit of $U_a^+(G_2)$

We are now ready to study the semiclassical limit of the algebra $A := U_q^+(G_2)$ studied in Chapter 2. Recall that A satisfies the following relations:

$$\begin{split} E_2 E_1 &= q^{-3} E_1 E_2 \\ E_3 E_2 &= q^{-3} E_2 E_3 \\ E_4 E_2 &= q^{-3} E_2 E_4 - \frac{q^4 - 2q^2 + 1}{q^4 + q^2 + 1} E_3^3 \\ E_5 E_1 &= q E_1 E_5 - (1 + q^2) E_3 \\ E_5 E_3 &= q^{-1} E_3 E_5 - (q + q^{-1} + q^{-3}) E_4 \\ E_5 E_3 &= q^{-1} E_3 E_5 - (q + q^{-1} + q^{-3}) E_4 \\ E_6 E_1 &= q^3 E_1 E_6 - q^3 E_5 \\ E_6 E_3 &= E_3 E_6 + (1 - q^2) E_5^2 \\ E_6 E_4 &= q^{-3} E_4 E_6 - \frac{q^4 - 2q^2 + 1}{q^4 + q^2 + 1} E_5^3 \\ E_6 E_5 &= q^{-3} E_5 E_6 . \end{split}$$

Set $U_i := (q-1)E_i$ for i = 1, 3, 4, 5, and $U_i := f(q)(q-1)E_i$ for i = 2, 6; where $f(q) = q^4 + q^2 + 1$. Then, A is now generated by U_1, \dots, U_6 subject to the relations:

$$U_{2}U_{1} = q^{-3}U_{1}U_{2}$$

$$U_{3}U_{2} = q^{-3}U_{2}U_{3}$$

$$U_{3}U_{1} = q^{-1}U_{1}U_{3} - q^{-3}(q-1)U_{2}$$

$$U_{4}U_{1} = U_{1}U_{4} + (1-q^{2})U_{3}^{2}$$

$$U_{4}U_{2} = q^{-3}U_{2}U_{4} - (q+1)^{2}(q-1)U_{3}^{3}$$

$$U_{4}U_{3} = q^{-3}U_{3}U_{4}$$

$$U_{5}U_{1} = qU_{1}U_{5} - (1+q^{2})(q-1)U_{3}$$

$$U_{5}U_{2} = U_{2}U_{5} + f(q)(1-q^{2})U_{3}^{2}$$

$$U_{5}U_{3} = q^{-1}U_{3}U_{5} - f(q)(q^{-2} - q^{-3})U_{4}$$

$$U_{5}U_{4} = q^{-3}U_{4}U_{5}$$

$$U_{6}U_{1} = q^{3}U_{1}U_{6} - f(q)(q^{4} - q^{3})U_{5} \qquad U_{6}U_{2} = q^{3}U_{2}U_{6} + f(q)^{2}(q^{2} - q^{4})U_{3}U_{5} + U_{6}U_{3} = U_{3}U_{6} + f(q)(1 - q^{2})U_{5}^{2} \qquad f(q)^{2}(q^{4} + q^{2} - 1)(q - 1)U_{4} + U_{6}U_{4} = q^{-3}U_{4}U_{6} - (q + 1)^{2}(q - 1)U_{5}^{3} \qquad U_{6}U_{5} = q^{-3}U_{5}U_{6}.$$

We are now going to find a "new" presentation for A that allows us to introduce a quantisation of A. We discuss this as follows. Let \widehat{A} be a $\mathbb{C}[z^{\pm 1}]$ -algebra generated by $\widehat{U}_1, \dots, \widehat{U}_6$ subject to the relations:

$$\begin{split} \widehat{U}_{2}\widehat{U}_{1} &= z^{-3}\widehat{U}_{1}\widehat{U}_{2} & \widehat{U}_{3}\widehat{U}_{2} &= z^{-3}\widehat{U}_{2}\widehat{U}_{3} \\ \widehat{U}_{3}\widehat{U}_{1} &= z^{-1}\widehat{U}_{1}\widehat{U}_{3} - z^{-3}(z-1)\widehat{U}_{2} & \widehat{U}_{4}\widehat{U}_{1} &= \widehat{U}_{1}\widehat{U}_{4} + (1-z^{2})\widehat{U}_{3}^{2} \\ \widehat{U}_{4}\widehat{U}_{2} &= z^{-3}\widehat{U}_{2}\widehat{U}_{4} - (z+1)^{2}(z-1)\widehat{U}_{3}^{3} & \widehat{U}_{4}\widehat{U}_{3} &= z^{-3}\widehat{U}_{3}\widehat{U}_{4} \\ \widehat{U}_{5}\widehat{U}_{1} &= z\widehat{U}_{1}\widehat{U}_{5} - (1+z^{2})(z-1)\widehat{U}_{3} & \widehat{U}_{5}\widehat{U}_{2} &= \widehat{U}_{2}\widehat{U}_{5} + f(z)(1-z^{2})\widehat{U}_{3}^{2} \\ \widehat{U}_{5}\widehat{U}_{3} &= z^{-1}\widehat{U}_{3}\widehat{U}_{5} - f(z)(z^{-2} - z^{-3})\widehat{U}_{4} & \widehat{U}_{5}\widehat{U}_{4} &= z^{-3}\widehat{U}_{4}\widehat{U}_{5} \\ \widehat{U}_{6}\widehat{U}_{1} &= z^{3}\widehat{U}_{1}\widehat{U}_{6} - f(z)(z^{4} - z^{3})\widehat{U}_{5} & \widehat{U}_{6}\widehat{U}_{2} &= z^{3}\widehat{U}_{2}\widehat{U}_{6} + f(z)^{2}(z^{2} - z^{4})\widehat{U}_{3}\widehat{U}_{5} + \\ \widehat{U}_{6}\widehat{U}_{3} &= \widehat{U}_{3}\widehat{U}_{6} + f(z)(1-z^{2})\widehat{U}_{5}^{2} & f(z)^{2}(z^{4} + z^{2} - 1)(z-1)\widehat{U}_{4} \\ \widehat{U}_{6}\widehat{U}_{4} &= z^{-3}\widehat{U}_{4}\widehat{U}_{6} - (z+1)^{2}(z-1)\widehat{U}_{5}^{3} & \widehat{U}_{6}\widehat{U}_{5} &= z^{-3}\widehat{U}_{5}\widehat{U}_{6}, \end{split}$$

where $f(z) = z^4 + z^2 + 1$. For $\lambda \in \mathbb{C}^*$, observe that the element $z - \lambda$ is central and not invertible in \widehat{A} . Hence, we can set $\mathcal{A}_{\lambda} := \widehat{A}/(z - \lambda)\widehat{A}$. Now, \mathcal{A}_q is the non-commutative algebra A, and $\mathcal{A}_1 = \mathbb{C}[X_1, \cdots, X_6]$ with $X_i := \widehat{U}_i + (z - 1)\widehat{A}$ is a Poisson algebra with the Poisson bracket defined as follows (see Appendix B.1 for the necessary details):

$$\{X_2, X_1\} = -3X_1X_2 \qquad \{X_3, X_1\} = -X_1X_3 - X_2$$

$$\{X_3, X_2\} = -3X_2X_3 \qquad \{X_4, X_1\} = -2X_3^2$$

$$\{X_4, X_2\} = -3X_2X_4 - 4X_3^3 \qquad \{X_4, X_3\} = -3X_3X_4$$

$$\{X_5, X_1\} = X_1X_5 - 2X_3 \qquad \{X_5, X_2\} = -6X_3^2$$

$$\{X_5, X_3\} = -X_3X_5 - 3X_4 \qquad \{X_5, X_4\} = -3X_4X_5$$

$$\{X_6, X_1\} = 3X_1X_6 - 3X_5 \qquad \{X_6, X_2\} = 3X_2X_6 + 9X_4 - 18X_3X_5$$
$$\{X_6, X_3\} = -6X_5^2 \qquad \{X_6, X_4\} = -3X_4X_6 - 4X_5^3$$
$$\{X_6, X_5\} = -3X_5X_6.$$

Therefore, A_1 is the semiclassical limit of the non-commutative algebra \widehat{A} , and A_q is a deformation of the Poisson algebra A_1 .



Henceforth, for simplicity, we set

$$\mathcal{A} := \mathcal{A}_1 = \mathbb{C}[X_1, \cdots, X_6].$$

One can write the Poisson algebra ${\mathcal A}$ as an iterated Poisson-Ore extension as:

$$\mathcal{A} = \mathbb{C}[X_1][X_2; \sigma_2]_P[X_3; \sigma_3, \delta_3]_P[X_4; \sigma_4, \delta_4]_P[X_5; \sigma_5, \delta_5]_P[X_6; \sigma_6, \delta_6]_P$$

where,

$$\begin{aligned} \sigma_2(X_1) &= -3X_1 & \sigma_3(X_1) = -X_1 & \sigma_3(X_2) = -3X_2 & \sigma_4(X_1) = 0 \\ \sigma_4(X_2) &= -3X_2 & \sigma_4(X_3) = -3X_3 & \sigma_5(X_1) = X_1 & \sigma_5(X_2) = 0 \\ \sigma_5(X_3) &= -X_3 & \sigma_5(X_4) = -3X_4 & \sigma_6(X_1) = 3X_1 & \sigma_6(X_2) = 3X_2 \\ \sigma_6(X_3) &= 0 & \sigma_6(X_4) = -3X_4 & \sigma_6(X_5) = -3X_5, \end{aligned}$$

 and

$$\begin{split} \delta_3(X_1) &= -X_2 & \delta_3(X_2) = 0 & \delta_4(X_1) = -2X_3^2 & \delta_4(X_2) = -4X_3^3 \\ \delta_4(X_3) &= 0 & \delta_5(X_1) = -2X_3 & \delta_5(X_2) = -6X_3^2 & \delta_5(X_3) = -3X_4 \\ \delta_5(X_4) &= 0 & \delta_6(X_1) = -3X_5 & \delta_6(X_2) = 9X_4 - 18X_3X_5 & \delta_6(X_3) = -6X_5^2 \\ \delta_6(X_4) &= -4X_5^3 & \delta_6(X_5) = 0. \end{split}$$

Note, σ_j is a Poisson derivation and δ_j is a Poisson σ_j -derivation of $\mathbb{C}[X_1][X_2; \sigma_2]_P \cdots [X_{j-1}; \sigma_{j-1}, \delta_{j-1}]_P$ for each $2 \le j \le 6$ (note, $\delta_2 = 0$). **5.2.1 PDDA of** \mathcal{A} . The algebra

$$\mathcal{A} = \mathbb{C}[X_1][X_2; \sigma_2]_P[X_3; \sigma_3, \delta_3]_P[X_4; \sigma_4, \delta_4]_P[X_5; \sigma_5, \delta_5]_P[X_6; \sigma_6, \delta_6]_P$$

satisfies H1 and H2 in Hypothesis 5.1.11, and so the theory of PDDA applies to \mathcal{A} . We construct the following elements of $Fract(\mathcal{A})$ (computations have been omitted here, but can be found in Appendix B.2):

$$\begin{split} X_{1,6} &= X_1 - \frac{1}{2} X_5 X_6^{-1} \\ X_{2,6} &= X_2 + \frac{3}{2} X_4 X_6^{-1} - 3 X_3 X_5 X_6^{-1} + X_5^3 X_6^{-2} \\ X_{3,6} &= X_3 - X_5^2 X_6^{-1} \\ X_{4,6} &= X_4 - \frac{2}{3} X_5^3 X_6^{-1} \\ X_{1,5} &= X_{1,6} - X_{3,6} X_{5,6}^{-1} + \frac{3}{4} X_{4,6} X_{5,6}^{-2} \\ X_{2,5} &= X_{2,6} - 3 X_{3,6}^2 X_{5,6}^{-1} + \frac{9}{2} X_{3,6} X_{4,6} X_{5,6}^{-2} - \frac{9}{4} X_{4,6}^2 X_{5,6}^{-3} \\ X_{3,5} &= X_{3,6} - \frac{3}{2} X_{4,6} X_{5,6}^{-1} \\ X_{1,4} &= X_{1,5} - \frac{1}{3} X_{3,5}^2 X_{4,5}^{-1} \\ X_{2,4} &= X_{2,5} - \frac{2}{3} X_{3,5}^3 X_{4,5}^{-1} \\ X_{1,3} &= X_{1,4} - \frac{1}{2} X_{2,4} X_{3,4}^{-1} \end{split}$$

$$T_{1} := X_{1,2} = X_{1,3}$$

$$T_{2} := X_{2,2} = X_{2,3} = X_{2,4}$$

$$T_{3} := X_{3,2} = X_{3,3} = X_{3,4} = X_{3,5}$$

$$T_{4} := X_{4,2} = X_{4,3} = X_{4,4} = X_{4,5} = X_{4,6}$$

$$T_{5} := X_{5,2} = X_{5,3} = X_{5,4} = X_{5,5} = X_{5,6} = X_{5}$$

$$T_{6} := X_{6,2} = X_{6,3} = X_{6,4} = X_{6,5} = X_{6,6} = X_{6}$$

From the theory of the PDDA, we have that for each $j \in \{2, \dots, 7\}$, the algebra $\mathcal{A}^{(j)} := \mathbb{C}\langle X_{i,j} \mid i = 1, \dots, 6\rangle$ is a subalgebra of $\operatorname{Fract}(\mathcal{A})$. Since $(X_{1,7}, \dots, X_{6,7}) := (X_1, \dots, X_6)$, it follows that $\mathcal{A}^{(7)} = \mathcal{A}$.

We recall the skew-symmetric matrix in Subsection 2.2.3 below:

$$M = \begin{bmatrix} 0 & 3 & 1 & 0 & -1 & -3 \\ -3 & 0 & 3 & 3 & 0 & -3 \\ -1 & -3 & 0 & 3 & 1 & 0 \\ 0 & -3 & -3 & 0 & 3 & 3 \\ 1 & 0 & -1 & -3 & 0 & 3 \\ 3 & 3 & 0 & -3 & -3 & 0 \end{bmatrix}$$

Observe that $\bar{\mathcal{A}} := \mathcal{A}^{(2)} = \mathbb{C}[T_1, \cdots, T_6]$ satisfies the relation $T_i T_j = \mu_{ij} T_j T_i$ for all $1 \leq i, j \leq 6$, with $\mu_{ij} \in M$, hence \mathcal{A} is a Poisson affine space. Set $\Omega_1 := T_1 T_3 T_5$ and $\Omega_2 := T_2 T_4 T_6$. One can verify that Ω_1 and Ω_2 are Poisson central elements of $\bar{\mathcal{A}}$. That is, $\{\Omega_i, T_j\} = 0$ for all i = 1, 2, and $1 \leq j \leq 6$. We now want to successively pull Ω_1 and Ω_2 from the Poisson affine space $\bar{\mathcal{A}}$ into the algebra \mathcal{A} using the PDDA of \mathcal{A} discussed above. Through a direct computation, one can confirm that:

$$\Omega_{1} := T_{1}T_{3}T_{5}
= X_{1,3}X_{3,3}X_{5,3}
= X_{1,4}X_{3,4}X_{5,4} - \frac{1}{2}X_{2,4}X_{5,4}
= X_{1,5}X_{3,5}X_{5,5} - \frac{1}{2}X_{2,5}X_{5,5}
= X_{1,6}X_{3,6}X_{5,6} - \frac{3}{2}X_{1,6}X_{4,6} - \frac{1}{2}X_{2,6}X_{5,6} + \frac{1}{2}X_{3,6}^{2}
= X_{1}X_{3}X_{5} - \frac{3}{2}X_{1}X_{4} - \frac{1}{2}X_{2}X_{5} + \frac{1}{2}X_{3}^{2},$$

and

$$\begin{split} \Omega_2 &:= T_2 T_4 T_6 \\ &= X_{2,4} X_{4,4} X_{6,4} \\ &= X_{2,5} X_{4,5} X_{6,5} - \frac{2}{3} X_{3,5}^3 X_{6,5} \\ &= X_{2,6} X_{4,6} X_{6,6} - \frac{2}{3} X_{3,6}^3 X_{6,6} \\ &= X_2 X_4 X_6 - \frac{2}{3} X_3^3 X_6 - \frac{2}{3} X_2 X_5^3 + 2 X_3^2 X_5^2 - 3 X_3 X_4 X_5 + \frac{3}{2} X_4^2. \end{split}$$

As a result, Ω_1 and Ω_2 are also Poisson central elements of $\mathcal{A}^{(j)}$ for each $2 \leq j \leq 7$ since $\operatorname{Fract}(\mathcal{A}^{(j)}) = \operatorname{Fract}(\overline{\mathcal{A}})$. We now want to show that the Poisson center of \mathcal{A} is a polynomial ring generated by Ω_1 and Ω_2 over \mathbb{C} . The following discussions (similar to that of the non-commutative algebra $U_q^+(G_2)$ in Subsection 2.2.3) will lead us to the proof.

Set $S_j := \{\lambda T_j^{i_j} T_{j+1}^{i_{j+1}} \cdots T_6^{i_6} \mid i_j, \cdots, i_6 \in \mathbb{N}\}$ for each $2 \leq j \leq 6$. One can observe that S_j is a multiplicative system of non-zero divisors of $\mathcal{A}^{(j)} = \mathbb{C}\langle X_{i,j} \mid$ for all $i = 1, \cdots, 6\rangle$. One can therefore localize $\mathcal{A}^{(j)}$ at S_j as follows:

$$\mathfrak{R}_j := \mathcal{A}^{(j)} S_j^{-1}.$$

Note, the set $\Sigma_j := \{T_j^n \mid n \in \mathbb{N}\}$ is a multiplicative set in both $\mathcal{A}^{(j)}$ and $\mathcal{A}^{(j+1)}$ for each

 $2 \leq j \leq 6$. It follows from the discussions in Subsection 5.1.8 that

$$\mathcal{A}^{(j)}\Sigma_j^{-1} = \mathcal{A}^{(j+1)}\Sigma_j^{-1}.$$

Similar to (2.2.1), one can verify that

$$\mathfrak{R}_j = \mathfrak{R}_{j+1} \Sigma_j^{-1}$$
, for all $2 \le j \le 6$.

Again, the localization

$$\mathfrak{R}_1 := \mathfrak{R}_2[T_1^{-1}]$$

also holds in \mathfrak{R}_2 since T_1 generates a multiplicative system in \mathfrak{R}_2 . In fact, \mathfrak{R}_1 is the Poisson torus associated to the Poisson affine space $\overline{\mathcal{A}}$. As a result, $\mathfrak{R}_1 = \mathbb{C}[T_1^{\pm 1}, \cdots, T_6^{\pm 1}]$, where $T_iT_j = \mu_{ij}T_jT_i$ for all $1 \leq i, j \leq 6$ and $\mu_{ij} \in M$. Similar to (2.2.2), we construct the following embeddings:

$$\mathcal{A} := \mathfrak{R}_7 \subset \mathfrak{R}_6 = \mathfrak{R}_7 \Sigma_6^{-1} \subset \mathfrak{R}_5 = \mathfrak{R}_6 \Sigma_5^{-1} \subset \mathfrak{R}_4 = \mathfrak{R}_5 \Sigma_4^{-1}$$
$$\subset \mathfrak{R}_3 = \mathfrak{R}_4 \Sigma_3^{-1} \subset \mathfrak{R}_2 = \mathfrak{R}_3 \Sigma_2^{-1} \subset \mathfrak{R}_1.$$
(5.2.1)

Note, the family $(X_{1,j}^{k_1}\cdots X_{6,j}^{k_6})$, where $k_i \in \mathbb{N}$ if i < j and $k_i \in \mathbb{Z}$ otherwise is a PBWbasis of \mathfrak{R}_j for all $1 \leq i, j \leq 7$. Therefore, the family $(T_1^{k_1}T_2^{k_2}T_3^{k_3}T_4^{k_4}T_5^{k_5}T_6^{k_6})_{k_1,\cdots,k_6\in\mathbb{Z}}$ is a basis of \mathfrak{R}_1 .

5.2.2 Lemma. $Z_P(\mathfrak{R}_1) = \mathbb{C}[\Omega_1^{\pm 1}, \Omega_2^{\pm 1}].$

Proof. Obviously, $\mathbb{C}[\Omega_1^{\pm 1}, \Omega_2^{\pm 1}] \subseteq Z_P(\mathfrak{R}_1)$. For the reverse inclusion, let $y \in Z_P(\mathfrak{R}_1)$. Then, y can be written in terms of the basis of \mathfrak{R}_1 as:

$$y = \sum_{(i,\cdots,n)\in\mathbb{Z}^6} a_{(i,\cdots,n)} T_1^i T_2^j T_3^k T_4^l T_5^m T_6^n.$$

One can verify that $\{y, T_1\} = (-3j - k + m + 3n)yT_1$. Since $y \in Z_P(\mathfrak{R}_1)$, it follows that

$$-3j - k + m + 3n = 0.$$

Similarly, $\{y, T_2\} = (3i - 3k - 3l + 3n)yT_2$. Since $y \in Z_P(\mathfrak{R}_1)$, we have:

$$3i - 3k - 3l + 3n = 0.$$

Following the same pattern for T_3, T_4, T_5 and T_6 , one can confirm that

$$i + 3j - 3l - m = 0,$$

 $3j + 3k - 3m - 3n = 0,$
 $-i + k + 3l - 3n = 0,$
 $-3i - 3j + 3l + 3m = 0.$

We already know the solution to this system of equations as i = k = m and j = l = n(see the proof of Lemma 2.2.4). One can therefore write

$$y = \sum_{(i,j)\in\mathbb{Z}^2} a_{(i,j)} T_1^i T_2^j T_3^i T_4^j T_5^i T_6^j = \sum_{(i,j)\in\mathbb{Z}^2} q^{\bullet} a_{(i,j)} T_1^i T_3^i T_5^j T_2^j T_4^j T_6^j = \sum_{(i,j)\in\mathbb{Z}^2} q^{\bullet} a_{(i,j)} \Omega_1^i \Omega_2^j.$$

This implies that $y \in \mathbb{C}[\Omega_1^{\pm 1}, \Omega_2^{\pm 1}]$ as expected.

5.2.3 Corollary. 1. $Z_P(\mathfrak{R}_3) = \mathbb{C}[\Omega_1, \Omega_2].$

2. $Z_P(\bar{\mathcal{A}}) = \mathbb{C}[\Omega_1, \Omega_2].$

Proof. 1. Clearly, $\mathbb{C}[\Omega_1, \Omega_2] \subseteq Z_P(\mathfrak{R}_3)$. For the reverse inclusion, let $y \in Z_P(\mathfrak{R}_3)$. Then, y can be written in terms of the basis of \mathfrak{R}_3 (recall, $T_i = X_{i,3}$) as:

$$y = \sum_{(i,\dots,n) \in \mathbb{N}^2 \times \mathbb{Z}^4} a_{(i,\dots,n)} T_1^i T_2^j T_3^k T_4^l T_5^m T_6^n.$$

Note, the generators T_1, \cdots, T_6 of \mathfrak{R}_3 satisfy the Poisson bracket of \mathfrak{R}_1 . Hence, following

procedures similar to the lemma above, one will arrive at the conclusion that i = k = mand j = l = n. Since $i, j \ge 0$; it follows that $y = \sum_{(i,j)\in\mathbb{N}^2} a_{(i,j)}T_1^i T_3^i T_5^i T_2^j T_4^j T_6^j = \sum_{(i,j)\in\mathbb{N}^2} a_{(i,j)}\Omega_1^i \Omega_2^j$. This implies that $y \in \mathbb{C}[\Omega_1, \Omega_2]$ as expected.

2. Similar to (1).

5.2.4 Lemma. $Z_P(\mathcal{A}) = \mathbb{C}[\Omega_1, \Omega_2].$

Proof. Since \mathfrak{R}_i is a localization of \mathfrak{R}_{i+1} , it follows that $Z_P(\mathfrak{R}_{i+1}) \subseteq Z_P(\mathfrak{R}_i)$. From (5.2.1), we have that $Z_P(\mathcal{A}) \subseteq Z_P(\mathfrak{R}_3)$. Observe that $\mathbb{C}[\Omega_1, \Omega_2] \subseteq Z_P(\mathcal{A}) \subseteq Z_P(\mathfrak{R}_3) = \mathbb{C}[\Omega_1, \Omega_2]$.

5.2.5 Poisson \mathcal{H} -invariant prime ideals of \mathcal{A} of at most height 2. We begin this section by showing that $\langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$ are Poisson prime ideals. Note, the data of the PDDA of \mathcal{A} and expressions for Ω_1 and Ω_2 in the previous subsection will be very helpful in the proofs of Lemmas 5.2.6 and 5.2.7. In fact, Lemmas 5.2.6 and 5.2.7 are similar to Lemmas 2.3.1 and 2.3.2 respectively, as a result, the strategies of their proofs are also similar. Recall that $\langle \Theta \rangle_R$ denotes an ideal generated by the element Θ in any arbitrary ring R. Where no doubt arises, we will simply write $\langle \Theta \rangle$.

Recall that ψ_j : P.Spec $(\mathcal{A}^{(j+1)}) \hookrightarrow$ P.Spec $(\mathcal{A}^{(j)})$ for all $j \in \{2, \dots, 7\}$, and $\psi := \psi_2 \circ \dots \circ \psi_6$: P.Spec $(\mathcal{A}) \hookrightarrow$ P.Spec $(\bar{\mathcal{A}})$. Let $Q \in$ P.Spec $(\mathcal{A}^{(j)})$. If $T_j = X_{j,j} \notin Q$, then we already know that $\psi_j^{-1}(Q) = QS_j^{-1} \cap \mathcal{A}^{(j+1)} = P \in$ P.Spec $(\mathcal{A}^{(j+1)})$ for all $j \in \{2, \dots, 7\}$ (see Subsection 5.1.8). Observe that $\langle T_1 \rangle$ and $\langle T_2 \rangle$ are both elements of P.Spec $(\bar{\mathcal{A}})$. Note, $\mathcal{A}^{(7)} = \mathcal{A}$ and $\mathcal{A}^{(2)} = \bar{\mathcal{A}}$.

The following result shows that $\langle T_1 \rangle \in \text{Im}(\psi)$ and that $\langle \Omega_1 \rangle$ is the Poisson prime ideal of \mathcal{A} such that $\psi(\langle \Omega_1 \rangle) = \langle T_1 \rangle$.

5.2.6 Lemma. $\langle \Omega_1 \rangle \in \mathsf{P}.\mathsf{Spec}(\mathcal{A}).$

Proof. We will prove this result in several steps by showing that:

1. $\langle T_1 \rangle \in \mathsf{P}.\mathsf{Spec}(\mathcal{A}^{(3)}).$

2.
$$\langle X_{1,4}T_3 - \frac{1}{2}T_2 \rangle \in \mathsf{P}.\mathsf{Spec}(\mathcal{A}^{(4)})$$
 [note, $\langle T_1 \rangle [T_3^{-1}] \cap \mathcal{A}^{(4)} = \langle X_{1,4}T_3 - \frac{1}{2}T_2 \rangle$]

3. $\langle X_{1,5}T_3 - \frac{1}{2}X_{2,5} \rangle \in \mathsf{P}.\mathsf{Spec}(\mathcal{A}^{(5)})$ [note, $\langle X_{1,4}T_3 - \frac{1}{2}T_2 \rangle [T_4^{-1}] \cap \mathcal{A}^{(5)} = \langle X_{1,5}T_3 - \frac{1}{2}X_{2,5} \rangle$].

4.
$$\langle X_{1,5}T_3 - \frac{1}{2}X_{2,5}\rangle[T_5^{-1}] \cap \mathcal{A}^{(6)} = \langle \Omega_1 \rangle_{\mathcal{A}^{(6)}}, \text{ hence } \langle \Omega_1 \rangle_{\mathcal{A}^{(6)}} \in \mathsf{P}.\mathsf{Spec}(\mathcal{A}^{(6)})$$

5.
$$\langle \Omega_1 \rangle_{\mathcal{A}^{(6)}}[T_6^{-1}] \cap \mathcal{A} = \langle \Omega_1 \rangle_{\mathcal{A}}$$
, hence $\langle \Omega_1 \rangle_{\mathcal{A}} \in \mathsf{P}.\mathsf{Spec}(\mathcal{A})$.

We proceed to prove the above claims.

1. One can easily verify that $\mathcal{A}^{(3)}/\langle T_1 \rangle$ is isomorphic to a Poisson affine space of rank 5. Hence, $\langle T_1 \rangle$ is a Poisson prime ideal in $\mathcal{A}^{(3)}$.

2. Set $I := \langle X_{1,4}T_3 - \frac{1}{2}T_2 \rangle$. One can verify that $\{X_{i,4}, I\} \subseteq I$ for all $i = 1, \dots, 6$. Therefore, I is a Poisson ideal in $\mathcal{A}^{(4)}$. In addition, $\mathcal{A}^{(4)}/I$ is isomorphic to a polynomial ring in five variables which is a domain, hence, I is a prime ideal. Since I is both Poisson and prime ideal, it is a Poisson prime ideal in $\mathcal{A}^{(4)}$.

3. Similar to (2).

4. Observe that $\Omega'_1 := X_{1,5}T_3 - \frac{1}{2}X_{2,5} = \Omega_1 T_5^{-1}$ in $\mathcal{A}^{(6)}[T_5^{-1}]$. Since $\langle \Omega'_1 \rangle \in$ $\mathsf{P}.\mathsf{Spec}(\mathcal{A}^{(5)}), \text{ we want to show that } \langle \Omega'_1 \rangle [T_5^{-1}] \cap \mathcal{A}^{(6)} = \langle \Omega_1 \rangle_{\mathcal{A}^{(6)}}. \text{ Observe that } \langle \Omega_1 \rangle_{\mathcal{A}^{(6)}} \subseteq$ $\langle \Omega'_1 \rangle [T_5^{-1}] \cap \mathcal{A}^{(6)}$. We establish the reverse inclusion. Let $y \in \langle \Omega'_1 \rangle [T_5^{-1}] \cap \mathcal{A}^{(6)}$. Then, $y \in \langle \Omega'_1 \rangle [T_5^{-1}]$. There exists $i \in \mathbb{N}$ such that $yT_5^i \in \langle \Omega'_1 \rangle$. Hence, $yT_5^i = \Omega'_1 v$, for some $v \in \mathcal{A}^{(5)}$. Furthermore, since $\mathcal{A}^{(5)}[T_5^{-1}] = \mathcal{A}^{(6)}[T_5^{-1}]$, there exists $j \in \mathbb{N}$ such that $vT_5^j = v'$ for some $v' \in \mathcal{A}^{(6)}$. It follows from $yT_5^i = \Omega_1'v$ that $yT_5^{i+j} = \Omega_1'vT_5^j = \Omega_1'v'$. Hence, $yT_5^{\delta} = \Omega_1'T_5v' = \Omega_1v'$, where $\delta = i + j + 1$ (note, $\Omega_1'T_5 = \Omega_1$ in $\mathcal{A}^{(6)}$). Let $S = \{s \in \mathbb{N} \mid \exists v' \in \mathcal{A}^{(6)} : yT_5^s = \Omega_1 v'\}. \text{ Since } \delta \in S, \text{ we have that } S \neq \emptyset. \text{ Let } s = s_0 \text{ be}$ the minimum element of S such that $yT_5^{s_0} = \Omega_1 v'$ for some $v' \in \mathcal{A}^{(6)}$. We want to show that $s_0 = 0$. Suppose that $s_0 > 0$. Since T_5 is irreducible, $yT_5^{s_0} = \Omega_1 v'$ implies that T_5 is a factor of Ω_1 or v'. Clearly, T_5 is not a factor of Ω_1 , hence, it must be a factor of v'. Now $\mathcal{A}^{(6)}$ can be viewed as a free $\mathbb{C}\langle X_{1,6}, X_{2,6}, X_{3,6}, T_4, T_6 \rangle$ -module with basis $\left(T_5^{\xi}\right)_{\xi \in \mathbb{N}}$. One can therefore write $v' = \sum_{\xi=1}^{n} \beta_{\xi} T_5^{\xi}$, where $\beta_{\xi} \in \mathbb{C}\langle X_{1,6}, X_{2,6}, X_{3,6}, T_4, T_6 \rangle$. Returning to $yT_5^{s_0} = \Omega_1 v'$, we have that $yT_5^{s_0} = \Omega_1 \sum_{\xi=1}^n \beta_\xi T_5^{\xi}$. This implies that $yT_5^{s_0-1} = \Omega_1 v''$, where $v'' = \sum_{\xi=1}^n \beta_\xi T_5^{\xi-1} \in \mathcal{A}^{(6)}$. Consequently, $s_0 - 1 \in S$, a contradiction! Therefore, $s_0 = 0$ and $y = \Omega_1 v' \in \langle \Omega_1 \rangle_{\mathcal{A}^{(6)}}$. Hence, $\langle \Omega'_1 \rangle [T_5^{-1}] \cap \mathcal{A}^{(6)} \subseteq \langle \Omega_1 \rangle_{\mathcal{A}^{(6)}}$ as expected.

5. The proof is similar to (4). We want to show that $\langle \Omega_1 \rangle_{\mathcal{A}} = \langle \Omega_1 \rangle_{\mathcal{A}^{(6)}} [T_6^{-1}] \cap \mathcal{A}$. Note, $\langle \Omega_1 \rangle_{\mathcal{A}} \subseteq \langle \Omega_1 \rangle_{\mathcal{A}^{(6)}} [T_6^{-1}] \cap \mathcal{A}$ is trivial. Let $y \in \langle \Omega_1 \rangle_{\mathcal{A}^{(6)}} [T_6^{-1}] \cap \mathcal{A}$. Then, there exists $i \in \mathbb{N}$ such that $yT_6^i = \Omega_1 v$ for some $v \in \mathcal{A}^{(6)}$. Furthermore, since $\mathcal{A}^{(6)} [T_6^{-1}] = \mathcal{A} [T_6^{-1}]$, there exists $j \in \mathbb{N}$ such that $vT_6^j = v'$ for some $v' \in \mathcal{A}$. It follows from $yT_6^i = \Omega_1 v$ that $yT_6^{\delta} = \Omega_1 v'$, where $\delta = i + j$. Similar to (4), one can easily verify that $\delta = 0$. Hence, $y = \Omega_1 v' \in \langle \Omega_1 \rangle_{\mathcal{A}}$. Consequently, $\langle \Omega_1 \rangle_{\mathcal{A}^{(6)}} [T_6^{-1}] \cap \mathcal{A} \subseteq \langle \Omega_1 \rangle_{\mathcal{A}}$.

Similarly, the following result also shows that $\langle T_2 \rangle \in \text{Im}(\psi)$, and that $\langle \Omega_2 \rangle$ is the Poisson prime ideal of \mathcal{A} such that $\psi(\langle \Omega_2 \rangle) = \langle T_2 \rangle$.

5.2.7 Lemma. $\langle \Omega_2 \rangle \in \mathsf{P}.\mathsf{Spec}(\mathcal{A}).$

Proof. We will also prove this result in several steps by showing that:

1.
$$\langle T_2 \rangle_{\mathcal{A}^{(3)}} \in \mathsf{P}.\mathsf{Spec}(\mathcal{A}^{(3)}).$$

2.
$$\langle T_2 \rangle_{\mathcal{A}^{(4)}} \in \mathsf{P}.\mathsf{Spec}(\mathcal{A}^{(4)}).$$

3.
$$\langle T_2 \rangle_{\mathcal{A}^{(4)}}[T_4^{-1}] \cap \mathcal{A}^{(5)} = \langle X_{2,5}T_4 - \frac{2}{3}T_3^3 \rangle$$
, hence $\langle X_{2,5}T_4 - \frac{2}{3}T_3^3 \rangle \in \mathsf{P}.\mathsf{Spec}(\mathcal{A}^{(5)})$.

4.
$$\langle X_{2,5}T_4 - \frac{2}{3}T_3^3 \rangle [T_5^{-1}] \cap \mathcal{A}^{(6)} = \langle \Omega_2' \rangle_{\mathcal{A}^{(6)}}$$
, hence $\langle X_{2,6}T_4 - \frac{2}{3}X_{3,6}^3 \rangle \in \mathsf{P}.\mathsf{Spec}(\mathcal{A}^{(6)})$.

5.
$$\langle X_{2,6}T_4 - \frac{2}{3}X_{3,6}^3 \rangle [T_6^{-1}] \cap \mathcal{A} = \langle \Omega_2 \rangle$$
, hence $\langle \Omega_2 \rangle \in \mathsf{P.Spec}(\mathcal{A})$.

We now proceed to prove the above claims.

1. Observe that $\mathcal{A}^{(3)}/\langle T_2 \rangle$ is isomorphic to a Poisson affine space of rank 5 which is a domain. Hence, $\langle T_2 \rangle$ is a Poisson prime ideal in $\mathcal{A}^{(3)}$.

2. Similar to (1).

3. Recall from the PDDA that $T_2 = X_{2,5} - \frac{2}{3}T_3^3T_4^{-1}$. We want to show that $\langle T_2 \rangle_{\mathcal{A}^{(4)}}[T_4^{-1}] \cap \mathcal{A}^{(5)} = \langle X_{2,5}T_4 - \frac{2}{3}T_3^3 \rangle$. Clearly, $\langle X_{2,5}T_4 - \frac{2}{3}T_3^3 \rangle \subseteq \langle T_2 \rangle_{\mathcal{A}^{(4)}}[T_4^{-1}] \cap \mathcal{A}^{(5)}$. For the reverse inclusion, let $y \in \langle T_2 \rangle_{\mathcal{A}^{(4)}}[T_4^{-1}] \cap \mathcal{A}^{(5)}$. Then, $y \in \langle T_2 \rangle_{\mathcal{A}^{(4)}}[T_4^{-1}]$. Hence, there exists $i \in \mathbb{N}$ such that $yT_4^i = vT_2$, for some $v \in \mathcal{A}^{(4)}$. Furthermore, since $\mathcal{A}^{(4)}[T_4^{-1}] = \mathcal{A}^{(5)}[T_4^{-1}]$, there exists $j \in \mathbb{N}$ such that $vT_4^j = v'$ for some $v' \in \mathcal{A}^{(5)}$. It follows from $yT_4^i = vT_2$ that $yT_4^{i+j} = vT_4^jT_2 = v' \left(X_{2,5} - \frac{2}{3}T_3^3T_4^{-1}\right)$. Consequently,

 $yT_4^{\delta} = \Omega_2'v', \text{ where } \Omega_2' := X_{2,5}T_4 - \frac{2}{3}T_3^3 \text{ and } \delta = i + j + 1. \text{ Let } S = \{s \in \mathbb{N} \mid \exists v' \in \mathcal{A}^{(5)} : yT_4^s = \Omega_2'v'\}. \text{ Since } \delta \in S, \text{ we have that } S \neq \emptyset. \text{ Let } s = s_0 \text{ be the minimum element of } S \text{ such that } yT_4^{s_0} = \Omega_2'v' \text{ for some } v' \in \mathcal{A}^{(5)}. \text{ Suppose that } s_0 > 0. \text{ We want to show that } s_0 = 0. \text{ Since } T_4 \text{ is irreducible, } yT_4^{s_0} = \Omega_2'v' \text{ implies that } T_4 \text{ is a factor of } \Omega_2' \text{ or } v'. \text{ Clearly, } T_4 \text{ is not a factor of } \Omega_2', \text{ hence, it must be a factor of } v'. \text{ Now } \mathcal{A}^{(5)} \text{ can be viewed as a free } \mathbb{C}\langle X_{1,5}, X_{2,5}, T_3, T_5, T_6\rangle - \text{module with basis } \left(T_4^{\xi}\right)_{\xi \in \mathbb{N}}. \text{ One can therefore write } v' = \sum_{\xi=1}^n \beta_\xi T_4^{\xi}, \text{ where } \beta_\xi \in \mathbb{C}\langle X_{1,5}, X_{2,5}, T_3, T_5, T_6\rangle. \text{ Given } yT_4^{s_0} = \Omega_2'v', \text{ we have that } yT_4^{s_0} = \Omega_2'\sum_{\xi=1}^n \beta_\xi T_4^{\xi}. \text{ This implies that } yT_4^{s_0-1} = \Omega_2'v'', \text{ where } v'' = \sum_{\xi=1}^n \beta_\xi T_4^{\xi-1} \in \mathcal{A}^{(5)}. \text{ Consequently, } s_0 - 1 \in S, \text{ a contradiction! Therefore, } s_0 = 0 \text{ and } y = \Omega_2'v' \in \langle \Omega_2'\rangle. \text{ Hence, } \langle T_2\rangle_{\mathcal{A}^{(4)}}[T_4^{-1}] \cap \mathcal{A}^{(5)} \subseteq \langle \Omega_2'\rangle \text{ as desired.}$

4. The proof is similar to point (3). Note, $\Omega'_2 = X_{2,5}T_4 - \frac{2}{3}T_3^3 = X_{2,6}T_4 - \frac{2}{3}X_{3,6}^3$. We want to show that $\langle \Omega'_2 \rangle_{\mathcal{A}^{(5)}}[T_5^{-1}] \cap \mathcal{A}^{(6)} = \langle \Omega'_2 \rangle_{\mathcal{A}^{(6)}}$. The inclusion $\langle \Omega'_2 \rangle_{\mathcal{A}^{(6)}} \subseteq \langle \Omega'_2 \rangle_{\mathcal{A}^{(5)}}[T_5^{-1}] \cap \mathcal{A}^{(6)}$ is trivial. For the reverse inclusion, let $y \in \langle \Omega'_2 \rangle_{\mathcal{A}^{(5)}}[T_5^{-1}] \cap \mathcal{A}^{(6)}$. This implies that there exists $i \in \mathbb{N}$ such that $yT_5^i = v\Omega'_2$ for some $v \in \mathcal{A}^{(5)}$. Since $\mathcal{A}^{(5)}[T_5^{-1}] = \mathcal{A}^{(6)}[T_5^{-1}]$, there exists $j \in \mathbb{N}$ such that $vT_5^j = v'$ for some $v' \in \mathcal{A}^{(6)}$. Therefore, $yT_5^{\delta} = v'\Omega'_2$, where $\delta = i + j$. An argument similar to point (3) above will reveal that $\delta = 0$. As a result, $y = v'\Omega'_2 = \langle \Omega'_2 \rangle_{\mathcal{A}^{(6)}}$. Consequently, $\langle \Omega'_2 \rangle_{\mathcal{A}^{(5)}}[T_5^{-1}] \cap \mathcal{A}^{(6)} \subseteq \langle \Omega'_2 \rangle_{\mathcal{A}^{(6)}}$ as expected.

5. Note, $\Omega'_2 = X_{2,5}T_4 - \frac{2}{3}T_3^3 = \Omega_2 T_6^{-1}$ in $\mathcal{A}[T_6^{-1}]$. We want to show that $\langle \Omega'_2 \rangle_{\mathcal{A}^{(6)}}[T_6^{-1}] \cap \mathcal{A}$ $\mathcal{A} = \langle \Omega_2 \rangle$. Clearly, $\langle \Omega_2 \rangle \subseteq \langle \Omega'_2 \rangle_{\mathcal{A}^{(6)}}[T_6^{-1}] \cap \mathcal{A}$. For the reverse inclusion, let $y \in \langle \Omega'_2 \rangle_{\mathcal{A}^{(6)}}[T_6^{-1}] \cap \mathcal{A}$. Then, $y \in \langle \Omega'_2 \rangle_{\mathcal{A}^{(6)}}[T_6^{-1}]$. Hence, there exists $i \in \mathbb{N}$ such that $yT_6^i = \Omega'_2 v$, for some $v \in \mathcal{A}^{(6)}$. Furthermore, since $\mathcal{A}^{(6)}[T_6^{-1}] = \mathcal{A}[T_6^{-1}]$, there exists $j \in \mathbb{N}$ such that $vT_6^j = v'$ for some $v' \in \mathcal{A}$. Now, $yT_6^i = \Omega'_2 v$ implies that $yT_6^{s_0} = \Omega'_2 v'T_6 = \Omega_2 v'$, where $s_0 = i + j + 1$. Note, $\Omega_2 = \Omega'_2 T_6$ in \mathcal{A} . Similar to point (3), one can easily show that $s_0 = 0$ and $y = \Omega_2 v' \in \langle \Omega_2 \rangle$. As a result, $\langle \Omega'_2 \rangle_{\mathcal{A}^{(6)}}[T_6^{-1}] \cap \mathcal{A} \subseteq \langle \Omega_2 \rangle$ as desired.

Observe that $\langle T_1, T_2 \rangle$ and $\langle T_2, T_3 \rangle$ are Poisson prime ideals of $\overline{\mathcal{A}}$. In the next lemma, we will show that $\langle T_1, T_2 \rangle, \langle T_2, T_3 \rangle \in \psi(\mathsf{P}.\mathsf{Spec}(\mathcal{A})).$

5.2.8 Lemma. $\langle T_1, T_2 \rangle, \langle T_2, T_3 \rangle \in \psi(\mathsf{P}.\mathsf{Spec}(\mathcal{A})).$

Proof. Let $J^{(j)} \in \mathsf{P}.\mathsf{Spec}(\mathcal{A}^{(j)})$ for all $2 \leq j \leq 6$. We already know that $J^{(j+1)} = J^{(j)}[T_j^{-1}] \cap \mathcal{A}^{(j+1)} \in \mathsf{P}.\mathsf{Spec}(\mathcal{A}^{(1+j)})$ provided $T_j \notin J^{(j)}$. Note, $J^{(7)} = J$ and $\mathcal{A}^{(7)} = \mathcal{A}$.

We begin by showing that $\langle T_1, T_2 \rangle \in \psi(\mathsf{P}.\mathsf{Spec}(\mathcal{A})).$

Set $J_{1,2}^{(3)} := \langle T_1, T_2 \rangle \in \mathsf{P}.\mathsf{Spec}(\mathcal{A}^{(3)})$. Observe that $T_3 \notin J_{1,2}^{(3)}$. Therefore, $J_{1,2}^{(3)}[T_3^{-1}] \cap \mathcal{A}^{(4)} = J_{1,2}^{(4)}$. Suppose that $T_4 \in J_{1,2}^{(4)}$. Then, since $J_{1,2}^{(3)}[T_3^{-1}] = J_{1,2}^{(4)}[T_3^{-1}]$, we have that $T_4 \in J_{1,2}^{(3)}[T_3^{-1}] \cap \mathcal{A}^{(4)} = J_{1,2}^{(4)}[T_3^{-1}] \cap \mathcal{A}^{(3)} = J_{1,2}^{(3)}$, a contradiction! Therefore, $T_4 \notin J_{1,2}^{(4)}$. Hence, $J_{1,2}^{(4)}[T_4^{-1}] \cap \mathcal{A}^{(5)} = J_{1,2}^{(5)}$. Suppose that $T_5 \in J_{1,2}^{(5)}$. Then, $T_5 \in J_{1,2}^{(4)}[T_4^{-1}] \cap \mathcal{A}^{(5)} = J_{1,2}^{(5)}[T_4^{-1}] \cap \mathcal{A}^{(4)} = J_{1,2}^{(4)}$, a contradiction! Therefore, $T_5 \notin J_{1,2}^{(5)}$. Hence, $J_{1,2}^{(5)}[T_5^{-1}] \cap \mathcal{A}^{(6)} = J_{1,2}^{(6)}$. Similarly, one can show that $T_6 \notin J_{1,2}^{(6)}$. Hence, $J_{1,2}^{(6)}[T_6^{-1}] \cap \mathcal{A} = J_{1,2}$. Therefore, there exists $J_{1,2} \in \mathsf{P}.\mathsf{Spec}(\mathcal{A})$ such that $\psi(J_{1,2}) = \langle T_1, T_2 \rangle$.

We finally show that $\langle T_2, T_3 \rangle \in \psi(\mathsf{P}.\mathsf{Spec}(\mathcal{A})).$

Set $J_{2,3}^{(4)} := \langle T_2, T_3 \rangle \in \mathsf{P}.\mathsf{Spec}(\mathcal{A}^{(4)})$. Similar to (1), one can verify that $T_j \notin J_{2,3}^{(j)}$ for all j = 4, 5, 6. By an induction on j (note, j = 4, 5, 6), we have that $J_{2,3}^{(j)}[T_j^{-1}] \cap \mathcal{A}^{(j+1)} = J_{2,3}^{(j+1)}$. Therefore, there exists $J_{2,3}^{(7)} = J_{2,3} \in \mathsf{P}.\mathsf{Spec}(\mathcal{A})$ such that $\psi(J_{2,3}) = \langle T_2, T_3 \rangle$.

In $\overline{\mathcal{A}}$, recall that $\Omega_1 = T_1T_3T_5$ and $\Omega_2 = T_2T_4T_6$. Observe that Ω_1, Ω_2 are both elements of $\langle T_1, T_2 \rangle$ and $\langle T_2, T_3 \rangle$. In the above lemma, we know that $J_{1,2}$ and $J_{2,3}$ are the elements of P.Spec(\mathcal{A}) such that $\psi(J_{1,2}) = \langle T_1, T_2 \rangle$ and $\psi(J_{2,3}) = \langle T_2, T_3 \rangle$. In the next lemma, we show that $J_{1,2}$ and $J_{2,3}$ contain Ω_1 and Ω_2 .

5.2.9 Lemma. Ω_1 and Ω_2 are elements of $J_{1,2}$ and $J_{2,3}$.

Proof. Recall that Ω_1 and Ω_2 are central elements of $\mathcal{A}^{(j)}$ for all $2 \leq j \leq 7$. Given the setup in the proof of Lemma 5.2.8, we know that $\Omega_1, \Omega_2 \in J_{1,2}^{(3)} = \langle T_1, T_2 \rangle$. By an induction on j (where j = 3, 4, 5, 6), one can confirm that $\Omega_1, \Omega_2 \in J_{1,2}^{(j)}[T_j^{-1}] \cap \mathcal{A}^{(j+1)} = J_{1,2}^{(j+1)}$. Therefore; $\Omega_1, \Omega_2 \in J_{1,2}$.

Similarly; $\Omega_1, \Omega_2 \in J_{2,3}^{(4)} = \langle T_2, T_3 \rangle$. By an induction on j (where j = 4, 5, 6), one can confirm that $\Omega_1, \Omega_2 \in J_{2,3}^{(j)}[T_j^{-1}] \cap \mathcal{A}^{(j+1)} = J_{2,3}^{(j+1)}$. Therefore, $\Omega_1, \Omega_2 \in J_{2,3}$.

We now want to find the height one Poisson \mathcal{H} -invariant prime ideals of \mathcal{A} , and show that the height 2 Poisson \mathcal{H} -invariant prime ideals of \mathcal{A} contain those of height one. Of course, the zero ideal $\langle 0 \rangle$ is the only height zero Poisson \mathcal{H} -invariant prime ideal.

Let the torus $\mathcal{H}:=(\mathbb{C}^*)^2$ acts by Poisson automorphisms on \mathcal{A} via:

$$h \cdot X_1 = \alpha_1 X_1 \qquad h \cdot X_2 = \alpha_1^3 \alpha_6 X_2 \qquad h \cdot X_3 = \alpha_1^2 \alpha_6 X_3$$
$$h \cdot X_4 = \alpha_1^3 \alpha_6^2 X_4 \qquad h \cdot X_5 = \alpha_1 \alpha_6 X_5 \qquad h \cdot X_6 = \alpha_6 X_6,$$

for all $h := (\alpha_1, \alpha_6) \in \mathcal{H}$. This \mathcal{H} -action is rational. Furthermore, Ω_1 and Ω_2 are Poisson \mathcal{H} -eigenvectors. Hence, $\langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$ are Poisson \mathcal{H} -invariant prime ideals.

Set $\theta_1 := id_{P.Spec(\bar{A})}$. For all $2 \leq j \leq 6$, define $\theta_j := \theta_{j-1} \circ \psi_j$. Then, $\theta_j :$ P.Spec $(A^{(j+1)}) \hookrightarrow P.Spec(\bar{A})$. The map θ_j is injective [35, Section 2.3]. Let $\langle X_{j,j+1} \rangle_P$ denote the smallest Poisson ideal in $A^{(1+j)}$ containing $X_{j,j+1}$. It follows from [35, Lemma 2.2] that there is a surjective Poisson algebra homomorphism $g_j : A^{(j)} \longrightarrow A^{(j+1)}/\langle X_{j,j+1} \rangle_P$ given by $g_j(X_{i,j}) = \overline{X_{i,j+1}} := X_{i,j+1} + \langle X_{j,j+1} \rangle_P$ for all $1 \leq i \leq 6$. Denote the kernel of g_j by ker (g_j) and the image of ψ by Im (ψ) . We have the following lemma.

5.2.10 Lemma. [35, Proposition 2.8] Let $P \in P.Spec(\overline{A})$. The following are equivalent:

- $P \in \operatorname{Im}(\psi)$,
- for all $2 \leq j \leq 6$, we have $P \in \text{Im}(\theta_{j-1})$ and either $X_{j,j} = X_{j,j+1} \notin \theta_{j-1}^{-1}(P)$ or $\text{ker}(g_j) \subseteq \theta_{j-1}^{-1}(P)$.

Note, the map ψ induces a canonical embedding from $\mathcal{H}-\mathsf{P}.\mathsf{Spec}(\mathcal{A})$ to $\mathcal{H}-\mathsf{P}.\mathsf{Spec}(\bar{\mathcal{A}})$. Observe that $\{\langle T_i \rangle \mid i = 1, \cdots, 6\}$ is the set of only height one Poisson $\mathcal{H}-\mathsf{invariant}$ prime ideals we have in $\bar{\mathcal{A}}$. Since ψ preserves the height of a prime ideal, if $\psi^{-1}(\langle T_i \rangle) \in$ $\mathsf{P}.\mathsf{Spec}(\mathcal{A})$, then it is a height one Poisson $\mathcal{H}-\mathsf{invariant}$ prime ideal in \mathcal{A} for all $1 \leq i \leq 6$. For example, we already know that $\psi^{-1}(\langle T_1 \rangle) = \langle \Omega_1 \rangle$ and $\psi^{-1}(\langle T_2 \rangle) = \langle \Omega_2 \rangle$. Therefore, $\langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$ are height one Poisson $\mathcal{H}-\mathsf{invariant}$ prime ideals in \mathcal{A} . We will show in the next lemma that $\langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$ are the only height one Poisson \mathcal{H} -invariant prime ideals we have in \mathcal{A} .

5.2.11 Lemma. For each $j \in \{3, 4, 5, 6\}, \langle T_j \rangle \notin \psi(\mathsf{P.Spec}(\mathcal{A})).$

Proof. Suppose that there exists $j \in \{3, 4, 5, 6\}$ such that $\langle T_j \rangle \in \psi(\mathsf{P.Spec}(\mathcal{A}))$. Then, there exists $P \in \mathsf{P.Spec}(\mathcal{A})$ such that $\psi(P) = \langle T_j \rangle_{\mathcal{A}^{(j)}}$, where $\psi := \psi_j \circ \cdots \psi_6$. Set $P^{(j)} := \langle T_j \rangle_{\mathcal{A}^{(j)}}$. Since $T_j \in P^{(j)}$, it follows from Theorem 5.2.10 that $\ker(g_j) \subseteq P^{(j)}$. The rest follows in cases.

- When j = 3, then $\ker(g_3) \subseteq P^{(3)} = \langle T_3 \rangle_{\mathcal{A}^{(3)}}$. One can easily deduce from the Poisson bracket of \mathcal{A} that $\{X_{3,4}, X_{1,4}\} = -X_{1,4}X_{3,4} - X_{2,4}$. This implies that $X_{2,4} = -\{X_{3,4}, X_{1,4}\} - X_{1,4}X_{3,4} \in \langle X_{3,4} \rangle_P = \langle T_3 \rangle_P$. It follows that $g_3(X_{2,3}) =$ $X_{2,4} + \langle T_3 \rangle_P = \overline{0}$. Hence, $X_{2,3} = T_2 \in \ker(g_3) \subseteq \langle T_3 \rangle_{\mathcal{A}^{(3)}}$, a contradiction! Consequently, $\langle T_3 \rangle \notin \psi(\mathsf{P}.\mathsf{Spec}(\mathcal{A}))$.
- When j = 4, then ker(g₄) ⊆ P⁽⁴⁾ = ⟨T₄⟩_{A⁽⁴⁾}. From the Poisson bracket of A, we have that X²_{3,5} = -¹/₂{X_{4,5}, X_{1,5}} ∈ ⟨X_{4,5}⟩_P = ⟨T₄⟩_P. Since g₄ is a homomorphism, it follows that g₄(X²_{3,4}) = (g₄(X_{3,4}))² = X²_{3,5} + ⟨T₄⟩_P = 0. Therefore, X²_{3,4} = T²₃ ∈ ker(g₄) ⊆ ⟨T₄⟩_{A⁽⁴⁾}, a contradiction! Hence, ⟨T₄⟩ ∉ ψ(P.Spec(A)).
- When j = 5, then ker(g₅) ⊆ P⁽⁵⁾ = ⟨T₅⟩_{A⁽⁵⁾}. One can deduce from the Poisson bracket of A that X_{4,6} = -¹/₃X_{3,6}X_{5,6} ¹/₃{X_{5,6}, X_{3,6}} ∈ ⟨X_{5,6}⟩_P = ⟨T₅⟩_P. It follows that g₅(X_{4,5}) = X_{4,6} + ⟨T₅⟩_P = 0. Hence, X_{4,6} = T₄ ∈ ker(g₅) ⊆ ⟨T₅⟩_{A⁽⁵⁾}, a contradiction! Therefore, ⟨T₅⟩ ∉ ψ(P.Spec(A)).
- Finally, when j = 6, then $\ker(g_6) \subseteq P^{(6)} = \langle T_6 \rangle_{\mathcal{A}^{(6)}}$. Similarly, one can verify that $X_5 = X_1 X_6 - \frac{1}{3} \{ X_6, X_1 \} \in \langle X_6 \rangle_P = \langle T_6 \rangle_P$. Now, $g_6(X_{5,6}) = X_5 + \langle T_6 \rangle_P = \overline{0}$. Therefore, $X_{5,6} = T_5 \in \ker(g_6) \subseteq \langle T_6 \rangle_{\mathcal{A}^{(6)}}$, a contradiction! Hence, $\langle T_6 \rangle \notin \psi(\mathsf{P}.\mathsf{Spec}(\mathcal{A}))$.

In conclusion, $\langle T_j \rangle \notin \psi(\mathsf{P}.\mathsf{Spec}(\mathcal{A}))$ for all $j \in \{3, 4, 5, 6\}$ as expected.

The corollary below is deduced from the above proof (i.e. the proof of Lemma 5.2.11).

5.2.12 Corollary. Let $P \in \psi(\mathsf{P}.\mathsf{Spec}(\mathcal{A}))$. If $T_j \in P$, then $T_{j-1}, \cdots, T_2 \in P$ for all $3 \leq j \leq 6$.

Recall from Lemma 5.2.8 that there exist $J_{1,2}$ and $J_{2,3}$ of P.Spec(\mathcal{A}) such that $\psi(J_{1,2}) = \langle T_1, T_2 \rangle$ and $\psi(J_{2,3}) = \langle T_2, T_3 \rangle$. As a result of Corollary 5.2.12, the Poisson ideals $\langle T_1, T_2 \rangle$ and $\langle T_2, T_3 \rangle$ are the only height two Poisson \mathcal{H} -invariant prime ideals of $\psi(\text{P.Spec}(\mathcal{A}))$. Since ψ preserves Poisson \mathcal{H} -invariant prime ideals and the height of a Poisson prime ideal, it implies that $J_{1,2}$ and $J_{2,3}$ are the only height two Poisson \mathcal{H} -invariant prime ideals of \mathcal{A} . It follows from Lemma 5.2.9 that the height two Poisson \mathcal{H} -invariant prime ideals of \mathcal{A} contain Ω_1 and Ω_2 .

5.2.13 Remark. Since the height two Poisson \mathcal{H} -invariant prime ideals of \mathcal{A} contain Ω_1 and Ω_2 , every non-zero Poisson \mathcal{H} -invariant prime ideal of \mathcal{A} will contain either Ω_1 or Ω_2 . Note, those Poisson \mathcal{H} -invariant primes of at least height 2 will contain both Ω_1 and Ω_2 .

5.2.14 \mathcal{H} -strata corresponding $\langle 0 \rangle$, $\langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$. In this subsection, we aim to find the \mathcal{H} -strata corresponding to $\langle 0 \rangle$, $\langle \Omega_1 \rangle$ and $\langle \Omega_2 \rangle$. We state the results in Propositions 5.2.15, 5.2.16 and 5.2.17. The proofs of these three propositions are similar to those of Propositions 2.4.4, 2.4.5 and 2.4.6 respectively. As a result, we will only prove Proposition 5.2.15, and leave the remaining ones for the reader to verify. Note, all Poisson ideals in \mathcal{A} shall be written as $\langle \Theta \rangle$, where $\Theta \in \mathcal{A}$. However, if we want to refer to a Poisson ideal in any other Poisson algebra, say R, then that Poisson ideal shall be written as $\langle \Theta \rangle_R$, where in this case, $\Theta \in R$.

5.2.15 Proposition. Let \mathcal{P} be the set of those unitary irreducible polynomials $P(\Omega_1, \Omega_2) \in \mathbb{C}[\Omega_1, \Omega_2]$ with $P(\Omega_1, \Omega_2) \neq \Omega_1$ and $P(\Omega_1, \Omega_2) \neq \Omega_2$. Then $\mathsf{P.Spec}_{\langle 0 \rangle}(\mathcal{A}) = \{\langle 0 \rangle\} \cup \{\langle P(\Omega_1, \Omega_2) \rangle \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle \mid \alpha, \beta \in \mathbb{C}^*\}.$

Proof. We claim that $P.Spec_{(0)}(\mathcal{A}) = \{Q \in P.Spec(\mathcal{A}) \mid \Omega_1, \Omega_2 \notin Q\}$. To establish this claim, let us assume that this is not the case. That is, suppose there exists $Q \in P.Spec_{(0)}(\mathcal{A})$ such that $\Omega_1, \Omega_2 \in Q$, then the product $\Omega_1\Omega_2$ which is Poisson

 \mathcal{H} -eigenvector belongs to Q. Consequently, $\Omega_1\Omega_2 \in \bigcap_{h\in\mathcal{H}} h \cdot Q = \langle 0 \rangle$, a contradiction! Hence, we have shown that $P.\operatorname{Spec}_{\langle 0 \rangle}(\mathcal{A}) \subseteq \{Q \in P.\operatorname{Spec}(\mathcal{A}) \mid \Omega_1, \Omega_2 \notin Q\}$. Conversely, suppose that $Q \in P.\operatorname{Spec}(\mathcal{A})$ such that $\Omega_1, \Omega_2 \notin Q$, then $\bigcap_{h\in\mathcal{H}} h \cdot Q$ is a Poisson \mathcal{H} -invariant prime ideal of \mathcal{A} , which contains neither Ω_1 nor Ω_2 . The only possibility for $\bigcap_{h\in\mathcal{H}} h \cdot Q$ is the zero ideal since every non-zero Poisson \mathcal{H} -invariant prime ideal of \mathcal{A} contains Ω_1 or Ω_2 (Remark 5.2.13). Thus, $\bigcap_{h\in\mathcal{H}} h \cdot Q = \langle 0 \rangle$. Hence, $Q \in P.\operatorname{Spec}_{\langle 0 \rangle}(\mathcal{A})$. Therefore, $\{Q \in P.\operatorname{Spec}(\mathcal{A}) \mid \Omega_1, \Omega_2 \notin Q\} \subseteq P.\operatorname{Spec}_{\langle 0 \rangle}(\mathcal{A})$. This confirms our claim.

Since $\Omega_1, \Omega_2 \in Z_P(\mathcal{A})$, we have that the set $\{\Omega_1^i \Omega_2^j \mid i, j \in \mathbb{N}\}$ is a multiplicative set in \mathcal{A} . We can now localize \mathcal{A} as $R := \mathcal{A}[\Omega_1^{-1}, \Omega_2^{-1}]$. Given $Q \in \mathsf{P.Spec}_{\langle 0 \rangle}(\mathcal{A})$, the map $\phi : Q \longrightarrow Q[\Omega_1^{-1}, \Omega_2^{-1}]$ is an increasing bijection from $\mathsf{P.Spec}_{\langle 0 \rangle}(\mathcal{A})$ onto $\mathsf{P.Spec}(R)$.

Let us verify that R is Poisson \mathcal{H} -simple before we describe P.Spec(R). Now, ϕ still induces a bijection between the set of those Poisson \mathcal{H} -invariant prime ideals of $\mathsf{P}.\mathsf{Spec}_{(0)}(\mathcal{A})$ and the set of Poisson $\mathcal{H}-\mathsf{invariant}$ prime ideals of $\mathsf{P}.\mathsf{Spec}(R)$. It is already known that the set of Poisson \mathcal{H} -invariant prime ideals of \mathcal{A} that contains neither Ω_1 nor Ω_2 consists only of the zero ideal $\{\langle 0 \rangle\}$ (Remark 5.2.13). This implies that $\langle 0 \rangle_R$ is the only Poisson \mathcal{H} -invariant prime ideal of R. Every Poisson \mathcal{H} -invariant proper ideal of R is contained in a Poisson \mathcal{H} -invariant prime ideal of R. Therefore, $\langle 0 \rangle_R$ is the only unique Poisson \mathcal{H} -invariant proper ideal of R. This confirms that R is Poisson \mathcal{H} -simple. It follows from [16, Theorem 4.2] that the extension and contraction maps provide a mutually inverse bijection between P.Spec(R) and $Spec(Z_P(R))$. From Lemma 5.2.4, $Z_P(\mathcal{A}) = \mathbb{C}[\Omega_1, \Omega_2]$, and so $Z_P(R) = \mathbb{C}[\Omega_1^{\pm 1}, \Omega_2^{\pm 1}]$. Since \mathbb{C} is algebraically closed, we have that $\operatorname{Spec}(Z_P(R)) = \{\langle 0 \rangle_{Z_P(R)}\} \cup \{\langle P(\Omega_1, \Omega_2) \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)} \mid P(\Omega_1, \Omega_2) \in \mathcal{P}\} \cup \{\langle \Omega_1 - \Omega_2 \rangle_{Z_P(R)}$ $\alpha, \Omega_2 - \beta \rangle_{Z_P(R)} \mid \alpha, \beta \in \mathbb{C}^* \}$. One can now recover P.Spec(R) from Spec($Z_P(R)$) as $\mathsf{follows:} \ \mathsf{P}.\mathsf{Spec}(R) = \{ \langle 0 \rangle_R \} \cup \{ \langle P(\Omega_1, \Omega_2) \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \mid P(\Omega_1, \Omega_2) \in \mathcal{P} \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \} \}$ $\alpha,\beta\in\mathbb{C}^*\}. \text{ It follows that }\mathsf{P}.\mathsf{Spec}_{\langle 0\rangle}(\mathcal{A})=\{\langle 0\rangle_R\cap\mathcal{A}\}\cup\{\langle P(\Omega_1,\Omega_2)\rangle_R\cap\mathcal{A}\mid P(\Omega_1,\Omega_2)\in\mathbb{C}^*\}.$ $\mathcal{P}\} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \cap \mathcal{A} \mid \alpha, \beta \in \mathbb{C}^* \}.$

Undoubtedly, $\langle 0 \rangle_R \cap \mathcal{A} = \langle 0 \rangle$. We now have to show that $\langle P(\Omega_1, \Omega_2) \rangle_R \cap \mathcal{A} = \langle P(\Omega_1, \Omega_2) \rangle$, $\forall P(\Omega_1, \Omega_2) \in \mathcal{P}$, and $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \cap \mathcal{A} = \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$, $\forall \alpha, \beta \in \mathbb{C}^*$,
in order to complete the proof.

Fix $P(\Omega_1, \Omega_2) \in \mathcal{P}$. Clearly, $\langle P(\Omega_1, \Omega_2) \rangle \subseteq \langle P(\Omega_1, \Omega_2) \rangle_R \cap \mathcal{A}$. To show the reverse inclusion, let $y \in \langle P(\Omega_1, \Omega_2) \rangle_R \cap \mathcal{A}$. Since $y \in \langle P(\Omega_1, \Omega_2) \rangle_R$, it implies that $y = dP(\Omega_1, \Omega_2)$ for some $d \in R$. Also, $d \in R$ implies that there exists $i, j \in \mathbb{N}$ such that $d = a\Omega_1^{-i}\Omega_2^{-j}$, where $a \in \mathcal{A}$. Therefore, $y = a\Omega_1^{-i}\Omega_2^{-j}P(\Omega_1, \Omega_2)$, which implies that $y\Omega_1^i\Omega_2^j = aP(\Omega_1, \Omega_2)$. Choose $(i, j) \in \mathbb{N}^2$ minimal (in the lexicographic order on \mathbb{N}^2) such that the equality holds. Without loss of generality, let us suppose that i > 0, then $aP(\Omega_1, \Omega_2) \in \langle \Omega_1 \rangle$. Since $\langle \Omega_1 \rangle$ is a prime ideal, it implies that $a \in \langle \Omega_1 \rangle$ or $P(\Omega_1, \Omega_2) \in \langle \Omega_1 \rangle$. Since $P(\Omega_1, \Omega_2) \in \mathcal{P}$, it implies that $P(\Omega_1, \Omega_2) \notin \langle \Omega_1 \rangle$. Hence, $a \in \langle \Omega_1 \rangle$, which implies that $y\Omega_1^i\Omega_2^j = t\Omega_1P(\Omega_1, \Omega_2)$. Finally, $y\Omega_1^{i-1}\Omega_2^j = tP(\Omega_1, \Omega_2)$. This clearly contradicts the minimality of (i, j), hence (i, j) = (0, 0). As a result, $y = aP(\Omega_1, \Omega_2) \in \mathcal{P}$ as desired.

Similarly, we show that $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \cap \mathcal{A} = \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle; \forall \alpha, \beta \in \mathbb{C}^*$. Fix $\alpha, \beta \in \mathbb{C}^*$. Obviously, $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle \subseteq \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \cap \mathcal{A}$. We establish the reverse inclusion. Let $y \in \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \cap \mathcal{A}$. Since $y \in \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R$, we have that $y = m_0(\Omega_1 - \alpha) + n_0(\Omega_2 - \beta)$, where $m_0, n_0 \in R$. Also, $m_0, n_0 \in R$ implies that there exists $i, j \in \mathbb{N}$ such that $m_0 = m\Omega_1^{-i}\Omega_2^{-j}$ and $n_0 = n\Omega_1^{-i}\Omega_2^{-j}$ for some $m, n \in \mathcal{A}$. Therefore, $y = m\Omega_1^{-i}\Omega_2^{-j}(\Omega_1 - \alpha) + n\Omega_1^{-i}\Omega_2^{-j}(\Omega_2 - \beta)$, which implies that $y\Omega_1^i\Omega_2^j = m(\Omega_1 - \alpha) + n(\Omega_2 - \beta)$. Choose $(i, j) \in \mathbb{N}^2$ minimal (in the lexicographic order on \mathbb{N}^2) such that the equality holds. Without loss of generality, suppose that i > 0 and let $f : \mathcal{A} \longrightarrow \mathcal{A}/\langle \Omega_2 - \beta \rangle$ be a canonical surjection. We have that $f(y)f(\Omega_1)^if(\Omega_2)^j = f(m)f(\Omega_1 - \alpha)$. It follows that $f(m)f(\Omega_1 - \alpha) \in \langle f(\Omega_1) \rangle$. Note, $f(\Omega_1 - \alpha) \notin \langle f(\Omega_1) \rangle$. It implies that $f(m) \in \langle f(\Omega_1) \rangle$. Therefore, $\exists \lambda \in \mathcal{A}$ such that $f(m) = f(\lambda)f(\Omega_1)$. Consequently, $f(y)f(\Omega_1)^if(\Omega_2)^j = f(\lambda)f(\Omega_1)f(\Omega_1 - \alpha)$. Since $f(\Omega_1) \neq 0$, it implies that $f(y)f(\Omega_1)^{i-1}f(\Omega_2)^j = f(\lambda)f(\Omega_1 - \alpha)$. Therefore, $y\Omega_1^{i-1}\Omega_2^j = \lambda(\Omega_1 - \alpha) + \lambda'(\Omega_2 - \beta)$, for some $\lambda' \in \mathcal{A}$. This contradicts the minimality of (i, j). Hence, (i, j) = (0, 0) and so $y = m(\Omega_1 - \alpha) + n(\Omega_2 - \beta) \in \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$. In conclusion, $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle_R \cap \mathcal{A} = \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle, \, \forall \alpha, \beta \in \mathbb{C}^*.$

5.2.16 Proposition. $\mathsf{P}.\mathsf{Spec}_{\langle\Omega_1\rangle}(\mathcal{A}) = \{\langle\Omega_1\rangle\} \cup \{\langle\Omega_1,\Omega_2-\beta\rangle \mid \beta \in \mathbb{C}^*\}.$

Proof. The proof is similar to that of Proposition 2.4.5.

5.2.17 Proposition. P.Spec $_{\langle \Omega_2 \rangle}(\mathcal{A}) = \{ \langle \Omega_2 \rangle \} \cup \{ \langle \Omega_1 - \alpha, \Omega_2 \rangle \mid \alpha \in \mathbb{C}^* \}.$

Proof. The proof is similar to that of Proposition 2.4.6.

5.2.18 Corollary. The Poisson ideal $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$ is primitive in \mathcal{A} for each $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0,0)\}$.

Proof. Since the Poisson ideal $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$ is maximal in its respective strata for each $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}$, it is also primitive (see Proposition 5.1.7).

5.2.19 Proposition. Let $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}$. The Poisson prime ideal $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$ is maximal in \mathcal{A} .

Proof. The proof is similar to that of Proposition 2.4.8. Suppose that there exists a maximal Poisson ideal I of \mathcal{A} such that $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle \subsetneq I \varsubsetneq \mathcal{A}$. Let J be the Poisson \mathcal{H} -invariant prime ideal in \mathcal{A} such that $I \in \mathsf{P.Spec}_J(\mathcal{A})$. By Propositions 5.2.15, 5.2.16 and 5.2.17, J cannot be $\langle 0 \rangle$, $\langle \Omega_1 \rangle$ or $\langle \Omega_2 \rangle$, since either of these will lead to a contradiction. Every non-zero Poisson \mathcal{H} -invariant prime ideal contains Ω_1 only or Ω_2 only or both (Remark 5.2.13). Since $J \neq \langle \Omega_1 \rangle$, $\langle \Omega_2 \rangle$, it implies that J contains both Ω_1 and Ω_2 . Moreover, since $J \subseteq I$, it implies that $\Omega_1, \Omega_2 \in I$. Given $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle \subset I$, we have that $\Omega_1 - \alpha, \Omega_2 - \beta \in I$. It follows that $\alpha, \beta \in I$, hence $I = \mathcal{A}$, a contradiction! This confirms that $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$ is a maximal Poisson ideal in \mathcal{A} .

5.3 Semiclassical limit of $A_{\alpha,\beta}$

In Chapter 3, we studied the simple quotients $A_{\alpha,\beta}$. Similarly, in this section, we will also study the simple quotients of $\mathcal{A} = \mathbb{C}[X_1, \cdots, X_6]$, which we denote by $\mathcal{A}_{\alpha,\beta}$ (note,

 $\alpha, \beta \in \mathbb{C}^2 \setminus \{(0,0)\}$). We will conclude that $\mathcal{A}_{\alpha,\beta}$ is the semiclassical of the noncommutative algebra $A_{\alpha,\beta}$, and close this section by finding a \mathbb{C} -basis for $\mathcal{A}_{\alpha,\beta}$.

Let $\alpha,\beta\in\mathbb{C}^2\setminus\{(0,0)\},$ and set

$$\mathcal{A}_{\alpha,\beta} := \frac{\mathcal{A}}{\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle}.$$

We denote the canonical image of X_i by $x_i := X_i + \langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$ for each $i \in \{1, \dots, 6\}$. The algebra $\mathcal{A}_{\alpha,\beta}$ is commutative, and satisfies the following two relations:

$$x_1 x_3 x_5 - \frac{3}{2} x_1 x_4 - \frac{1}{2} x_2 x_5 + \frac{1}{2} x_3^2 = \alpha,$$
(5.3.1)

$$x_2 x_4 x_6 - \frac{2}{3} x_3^3 x_6 - \frac{2}{3} x_2 x_5^3 + 2 x_3^2 x_5^2 - 3 x_3 x_4 x_5 + \frac{3}{2} x_4^2 = \beta.$$
 (5.3.2)

We also have the following extra relations in $\mathcal{A}_{\alpha,\beta}.$

5.3.1 Lemma.

(1)
$$x_3^2 = 2\alpha + 3x_1x_4 + x_2x_5 - 2x_1x_3x_5$$
.

$$(2) \ x_4^2 = \frac{2}{3}\beta + \frac{8}{9}\alpha x_3 x_6 + \frac{4}{3}x_1 x_3 x_4 x_6 + \frac{4}{9}x_2 x_3 x_5 x_6 - \frac{16}{9}\alpha x_1 x_5 x_6 - \frac{8}{3}x_1^2 x_4 x_5 x_6 + \frac{16}{9}x_1^2 x_3 x_5^2 x_6 - \frac{8}{9}x_2 x_5^3 - \frac{8}{3}\alpha x_5^2 - 4x_1 x_4 x_5^2 + \frac{8}{3}x_1 x_3 x_5^3 + 2x_3 x_4 x_5 - \frac{2}{3}x_2 x_4 x_6 - \frac{8}{9}x_1 x_2 x_5^2 x_6.$$

$$(3) \ x_3^2 x_4 = 2\alpha x_4 + x_2 x_4 x_5 + 2\beta x_1 + \frac{8}{3}\alpha x_1 x_3 x_6 + 4x_1^2 x_3 x_4 x_6 + \frac{4}{3}x_1 x_2 x_3 x_5 x_6 - 8x_1^3 x_4 x_5 x_6 - \frac{8}{3}x_1^2 x_2 x_5^2 x_6 + \frac{16}{3}x_1^3 x_3 x_5^2 x_6 - \frac{8}{3}x_1 x_2 x_5^3 - 8\alpha x_1 x_5^2 - 12x_1^2 x_4 x_5^2 + 8x_1^2 x_3 x_5^3 + 4x_1 x_3 x_4 x_5 - 2x_1 x_2 x_4 x_6 - \frac{16}{3}\alpha x_1^2 x_5 x_6.$$

$$(4) \ x_3 x_4^2 = \frac{2}{3} \beta x_3 + \frac{16}{9} \alpha^2 x_6 + \frac{16}{3} \alpha x_1 x_4 x_6 + \frac{16}{9} \alpha x_2 x_5 x_6 + \frac{16}{9} \alpha x_1 x_3 x_5 x_6 + \frac{4}{9} x_2^2 x_5^2 x_6 \\ + \frac{8}{9} x_1 x_2 x_3 x_5^2 x_6 - \frac{64}{9} \alpha x_1^3 x_5 x_6^2 - \frac{160}{9} \alpha x_1^2 x_5^2 x_6 - \frac{80}{3} x_1^3 x_4 x_5^2 x_6 - \frac{64}{9} x_1^2 x_2 x_5^3 x_6 \\ - \frac{8}{9} x_2 x_3 x_5^3 - \frac{8}{3} \alpha x_3 x_5^2 + 4 x_1 x_3 x_4 x_5^2 + \frac{160}{9} x_1^3 x_3 x_5^3 x_6 - 16 x_1^2 x_4 x_5^3 - \frac{8}{3} x_1 x_2 x_4^4 x_5 \\ - \frac{4}{3} x_1 x_2 x_4 x_5 x_6 + \frac{8}{3} \beta x_1^2 x_6 + \frac{32}{9} \alpha x_1^2 x_3 x_6^2 + \frac{16}{3} x_1^3 x_3 x_4 x_6^2 + \frac{16}{9} x_1^2 x_2 x_3 x_5 x_6^2 \\ - \frac{32}{3} x_1^4 x_4 x_5 x_6^2 - \frac{8}{3} x_1^2 x_2 x_4 x_6^2 + 4 \alpha x_4 x_5 + 2 x_2 x_4 x_5^2 + 4 \beta x_1 x_5 + \frac{64}{9} x_1^4 x_3 x_5^2 x_6^2 \\ - \frac{2}{3} x_2 x_3 x_4 x_6 - \frac{32}{3} \alpha x_1 x_5^3 + \frac{32}{3} x_1^2 x_3 x_5^4 + \frac{32}{3} x_1^2 x_3 x_4 x_5 x_6 - \frac{32}{9} x_1^3 x_2 x_5^2 x_6^2. \end{aligned}$$

Now, the commutative algebra $\mathcal{A}_{\alpha,\beta}$ is a Poisson \mathbb{C} -algebra with the Poisson bracket defined as follows:

$$\{x_2, x_1\} = -3x_1x_2 \qquad \{x_3, x_1\} = -x_1x_3 - x_2 \qquad \{x_3, x_2\} = -3x_2x_3 \{x_4, x_1\} = -2x_3^2 \qquad \{x_4, x_2\} = -3x_2x_4 - 4x_3^3 \quad \{x_4, x_3\} = -3x_3x_4 \{x_5, x_1\} = x_1x_5 - 2x_3 \quad \{x_5, x_2\} = -6x_3^2 \qquad \{x_5, x_3\} = -x_3x_5 - 3x_4 \{x_5, x_4\} = -3x_4x_5 \qquad \{x_6, x_1\} = 3x_1x_6 - 3x_5 \qquad \{x_6, x_2\} = 3x_2x_6 + 9x_4 - 18x_3x_5 \{x_6, x_3\} = -6x_5^2 \qquad \{x_6, x_4\} = -3x_4x_6 - 4x_5^3 \quad \{x_6, x_5\} = -3x_5x_6.$$

Since $\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$ is a maximal Poisson prime ideal of \mathcal{A} (Proposition 5.2.19), $\mathcal{A}_{\alpha,\beta}$ is a simple domain. Moreover, $\mathcal{A}_{\alpha,\beta}$ is noetherian since it is a factor algebra of a noetherian ring \mathcal{A} .

5.3.2 Remark. The Poisson algebra $\mathcal{A}_{\alpha,\beta}$ is the semiclassical limit of the non-commutative algebra $A_{\alpha,\beta}$.

Let $\alpha, \beta \neq 0$. Recall that $\Omega_1 = T_1 T_3 T_5$ and $\Omega_2 = T_2 T_4 T_6$ in \overline{A} . There exists a multiplicative set $S_{\alpha,\beta}$ such that

$$\mathcal{A}_{\alpha,\beta}S_{\alpha,\beta}^{-1} \cong \mathscr{P}_{\alpha,\beta} := \frac{\mathfrak{R}_1}{\langle T_1T_3T_5 - \alpha, T_2T_4T_6 - \beta \rangle},$$

where $\mathfrak{R}_1=\mathbb{C}[T_1^{\pm 1},\cdots,T_6^{\pm 1}]$ is a Poisson torus associated to the Poisson affine space

 $\overline{\mathcal{A}}$. Let $t_i := T_i + \langle T_1 T_3 T_5 - \alpha, T_2 T_4 T_6 - \beta \rangle$ denote the canonical image of T_i in $\mathscr{P}_{\alpha,\beta}$ for each $1 \leq i \leq 6$. The algebra $\mathscr{P}_{\alpha,\beta}$ is a Poisson torus generated by $t_1^{\pm 1}, \cdots, t_6^{\pm 1}$ subject to the relations:

$$t_1 = \alpha t_3^{-1} t_5^{-1}$$
 and $t_2 = \beta t_4^{-1} t_5^{-1}$

One can verify that $\mathscr{P}_{\alpha,\beta} \cong \mathbb{C}[t_3^{\pm 1}, t_4^{\pm 1}, t_5^{\pm 1}, t_6^{\pm 1}]$. One can also verify that this isomorphism holds when either $\alpha = 0$ or $\beta = 0$ (see Section 3.1 for a similar construction). Henceforth, we will identify $\mathscr{P}_{\alpha,\beta}$ with $\mathbb{C}[t_3^{\pm 1}, t_4^{\pm 1}, t_5^{\pm 1}, t_6^{\pm 1}]$ for all $(\alpha, \beta) \in \mathbb{C} \setminus \{(0, 0)\}$.

Set $\mathcal{A}_{\beta} := \mathcal{A}/\langle \Omega_2 - \beta \rangle$, $\beta \in \mathbb{C}$. Denote the canonical image of X_i in \mathcal{A}_{β} by $\widehat{x}_i := X_i + \langle \Omega_2 - \beta \rangle$ for each $1 \leq i \leq 6$. It can be verified that $\mathcal{A}_{\alpha,\beta} \cong \mathcal{A}_{\beta}/\langle \widehat{\Omega}_1 - \alpha \rangle$. Note, \mathcal{A}_{β} satisfies the relation:

$$\widehat{x_4}^2 = \frac{2}{3}\beta - \frac{2}{3}\widehat{x_2}\widehat{x_4}\widehat{x_6} + \frac{4}{9}\widehat{x_3}^3\widehat{x_6} + \frac{4}{9}\widehat{x_2}\widehat{x_5}^3 - \frac{4}{3}\widehat{x_3}^2\widehat{x_5}^2 + 2\widehat{x_3}\widehat{x_4}\widehat{x_5}.$$
 (5.3.3)

5.3.3 Proposition. The set $\mathfrak{F} = \{ \widehat{x_1}^i \widehat{x_2}^j \widehat{x_3}^k \widehat{x_4}^\xi \widehat{x_5}^l \widehat{x_6}^m \mid (\xi, i, j, k, l, m) \in \{0, 1\} \times \mathbb{N}^5 \}$ is a \mathbb{C} -basis of \mathcal{A}_{β} .

Proof. Since $(\prod_{s=1}^{6} X_s^{i_s})_{i_s \in \mathbb{N}}$ is a basis of \mathcal{A} over \mathbb{C} , we have that $(\prod_{s=1}^{6} \hat{x}_s^{i_s})_{i_s \in \mathbb{N}}$ is a spanning set of \mathcal{A}_{β} over \mathbb{C} . We want to show that \mathfrak{F} is a spanning set of \mathcal{A}_{β} . It is sufficient to do that by showing that $\hat{x}_1^{i_1} \hat{x}_2^{i_2} \hat{x}_3^{i_3} \hat{x}_4^{i_4} \hat{x}_5^{i_5} \hat{x}_6^{i_6}$ can be written as a finite linear combination of the elements of \mathfrak{F} over \mathbb{C} for all $i_1, \dots, i_6 \in \mathbb{N}$. We do this by an induction on i_4 . The result is clear when $i_4 = 0$. For $i_4 \geq 0$, suppose that

$$\widehat{x_1}^{i_1} \widehat{x_2}^{i_2} \widehat{x_3}^{i_3} \widehat{x_4}^{i_4} \widehat{x_5}^{i_5} \widehat{x_6}^{i_6} = \sum_{(\xi,\underline{v})\in I} a_{(\xi,\underline{v})} \widehat{x_1}^{i_1} \widehat{x_2}^{j_1} \widehat{x_3}^{k_1} \widehat{x_4}^{\xi} \widehat{x_5}^{l_1} \widehat{x_6}^{m_1},$$

where $\underline{v} := (i, j, k, l, m) \in \mathbb{N}^5$, I is a finite subset of $\{0, 1\} \times \mathbb{N}^5$, and $a_{(\xi, \underline{v})}$ are complex numbers. It follows that

$$\widehat{x_{1}}^{i_{1}}\widehat{x_{2}}^{i_{2}}\widehat{x_{3}}^{i_{3}}\widehat{x_{4}}^{i_{4}+1}\widehat{x_{5}}^{i_{5}}\widehat{x_{6}}^{i_{6}} = \sum_{(\xi,\underline{v})\in I} a_{(\xi,\underline{v})}\widehat{x_{1}}^{i}\widehat{x_{2}}^{j}\widehat{x_{3}}^{k}\widehat{x_{4}}^{\xi+1}\widehat{x_{5}}^{l}\widehat{x_{6}}^{m}$$

We have to show that $\hat{x_1}^i \hat{x_2}^j \hat{x_3}^k \hat{x_4}^{\xi+1} \hat{x_5}^l \hat{x_6}^m \in \text{Span}(\mathfrak{F})$ for all $(\xi, \underline{v}) \in I$. The result is obvious when $\xi = 0$. For $\xi = 1$, then using (5.3.3), one can verify that $\hat{x_1}^i \hat{x_2}^j \hat{x_3}^k \hat{x_4}^2 \hat{x_5}^l \hat{x_6}^m \in \text{Span}(\mathfrak{F})$. Consequently, $\hat{x_1}^{i_1} \hat{x_2}^{i_2} \hat{x_3}^{i_3} \hat{x_4}^{i_4+1} \hat{x_5}^{i_5} \hat{x_6}^{i_6} \in \text{Span}(\mathfrak{F})$. Therefore, \mathfrak{F} spans \mathcal{A}_{β} .

We proceed to show that \mathfrak{F} is a linearly independent set. Note, the ordering $<_4$ in the proof of Proposition 3.2.1 (see item \clubsuit) will be helpful in this part of the proof. Suppose that

$$\sum_{(\xi,\underline{v})\in I} a_{(\xi,\underline{v})} \widehat{x_1}^i \widehat{x_2}^j \widehat{x_3}^k \widehat{x_4}^\xi \widehat{x_5}^l \widehat{x_6}^m = 0.$$

It follows that

$$\sum_{(\xi,\underline{\nu})\in I} a_{(\xi,\underline{\nu})} X_1^i X_2^j X_3^k X_4^{\xi} X_5^l X_6^m = \nu(\Omega_2 - \beta),$$

where $\nu \in \mathcal{A}$. Write $\nu = \sum_{(i,\dots,n)\in J} b_{(i,\dots,n)} X_1^i X_2^j X_3^k X_4^l X_5^m X_6^n$, where J is a finite subset of \mathbb{N}^6 , and $b_{(i,\dots,n)}$ are complex numbers. From Subsection 5.2.1, we have that $\Omega_2 = X_2 X_4 X_6 - \frac{2}{3} X_3^3 X_6 - \frac{2}{3} X_2 X_5^3 + 2 X_3^2 X_5^2 - 3 X_3 X_4 X_5 + \frac{3}{2} X_4^2$. It follows that:

$$\begin{split} \sum_{(\xi,\underline{v})\in I} a_{(\xi,\underline{v})} X_1^i X_2^j X_3^k X_4^{\xi} X_5^l X_6^m &= \sum_{(i,\cdots,n)\in J} b_{(i,\cdots,n)} X_1^i X_2^j X_3^k X_4^l X_5^m X_6^n (\Omega_2 - \beta) \\ &= \sum_{(i,\cdots,n)\in J} \frac{3}{2} b_{(i,\cdots,n)} X_1^i X_2^j X_3^k X_4^{l+2} X_5^m X_6^n + \mathsf{LT}_{<_4}, \end{split}$$

where $LT_{<_4}$ contains lower order terms with respect to $<_4$ (see item 4 in the proof of Proposition 3.2.1). Moreover, $LT_{<_4}$ vanishes when $b_{(i,\cdots,n)} = 0$ for all $(i,\cdots,n) \in J$. One can easily confirm this when the previous line of equality (right hand side) is fully expanded.

Suppose that there exists $(i, j, k, l, m, n) \in J$ such that $b_{(i,j,k,l,m,n)} \neq 0$. Let (i', j', k', l', m', n') be the greatest element of J with respect to $<_4$ such that $b_{(i',j',k',l',m',n')} \neq 0$. Note, the family $(X_1^i X_2^j X_3^k X_4^l X_5^m X_6^n)_{i,\cdots,n\in\mathbb{N}}$ is a basis for \mathcal{A} and $LT_{<_4}$ contains lower order terms. Hence, identifying the coefficients of $X_1^{i'} X_2^{j'} X_3^{k'} X_4^{l'+2} X_5^{m'} X_6^{n'}$, we have $\frac{3}{2} b_{(i',j',k',l',m',n')} = 0$. Therefore, $b_{(i',j',k',l',m',n')} = 0$, a contradiction! As a result, $b_{(i,j,k,l,m,n)} = 0$ for all $(i, j, k, l, m, n) \in J$, and $\sum_{(\xi,\underline{v})\in I} a_{(\xi,\underline{v})} X_1^i X_2^j X_3^k X_4^{\xi} X_5^l X_6^m = 0$. Consequently, $a_{(\xi,i,j,k,l)} = 0$ for all $(\xi, i, j, k, l) \in I$. We are now ready to find a basis for $\mathcal{A}_{\alpha,\beta}$.

5.3.4 Proposition. The set $\mathfrak{P} = \{x_1^i x_2^j x_3^{\epsilon_1} x_4^{\epsilon_2} x_5^k x_6^l \mid (\epsilon_1, \epsilon_2, i, j, k, l) \in \{0, 1\}^2 \times \mathbb{N}^4\}$ is a \mathbb{C} -basis of $\mathcal{A}_{\alpha,\beta}$.

Proof. Since the set $\mathfrak{F} = \{\widehat{x_1}^{i_1} \widehat{x_2}^{i_2} \widehat{x_3}^{i_3} \widehat{x_4}^{\xi} \widehat{x_5}^{i_5} \widehat{x_6}^{i_6} \mid (i_1, i_2, i_3, \xi, i_5, i_6) \in \{0, 1\} \times \mathbb{N}^4\}$ is a \mathbb{C} -basis of \mathcal{A}_{β} over \mathbb{C} (Proposition 5.3.3) and $\mathcal{A}_{\alpha,\beta}$ is identified with $\mathcal{A}_{\beta}/\langle \widehat{\Omega}_1 - \alpha \rangle$, it follows that $(x_1^{i_1} x_2^{i_2} x_3^{i_3} x_4^{\xi} x_5^{i_5} x_6^{i_6})_{(i_1, i_2, i_3, \xi, i_5, i_6) \in \{0, 1\} \times \mathbb{N}^5}$ is a spanning set of $\mathcal{A}_{\alpha,\beta}$ over \mathbb{C} . We want to show that \mathfrak{P} spans $\mathcal{A}_{\alpha,\beta}$ by showing that $e_1^{i_1} e_2^{i_2} e_3^{i_3} e_4^{\xi} e_5^{i_5} e_6^{i_6}$ can be written as a finite linear combination of the elements of \mathfrak{P} over \mathbb{C} for all $(i_1, i_2, i_3, \xi, i_5, i_6) \in \{0, 1\} \times \mathbb{N}^5$. By Proposition 5.3.3, it is sufficient to do this by an induction on i_3 . The result is obvious when $i_3 = 0$ or 1. For $i_3 \ge 1$, suppose that

$$x_1^{i_1} x_2^{i_2} x_3^{i_3} x_4^{\xi} x_5^{i_5} x_6^{i_6} = \sum_{(\epsilon_1, \epsilon_2, \underline{\nu}) \in I} a_{(\epsilon_1, \epsilon_2, \underline{\nu})} x_1^{i} x_2^{j} x_3^{\epsilon_1} x_4^{\epsilon_2} x_5^{k} x_6^{l_5} x_6^{\epsilon_5} x_6^$$

where $\underline{v} := (i, j, k, l) \in \mathbb{N}^4$, $a_{(\epsilon_1, \epsilon_2, \underline{v})}$ are complex numbers, and I is a finite subset of $\{0, 1\}^2 \times \mathbb{N}^4$. It follows from the inductive hypothesis that

$$x_1^{i_1} x_2^{i_2} x_3^{i_3+1} x_4^{\xi} x_5^{i_5} x_6^{i_6} = \sum_{(\epsilon_1, \epsilon_2, \underline{v}) \in I} a_{(\epsilon_1, \epsilon_2, \underline{v})} x_1^i x_2^j x_3^{\epsilon_1+1} x_4^{\epsilon_2} x_5^k x_6^l.$$

We need to show that $x_1^i x_2^j x_3^{\epsilon_1+1} x_4^{\epsilon_2} x_5^k x_6^l \in \text{Span}(\mathfrak{P})$ for all $(\epsilon_1, \epsilon_2, \underline{v}) \in I$. The result is obvious when $(\epsilon_1, \epsilon_2) = (0, 0), (0, 1)$. Using Lemma 5.3.1(1),(3), one can also show that $x_1^i x_2^j x_3^{\epsilon_1+1} x_4^{\epsilon_2} x_5^k x_6^l \in \text{Span}(\mathfrak{P})$ for all $(\epsilon_1, \epsilon_2) = (1, 0), (1, 1)$; and $i, j, k, l \in \mathbb{N}$. Therefore, $x_1^{i_1} x_2^{i_2} x_3^{i_3+1} x_4^{\xi_3} x_5^{i_5} x_6^{i_6} \in \text{Span}(\mathfrak{P})$. As a result, \mathfrak{P} spans $\mathcal{A}_{\alpha,\beta}$.

We proceed to show that \mathfrak{F} is a linearly independent set. Note, the ordering $<_3$ in the proof of Proposition 3.2.2 (see item \clubsuit) will be helpful in this part of the proof. Suppose that

$$\sum_{(\epsilon_1,\epsilon_2,\underline{v})\in I} a_{(\epsilon_1,\epsilon_2,\underline{v})} x_1^i x_2^j x_3^{\epsilon_1} x_4^{\epsilon_2} x_5^k x_6^l = 0.$$

Then, we have that

$$\sum_{(\epsilon_1,\epsilon_2,\underline{v})\in I} a_{(\epsilon_1,\epsilon_2,\underline{v})} \widehat{x_1}^i \widehat{x_2}^j \widehat{x_3}^{\epsilon_1} \widehat{x_4}^{\epsilon_2} \widehat{e_5}^k \widehat{e_6}^l = (\widehat{\Omega}_1 - \alpha)\nu$$

in \mathcal{A}_{β} , where $\nu \in \mathcal{A}_{\beta}$. Set $\underline{w} := (i, j, k, l, m) \in \mathbb{N}^5$, and let J_1 and J_2 be finite subsets of \mathbb{N}^5 . One can write ν in terms of the basis \mathfrak{F} of \mathcal{A}_{β} as

$$\nu = \sum_{\underline{w}\in J_1} b_{\underline{w}} \widehat{x_1}^i \widehat{x_2}^j \widehat{x_3}^k \widehat{x_4} \widehat{x_5}^l \widehat{x_6}^m + \sum_{\underline{w}\in J_2} c_{\underline{w}} \widehat{x_1}^i \widehat{x_2}^j \widehat{x_3}^k \widehat{x_5}^l \widehat{x_6}^m,$$

where $b_{\underline{w}}$ and $c_{\underline{w}}$ are complex numbers. Note, $\widehat{\Omega}_1 = \widehat{x}_1 \widehat{x}_3 \widehat{x}_5 - \frac{3}{2} \widehat{x}_1 \widehat{x}_4 - \frac{1}{2} \widehat{x}_2 \widehat{x}_5 + \frac{1}{2} \widehat{x}_3^2$. Given this expression, and the relation (5.3.3), one can express the above equality as follows:

$$\begin{split} \sum_{(\epsilon_1,\epsilon_2,\underline{v})\in I} a_{(\epsilon_1,\epsilon_2,\underline{v})} \widehat{x_1}^i \widehat{x_2}^j \widehat{x_3}^{\epsilon_1} \widehat{x_4}^{\epsilon_2} \widehat{e_5}^k \widehat{e_6}^l &= \sum_{\underline{w}\in J_1} b_{\underline{w}} \widehat{x_1}^i \widehat{x_2}^j \widehat{x_3}^k \widehat{x_4} \widehat{x_5}^l \widehat{x_6}^m (\widehat{\Omega}_1 - \alpha) \\ &+ \sum_{\underline{w}\in J_2} c_{\underline{w}} \widehat{x_1}^i \widehat{x_2}^j \widehat{x_3}^k \widehat{x_5}^l \widehat{x_6}^m (\widehat{\Omega}_1 - \alpha) \\ &= \sum_{\underline{w}\in J_1} \frac{1}{2} b_{\underline{w}} \widehat{x_1}^i \widehat{x_2}^j \widehat{x_3}^{k+2} \widehat{x_4} \widehat{x_5}^l \widehat{x_6}^m \\ &- \sum_{\underline{w}\in J_1} \frac{2}{3} b_{\underline{w}} \widehat{x_1}^{i+1} \widehat{x_2}^j \widehat{x_3}^{k+3} \widehat{x_5}^l \widehat{x_6}^{m+1} \\ &- \sum_{\underline{w}\in J_2} \frac{3}{2} c_{\underline{w}} \widehat{x_1}^{i+1} \widehat{x_2}^j \widehat{x_3}^k \widehat{x_4} \widehat{x_5}^l \widehat{x_6}^m \\ &+ \sum_{\underline{w}\in J_2} \frac{1}{2} c_{\underline{w}} \widehat{x_1}^i \widehat{x_2}^j \widehat{x_3}^{k+2} \widehat{x_5}^l \widehat{x_6}^m + \Upsilon, \end{split}$$

where Υ is defined on the next page.

$$\begin{split} \Upsilon &= \sum_{\underline{w} \in J_1} r_1 b_{\underline{w}} \widehat{x_1}^{i+1} \widehat{x_2}^j \widehat{x_3}^k \widehat{x_5}^l \widehat{x_6}^m + \sum_{\underline{w} \in J_1} r_2 b_{\underline{w}} \widehat{x_1}^{i+1} \widehat{x_2}^{1+j} \widehat{x_3}^k \widehat{x_4} \widehat{x_5}^l \widehat{x_6}^{m+1} \\ &+ \sum_{\underline{w} \in J_1} r_3 b_{\underline{w}} \widehat{x_1}^{i+1} \widehat{x_2}^{j+1} \widehat{x_3}^k \widehat{x_5}^{l+3} \widehat{x_6}^m + \sum_{\underline{w} \in J_1} r_4 b_{\underline{w}} \widehat{x_1}^{i+1} \widehat{x_2}^j \widehat{x_3}^{k+2} \widehat{x_5}^{l+2} \widehat{x_6}^m \\ &+ \sum_{\underline{w} \in J_1} r_5 b_{\underline{w}} \widehat{x_1}^{i+1} \widehat{x_2}^j \widehat{x_3}^{k+1} \widehat{x_4} \widehat{x_5}^{l+1} \widehat{x_6}^m + \sum_{\underline{w} \in J_1} r_6 b_{\underline{w}} \widehat{x_1}^i \widehat{x_2}^{j+1} \widehat{x_3}^k \widehat{x_4} \widehat{x_5}^{l+1} \widehat{x_6}^m \\ &+ \sum_{\underline{w} \in J_1} r_7 b_{\underline{w}} \beta \widehat{x_1}^i \widehat{x_2}^j \widehat{x_3}^k \widehat{x_4} \widehat{x_5}^l \widehat{x_6}^m + \sum_{\underline{w} \in J_1} r_8 b_{\underline{w}} \widehat{x_1}^{i+1} \widehat{x_2}^j \widehat{x_3}^{k+1} \widehat{x_4} \widehat{x_5}^{l+1} \widehat{x_6}^m \\ &+ \sum_{\underline{w} \in J_2} r_9 c_{\underline{w}} \widehat{x_1}^{i+1} \widehat{x_2}^j \widehat{x_3}^{k+1} \widehat{x_5}^{l+1} \widehat{x_6}^m + \sum_{\underline{w} \in J_2} r_{10} c_{\underline{w}} \widehat{x_1}^i \widehat{x_2}^{j+1} \widehat{x_3}^k \widehat{x_5}^{l+1} \widehat{x_6}^m \\ &+ \sum_{\underline{w} \in J_2} r_{11} c_{\underline{w}} \alpha \widehat{x_1}^i \widehat{x_2}^j \widehat{x_3}^k \widehat{x_5}^l \widehat{x_6}^m. \end{split}$$

Note: r_1, \dots, r_{11} are some non-zero rational numbers.

Observe that Υ contains lower order terms with respect to $<_3$ (defined in \blacklozenge , see the proof of Proposition 3.2.2) in each monomial type (note, there are two different types of monomials in the basis of $\mathcal{A}_eta;$ one with $\widehat{x_4}$ and the other without $\widehat{x_4}$). Now, suppose that there exists $(i,j,k,l,m) \in J_1$ and $(i,j,k,l,m) \in J_2$ such that $b_{(i,j,k,l,m)}
eq 0$ and $c_{(i,j,k,l,m)} \neq 0$. Let $(v_1, v_2, v_3, v_5, v_6)$ and $(w_1, w_2, w_3, w_5, w_6)$ be the greatest elements of J_1 and J_2 respectively with respect to $<_3$ such that $b_{(v_1,v_2,v_3,v_5,v_6)}$ and $c_{(w_1,w_2,w_3,w_5,w_6)}$ are non-zero. Since $\mathfrak F$ is a linear basis for $\mathcal A_eta$ and Υ contains lower order terms with respect to $<_3$, we have the following: if $w_3 - v_3 < 2$, then identifying the coefficients of $\widehat{x_1}^{v_1} \widehat{x_2}^{v_2} \widehat{x_3}^{v_3+2} \widehat{x_4} \widehat{x_5}^{v_5} \widehat{x_6}^{v_6}$, we have $\frac{1}{2} b_{(v_1, v_2, v_3, v_5, v_6)} = 0$, a contradiction! Finally, if $w_3 - v_3 \ge 2$, then identifying the coefficients of $\widehat{x_1}^{w_1} \widehat{x_2}^{w_2} \widehat{x_3}^{w_3+2} \widehat{x_5}^{w_5} \widehat{x_6}^{w_6}$, we have $\frac{1}{2}c_{(w_1,w_2,w_3,w_5,w_6)} = 0$, another contradiction! This implies that either all $b_{(i,j,k,m,n)}$ or all $c_{(i,j,k,m,n)}$ are zero. Without loss of generality, suppose that there exists $(i,j,k,m,n)\in J_2$ such that $c_{(i,j,k,m,n)}$ is not zero. Then, $b_{(i,j,k,m,n)}$ are all zero. Let (w_1,w_2,w_3,w_5,w_6) be the greatest element of J_2 such that $c_{(w_1,w_2,w_3,w_5,w_6)} \neq 0$. Identifying the coefficients of $\widehat{x_1}^{w_1} \widehat{x_2}^{w_2} \widehat{x_3}^{w_3+2} \widehat{x_5}^{w_5} \widehat{x_6}^{w_6}$ in the above equality, we have that $\frac{1}{2} c_{(w_1, w_2, w_3, w_5, w_6)} = 0$, a contradiction! We can therefore conclude that $b_{(i,j,k,m,n)}$ and $c_{(i,j,k,m,n)}$ are all zero.

Consequently,

$$\sum_{\substack{(\epsilon_1,\epsilon_2,\underline{v})\in I}} a_{(\epsilon_1,\epsilon_2,\underline{v})} \widehat{x_1}^i \widehat{x_2}^j \widehat{x_3}^{\epsilon_1} \widehat{x_4}^{\epsilon_2} \widehat{x_5}^k \widehat{x_6}^l = 0.$$

Since \mathfrak{F} is a basis for \mathcal{A}_{β} , it implies that $a_{(\epsilon_1,\epsilon_2,\underline{v})} = 0$ for all $(\epsilon_1,\epsilon_2,\underline{v}) \in I$. Therefore, \mathfrak{P} is a linearly independent set.

5.3.5 Corollary. Let $\underline{v} = (i, j, k, l) \in \mathbb{N}^2 \times \mathbb{Z}^2$, I represent a finite subset of $\{0, 1\}^2 \times \mathbb{N}^2 \times \mathbb{Z}^2$, and $(a_{(\epsilon_1, \epsilon_2, \underline{v})})_{(\epsilon_1, \epsilon_2, \underline{v}) \in I}$ be a family of complex numbers. If

$$\sum_{(\epsilon_1,\epsilon_2,\underline{v})\in I} a_{(\epsilon_1,\epsilon_2,\underline{v})} x_1^i x_2^j e_3^{\epsilon_1} x_4^{\epsilon_2} t_5^k t_6^l = 0,$$

then $a_{(\epsilon_1,\epsilon_2,\underline{v})} = 0$ for all $(\epsilon_1,\epsilon_2,\underline{v}) \in I$.

Proof. The result is obvious when $k, l \ge 0$ due to Proposition 5.3.4. When k (resp. l) is negative, then one can multiply the above equality enough times by t_5 (resp. t_6) to kill all the negative powers and then apply Proposition 5.3.4 to complete the proof.

Chapter 6

Poisson derivations of the semiclassical limits of $A_{\alpha,\beta}$

In this chapter, we compute the Poisson derivations of $\mathcal{A}_{\alpha,\beta}$. We employ the same strategy used in computing the derivations of the non-commutative algebra $A_{\alpha,\beta}$ in Chapter 4. We begin by finding the Poisson derivations of a Poisson group algebra by following a pattern similar to [40]. We conclude that every Poisson derivation of the Poisson group algebra is the sum of an inner Poisson derivation and a scalar Poisson derivation. Since the Poisson torus is a Poisson group algebra, we have that every Poisson derivation of a Poisson torus is the sum of an inner and a central/scalar Poisson derivation. Knowing the Poisson derivations of the Poisson torus, we embed each of the Poisson algebra $\mathcal{A}_{lpha,eta}$ into a suitable Poisson torus, and then extend every Poisson derivation of $\mathcal{A}_{lpha,eta}$ uniquely to the Poisson torus. We then restrict the Poisson derivations of the Poisson torus back to $\mathcal{A}_{\alpha,\beta}$, and conclude that the Poisson derivations of $\mathcal{A}_{\alpha,\beta}$ are all inner when α and β are non-zero. However, when either α or β is zero, we conclude that every Poisson derivation of $\mathcal{A}_{lpha,eta}$ is the sum of an inner and a scalar Poisson derivation. More precisely, the first Hochschild cohomology group of $\mathcal{A}_{lpha,eta}$ is of dimension 0 when lpha and eta are non-zero and 1 when either lpha or eta is zero. The results in this chapter are congruent to their non-commutative counterparts.

6.1 Poisson derivations of Poisson group algebras

Osborn and Passman have studied the derivations of the twisted group algebras [40, §1&2]. In this section, we produce the Poisson version of their results. That is, we will study the Poisson derivations of the Poisson group algebras. Where applicable, we will maintain their notations. Let G be a finitely generated abelian group and K be a field with characteristic zero. A Poisson group algebra $K_P^{\lambda}[G]$ is a commutative K-algebra which has a copy $\overline{G} := \{\overline{g} \mid g \in G\}$ of G as a basis and satisfies the Poisson bracket via $\{\overline{x}, \overline{y}\} = \lambda(x, y)\overline{x}\overline{y} = \lambda(x, y)\overline{x}\overline{y}$ for all $x, y \in G$; and $\lambda : G \times G \longrightarrow K$ (note, $\overline{x}\overline{y} = \overline{xy}$). The map λ satisfies the following properties: $\lambda(y, x) = -\lambda(x, y)$ and $\lambda(x, yz) = \lambda(x, y) + \lambda(x, z)$. Obviously, $\lambda(x, y) = 0$ if and only if $\{\overline{x}, \overline{y}\} = 0$.

For example, take the additive group $G = \mathbb{Z}^2$, and let $\mathbb{C}[G]$ be a group algebra generated by $x^{\pm 1}, y^{\pm 1}$ over \mathbb{C} with a basis $\overline{G} = \{x^i y^j \mid (i, j) \in G\}$. One can define a Poisson structure on $\mathbb{C}[G]$ via $\{x^i y^j, x^k y^l\} = \lambda((i, j), (k, l))x^{i+k}y^{j+l}$, where $\lambda((i, j), (k, l)) := il - jk$. In general, $K_P^{\lambda}[\mathbb{Z}^n]$ is a Poisson torus of rank n over the field K for some $\lambda : \mathbb{Z}^n \times \mathbb{Z}^n \to K$, where \mathbb{Z}^n is the usual additive group.

For any $\gamma \in K_P^{\lambda}[G]$, one can write γ as $\gamma = \sum_g c_g \bar{g}$ with $g \in G$ and $c_g \in K$. Note, $c_g = 0$ for almost all $c_g \in K$. The set $\operatorname{supp}(\gamma) := \{g \in G \mid c_g \neq 0 \text{ in } \gamma\}$ is called the support of γ . Furthermore, set $C := \{g \in G \mid \{\bar{g}, \bar{x}\} = 0 \text{ for all } x \in G\}$ and $\Delta(\bar{x}) := \{g \in G \mid \{\bar{g}, \bar{x}\} = 0\}$. One can observe that C and $\Delta(\bar{x})$ are both subgroups of G. If g_1, \dots, g_n are the generators of G, then $C = \bigcap_{i=1}^n \Delta(g_i)$.

6.1.1 Lemma. The Poisson center $Z_P(K_P^{\lambda}[G])$ of $K_P^{\lambda}[G]$ is $K_P^{\lambda}[C]$.

Proof. Clearly, $K_P^{\lambda}[C] \subseteq Z_P(K_P^{\lambda}[G])$. For the reverse inclusion, take $\gamma = \sum_g c_g \bar{g} \in Z_P(K_P^{\lambda}[G])$. It follows that $0 = \{\gamma, \bar{x}\} = \sum_g c_g\{\bar{g}, \bar{x}\} = \sum_g c_g\lambda(g, x)\bar{g}\bar{x}$ for any $x \in G$. Consequently, $\lambda(g, x) = 0$ for all $g \in \operatorname{supp}(\gamma)$. This implies that $\operatorname{supp}(\gamma) \subseteq C$, hence, $\gamma \in K_P^{\lambda}[C]$.

6.1.2 Remark. Let e be the identity element of G. One can easily observe that $Z_P(K_P^{\lambda}[G]) = K$ if and only if $C = \{e\}$.

Central and inner Poisson derivations. Let $\theta : (G, \cdot) \longrightarrow (K_P^{\lambda}[C], +)$ be a group homomorphism. That is, $\theta(xy) = \theta(x) + \theta(y)$ for all $x, y \in G$. Define the K-linear operator $\mathcal{D} := \mathcal{D}_{\theta}$ by $\mathcal{D}(\bar{x}) = \theta(x)\bar{x}$ for all $x \in G$. We claim that \mathcal{D} is a Poisson derivation of $K_P^{\lambda}[G]$. To establish this claim, we need to show that $\mathcal{D}(\bar{x}\bar{y}) = \mathcal{D}(\bar{x})\bar{y} + \bar{x}\mathcal{D}(\bar{y})$ and $\mathcal{D}(\{\bar{x},\bar{y}\}) = \{\mathcal{D}(\bar{x}),\bar{y}\} + \{\bar{x},\mathcal{D}(\bar{y})\}$ for all $x, y \in G$. Now, $\mathcal{D}(\bar{x}\bar{y}) = \theta(xy)\bar{x}\bar{y} =$ $\theta(xy)\bar{x}\bar{y} = \theta(x)\bar{x}\bar{y} + \theta(y)\bar{x}\bar{y} = \mathcal{D}(\bar{x})\bar{y} + \bar{x}\mathcal{D}(\bar{y})$. Secondly, $\mathcal{D}(\{\bar{x},\bar{y}\}) = \lambda(x,y)\mathcal{D}(\bar{x}\bar{y}) =$ $\theta(xy)\lambda(x,y)\bar{x}\bar{y} = \theta(xy)\{\bar{x},\bar{y}\} = [\theta(x) + \theta(y)]\{\bar{x},\bar{y}\} = \theta(x)\{\bar{x},\bar{y}\} + \theta(y)\{\bar{x},\bar{y}\} =$ $\{\theta(x)\bar{x},\bar{y}\} + \{\bar{x},\theta(y)\bar{y}\} = \{\mathcal{D}(\bar{x}),\bar{y}\} + \{\bar{x},\mathcal{D}(\bar{y})\}$ (note: $\{\theta(x),\bar{y}\} = \{\bar{x},\theta(y)\} = 0$, since $\theta(x)$ and $\theta(y)$ are Poisson central elements). This establishes our claim. Call any derivation \mathcal{D} in line with this construction a *central Poisson derivation*. If $Z_P(K_P^{\lambda}[G]) =$ K, then call \mathcal{D} a *scalar Poisson derivation*. Observe that $\mathcal{D}(\bar{x}) \in K_P^{\lambda}[Cx]$ for all $x \in G$.

Let $\gamma = \sum_{g} c_g \bar{g} \in K_P^{\lambda}[G]$, where $c_g \in K$, and $\hom_{\gamma} := \{\gamma, -\}$. It is well known that $\hom_{\gamma} : K_P^{\lambda}[G] \longrightarrow K_P^{\lambda}[G]$ is a derivation called the *hamiltonian derivation associated to* γ . Moreover, $\hom_{\gamma}(\bar{x}) = \{\gamma, \bar{x}\} = \sum_{g} \lambda(g, x) c_g \bar{g} \bar{x} \in K_P^{\lambda}[Gx]$ for all $x \in G$. Observe that the elements of $C \cap \operatorname{supp}(\gamma)$ do not have any effect on the map \hom_{γ} . That is, $\hom_{\gamma} = \hom_{\gamma+\mu\bar{t}}$ for all $t \in C \cap \operatorname{supp}(\gamma)$ and $\mu \in K$. As a result, one can always assume that $C \cap \operatorname{supp}(\gamma) = \emptyset$. Therefore, $\hom_{\gamma}(\bar{x}) \in K_P^{\lambda}[(G \setminus C)x]$ for all $x \in G$. Let us call the hamiltonian derivation \hom_{γ} an *inner Poisson derivation*. We have the following theorem.

6.1.3 Theorem. Every Poisson derivation of $K_P^{\lambda}[G]$ is uniquely the sum of an inner Poisson derivation and a central Poisson derivation.

Proof. Let \mathcal{D} be a Poisson derivation of $K_P^{\lambda}[G]$. Then, for $x \in G$, we have that $\mathcal{D}(\bar{x}) \in K_P^{\lambda}[G]$. Hence, $\mathcal{D}(\bar{x}) = \sum_{h \in G} b_h(x)\bar{h} = \sum_{h \in G} b_h(x)\bar{h}\bar{x}^{-1}\bar{x}$. Now, the map $G \to G$ with $h \mapsto hx^{-1}$ is bijective, and so

$$\mathcal{D}(\bar{x}) = \sum_{g} a_g(x) \bar{g} \bar{x},$$

where $g := hx^{-1}$ and $a_g(x) := b_{gx}(x)$. Note; $a_g : G \longrightarrow K$ and $a_g(x) = 0$ for almost all

 $x\in G.$

Since \mathcal{D} is a Poisson derivation, we have that $\mathcal{D}(\bar{x}\bar{y}) = \mathcal{D}(\bar{x})\bar{y} + \bar{x}\mathcal{D}(\bar{y})$ for all $x, y \in G$. As a result,

$$\sum_{g} a_g(xy)\bar{g}\bar{x}\bar{y} = \sum_{g} a_g(x)\bar{g}\bar{x}\bar{y} + \sum_{g} a_g(y)\bar{g}\bar{x}\bar{y} = \sum_{g} [a_g(x) + a_g(y)]\bar{g}\bar{x}\bar{y}.$$

Identifying the coefficients in the above equality reveals that

$$a_g(xy) = a_g(x) + a_g(y).$$

Secondly, $\mathcal{D}(\{\bar{x}, \bar{y}\}) = \{\mathcal{D}(\bar{x}), \bar{y}\} + \{\bar{x}, \mathcal{D}(\bar{y})\}$. Now,

$$\mathcal{D}(\{\bar{x},\bar{y}\}) = \lambda(x,y)\mathcal{D}(\bar{x}\bar{y}) = \sum_{g} \lambda(x,y)a_g(xy)\bar{g}\bar{x}\bar{y}.$$
(6.1.1)

On the other hand,

$$\begin{aligned} \{\mathcal{D}(\bar{x}), \bar{y}\} + \{\bar{x}, \mathcal{D}(\bar{y})\} &= \sum_{g} a_{g}(x)\{\bar{g}\bar{x}, \bar{y}\} + \sum_{g} a_{g}(y)\{\bar{x}, \bar{g}\bar{y}\} \\ &= \sum_{g} a_{g}(x)\{\bar{g}\bar{x}, \bar{y}\} + \sum_{g} a_{g}(y)\{\bar{x}, \bar{g}\bar{y}\} \\ &= \sum_{g} [a_{g}(x)\lambda(gx, y) + a_{g}(y)\lambda(x, gy)]\bar{g}\bar{x}\bar{y} \\ &= \sum_{g} a_{g}(x)[\lambda(g, y) + \lambda(x, y)] + a_{g}(y)[\lambda(x, g) + \lambda(x, y)]\bar{g}\bar{x}\bar{y} \\ &= \sum_{g} [\lambda(x, y)a_{g}(xy) + a_{g}(x)\lambda(g, y) - a_{g}(y)\lambda(g, x)]\bar{g}\bar{x}\bar{y}. \end{aligned}$$

$$(6.1.2)$$

Since $\mathcal{D}(\{\bar{x}, \bar{y}\} = \{\mathcal{D}(\bar{x}), \bar{y}\} + \{\bar{x}, \mathcal{D}(\bar{y})\}$, comparing (6.1.1) to (6.1.2) reveals that

$$\lambda(x,y)a_g(xy) = \lambda(x,y)a_g(xy) + a_g(x)\lambda(g,y) - a_g(y)\lambda(g,x).$$

This implies that

$$a_g(x)\lambda(g,y) = a_g(y)\lambda(g,x).$$
(6.1.3)

Suppose that $g \in C$, it follows that $\lambda(g, y) = \lambda(g, x) = 0$ for all $x, y \in G$. Since $a_g(xy) = a_g(x) + a_g(y)$, the map $\theta : (G, \cdot) \longrightarrow (K_P^{\lambda}[C], +)$ given by $\theta(x) = \sum_{g \in C} a_g(x)\overline{g}$ is a group homomorphism. Hence, θ defines a central Poisson derivation \mathcal{D}_{θ} of $K_P^{\lambda}[G]$, where

$$\mathcal{D}_{\theta}(\bar{x}) = \sum_{g \in C} a_g(x) \bar{g} \bar{x}.$$
(6.1.4)

Now, let $g \notin C$. There exists $y \in G$ such that $\lambda(g, y) \neq 0$. Fix y and define

$$c_g := \frac{a_g(y)}{\lambda(g, y)}.$$

Take any arbitrary element $x \in G$. It follows that

$$c_g\lambda(g,x) = \frac{a_g(y)\lambda(g,x)}{\lambda(g,y)}$$

From (6.1.3), we have that

$$c_g\lambda(g,x) = \frac{a_g(y)\lambda(g,x)}{\lambda(g,y)} = \frac{a_g(x)\lambda(g,y)}{\lambda(g,y)} = a_g(x),$$

for all $x \in G$.

Define $\gamma \in K^{\lambda}_{P}[G]$ as $\gamma := \sum_{g \not\in C} c_{g} \bar{g}.$ Then,

$$\mathsf{ham}_{\gamma}(\bar{x}) = \{\gamma, \bar{x}\} = \sum_{g \notin C} c_g \lambda(g, x) \bar{g} \bar{x} = \sum_{g \notin C} a_g(x) \bar{g} \bar{x}.$$
(6.1.5)

From (6.1.4) and (6.1.5), one can conclude that every Poisson derivation \mathcal{D} of $K_P^{\lambda}[G]$ can be written as $\mathcal{D} = \mathcal{D}_{\theta} + \text{ham}_{\gamma}$. This decomposition of \mathcal{D} into an inner Poisson derivation (ham_{γ}) and a central Poisson derivation (\mathcal{D}_{θ}) is actually unique. This is because $K_P^{\lambda}[Gx]$ can be decomposed as $K_P^{\lambda}[Gx] = K_P^{\lambda}[Cx] \oplus K_P^{\lambda}[(G \setminus C)x]$. Now, every central Poisson derivation maps \bar{x} to an element of the subspace $K_P^{\lambda}[Cx]$, and every inner Poisson derivation maps \bar{x} to an element of the subspace $K_P^{\lambda}[(G \setminus C)x]$.

6.1.4 Corollary. Suppose that $C = \{e\}$ (equivalently, $Z_P(K_P^{\lambda}[G]) = K$). Then, every Poisson derivation of $K_P^{\lambda}[G]$ is uniquely the sum of an inner and a scalar Poisson derivation.

6.2 Preliminaries on the Poisson derivations of $A_{\alpha,\beta}$

Let $2 \leq j \leq 7$ and $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}$. Set

$$\mathcal{A}^{(j)}_{lpha,eta}:=rac{\mathcal{A}^{(j)}}{\langle\Omega_1-lpha,\Omega_2-eta
angle},$$

where $\mathcal{A}^{(j)}$ is defined in Subsection 5.2.1, and Ω_1 and Ω_2 are the generators of the center of $\mathcal{A}^{(j)}$ (Subsection 5.2.1). Recall that $\mathcal{A}^{(7)} = \mathcal{A} = \mathbb{C}[X_1, \cdots, X_6]$. It follows that $\mathcal{A}^{(7)}_{\alpha,\beta} = \mathcal{A}_{\alpha,\beta}$. For each $2 \leq j \leq 7$, denote the canonical images of the generators $X_{i,j}$ of $\mathcal{A}^{(j)}$ in $\mathcal{A}^{(j)}_{\alpha,\beta}$ by $x_{i,j}$ for all $1 \leq i \leq 6$. Since the data of the PDDA of $\mathcal{A}_{\alpha,\beta}$ is going to be useful in this section, we present them below (note, we deduce them from that of \mathcal{A} in Subsection 5.2.1):

$$\begin{aligned} x_{1,6} &= x_1 - \frac{1}{2} x_5 x_6^{-1} \\ x_{2,6} &= x_2 + \frac{3}{2} x_4 x_6^{-1} - 3 x_3 x_5 x_6^{-1} + x_5^3 x_6^{-2} \\ x_{3,6} &= x_3 - x_5^2 x_6^{-1} \\ x_{4,6} &= x_4 - \frac{2}{3} x_5^3 x_6^{-1} \\ x_{1,5} &= x_{1,6} - x_{3,6} x_{5,6}^{-1} + \frac{3}{4} x_{4,6} x_{5,6}^{-2} \\ x_{2,5} &= x_{2,6} - 3 x_{3,6}^2 x_{5,6}^{-1} + \frac{9}{2} x_{3,6} x_{4,6} x_{5,6}^{-2} - \frac{9}{4} x_{4,6}^2 x_{5,6}^{-3} \\ x_{3,5} &= x_{3,6} - \frac{3}{2} x_{4,6} x_{5,6}^{-1} \end{aligned}$$

$$\begin{aligned} x_{1,4} &= x_{1,5} - \frac{1}{3} x_{3,5}^2 x_{4,5}^{-1} \\ x_{2,4} &= x_{2,5} - \frac{2}{3} x_{3,5}^3 x_{4,5}^{-1} \\ x_{1,3} &= x_{1,4} - \frac{1}{2} x_{2,4} x_{3,4}^{-1} \\ t_1 &:= x_{1,2} = x_{1,3} \\ t_2 &:= x_{2,2} = x_{2,3} = x_{2,4} \\ t_3 &:= x_{3,2} = x_{3,3} = x_{3,4} = x_{3,5} \\ t_4 &:= x_{4,2} = x_{4,3} = x_{4,4} = x_{4,5} = x_{4,6} \\ t_5 &:= x_{5,2} = x_{5,3} = x_{5,4} = x_{5,5} = x_{5,6} = x_5 \\ t_6 &:= x_{6,2} = x_{6,3} = x_{6,4} = x_{6,5} = x_{6,6} = x_6 \end{aligned}$$

Note, the t_i are the canonical images of T_i in $\mathcal{A}_{\alpha,\beta}^{(2)}$ for all $1 \leq i \leq 6$. For each $2 \leq j < 7$, define $S_j := \{\lambda t_j^{i_j} t_{j+1}^{i_{j+1}} \cdots t_6^{i_6} \mid i_j, \cdots, i_6 \in \mathbb{N}, \ \lambda \in \mathbb{C}^*\}$. One can observe that S_j is a multiplicative system of non-zero divisors (or regular elements) of $\mathcal{A}_{\alpha,\beta}^{(j)}$. As a result, one can localize $\mathcal{A}_{\alpha,\beta}^{(j)}$ at S_j . Let us denote this localization by \mathcal{R}_j . That is,

$$\mathcal{R}_j := \mathcal{A}_{\alpha,\beta}^{(j)} S_j^{-1}.$$

Again, set $\Sigma_j := \{t_j^k \mid k \in \mathbb{N}\}$, with $2 \leq j \leq 6$. Then, Σ_j is a multiplicative set in both $\mathcal{A}_{\alpha,\beta}^{(j)}$ and $\mathcal{A}_{\alpha,\beta}^{(j+1)}$. Therefore,

$$\mathcal{A}_{\alpha,\beta}^{(j)}\Sigma_j^{-1} = \mathcal{A}_{\alpha,\beta}^{(j+1)}\Sigma_j^{-1}.$$

One can also verify that (similar to (2.2.1)):

$$\mathcal{R}_j = \mathcal{R}_{j+1} \Sigma_j^{-1}, \tag{6.2.1}$$

for all $2 \leq j \leq 6$, with the convention that $\mathcal{R}_7 := \mathcal{A}_{\alpha,\beta}$.

Similar to (4.1.2), we have the following tower of algebras:

$$\mathcal{R}_7 = \mathcal{A}_{\alpha,\beta} \subset \mathcal{R}_6 = \mathcal{R}_7 \Sigma_6^{-1} \subset \mathcal{R}_5 = \mathcal{R}_6 \Sigma_5^{-1} \subset \mathcal{R}_4 = \mathcal{R}_5 \Sigma_4^{-1} \subset \mathcal{R}_3.$$
(6.2.2)

Observe that $\mathcal{R}_3 = \mathcal{A}_{\alpha,\beta}^{(3)} S_3^{-1} = \mathcal{R}_4 \Sigma_3^{-1}$ is the Poisson torus $\mathscr{P}_{\alpha,\beta} = \mathbb{C}[t_3^{\pm 1}, t_4^{\pm 1}, t_5^{\pm 1}, t_6^{\pm 1}]$ in Section 5.3.

Linear bases for $\mathcal{R}_3, \mathcal{R}_4$ and \mathcal{R}_5 .

We aim to find a basis for \mathcal{R}_j for each j = 3, 4, 5. Since \mathcal{R}_3 is a Poisson torus generated by $t_3^{\pm 1}, \dots, t_6^{\pm 1}$ over \mathbb{C} , the set $\{t_3^i t_4^j t_5^k t_6^l \mid i, j, k, l \in \mathbb{Z}\}$ is a basis of \mathcal{R}_3 .

For simplicity, we set

$$f_{1} := x_{1,4} F_{1} := X_{1,4}$$

$$z_{1} := x_{1,5} Z_{1} := X_{1,5}$$

$$z_{2} := x_{2,5} Z_{2} := X_{2,5}.$$

Basis for \mathcal{R}_4 . Observe that

$$\mathcal{A}_{\alpha,\beta}^{(4)} = \frac{\mathcal{A}^{(4)}}{\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle},$$

where $\Omega_1 = F_1 T_3 T_5 - \frac{1}{2} T_2 T_5$ and $\Omega_2 = T_2 T_4 T_6$ in $\mathcal{A}^{(4)}$ (Subsection 5.2.1). Set

$$\mathcal{A}_{\beta}^{(4)}S_4^{-1} := \frac{\mathcal{A}^{(4)}S_4^{-1}}{\langle \Omega_2 - \beta \rangle},$$

where $\beta \in \mathbb{C}$. We will denote the canonical images of $X_{i,4}$ (resp. T_i) in $\mathcal{A}_{\beta}^{(4)}$ by $\widehat{x_{i,4}}$ (resp. $\widehat{t_i}$) for all $1 \leq i \leq 6$. Observe that $\widehat{t_2} = \beta \widehat{t_6}^{-1} \widehat{t_4}^{-1}$ in $\mathcal{A}_{\beta}^{(4)} S_4^{-1}$. As usual, one can identify \mathcal{R}_4 with $\mathcal{A}_{\beta}^{(4)} S_4^{-1} / \langle \Omega_1 - \alpha \rangle$.

The proofs of the following two propositions are similar to that of Propositions 4.1.2 and 4.1.4, nonetheless, we will still prove the results.

6.2.1 Proposition. The set $\mathfrak{P}_4 = \{f_1^{i_1}t_4^{i_4}t_5^{i_5}t_6^{i_6}, t_3^{i_3}t_4^{i_4}t_5^{i_5}t_6^{i_6} \mid (i_1, i_3, i_4, i_5, i_6) \in \mathbb{N}^2 \times \mathbb{Z}^3\}$ is a \mathbb{C} -basis of \mathcal{R}_4 .

Proof. One can verify that $(\hat{f}_1^{k_1}\hat{t}_3^{k_3}\hat{t}_4^{k_4}\hat{t}_5^{k_5}\hat{t}_6^{k_6})_{(k_1,k_3,\cdots,k_6)\in\mathbb{N}^2\times\mathbb{Z}^3}$ is a basis of $\mathcal{A}_{\beta}^{(4)}S_4^{-1}$ (the proof is similar to that of Proposition 4.1.1). Since $\mathcal{A}_{\beta}^{(4)}S_4^{-1} = \mathcal{A}^{(4)}S_4^{-1}/\langle\Omega_2 - \beta\rangle$, the family $(f_1^{k_1}t_3^{k_3}t_4^{k_4}t_5^{k_5}t_6^{k_6})_{(k_1,k_3,\cdots,k_6)\in\mathbb{N}^2\times\mathbb{Z}^3}$ spans \mathcal{R}_4 . We show that \mathfrak{P}_4 is a spanning set of \mathcal{R}_4 by showing that $f_1^{k_1}t_3^{k_3}t_4^{k_4}t_5^{k_5}t_6^{k_6}$ can be written as a finite linear combination of the elements of \mathfrak{P}_4 for all $(k_1,k_3,\cdots,k_6)\in\mathbb{N}^2\times\mathbb{Z}^3$. It is sufficient to do this by an induction on k_1 . The result is clear when $k_1 = 0$. Assume that the statement is true for $k_1 \ge 0$. That is,

$$f_1^{k_1} t_3^{k_3} t_4^{k_4} t_5^{k_5} t_6^{k_6} = \sum_{\underline{i} \in I_1} a_{\underline{i}} f_1^{i_1} t_4^{i_4} t_5^{i_5} t_6^{i_6} + \sum_{\underline{j} \in I_2} b_{\underline{j}} t_3^{i_3} t_4^{i_4} t_5^{i_5} t_6^{i_6},$$

where $\underline{i} = (i_1, i_4, i_5, i_6) \in I_1 \subset \mathbb{N} \times \mathbb{Z}^3$ and $\underline{j} = (i_3, i_4, i_5, i_6) \in I_2 \subset \mathbb{N} \times \mathbb{Z}^3$. Note, $a_{\underline{i}}$ and b_j are all complex numbers.

$$f_1^{k_1+1}t_3^{k_3}t_4^{k_4}t_5^{k_5}t_6^{k_6} = f_1(f_1^{k_1}t_3^{k_3}t_4^{k_4}t_5^{k_5}t_6^{k_6}) = \sum_{\underline{i}\in I_1}a_{\underline{i}}f_1^{i_1+1}t_4^{i_4}t_5^{i_5}t_6^{i_6} + \sum_{\underline{j}\in I_2}b_{\underline{j}}f_1t_3^{i_3}t_4^{i_4}t_5^{i_5}t_6^{i_6}.$$

Clearly, the monomial $f_1^{i_1+1}t_4^{i_4}t_5^{i_5}t_6^{i_6} \in \text{Span}(\mathfrak{P}_4)$. We have to also show that $f_1t_3^{i_3}t_4^{i_4}t_5^{i_5}t_6^{i_6} \in \text{Span}(\mathfrak{P}_4)$ for all $i_3 \in \mathbb{N}$ and $i_4, i_5, i_6 \in \mathbb{Z}$. This can easily be achieved by induction on i_3 , and the use of the relation $f_1t_3 = \alpha t_5^{-1} + \frac{1}{2}\beta t_4^{-1}t_6^{-1}$. Therefore, by the principle of mathematical induction, \mathfrak{P}_4 is a spanning set of \mathcal{R}_4 over \mathbb{C} .

We prove that \mathfrak{P}_4 is a linearly independent set. Suppose that

$$\sum_{\underline{i}\in I_1} a_{\underline{i}} f_1^{i_1} t_4^{i_4} t_5^{i_5} t_6^{i_6} + \sum_{\underline{j}\in I_2} b_{\underline{j}} t_3^{i_3} t_4^{i_4} t_5^{i_5} t_6^{i_6} = 0.$$

It follows that there exists $\nu\in \mathcal{A}_{\beta}^{(4)}S_{4}^{-1}$ such that

$$\sum_{\underline{i}\in I_1} a_{\underline{i}} \widehat{f_1}^{i_1} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} + \sum_{\underline{j}\in I_2} b_{\underline{j}} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} = \left(\widehat{\Omega}_1 - \alpha\right) \nu.$$

Write $\nu = \sum_{\underline{l}\in J} c_{\underline{l}} \widehat{f_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6}$, with $\underline{l} = (i_1, i_3, i_4, i_5, i_6) \in J \subset \mathbb{N}^2 \times \mathbb{Z}^3$, and $c_{\underline{l}} \in \mathbb{C}$. One can easily deduce that $\widehat{\Omega}_1 = \widehat{f_1} \widehat{t_3} \widehat{t_5} - \frac{1}{2} \widehat{t_2} \widehat{t_5} = \widehat{f_1} \widehat{t_3} \widehat{t_5} - \frac{1}{2} \beta \widehat{t_6}^{-1} \widehat{t_4}^{-1} \widehat{t_5}$ (note, $\widehat{t_2} = \beta \widehat{t_6}^{-1} \widehat{t_4}^{-1}$). It follows that

$$\begin{split} \sum_{\underline{i}\in I_1} a_{\underline{i}} \widehat{f_1}^{i_1} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} + \sum_{\underline{j}\in I_2} b_{\underline{j}} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} &= \sum_{\underline{l}\in J} c_{\underline{l}} \widehat{f_1}^{i_1+1} \widehat{t_3}^{i_3+1} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5+1} \widehat{t_6}^{i_6} \\ &- \sum_{\underline{l}\in J} \frac{1}{2} \beta c_{\underline{l}} \widehat{f_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4-1} \widehat{t_5}^{i_5+1} \widehat{t_6}^{i_6-1} \\ &- \sum_{\underline{l}\in J} \alpha c_{\underline{l}} \widehat{f_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6}. \end{split}$$

Suppose that there exists $(i_1, i_3, i_4, i_5, i_6) \in J$ such that $c_{(i_1, i_3, i_4, i_5, i_6)} \neq 0$.

Let $(w_1, w_3, w_4, w_5, w_6) \in J$ be the greatest element (in the lexicographic order on $\mathbb{N}^2 \times \mathbb{Z}^3$) of J such that $c_{(w_1, w_3, w_4, w_5, w_6)} \neq 0$. Since $\left(\widehat{f_1}^{k_1} \widehat{t_3}^{k_3} \widehat{t_4}^{k_4} \widehat{t_5}^{k_5} \widehat{t_6}^{k_6}\right)_{(k_1, k_3, \cdots, k_6) \in \mathbb{N}^2 \times \mathbb{Z}^3}$ is a basis of $\mathcal{A}^{(4)} S_4^{-1}$, it implies that the coefficients of $\widehat{f_1}^{w_1+1} \widehat{t_3}^{w_3+1} \widehat{t_4}^{w_4} \widehat{t_5}^{w_5+1} \widehat{t_6}^{w_6}$ in the above equality can be identified as: $c_{(w_1, w_3, w_4, w_5, w_6)} = 0$. Hence, $c_{(w_1, w_3, w_4, w_5, w_6)} = 0$, a contradiction! Therefore, $c_{(i_1, i_3, i_4, i_5, i_6)} = 0$ for all $(i_1, i_3, i_4, i_5, i_6) \in J$. This further implies that

$$\sum_{\underline{i}\in I_1} a_{\underline{i}} \widehat{f_1}^{i_1} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} + \sum_{\underline{j}\in I_2} b_{\underline{j}} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} = 0.$$

Consequently, $a_{\underline{i}}$ and b_j are all zero. In conclusion, \mathfrak{P}_4 is a linearly independent set.

Basis for \mathcal{R}_5 . We will identify \mathcal{R}_5 with $\mathcal{A}_{\alpha}^{(5)}S_5^{-1}/\langle \widehat{\Omega}_2 - \beta \rangle$, where $\mathcal{A}_{\alpha}^{(5)}S_5^{-1} = \mathcal{A}^{(5)}S_5^{-1}/\langle \Omega_1 - \alpha \rangle$. Note, the canonical images of $X_{i,5}$ (resp. T_i) in $\mathcal{A}_{\alpha}^{(5)}$ will be denoted by $\widehat{x_{i,5}}$ (resp. $\widehat{t_i}$) for all $1 \leq i \leq 6$. We now find a basis for $\mathcal{A}_{\alpha}^{(5)}S_5^{-1}$. Recall from Subsection 5.2.1 that $\Omega_1 = Z_1T_3T_5 - \frac{1}{2}Z_2T_5$ and $\Omega_2 = Z_2T_4T_6 - \frac{2}{3}T_3^3T_6$ in $\mathcal{A}^{(5)}$ (remember, $Z_1 := X_{1,5}$ and $Z_2 := X_{2,5}$). Since $z_2t_4t_6 - \frac{2}{3}t_3^3t_6 = \beta$ and $\widehat{z_1}\widehat{t_3}\widehat{t_5} - \frac{1}{2}\widehat{z_2}\widehat{t_5} = \alpha$ in \mathcal{R}_5 and $\mathcal{A}_{\alpha}^{(5)}S_5^{-1}$ respectively, we have the relation $\widehat{z_2} = 2\left(\widehat{z_1}\widehat{t_3} - \alpha\widehat{t_5}^{-1}\right)$ in $\mathcal{A}_{\alpha}^{(5)}S_5^{-1}$ and, in R_5 , we have the following two relations:

$$z_2 = 2(z_1 t_3 - \alpha t_5^{-1}). \tag{6.2.3}$$

$$t_3^3 = \frac{3}{2}(z_2t_4 - \beta t_6^{-1}) = 3z_1t_3t_4 - \frac{3}{2}\beta t_6^{-1} - 3\alpha t_4t_5^{-1}.$$
 (6.2.4)

6.2.2 Proposition. The set $\mathfrak{P}_5 = \left\{ z_1^{i_1} t_3^{\xi} t_4^{i_4} t_5^{i_5} t_6^{i_6} \mid (\xi, i_1, i_4, i_5, i_6) \in \{0, 1, 2\} \times \mathbb{N}^2 \times \mathbb{Z}^2 \right\}$ is a basis of \mathcal{R}_5 .

Proof. One can easily show that the family $\left(\widehat{z_1}^{k_1}\widehat{t_3}^{k_3}\widehat{t_4}^{k_4}\widehat{t_5}^{k_5}\widehat{t_6}^{k_6}\right)_{(k_1,k_3,k_4,k_5,k_6)\in\mathbb{N}^2\times\mathbb{Z}^2}$ is a basis of $\mathcal{A}_{\alpha}^{(5)}S_5^{-1}/\langle\widehat{\Omega}_2 - \beta\rangle$ (the proof is similar to that of Proposition 4.1.3). Since \mathcal{R}_5 is identified with $\mathcal{A}_{\alpha}^{(5)}S_5^{-1}/\langle\widehat{\Omega}_2 - \beta\rangle$, we show that $z_1^{k_1}t_3^{k_3}t_4^{k_4}t_5^{k_5}t_6^{k_6}$ can be written as a finite linear combination of the elements of \mathfrak{P}_5 for all $(k_1, k_3, k_4, k_5, k_6) \in \mathbb{N}^3 \times \mathbb{Z}^2$. It is sufficient to do this by an induction on k_3 . The result is obvious when $k_3 = 0, 1$ or 2. For $k_3 \geq 2$, suppose that

$$z_1^{k_1} t_3^{k_3} t_4^{k_4} t_5^{k_5} t_6^{k_6} = \sum_{(\xi,\underline{i})\in I} a_{(\xi,\underline{i})} z_1^{i_1} t_3^{\xi} t_4^{i_4} t_5^{i_5} t_6^{i_6},$$

where I is a finite subset of $\{0, 1, 2\} \times \mathbb{N}^2 \times \mathbb{Z}^2$, and $a_{(\xi,\underline{i})}$ are all complex numbers. It follows that

$$z_1^{k_1}t_3^{k_3+1}t_4^{k_4}t_5^{k_5}t_6^{k_6} = \left(z_1^{k_1}t_3^{k_3}t_4^{k_4}t_5^{k_5}t_6^{k_6}\right)t_3 = \sum_{(\xi,\underline{i})\in I} a_{(\xi,\underline{i})}z_1^{i_1}t_3^{\xi+1}t_4^{i_4}t_5^{i_5}t_6^{i_6}.$$

Now, $z_1^{i_1}t_3^{\xi+1}t_4^{i_4}t_5^{i_5}t_6^{i_6} \in \text{Span}(\mathfrak{P}_5)$ when $\xi = 0, 1$. For $\xi = 2$, one can easily verify that $z_1^{i_1}t_3^3t_4^{i_4}t_5^{i_5}t_6^{i_6} \in \text{Span}(\mathfrak{P}_5)$ by using the relation in (6.2.4). Therefore, by the principle of mathematical induction, \mathfrak{P}_5 spans \mathcal{R}_5 .

We now prove that \mathfrak{P}_5 is a linearly independent set. Suppose that

$$\sum_{(\xi,\underline{i})\in I} a_{(\xi,\underline{i})} z_1^{i_1} t_3^{\xi} t_4^{i_4} t_5^{i_5} t_6^{i_6} = 0.$$

Since \mathcal{R}_5 is identified with $\mathcal{A}_{\alpha}^{(5)}S_5^{-1}/\langle \widehat{\Omega}_2 - \beta \rangle$, we have that

$$\sum_{(\xi,\underline{i})\in I} a_{(\xi,\underline{i})} \widehat{z_1}^{i_1} \widehat{t_3}^{\xi} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} = \langle \widehat{\Omega}_2 - \beta \rangle \nu,$$

where $\nu \in \mathcal{A}_{\alpha}^{(5)}S_5^{-1}$. Write $\nu = \sum_{\underline{j}\in J} b_{\underline{j}}\widehat{z_1}^{i_1}\widehat{t_3}^{i_3}\widehat{t_4}^{i_4}\widehat{t_5}^{i_5}\widehat{t_6}^{i_6}$, with $\underline{j} = (i_1, i_3, i_4, i_5, i_6) \in J \subset \mathbb{N}^3 \times \mathbb{Z}^2$ and $b_{\underline{j}} \in \mathbb{C}$. Given that $\Omega_2 = Z_2 T_4 T_6 - \frac{2}{3}T_3^3 T_6$ in $\mathcal{A}^{(5)}$ and the relation (6.2.3), one can deduce that

$$\widehat{\Omega}_2 = \widehat{z_2}\widehat{t_4}\widehat{t_6} - \frac{2}{3}\widehat{t_3}^3\widehat{t_6} = 2\widehat{z_1}\widehat{t_3}\widehat{t_4}\widehat{t_6} - 2\alpha\widehat{t_4}\widehat{t_5}^{-1}\widehat{t_6} - \frac{2}{3}\widehat{t_3}^3\widehat{t_6}.$$

Therefore,

$$\begin{split} \sum_{(\xi,\underline{i})\in I} a_{(\xi,\underline{i})} \widehat{z_1}^{i_1} \widehat{t_3}^{\xi} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} &= \sum_{\underline{j}\in J} 2b_{\underline{j}} \widehat{z_1}^{i_1+1} \widehat{t_3}^{i_3+1} \widehat{t_4}^{i_4+1} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6+1} \\ &- \sum_{\underline{j}\in J} \frac{2}{3} b_{\underline{j}} \widehat{z_1}^{i_1} \widehat{t_3}^{i_3+3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6+1} \\ &- \sum_{\underline{j}\in J} 2\alpha b_{\underline{j}} \widehat{z_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4+1} \widehat{t_5}^{i_5-1} \widehat{t_6}^{i_6+1} \\ &- \sum_{\underline{j}\in J} \beta b_{\underline{j}} \widehat{z_1}^{i_1} \widehat{t_3}^{i_3} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6}. \end{split}$$

Suppose that there exists $(i_1, i_3, i_4, i_5, i_6) \in J$ such that $b_{(i_1, i_3, i_4, i_5, i_6)} \neq 0$.

Let $(w_1, w_3, w_4, w_5, w_6) \in J$ be the greatest element (in the lexicographic order on $\mathbb{N}^3 \times \mathbb{Z}^2$) of J such that $b_{(w_1, w_3, w_4, w_5, w_6)} \neq 0$. Given that $\left(\widehat{z_1}^{k_1}\widehat{t_3}^{k_3}\widehat{t_4}^{k_4}\widehat{t_5}^{k_5}\widehat{t_6}^{k_6}\right)_{(k_1, k_3, \cdots, k_6) \in \mathbb{N}^3 \times \mathbb{Z}^2}$ is a basis of $\mathcal{A}_{\alpha}^{(5)}S_5^{-1}$, one can identify the coefficients of $\widehat{z_1}^{w_1+1}\widehat{t_3}^{w_3+1}\widehat{t_4}^{w_4+1}\widehat{t_5}^{w_5}\widehat{t_6}^{w_6+1}$ in the above equality as: $2b_{(w_1, w_3, w_4, w_5, w_6)} = 0$. Hence, $b_{(w_1, w_3, w_4, w_5, w_6)} = 0$, a contradiction! Therefore, $b_{(i_1, i_3, i_4, i_5, i_6)} = 0$ for all $(i_1, i_3, i_4, i_5, i_6) \in J$. Consequently,

$$\sum_{(\xi,\underline{i})\in I} a_{(\xi,\underline{i})} \widehat{z_1}^{i_1} \widehat{t_3}^{\xi} \widehat{t_4}^{i_4} \widehat{t_5}^{i_5} \widehat{t_6}^{i_6} = 0.$$

It follows that $a_{(\xi,\underline{i})} = 0$ for all $(\xi,\underline{i}) \in I$. As a result, \mathfrak{P}_5 is a linearly independent set.

6.2.3 Corollary. Let I be a finite subset of $\{0, 1, 2\} \times \mathbb{N} \times \mathbb{Z}^3$ and $(a_{(\xi,\underline{i})})_{\underline{i} \in I}$ be a family of complex numbers. If

$$\sum_{(\xi,\underline{i})\in I} a_{(\xi,\underline{i})} z_1^{i_1} t_3^{\xi} t_4^{i_4} t_5^{i_5} t_6^{i_6} = 0,$$

then $a_{(\xi,\underline{i})} = 0$ for all $(\xi,\underline{i}) \in I$.

Proof. When $i_4 \ge 0$, then the result is obvious as a result of Proposition 6.2.2. For $i_4 < 0$, multiply both sides of the equality enough times by t_4 to kill all the negative powers of t_4 , and then apply Proposition 6.2.2 to complete the proof.

6.2.4 Remark. We were not successful in finding a basis for \mathcal{R}_6 . However, this has no effect on our main result in this chapter. Since $\mathcal{R}_7 = \mathcal{A}_{\alpha,\beta}$, we already have a basis for \mathcal{R}_7 (Proposition 5.3.4).

6.2.5 Lemma. $Z_P(\mathcal{R}_i) = \mathbb{C}$ for each $3 \leq i \leq 7$.

Proof. Similar to that of Lemma 4.1.7.

6.2.6 Remark. Recall the notations:

$f_1 := x_{1,4}$	$F_1 := X_{1,4}$
$z_1 := x_{1,5}$	$Z_1 := X_{1,5}$
$z_2 := x_{2,5}$	$Z_2 := X_{2,5}.$

Let $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}$. Given the above notations, we present the following selected data of the PDDA of $\mathcal{A}_{\alpha,\beta}$, listed at the beginning of this section, in a manner that would be very useful in Subsections 6.3.1. They are as follows:

$$f_{1} = t_{1} + \frac{1}{2}t_{2}t_{3}^{-1} \qquad \qquad x_{3,6} = t_{3} + \frac{3}{2}t_{4}t_{5}^{-1}$$
$$z_{1} = f_{1} + \frac{1}{3}t_{3}^{2}t_{4}^{-1} \qquad \qquad x_{1} = x_{1,6} + \frac{1}{2}t_{5}t_{6}^{-1}$$

$$z_{2} = t_{2} + \frac{2}{3}t_{3}^{3}t_{4}^{-1} \qquad \qquad x_{3} = x_{3,6} + t_{5}^{2}t_{6}^{-1}$$
$$x_{1,6} = z_{1} + x_{3,6}t_{5}^{-1} - \frac{3}{4}t_{4}t_{5}^{-2} \qquad \qquad x_{4} = t_{4} + \frac{2}{3}t_{5}^{3}t_{6}^{-1}.$$

The remark below will also be helpful in Subsection 6.3.1.

6.2.7 Remark. Fix $n \in \mathbb{N}$. Let $R_s = \mathbb{K}\langle u_1, \cdots, u_s^{\pm 1}, u_{s+1}^{\pm 1}, \cdots, u_n^{\pm 1} \rangle$ be a commutative algebra generated by the elements $u_1, \cdots, u_s^{\pm 1}, u_{s+1}^{\pm 1}, \cdots, u_n^{\pm 1}$ over the field \mathbb{K} , with $1 \leq s \leq n+1$ (note, $R_{n+1} := \mathbb{K}\langle u_1, \cdots, u_n \rangle$). Then, for all $f, g \in R_s$, we have that

$$\{f,g\} = \sum_{i,j=1}^{n} \{u_i, u_j\} \frac{\partial f}{\partial u_i} \frac{\partial g}{\partial u_j}.$$

As a result,

$$\{u_1^{i_1}\cdots u_n^{i_n}, u_j\} = i_1 u_1^{i_1-1} u_2^{i_2} \cdots u_n^{i_n} \{u_1, u_j\} + i_2 u_1^{i_1} u_2^{i_2-1} u_3^{i_3} \cdots u_n^{i_n} \{u_2, u_j\}$$

+ \dots + i_n u_1^{i_1} \dots u_{n-1}^{i_{n-1}} u_n^{i_n-1} \{u_n, u_j\},

for all $(i_1, \dots, i_n) \in \mathbb{Z}^n$. Note, $\{u_i, u_j\} = 0$ whenever i = j.

6.3 Poisson derivations of $A_{\alpha,\beta}$

We are now going to study the Poisson derivations of $\mathcal{A}_{\alpha,\beta}$. We will begin with the case where both α and β are non-zero, and then proceed to look at the case where either α or β is zero.

6.3.1 Poisson derivations of $\mathcal{A}_{\alpha,\beta}$ $(\alpha, \beta \neq 0)$. Throughout this subsection, we assume that α and β are non-zero. Let $\text{Der}_P(\mathcal{A})$ be the collection of all the Poisson derivations of $\mathcal{A}_{\alpha,\beta}$ and $\mathcal{D} \in \text{Der}_P(\mathcal{A})$. Then, it follows from (6.2.1) that \mathcal{D} extends uniquely to a Poisson derivation of each of the series of algebras in (6.2.2) via localization. Hence, \mathcal{D} is a Poisson derivation of the Poisson torus $\mathcal{R}_3 = \mathbb{C}[t_3^{\pm 1}, t_4^{\pm 1}, t_5^{\pm 1}, t_6^{\pm 1}]$. As a result, \mathcal{D} can be written as

$$\mathcal{D} = \mathsf{ham}_x + \rho,$$

where ρ is a scalar Poisson derivation of \mathcal{R}_3 defined as $\rho(t_i) = \lambda_i t_i$, i = 3, 4, 5, 6. Note, $\lambda_i \in Z_P(\mathcal{R}_3) = \mathbb{C}$ and $ham_x := \{x, -\} : \mathcal{R}_3 \to \mathcal{R}_3$ with $x \in \mathcal{R}_3$ (see Corollary 6.1.4).

We aim to describe \mathcal{D} as a Poisson derivation of $\mathcal{A}_{\alpha,\beta}$. We do this in several steps. We first describe \mathcal{D} as a Poisson derivation of \mathcal{R}_4 .

6.3.2 Lemma. 1. $x \in \mathcal{R}_4$.

- 2. $\lambda_5 = \lambda_4 + \lambda_6$, $\rho(f_1) = -(\lambda_3 + \lambda_5)f_1$ and $\rho(t_2) = -\lambda_5 t_2$.
- 3. $\mathcal{D}(x_{\kappa,4}) = \max_x(x_{\kappa,4}) + \lambda_{\kappa}x_{\kappa,4}$ for all $\kappa \in \{1, \dots, 6\}$, where $\lambda_1 := -(\lambda_3 + \lambda_5)$ and $\lambda_2 := -\lambda_5$.

Proof. 1. Observe that $Q := \mathbb{C}[t_4^{\pm 1}, t_5^{\pm 1}, t_6^{\pm 1}]$ is a subalgebra of both \mathcal{R}_3 and \mathcal{R}_4 . Furthermore, $\mathcal{R}_3 = \mathcal{R}_4[t_3^{-1}]$. One can easily verify that $z := t_4 t_5^{-1} t_6$ is a Poisson central element of Q. Since \mathcal{R}_3 is a Poisson torus, it can be presented as a free Q-module with basis $(t_3^j)_{j\in\mathbb{Z}}$. One can therefore write $x \in \mathcal{R}_3$ as: $x = \sum_{j\in\mathbb{Z}} b_j t_3^j$, where $b_j \in Q$. Decompose x as follows: $x = x_- + x_+$, where $x_+ := \sum_{j\geq 0} b_j t_3^j$ and $x_- := \sum_{j<0} b_j t_3^j$. Clearly, $x_+ \in \mathcal{R}_4$. We now want to show that $x_- \in \mathcal{R}_4$. Write $x_- = \sum_{j=-1}^{-m} b_j t_3^j$ for some $m \in \mathbb{N}_{>0}$.

Now, $\mathcal{D}(z) = \operatorname{ham}_{x}(z) + \rho(z) = \operatorname{ham}_{x_{-}}(z) + \operatorname{ham}_{x_{+}}(z) + (\lambda_{4} - \lambda_{5} + \lambda_{6})z \in \mathcal{R}_{4}$. We have that $\operatorname{ham}_{x_{+}}(z) + (\lambda_{4} - \lambda_{5} + \lambda_{6})z \in \mathcal{R}_{4}$, hence, $\operatorname{ham}_{x_{-}}(z) \in \mathcal{R}_{4}$. Note: $\{t_{3}, z\} = 2zt_{3}$, and $\{\gamma, z\} = 0$ for all $\gamma \in \mathcal{Q}$ since z is Poisson central in \mathcal{Q} . One can therefore express $\operatorname{ham}_{x_{-}}(z)$ as follows:

$$\mathsf{ham}_{x_{-}}(z) = \{x_{-}, z\} = \sum_{j=-1}^{-m} b_{j}\{t_{3}^{j}, z\} = \sum_{j=-1}^{-m} 2jb_{j}zt_{3}^{j} \in \mathcal{R}_{4}$$

Let $n \in \mathbb{N}_{>0}$, and set

$$\ell^{(n)} := \underbrace{\{\{\cdots \{ x_{-}, \underline{z}\}, z\}, \cdots, z\}}_{n-\text{times}} \in \mathcal{R}_4.$$

We claim that

$$\ell^{(n)} = \sum_{j=-1}^{-m} (2j)^n z^n b_j t_3^j,$$

for all $n \in \mathbb{N}_{>0}$. Observe that

$$\ell^{(1)} = \mathsf{ham}_{x_{-}}(z) = \sum_{j=-1}^{-m} 2jb_j zt_3^j,$$

hence, the result is true for n=1. Suppose that the result is true for $n\geq 1.$ Then,

$$\ell^{(n+1)} = \{\ell^{(n)}, z\} = \sum_{j=-1}^{-m} (2j)^n z^n b_j \{t_3^j, z\} = \sum_{j=-1}^{-m} (2j)^{n+1} z^{n+1} b_j t_3^j$$

as expected. By the principle of mathematical induction, the claim is proved.

Given that $\ell^{(n)} = \sum_{j=-1}^{-m} (2j)^n z^n b_j t_3^j,$ it follows that

$$\mu_n = \sum_{j=-1}^{-m} (2j)^n b_j t_3^j, \text{ where } \mu_n := \ell^{(n)} z^{-n} \in \mathcal{R}_4.$$

The above equality can be written as a matrix equation in the form:

$$\begin{bmatrix} -2 & -4 & -6 & \cdots & -2m \\ (-2)^2 & (-4)^2 & (-6)^2 & \cdots & (-2m)^2 \\ (-2)^3 & (-4)^3 & (-6)^3 & \cdots & (-2m)^3 \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ (-2)^m & (-4)^m & (-6)^m & \cdots & (-2m)^m \end{bmatrix} \begin{bmatrix} b_{-1}t_3^{-1} \\ b_{-2}t_3^{-2} \\ b_{-3}t_3^{-3} \\ \vdots \\ b_{-m}t_3^{-m} \end{bmatrix} = \begin{bmatrix} \mu_1 \\ \mu_2 \\ \mu_3 \\ \vdots \\ \mu_m \end{bmatrix}$$

One can observe that the coefficient matrix

$$\begin{bmatrix} -2 & -4 & -6 & \cdots & -2m \\ (-2)^2 & (-4)^2 & (-6)^2 & \cdots & (-2m)^2 \\ (-2)^3 & (-4)^3 & (-6)^3 & \cdots & (-2m)^3 \\ \vdots & \vdots & \vdots & \ddots & \vdots \\ (-2)^m & (-4)^m & (-6)^m & \cdots & (-2m)^m \end{bmatrix}$$

is similar to a Vandermonde matrix (since the terms in each column form a geometric sequence) which is well known to be invertible. This therefore implies that each $b_j t_3^j$ is a linear combination of the $\mu_n \in \mathcal{R}_4$. As a result, $b_j t_3^j \in \mathcal{R}_4$ for all $j \in \{-1, \dots, -m\}$. Consequently, $x_- = \sum_{j=-1}^{-m} b_j t_3^j \in \mathcal{R}_4$ as desired.

2. Recall that $\rho(t_{\kappa}) = \lambda_{\kappa} t_{\kappa}$ for all $\kappa \in \{3, 4, 5, 6\}$ and $\lambda_{\kappa} \in \mathbb{C}$. From Remark 6.2.6, we have that $f_1 = t_1 + \frac{1}{2} t_2 t_3^{-1}$. Again, recall from Section 5.3 that $t_1 = \alpha t_3^{-1} t_5^{-1}$ and $t_2 = \beta t_4^{-1} t_6^{-1}$ in $\mathcal{R}_3 = \mathscr{P}_{\alpha,\beta}$. As a result, $f_1 = \alpha t_3^{-1} t_5^{-1} + \frac{1}{2} \beta t_3^{-1} t_4^{-1} t_6^{-1}$. Therefore,

$$\rho(f_1) = -(\lambda_3 + \lambda_5)\alpha t_3^{-1} t_5^{-1} - \frac{1}{2}(\lambda_3 + \lambda_4 + \lambda_6)\beta t_3^{-1} t_4^{-1} t_6^{-1}.$$
 (6.3.1)

Also, $\rho(f_1) \in \mathcal{R}_4$ implies that $\rho(f_1)$ can be written in terms of the basis \mathfrak{P}_4 of \mathcal{R}_4 (Proposition 6.2.1) as:

$$\rho(f_1) = \sum_{r>0} a_r f_1^r + \sum_{s\ge 0} b_s t_3^s, \tag{6.3.2}$$

where a_r and b_s belong to $\mathcal{Q}=\mathbb{C}[t_4^{\pm 1},t_5^{\pm 1},t_6^{\pm 1}].$

$$f_1^r = \left(\alpha t_3^{-1} t_5^{-1} + \frac{1}{2}\beta t_3^{-1} t_4^{-1} t_6^{-1}\right)^r = \sum_{i=0}^r \binom{r}{i} (\alpha)^i (\beta/2)^{r-1} t_3^{-r} t_4^{i-r} t_5^{-i} t_6^{i-r}$$
$$= c_r t_3^{-r}, \tag{6.3.3}$$

where

$$c_r = \sum_{i=0}^r \binom{r}{i} (\alpha)^i (\beta/2)^{r-i} t_4^{i-r} t_5^{-i} t_6^{i-r} \in \mathcal{Q} \setminus \{0\}.$$
 (6.3.4)

Substitute (6.3.3) into (6.3.2) to obtain

$$\rho(f_1) = \sum_{r>0} a_r c_r t_3^{-r} + \sum_{s\ge 0} b_s t_3^s.$$
(6.3.5)

One can rewrite (6.3.1) as

$$\rho(f_1) = dt_3^{-1},\tag{6.3.6}$$

where $d = -(\lambda_5 + \lambda_3)\alpha t_5^{-1} - \frac{1}{2}(\lambda_6 + \lambda_4 + \lambda_3)\beta t_4^{-1}t_6^{-1} \in \mathcal{Q}$. Comparing (6.3.5) to (6.3.6) shows that $b_s = 0$ for all $s \ge 0$, and $a_r c_r = 0$ for all $r \ne 1$. Therefore, $\rho(f_1) = a_1 c_1 t_3^{-1}$. Moreover, from (6.3.4), $c_1 = \frac{1}{2}\beta t_4^{-1}t_6^{-1} + \alpha t_5^{-1}$. Hence,

$$\rho(f_1) = a_1 c_1 t_3^{-1} = a_1 \left(\frac{1}{2} \beta t_4^{-1} t_6^{-1} + \alpha t_5^{-1} \right) t_3^{-1} = a_1 \alpha t_3^{-1} t_5^{-1} + \frac{1}{2} a_1 \beta t_3^{-1} t_4^{-1} t_6^{-1}.$$
 (6.3.7)

Comparing (6.3.7) to (6.3.1) reveals that $a_1 = -(\lambda_5 + \lambda_3) = -(\lambda_6 + \lambda_4 + \lambda_3)$. Consequently, $\lambda_5 = \lambda_6 + \lambda_4$. Hence, $\rho(f_1) = -(\lambda_5 + \lambda_3)\alpha t_3^{-1}t_5^{-1} - \frac{1}{2}(\lambda_5 + \lambda_3)\beta t_3^{-1}t_4^{-1}t_6^{-1} = -(\lambda_5 + \lambda_3)f_1$. Finally, since $t_2 = \beta t_4^{-1}t_6^{-1}$ in \mathcal{R}_4 , it follows that

$$\rho(t_2) = -(\lambda_6 + \lambda_4)\beta t_4^{-1} t_6^{-1} = -(\lambda_6 + \lambda_4)t_2 = -\lambda_5 t_2.$$

3. Set $\lambda_1 := -(\lambda_3 + \lambda_5)$ and $\lambda_2 := -\lambda_5$, it follows from points (1) and (2) that $\mathcal{D}(x_{\kappa,4}) = \operatorname{ham}_x(x_{\kappa,4}) + \rho(x_{\kappa,4}) = \operatorname{ham}_x(x_{\kappa,4}) + \lambda_{\kappa}x_{\kappa,4}$ for all $\kappa \in \{1, \dots, 6\}$. In conclusion, $\mathcal{D} = \operatorname{ham}_x + \rho$, with $x \in \mathcal{R}_4$.

We now proceed to describe \mathcal{D} as a Poisson derivation of \mathcal{R}_5 .

6.3.3 Lemma. 1. $x \in \mathcal{R}_5$.

2.
$$\lambda_4 = 3\lambda_3 + \lambda_5$$
, $\lambda_6 = -3\lambda_3$, $\rho(z_1) = -(\lambda_3 + \lambda_5)z_1$ and $\rho(z_2) = -\lambda_5 z_2$.

3. Set $\lambda_1 := -(\lambda_3 + \lambda_5)$ and $\lambda_2 := -\lambda_5$, then $\mathcal{D}(x_{\kappa,5}) = \operatorname{ham}_x(x_{\kappa,5}) + \lambda_{\kappa}x_{\kappa,5}$ for all $\kappa \in \{1, \dots, 6\}$.

Proof. In this proof, we denote $\underline{v} := (i, j, k, l) \in \mathbb{N} \times \mathbb{Z}^3$.

1. We already know that $x \in \mathcal{R}_4 = \mathcal{R}_5[t_4^{-1}]$. Given the basis \mathfrak{P}_5 of \mathcal{R}_5 (Proposition 6.2.2), x can be written as $x = \sum_{(\xi,\underline{v})\in I} a_{(\xi,\underline{v})} z_1^i t_3^{\xi} t_4^j t_5^k t_6^l$, where I is a finite subset of

 $\{0,1,2\} \times \mathbb{N} \times \mathbb{Z}^3$ and $a_{(\xi,\underline{v})}$ are complex numbers. Write $x = x_- + x_+$, where

$$x_+ = \sum_{\substack{(\xi,\underline{\upsilon})\in I\\j\geq 0}} a_{(\xi,\underline{\upsilon})} z_1^i t_3^\xi t_4^j t_5^k t_6^l \quad \text{and} \quad x_- = \sum_{\substack{(\xi,\underline{\upsilon})\in I\\j< 0}} a_{(\xi,\underline{\upsilon})} z_1^i t_3^\xi t_4^j t_5^k t_6^l$$

Suppose that there exists a minimum $j_0 < 0$ such that $a_{(\xi,i,j_0,k,l)} \neq 0$ for some $(\xi, i, j_0, k, l) \in I$ and $a_{(\xi,i,j,k,l)} = 0$ for all $(\xi, i, j_0, k, l) \in I$ with $j < j_0$. Given this assumption, write

$$x_- = \sum_{\substack{(\xi,\underline{\upsilon}) \in I\\ j_0 \leq j \leq -1}} a_{(\xi,\underline{\upsilon})} z_1^i t_3^\xi t_4^j t_5^k t_6^l.$$

Let s = 3, 6. Then, $\mathcal{D}(t_s) = \operatorname{ham}_{x_+}(t_s) + \operatorname{ham}_{x_-}(t_s) + \rho(t_s) \in \mathcal{R}_5$ for each s = 3, 6. This implies that $\operatorname{ham}_{x_-}(t_s) \in \mathcal{R}_5$, since $\operatorname{ham}_{x_+}(t_s) + \rho(t_s) = \operatorname{ham}_{x_+}(t_s) + \lambda_s t_s \in \mathcal{R}_5$. Set $\underline{w} := (i, j, k, l) \in \mathbb{N}^2 \times \mathbb{Z}^2$. One can therefore write $\operatorname{ham}_{x_-}(t_s) \in \mathcal{R}_5$ in terms of the basis \mathfrak{P}_5 of \mathcal{R}_5 as:

$$ham_{x_{-}}(t_{s}) = \sum_{(\xi,\underline{w})\in J} b_{(\xi,\underline{w})} z_{1}^{i} t_{3}^{\xi} t_{4}^{j} t_{5}^{k} t_{6}^{l}, \qquad (6.3.8)$$

where J is a finite subset of $\{0, 1, 2\} \times \mathbb{N}^2 \times \mathbb{Z}^2$ and $b_{(\xi,\underline{w})}$ are all complex numbers.

When s = 6, then using Remark 6.2.7, one can also express ham_{x-}(t_6) as:

$$\mathsf{ham}_{x_{-}}(t_{6}) = \sum_{\substack{(\xi,\underline{\upsilon}) \in I \\ j_{0} \leq j \leq -1}} 3(k+j-i) a_{(\xi,\underline{\upsilon})} z_{1}^{i} t_{3}^{\xi} t_{4}^{j} t_{5}^{k} t_{6}^{l+1} + 1 \leq j \leq j \leq -1} z_{1}^{j} z_{1}^{\xi} z_{1}^{j} z_{5}^{k} z_{6}^{j} z_{6}^{k} z_{6}^{l+1} + 1 \leq j \leq j \leq -1} z_{1}^{j} z$$

Comparing this expression for $ham_{x_{-}}(t_6)$ to (6.3.8) (when s = 6), we have that

$$\sum_{\substack{(\xi,\underline{\upsilon})\in I\\j_0\leq j\leq -1}} 3(k+j-i)a_{(\xi,\underline{\upsilon})}z_1^i t_3^\xi t_4^j t_5^k t_6^{l+1} = \sum_{(\xi,\underline{w})\in J} b_{(\xi,\underline{w})}z_1^i t_3^\xi t_4^j t_5^k t_6^l$$

As \mathfrak{P}_5 is a basis for \mathcal{R}_5 (Proposition 6.2.2), we deduce from Corollary 6.2.3 that $\left(z_1^i t_3^{\xi} t_4^j t_5^k t_6^l\right)_{(i \in \mathbb{N}; j, k, l \in \mathbb{Z}; \xi \in \{0, 1, 2\})}$ is a basis for $\mathcal{R}_5[t_4^{-1}]$. Now, at $j = j_0$, denote $\underline{v} = (i, j, k, l)$ by $\underline{v}_0 := (i, j_0, k, l)$. Since $\underline{v}_0 \in \mathbb{N} \times \mathbb{Z}^3$ (with $j_0 < 0$) and $\underline{w} = (i, j, k, l) \in \mathbb{Z}$ $\mathbb{N}^2 imes \mathbb{Z}^2$ (with $j \ge 0$), it follows from the above equality that, at \underline{v}_0 , we must have

$$k = i - j_0, \tag{6.3.9}$$

for some $(\xi, \underline{v}_0) \in I$.

Similarly, when s = 3, then using Remark 6.2.7, one can also express ham_{x-}(t₃) as:

$$\begin{split} & \mathsf{ham}_{x_{-}}(t_{3}) = -\sum \left[\frac{3}{2} \beta (3i - k - 3j_{0}) a_{2,i,j_{0},k,l+1} + 2(i+1) \alpha a_{(0,i+1,j_{0},k+1,l)} \right] z_{1}^{i} t_{4}^{j_{0}} t_{5}^{k} t_{6}^{l} \\ & \quad + \sum \left[(3i - k - 3j_{0}) a_{(0,i,j_{0},k,l)} - 2(i+1) \alpha a_{(1,i+1,j_{0},k+1,l)} \right] z_{1}^{i} t_{3} t_{4}^{j_{0}} t_{5}^{k} t_{6}^{l} \\ & \quad + \sum \left[(3i - k - 3j_{0}) a_{(1,i,j_{0},k,l)} - 2(i+1) \alpha a_{(2,i+1,j_{0},k+1,l)} \right] z_{1}^{i} t_{3}^{2} t_{4}^{j_{0}} t_{5}^{k} t_{6}^{l} + \mathcal{K}, \end{split}$$

where $\mathcal{K} \in \text{Span}\left(\mathfrak{P}_5 \setminus \{z_1^i t_3^{\xi} t_4^{j_0} t_5^k t_6^l \mid (\xi, i, j_0, k, l) \in \{0, 1, 2\} \times \mathbb{N} \times \mathbb{Z}^3\}\right)$ (note, one will need the following two expressions $z_2 = 2(z_1 t_3 - \alpha t_5^{-1})$ and $t_3^3 = 3z_1 t_3 t_4 - 3\alpha t_4 t_5^{-1} - \frac{3\beta}{2} t_6^{-1}$ from (6.2.3) and (6.2.4) to express some of the monomials in terms of the basis \mathfrak{P}_5 of \mathcal{R}_5). Comparing this expression for $\max_{x_-}(t_3)$ to (6.3.8) (when s = 3) reveals that:

$$\begin{split} \sum_{(\xi,\underline{w})\in J} b_{(\xi,\underline{w})} z_1^i t_3^{\xi} t_4^j t_5^k t_6^l \\ &= -\sum \left[\frac{3}{2} \beta (3i - k - 3j_0) a_{(2,i,j_0,k,l+1} + 2(i+1) \alpha a_{(0,i+1,j_0,k+1,l)} \right] z_1^i t_4^{j_0} t_5^k t_6^l \\ &+ \sum \left[(3i - k - 3j_0) a_{(0,i,j_0,k,l)} - 2(i+1) \alpha a_{(1,i+1,j_0,k+1,l)} \right] z_1^i t_3 t_4^{j_0} t_5^k t_6^l \\ &+ \sum \left[(3i - k - 3j_0) a_{(1,i,j_0,k,l)} - 2(i+1) \alpha a_{(2,i+1,j_0,k+1,l)} \right] z_1^i t_3^2 t_4^{j_0} t_5^k t_6^l + \mathcal{K}. \end{split}$$

We have already established that $(z_1^i t_3^{\xi} t_4^j t_5^k t_6^l)_{(i \in \mathbb{N}; j, k, l \in \mathbb{Z}; \xi \in \{0, 1, 2\})}$ is a basis for $\mathcal{R}_5[t_4^{-1}]$. Since $\underline{v}_0 = (i, j_0, k, l) \in \mathbb{N} \times \mathbb{Z}^3$ (with $j_0 < 0$) and $\underline{w} = (i, j, k, l) \in \mathbb{N}^2 \times \mathbb{Z}^2$ (with $j \geq 0$), it follows from the above equality that, at \underline{v}_0 , we must have

$$\frac{3}{2}\beta(3i-k-3j_0)a_{(2,i,j_0,k,l+1)} + 2(i+1)\alpha a_{(0,i+1,j_0,k+1,l)} = 0,$$
(6.3.10)

$$(3i - k - 3j_0)a_{(0,i,j_0,k,l)} - 2(i+1)\alpha a_{(1,i+1,j_0,k+1,l)} = 0,$$
(6.3.11)

$$(3i - k - 3j_0)a_{(1,i,j_0,k,l)} - 2(i+1)\alpha a_{(2,i+1,j_0,k+1,l)} = 0.$$
 (6.3.12)

Suppose that there exists $(\xi, i, j_0, k, l) \in I$ such that $3i - k - 3j_0 = 0$. Then,

$$k = 3(i - j_0). \tag{6.3.13}$$

Comparing (6.3.13) to (6.3.9) clearly shows that $i - j_0 = 0$ which implies that $i = j_0 < 0$, a contradiction (note: $i \ge 0$). Therefore, $3i - k - 3j_0 \ne 0$ for all $(\xi, i, j, k) \in I$.

Now, observe that if there exists $\xi \in \{0, 1, 2\}$ such that $a_{(\xi,i,j_0,k,l)} = 0$ for all $(i, j_0, k, l) \in \mathbb{N} \times \mathbb{Z}^3$, then one can easily deduce from equations (6.3.10), (6.3.11) and (6.3.12) that $a_{(\xi,i,j_0,k,l)} = 0$ for all $(\xi, i, j_0, k, l) \in I$. This contradicts our initial assumption. Therefore, there exists some $(i, j_0, k, l) \in \mathbb{N} \times \mathbb{Z}^3$ such that $a_{(\xi,i,j_0,k,l)} \neq 0$ for each $\xi \in \{0, 1, 2\}$. Without loss of generality, let (u, j_0, v, w) be the greatest element in the lexicographic order on $\mathbb{N} \times \mathbb{Z}^3$ such that $a_{(0,u,j_0,v,w)} \neq 0$ and $a_{(0,i,j_0,k,l)} = 0$ for all i > u.

From (6.3.11), at $(i, j_0, k, l) = (u, j_0, v, w)$, we have:

$$(3u - v - 3j_0)a_{(0,u,j_0,v,w)} - 2(u+1)\alpha a_{(1,u+1,j_0,v+1,w)} = 0.$$

From (6.3.12), at $(i, j_0, k, l) = (u + 1, j_0, v + 1, w)$, we have:

$$(3u - v - 3j_0)a_{(1,u+1,j_0,v+1,w)} - 2(u+1)\alpha a_{(2,u+2,j_0,v+2,w)} = 0.$$

Finally, from (6.3.10), at $(i, j_0, k, l) = (u + 2, j_0, v + 2, w - 1)$, we have:

$$\frac{3}{2}\beta(3u - v - 3j_0)a_{(2,u+2,j_0,v+2,w)} + 2(u+1)\alpha a_{(0,u+3,j_0,v+3,w-1)} = 0$$

Since $3i - k - 3j_0 \neq 0$ for all $(i, j_0, k, l) \in I$ and u + 3 > u, it follows from the above list of equations (starting from the last one) that

$$a_{(0,u+3,j_0,v+3,w-1)} = 0 \Rightarrow a_{(2,u+2,j_0,v+2,w)} = 0 \Rightarrow a_{(1,u+1,j_0,v+1,w)} = 0 \Rightarrow a_{(0,u,j_0,v,w)} = 0,$$

a contradiction! Hence, $a_{(0,i,j_0,k,l)} = 0$ for all $(i, j_0, k, l) \in \mathbb{N} \times \mathbb{Z}^3$. From (6.3.10), (6.3.11) and (6.3.12), one can easily conclude that $a_{(\xi,i,j_0,k,l)} = 0$ for all $(\xi, i, j_0, k, l) \in I$. This contradicts our initial assumption, hence $x_- = 0$. Consequently, $x = x_+ \in \mathcal{R}_5$ as desired.

2. From Remark 6.2.6, we have that $z_2 = t_2 + \frac{2}{3}t_3^3t_4^{-1}$. Since $\rho(t_{\kappa}) = \lambda_{\kappa}t_{\kappa}$, $\kappa \in \{2, 3, 4, 5, 6\}$, with $\lambda_2 := -\lambda_5$ (see Lemma 6.3.2), it follows that

$$\rho(z_2) = -\lambda_5 t_2 + \frac{2}{3} (3\lambda_3 - \lambda_4) t_3^3 t_4^{-1} = -\lambda_5 z_2 + \frac{2}{3} (3\lambda_3 - \lambda_4 + \lambda_5) t_3^3 t_4^{-1}.$$

Furthermore,

$$\mathcal{D}(z_2) = \mathsf{ham}_x(z_2) + \rho(z_2) = \mathsf{ham}_x(z_2) - \lambda_5 z_2 + \frac{2}{3}(3\lambda_3 - \lambda_4 + \lambda_5)t_3^3 t_4^{-1} \in \mathcal{R}_5.$$

We have that $(3\lambda_3 - \lambda_4 + \lambda_5)t_3^3t_4^{-1} \in \mathcal{R}_5$, since $ham_x(z_2) - \lambda_5 z_2 \in \mathcal{R}_5$. This implies that $(3\lambda_3 - \lambda_4 + \lambda_5)t_3^3 \in \mathcal{R}_5 t_4$. Set $w := 3\lambda_3 - \lambda_4 + \lambda_5$. Suppose that $w \neq 0$. From (6.2.4), we have:

$$t_3^3 = 3z_1t_3t_4 - \frac{3}{2}\beta t_6^{-1} - 3\alpha t_4t_5^{-1}.$$

It follows that

$$wt_3^3 = 3wz_1t_3t_4 - 3w\alpha t_4t_5^{-1} - \frac{3}{2}w\beta t_6^{-1} \in \mathcal{R}_5t_4.$$

Since t_3^3 , $t_4t_5^{-1}$ and $z_1t_3t_4$ are all elements of \mathcal{R}_5t_4 , it implies that $t_6^{-1} \in \mathcal{R}_5t_4$. Hence, $1 \in \mathcal{R}_5t_4t_6$. Using the basis \mathfrak{P}_5 of \mathcal{R}_5 (Proposition 6.2.2), this leads to a contradiction. Therefore, w = 0. That is, $w = 3\lambda_3 - \lambda_4 + \lambda_5 = 0$, and so $\lambda_4 = 3\lambda_3 + \lambda_5$. This further implies that $\rho(z_2) = -\lambda_5 z_2$ as desired.

Again, from Lemma 6.3.2, we have that $\rho(f_1) = -(\lambda_3 + \lambda_5)f_1$. Recall from Remark 6.2.6 that $z_1 = f_1 + \frac{1}{3}t_3^2t_4^{-1}$. It follows that

$$\rho(z_1) = -(\lambda_3 + \lambda_5)f_1 + \frac{1}{3}(2\lambda_3 - \lambda_4)t_3^2t_4^{-1} = -(\lambda_3 + \lambda_5)z_1 + \frac{1}{3}(3\lambda_3 - \lambda_4 + \lambda_5)t_3^2t_4^{-1} = -(\lambda_3 + \lambda_5)z_1 + \frac{1}{3}(3\lambda_3 - (3\lambda_3 + \lambda_5) + \lambda_5)t_3^2t_4^{-1} = -(\lambda_3 + \lambda_5)z_1.$$

Finally, we know that $\rho(t_6) = \lambda_6 t_6$. From the relation (6.2.4), we have:

$$t_3^3 = 3z_1t_3t_4 - 3\alpha t_4t_5^{-1} - \frac{3\beta}{2}t_6^{-1}.$$

This implies that

$$t_6^{-1} = \frac{2}{3\beta} (3z_1 t_3 t_4 - 3\alpha t_4 t_5^{-1} - t_3^3).$$

Apply ρ to this relation to obtain

$$-\lambda_6 t_6^{-1} = 3\lambda_3 \left(\frac{2}{3\beta} \left(3z_1 t_3 t_4 - 3\alpha t_4 t_5^{-1} - t_3^3 \right) \right).$$

Clearly, $\lambda_6 = -3\lambda_3$ as desired.

3. Set $\lambda_1 := -(\lambda_3 + \lambda_5)$ and $\lambda_2 := -\lambda_5$. It follows from points (1) and (2) that $\mathcal{D}(x_{\kappa,5}) = \operatorname{ham}_x(x_{\kappa,5}) + \rho(x_{\kappa,5}) = \operatorname{ham}_x(x_{\kappa,5}) + \lambda_{\kappa}x_{\kappa,5}$ for all $\kappa \in \{1, \dots, 6\}$. In conclusion, $\mathcal{D} = \operatorname{ham}_x + \rho$ with $x \in \mathcal{R}_5$.

We are now ready to describe \mathcal{D} as a Poisson derivation of $\mathcal{A}_{\alpha,\beta}$.

6.3.4 Lemma. 1. $x \in \mathcal{A}_{\alpha,\beta}$.

- 2. $\rho(x_{\kappa}) = 0$ for all $\kappa \in \{1, \dots, 6\}$.
- 3. $\mathcal{D} = ham_x$.

Proof. In this proof, we denote $\underline{v} := (i, j, k, l) \in \mathbb{N}^2 \times \mathbb{Z}^2$. Also, recall from the PDDA of $\mathcal{A}_{\alpha,\beta}$ at the beginning of this section that $t_5 = x_5$ and $t_6 = x_6$.

1. Given the basis \mathfrak{P} of $\mathcal{A}_{\alpha,\beta}$ (Proposition 5.3.4), one can write $x \in \mathcal{R}_5 = \mathcal{A}_{\alpha,\beta}[t_5^{-1},t_6^{-1}]$ as

$$x = \sum_{(\epsilon_1, \epsilon_2, \underline{\upsilon}) \in I} a_{(\epsilon_1, \epsilon_2, \underline{\upsilon})} x_1^i x_2^j x_3^{\epsilon_1} x_4^{\epsilon_2} t_5^k t_6^l,$$

where I is a finite subset of $\{0,1\}^2 \times \mathbb{N}^2 \times \mathbb{Z}^2$ and $a_{(\epsilon_1,\epsilon_2,\underline{v})}$ are complex numbers. Write $x = x_- + x_+$, where

$$x_{+} = \sum_{\substack{(\epsilon_{1},\epsilon_{2},\underline{\upsilon})\in I\\k,\ l\geq 0}} a_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon})} x_{1}^{i} x_{2}^{j} x_{3}^{\epsilon_{1}} x_{4}^{\epsilon_{2}} t_{5}^{k} t_{6}^{l}$$

and

$$x_{-} = \sum_{\substack{(\epsilon_1, \epsilon_2, \underline{\upsilon}) \in I \\ k < 0 \text{ or } l < 0}} a_{(\epsilon_1, \epsilon_2, \underline{\upsilon})} x_1^i x_2^j x_3^{\epsilon_1} x_4^{\epsilon_2} t_5^k t_6^l.$$

Suppose that there exists a minimum negative integer k_0 or l_0 such that $a_{(\epsilon_1,\epsilon_2,i,j,k_0,l)} \neq 0$ or $a_{(\epsilon_1,\epsilon_2,i,j,k,l_0)} \neq 0$ for some $(\epsilon_1, \epsilon_2, i, j, k_0, l), (\epsilon_1, \epsilon_2, i, j, k, l_0) \in I$; and $a_{(\epsilon_1,\epsilon_2,i,j,k,l)} = 0$ whenever $k < k_0$ or $l < l_0$. Write

$$x_{-} = \sum_{\substack{(\epsilon_1, \epsilon_2, \underline{\upsilon}) \in I \\ k_0 \le k \le -1 \text{ or } l_0 \le l \le -1}} a_{(\epsilon_1, \epsilon_2, \underline{\upsilon})} x_1^i x_2^j x_3^{\epsilon_1} x_4^{\epsilon_2} t_5^k t_6^l,$$

Now, $\mathcal{D}(x_3) = \max_{x_+}(x_3) + \max_{x_-}(x_3) + \rho(x_3) \in \mathcal{A}_{\alpha,\beta}$. From Remark 6.2.6, we have that $x_3 = x_{3,6} + t_5^2 t_6^{-1}$ and $x_{3,6} = t_3 + \frac{3}{2} t_4 t_5^{-1}$. Putting these two together gives

$$x_3 = t_3 + \frac{3}{2}t_4t_5^{-1} + t_5^2t_6^{-1}.$$

Again, from Remark 6.2.6, we also have that $t_4 = x_4 - \frac{2}{3}t_5^3t_6^{-1}$. Note, $\rho(t_{\kappa}) = \lambda_{\kappa}t_{\kappa}$, $\kappa = 3, 4, 5, 6$.

Now,

$$\rho(x_3) = \lambda_3 t_3 + \frac{3}{2} (\lambda_4 - \lambda_5) t_4 t_5^{-1} + (2\lambda_5 - \lambda_6) t_5^2 t_6^{-1}
= \lambda_3 \left(x_{3,6} - \frac{3}{2} t_4 t_5^{-1} \right) + \frac{3}{2} (\lambda_4 - \lambda_5) t_4 t_5^{-1} + (2\lambda_5 - \lambda_6) t_5^2 t_6^{-1}
= \lambda_3 x_{3,6} - \frac{3}{2} (\lambda_3 - \lambda_4 + \lambda_5) t_4 t_5^{-1} + (2\lambda_5 - \lambda_6) t_5^2 t_6^{-1}
= \lambda_3 (x_3 - t_5^2 t_6^{-1}) - \frac{3}{2} (\lambda_3 - \lambda_4 + \lambda_5) \left(x_4 - \frac{2}{3} t_5^3 t_6^{-1} \right) t_5^{-1} + (2\lambda_5 - \lambda_6) t_5^2 t_6^{-1}
= \lambda_3 x_3 + \alpha_1 x_4 t_5^{-1} + \alpha_2 t_5^2 t_6^{-1},$$
(6.3.14)

where $\alpha_1 = \frac{3}{2}(\lambda_4 - \lambda_3 - \lambda_5)$ and $\alpha_2 = (3\lambda_5 - \lambda_4 - \lambda_6)$. Therefore, $\mathcal{D}(x_3) = \max_{x_+}(x_3) + \max_{x_-}(x_3) + \lambda_3 x_3 + \alpha_1 x_4 t_5^{-1} + \alpha_2 t_5^2 t_6^{-1} \in \mathcal{A}_{\alpha,\beta}$. It follows that $\mathcal{D}(x_3) t_5 t_6 = \max_{x_+}(x_3) t_5 t_6 + \max_{x_-}(x_3) t_5 t_6 + \lambda_3 x_3 t_5 t_6 + \alpha_1 x_4 t_6 + \alpha_2 t_5^3 \in \mathcal{A}_{\alpha,\beta}$. Hence, $\max_{x_-}(x_3) t_5 t_6 \in \mathcal{A}_{\alpha,\beta}$, since $\max_{x_+}(x_3) t_5 t_6 + \lambda_3 x_3 t_5 t_6 + \alpha_1 x_4 t_6 + \alpha_2 t_5^3 \in \mathcal{A}_{\alpha,\beta}$.

Using Remark 6.2.7, one can verify that

$$\begin{aligned} \mathsf{ham}_{x_{-}}(x_{3})t_{5}t_{6} &= \sum_{(\epsilon_{1},\epsilon_{2},\underline{v})\in I} a_{(\epsilon_{1},\epsilon_{2},\underline{v})} \left((i+3j-3\epsilon_{2}-k)x_{1}^{i}x_{2}^{j}x_{3}^{\epsilon_{1}+1}x_{4}^{\epsilon_{2}}t_{5}^{k+1}t_{6}^{l+1} \right. \\ &\left. - 3kx_{1}^{i}x_{2}^{j}x_{3}^{\epsilon_{1}}x_{4}^{\epsilon_{2}+1}t_{5}^{k}t_{6}^{l+1} + ix_{1}^{i-1}x_{2}^{j+1}x_{3}^{\epsilon_{1}}x_{4}^{\epsilon_{2}}t_{5}^{k+1}t_{6}^{l+1} \right. \\ &\left. - 6lx_{1}^{i}x_{2}^{j}x_{3}^{\epsilon_{1}}x_{4}^{\epsilon_{2}}t_{5}^{k+3}t_{6}^{l} \right). \end{aligned}$$

$$(6.3.15)$$

Assume that there exists l < 0 such that $a_{(\epsilon_1,\epsilon_2,i,j,k,l)} \neq 0$. It follows from our initial assumption that $a_{(\epsilon_1,\epsilon_2,i,j,k,l_0)} \neq 0$. Now, at $l = l_0$, denote $\underline{v} = (i, j, k, l)$ by $\underline{v}_0 := (i, j, k, l_0)$. From (6.3.15), we have that

$$\mathsf{ham}_{x_{-}}(x_{3})t_{5}t_{6} = -\sum_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon}_{0})\in I} 6l_{0}a_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon}_{0})}x_{1}^{i}x_{2}^{j}x_{3}^{\epsilon_{1}}x_{4}^{\epsilon_{2}}t_{5}^{k+3}t_{6}^{l_{0}} + \mathcal{J}_{1},$$

where $\mathcal{J}_1 \in \text{Span}\left(\mathfrak{P} \setminus \{x_1^i x_2^j x_3^{\epsilon_1} x_4^{\epsilon_2} t_5^k t_6^{l_0} \mid \epsilon_1, \epsilon_2 \in \{0, 1\}, \ k \in \mathbb{Z} \text{ and } i, j \in \mathbb{N}\}\right)$. Set $\underline{w} := (i, j, k, l) \in \mathbb{N}^4$. One can also write $\max_{x_-}(x_3) t_5 t_6 \in \mathcal{A}_{\alpha,\beta}$ in terms of the basis \mathfrak{P} of $\mathcal{A}_{\alpha,\beta}$ (Proposition 5.3.4) as:

$$\mathsf{ham}_{x_{-}}(x_{3})t_{5}t_{6} = \sum_{(\epsilon_{1},\epsilon_{2},\underline{w})\in J} b_{(\epsilon_{1},\epsilon_{2},\underline{w})}x_{1}^{i}x_{2}^{j}x_{3}^{\epsilon_{1}}x_{4}^{\epsilon_{2}}t_{5}^{k}t_{6}^{l},$$
(6.3.16)

where J is a finite subset of $\{0,1\}^2 \times \mathbb{N}^4$ and $b_{(\epsilon_1,\epsilon_2,\underline{w})} \in \mathbb{C}$. It follows that

$$\sum_{(\epsilon_1,\epsilon_2,\underline{w})\in J} b_{(\epsilon_1,\epsilon_2,\underline{w})} x_1^i x_2^j x_3^{\epsilon_1} x_4^{\epsilon_2} t_5^k t_6^l = -\sum_{(\epsilon_1,\epsilon_2,\underline{w}_0)\in I} 6l_0 a_{(\epsilon_1,\epsilon_2,\underline{w}_0)} x_1^i x_2^j x_3^{\epsilon_1} x_4^{\epsilon_2} t_5^{k+3} t_6^{l_0} + \mathcal{J}_1$$

As \mathfrak{P} is a basis for $\mathcal{A}_{\alpha,\beta}$, we deduce from Corollary 5.3.5 that

 $(x_1^i x_2^j x_3^{\epsilon_1} x_4^{\epsilon_2} t_5^k t_6^l)_{((\epsilon_1, \epsilon_2, \underline{v}) \in \{0, 1\}^2 \times \mathbb{N}^2 \times \mathbb{Z}^2)} \text{ is a basis for } \mathcal{A}_{\alpha, \beta}[t_5^{-1}, t_6^{-1}]. \text{ Since } \underline{v}_0 = (i, j, k, l_0) \in \mathbb{N}^2 \times \mathbb{Z}^2 \text{ (with } l_0 < 0) \text{ and } \underline{w} = (i, j, k, l) \in \mathbb{N}^4 \text{ (with } l \ge 0) \text{ in the above equality, we must have}$

$$6l_0 a_{(\epsilon_1, \epsilon_2, \underline{v}_0)} = 0.$$

Note, $l_0 \neq 0$, it follows that $a_{(\epsilon_1, \epsilon_2, \underline{v}_0)} = a_{(\epsilon_1, \epsilon_2, i, j, k, l_0)}$ are all zero. This is a contradiction. Therefore, $l \geq 0$ (i.e. there is no negative exponent for t_6).

Given that $l \ge 0$, it follows from our initial assumption that there exists $k = k_0 < 0$ such that $a_{(\epsilon_1, \epsilon_2, i, j, k_0, l)} \ne 0$. The rest of the proof will show that this assumption cannot also hold. Set $\underline{v}_0 := (i, j, k_0, l) \in \mathbb{N}^2 \times \mathbb{Z} \times \mathbb{N}$. From (6.3.15), we have that

$$\mathsf{ham}_{x_{-}}(x_{3})t_{5}t_{6} = -\sum_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon}_{0})\in I} 3ka_{(\epsilon_{1},\epsilon_{2},\underline{\upsilon}_{0})}x_{1}^{i}x_{2}^{j}x_{3}^{\epsilon_{1}}x_{4}^{\epsilon_{2}+1}t_{5}^{k_{0}}t_{6}^{l+1} + V_{5}x_{6}^{k_{0}}x_{6}^{l+1} + V_{5}x_{6}^{l+1} + V_{$$

where $V \in \mathcal{J}_2 := \operatorname{Span}\left(\mathfrak{P} \setminus \{x_1^i x_2^j x_3^{\epsilon_1} x_4^{\epsilon_2} t_5^{k_0} t_6^l \mid \epsilon_1, \epsilon_2 \in \{0, 1\} \text{ and } i, j, l \in \mathbb{N}\}\right)$. It follows that:
$$\begin{aligned} \mathsf{ham}_{x_{-}}(x_{3})t_{5}t_{6} &= \\ &-\sum_{(\epsilon_{1},\epsilon_{2},\underline{v})\in I} 3k_{0}a_{(0,0,\underline{v})}x_{1}^{i}x_{2}^{j}x_{4}t_{5}^{k_{0}}t_{6}^{l+1} - \sum_{(\epsilon_{1},\epsilon_{2},\underline{v})\in I} 3k_{0}a_{(1,0,\underline{v})}x_{1}^{i}x_{2}^{j}x_{3}x_{4}t_{5}^{k_{0}}t_{6}^{l+1} \\ &-\sum_{(\epsilon_{1},\epsilon_{2},\underline{v})\in I} 3k_{0}a_{(0,1,\underline{v})}x_{1}^{i}x_{2}^{j}x_{4}^{2}t_{5}^{k_{0}}t_{6}^{l+1} - \sum_{(\epsilon_{1},\epsilon_{2},\underline{v})\in I} 3k_{0}a_{(1,1,\underline{v})}x_{1}^{i}x_{2}^{j}x_{3}x_{4}^{2}t_{5}^{k_{0}}t_{6}^{l+1} + V. \end{aligned}$$

$$(6.3.17)$$

Write the relations in Lemma 5.3.1(2),(4) as follows:

$$x_4^2 = \frac{2}{3}\beta - \frac{2}{3}x_2x_4x_6 + \frac{8}{9}\alpha x_3x_6 + \frac{4}{3}x_1x_3x_4x_6 + L_1,$$
(6.3.18)

$$x_{3}x_{4}^{2} = \frac{2}{3}\beta x_{3} - \frac{2}{3}x_{2}x_{3}x_{4}x_{6} + \frac{16}{9}\alpha^{2}x_{6} + \frac{16}{3}\alpha x_{1}x_{4}x_{6} + \frac{8}{3}\beta x_{1}^{2}x_{6} - \frac{8}{3}x_{1}^{2}x_{2}x_{4}x_{6}^{2} + \frac{32}{9}\alpha x_{1}^{2}x_{3}x_{6}^{2} + \frac{16}{3}x_{1}^{3}x_{3}x_{4}x_{6}^{2} + L_{2},$$
(6.3.19)

where L_1 and L_2 are some elements of the ideal $\mathcal{A}_{\alpha,\beta}t_5 \subseteq \mathcal{J}_2$. Substitute (6.3.18) and (6.3.19) into (6.3.17) and simplify to obtain:

$$\begin{aligned} \mathsf{ham}_{x_{-}}(e_{3})t_{5}t_{6} &= \sum [\lambda_{1,1}\beta a_{(0,1,i,j,k_{0},l-1)} + \lambda_{1,2}\alpha^{2}a_{(1,1,i,j,k_{0},l-2)} \\ &+ \lambda_{1,3}\beta a_{(1,1,i-2,j,k_{0},l-2)}]x_{1}^{i}x_{2}^{j}t_{5}^{k_{0}}t_{6}^{l} \\ &+ \sum [\lambda_{2,1}\alpha a_{(0,1,i,j,k_{0},l-2)} + \lambda_{2,2}\beta a_{(1,1,i,j,k_{0},l-1)} \\ &+ \lambda_{2,3}\alpha a_{(1,1,i-2,j,k_{0},l-3)}]x_{1}^{i}x_{2}^{j}x_{3}t_{5}^{k_{0}}t_{6}^{l} \\ &+ \sum [\lambda_{3,1}a_{(0,1,i,j-1,k_{0},l-2)} + \lambda_{3,2}\alpha a_{(1,1,i-1,j,k_{0},l-2)} \\ &+ \lambda_{3,3}a_{(1,1,i-2,j-1,k_{0},l-3)} + \lambda_{3,4}a_{(0,0,i,j,k_{0},l-1)}]x_{1}^{i}x_{2}^{j}x_{4}t_{5}^{k_{0}}t_{6}^{l} \\ &+ \sum [\lambda_{4,1}a_{(0,1,i-1,j,k_{0},l-2)} + \lambda_{4,2}a_{(1,1,i,j-1,k_{0},l-2)} \\ &+ \lambda_{4,3}a_{(1,1,i-3,j,k_{0},l-3)} + \lambda_{4,4}a_{(1,0,i,j,k_{0},l-1)}]x_{1}^{i}x_{2}^{j}x_{3}x_{4}t_{5}^{k_{0}}t_{6}^{l} + V', \end{aligned}$$

$$(6.3.20)$$

where $V' \in \mathcal{J}_2$. Also, $\lambda_{s,t} := \lambda_{s,t}(j, k_0, l)$ are some families of complex numbers which are non-zero for all $s, t \in \{1, 2, 3, 4\}$ and $j, l \in \mathbb{N}$, except $\lambda_{1,4}$ and $\lambda_{2,4}$ which are assumed to be zero since they do not exist in the above expression. Note, although each $\lambda_{s,t}$ depends on j, k_0, l , we have not made this dependency explicit in the above expression since the minimum requirement we need to complete the proof is for all the $\lambda_{s,t}$ existing in the above expression to be non-zero, which we already have.

Observe that (6.3.20) and (6.3.16) are equal, hence,

$$\begin{split} \sum_{(\epsilon_1,\epsilon_2,\underline{w})\in J} b_{(\epsilon_1,\epsilon_2,\underline{w})} x_1^i x_2^j x_3^{\epsilon_1} x_4^{\epsilon_2} t_5^k t_6^l &= \sum [\lambda_{1,1} \beta a_{(0,1,i,j,k_0,l-1)} + \lambda_{1,2} \alpha^2 a_{(1,1,i,j,k_0,l-2)} \\ &+ \lambda_{1,3} \beta a_{(1,1,i-2,j,k_0,l-2)}] x_1^i x_2^j t_5^{k_0} t_6^l \\ &+ \sum [\lambda_{2,1} \alpha a_{(0,1,i,j,k_0,l-2)} + \lambda_{2,2} \beta a_{(1,1,i,j,k_0,l-1)} \\ &+ \lambda_{2,3} \alpha a_{(1,1,i-2,j,k_0,l-3)}] x_1^i x_2^j x_3 t_5^{k_0} t_6^l \\ &+ \sum [\lambda_{3,1} a_{(0,1,i,j-1,k_0,l-2)} + \lambda_{3,2} \alpha a_{(1,1,i-1,j,k_0,l-2)} \\ &+ \lambda_{3,3} a_{(1,1,i-2,j-1,k_0,l-3)} + \lambda_{3,4} a_{(0,0,i,j,k_0,l-1)}] x_1^i x_2^j x_4 t_5^{k_0} t_6^l \\ &+ \sum [\lambda_{4,1} a_{(0,1,i-1,j,k_0,l-2)} + \lambda_{4,2} a_{(1,1,i,j-1,k_0,l-2)} \\ &+ \lambda_{4,3} a_{(1,1,i-3,j,k_0,l-3)} + \lambda_{4,4} a_{(1,0,i,j,k_0,l-1)}] x_1^i x_2^j x_3 x_4 t_5^{k_0} t_6^l + V'. \end{split}$$

We have previously established that $(x_1^i x_2^j x_3^{\epsilon_1} x_4^{\epsilon_2} t_5^k t_6^l)_{((\epsilon_1, \epsilon_2, \underline{v}) \in \{0,1\}^2 \times \mathbb{N}^2 \times \mathbb{Z}^2)}$ is a basis for $\mathcal{A}_{\alpha,\beta}[t_5^{-1}, t_6^{-1}]$ (note, in this part of the proof $l \ge 0$). Since $\underline{v}_0 = (i, j, k_0, l) \in \mathbb{N}^2 \times \mathbb{Z} \times \mathbb{N}$ (with $k_0 < 0$) and $\underline{w} = (i, j, k, l) \in \mathbb{N}^4$ (with $k \ge 0$) in the above equality, it follows that

$$\lambda_{1,1}\beta a_{(0,1,i,j,k_0,l-1)} + \lambda_{1,2}\alpha^2 a_{(1,1,i,j,k_0,l-2)} + \lambda_{1,3}\beta a_{(1,1,i-2,j,k_0,l-2)} = 0,$$
(6.3.21)

$$\lambda_{2,1}\alpha a_{(0,1,i,j,k_0,l-2)} + \lambda_{2,2}\beta a_{(1,1,i,j,k_0,l-1)} + \lambda_{2,3}\alpha a_{(1,1,i-2,j,k_0,l-3)} = 0,$$
(6.3.22)

 $\lambda_{3,1}a_{(0,1,i,j-1,k_0,l-2)} + \lambda_{3,2}\alpha a_{(1,1,i-1,j,k_0,l-2)} + \lambda_{3,3}a_{(1,1,i-2,j-1,k_0,l-3)}$

$$+\lambda_{3,4}a_{(0,0,i,j,k_0,l-1)} = 0, (6.3.23)$$

$$\lambda_{4,1}a_{(0,1,i-1,j,k_0,l-2)} + \lambda_{4,2}a_{(1,1,i,j-1,k_0,l-2)} + \lambda_{4,3}a_{(1,1,i-3,j,k_0,l-3)} + \lambda_{4,4}a_{(1,0,i,j,k_0,l-1)} = 0.$$
(6.3.24)

From (6.3.21) and (6.3.22), one can easily deduce that

$$a_{(0,1,i,j,k_0,l)} = -\frac{\alpha^2 \lambda_{1,2}}{\beta \lambda_{1,1}} a_{(1,1,i,j,k_0,l-1)} - \frac{\lambda_{1,3}}{\lambda_{1,1}} a_{(1,1,i-2,j,k_0,l-1)},$$
(6.3.25)

$$a_{(1,1,i,j,k_0,l)} = -\frac{\alpha\lambda_{2,1}}{\beta\lambda_{2,2}}a_{(0,1,i,j,k_0,l-1)} - \frac{\alpha\lambda_{2,3}}{\beta\lambda_{2,2}}a_{(1,1,i-2,j,k_0,l-2)}.$$
(6.3.26)

Note, $a_{(\epsilon_1,\epsilon_2,i,j,k_0,l)} := 0$ whenever i < 0 or j < 0 or l < 0 for all $\epsilon_1, \epsilon_2 \in \{0,1\}$.

Claim. The coefficients $a_{(0,1,i,j,k_0,l)}$ and $a_{(1,1,i,j,k_0,l)}$ are all zero for all $l \ge 0$. We now justify the claim by an induction on l. From (6.3.25) and (6.3.26), the result is obviously true when l = 0. For $l \ge 0$, assume that $a_{(0,1,i,j,k_0,l)} = a_{(1,1,i,j,k_0,l)} = 0$. Then, it follows from (6.3.25) and (6.3.26) that

$$a_{(0,1,i,j,k_0,l+1)} = -\frac{\alpha^2 \lambda_{1,2}}{\beta \lambda_{1,1}} a_{(1,1,i,j,k_0,l)} - \frac{\lambda_{1,3}}{\lambda_{1,1}} a_{(1,1,i-2,j,k_0,l)}$$

$$a_{(1,1,i,j,k_0,l+1)} = -\frac{\alpha\lambda_{2,1}}{\beta\lambda_{2,2}}a_{(0,1,i,j,k_0,l)} - \frac{\alpha\lambda_{2,3}}{\beta\lambda_{2,2}}a_{(1,1,i-2,j,k_0,l-1)}$$

From the inductive hypothesis, $a_{(1,1,i,j,k_0,l)} = a_{(1,1,i-2,j,k_0,l)} = a_{(0,1,i,j,k_0,l)} = a_{(1,1,i-2,j,k_0,l-1)}$ = 0. Hence, $a_{(1,1,i,j,k_0,l+1)} = a_{(0,1,i,j,k_0,l+1)} = 0$. By the principle of mathematical induction, $a_{(0,1,i,j,k_0,l)} = a_{(1,1,i,j,k_0,l)} = 0$ for all $l \ge 0$ as desired. Given that the families $a_{(0,1,i,j,k_0,l)}$ and $a_{(1,1,i,j,k_0,l)}$ are all zero, it follows from (6.3.23) and (6.3.24) that $a_{(0,0,i,j,k_0,l)}$ and $a_{(1,0,i,j,k_0,l)}$ are also zero for all $(i, j, k_0, l) \in \mathbb{N}^2 \times \mathbb{Z} \times \mathbb{N}$. Since $a_{(\epsilon_1,\epsilon_2,i,j,k_0,l)}$ are all zero, it contradicts our assumption. Hence, $x_- = 0$. Consequently, $x = x_+ \in \mathcal{A}_{\alpha,\beta}$ as desired.

2. From Remark 6.2.6, we have that $x_4 = x_{4,6} + \frac{2}{3}t_5^3t_6^{-1} = t_4 + \frac{2}{3}t_5^3t_6^{-1}$. Again, from Lemma 6.3.4, we have that $\lambda_4 = 3\lambda_3 + \lambda_5$ and $\lambda_6 = -3\lambda_3$. Therefore,

$$\rho(x_4) = \lambda_4 t_4 + \frac{2}{3} (3\lambda_5 - \lambda_6) t_5^3 t_6^{-1}$$

= $(3\lambda_3 + \lambda_5) x_{4,6} + 2(\lambda_3 + \lambda_5) t_5^3 t_6^{-1}$
= $(3\lambda_3 + \lambda_5) \left(x_4 - \frac{2}{3} t_5^3 t_6^{-1} \right) + 2(\lambda_3 + \lambda_5) t_5^3 t_6^{-1}$
= $(3\lambda_3 + \lambda_5) x_4 + \frac{4}{3} \lambda_5 t_5^3 t_6^{-1}$.

Hence,

$$\mathcal{D}(x_4) = \mathsf{ham}_x(x_4) + \rho(x_4) = \mathsf{ham}_x(x_4) + (3\lambda_3 + \lambda_5)x_4 + \frac{4}{3}\lambda_5 t_5^3 t_6^{-1} \in \mathcal{A}_{\alpha,\beta}.$$

It follows that $\lambda_5 t_5^3 t_6^{-1} \in \mathcal{A}_{\alpha,\beta}$, since $\operatorname{ham}_x(x_4) + (3\lambda_3 + \lambda_5)x_4 \in \mathcal{A}_{\alpha,\beta}$. Consequently, $\lambda_5 t_5^3 \in \mathcal{A}_{\alpha,\beta} t_6$. Clearly, $\lambda_5 = 0$, otherwise, there will be a contradiction using the basis of $\mathcal{A}_{\alpha,\beta}$ (Proposition 5.3.4). Therefore, $\rho(x_4) = 3\lambda_3 x_4$ and $\rho(t_5) = 0$. We already know from Lemma 6.3.4 that $\rho(t_6) = -3\lambda_3 t_6$. From (6.3.14), we have $\rho(x_3) = \lambda_3 x_3 + \frac{3}{2}(\lambda_4 - \lambda_3 - \lambda_5)x_4 t_5^{-1} + (3\lambda_5 - \lambda_4 - \lambda_6)t_5^2 t_6^{-1}$. Given that $\lambda_4 = 3\lambda_3$, $\lambda_5 = 0$ and $\lambda_6 = -3\lambda_3$, we have that $\rho(x_3) = \lambda_3 x_3 + 3\lambda_3 x_4 t_5^{-1}$. Now, $\mathcal{D}(x_3) = \operatorname{ham}_x(x_3) + \rho(x_3) = \operatorname{ham}_x(x_3) + \lambda_3 x_3 + 3\lambda_3 x_4 t_5^{-1}$. Now, $\mathcal{D}(x_3) = \operatorname{ham}_x(x_3) + \rho(x_3) = \operatorname{ham}_x(x_3) + \lambda_3 x_3 + 3\lambda_3 x_4 t_5^{-1}$. Therefore, $\lambda_3 = 0$, otherwise, there will be a contradiction using the basis of $\mathcal{A}_{\alpha,\beta}$. We now have that $\rho(x_3) = \rho(x_4) = \rho(x_5) = \rho(x_6) = 0$. We finish the proof by showing that $\rho(x_1) = \rho(x_2) = 0$. Recall from (5.3.2) that

$$x_2x_4x_6 - \frac{2}{3}x_3^3x_6 - \frac{2}{3}x_2x_5^3 + 2x_3^2x_5^2 - 3x_3x_4x_5 + \frac{3}{2}x_4^2 = \beta$$

Apply ρ to this relation to obtain $\rho(x_2)x_4x_6 - \frac{2}{3}\rho(x_2)x_5^3 = 0$. This implies that $\rho(x_2)\left(x_4x_6 - \frac{2}{3}x_5^3\right) = 0$. Since $x_4x_6 - \frac{2}{3}x_5^3 \neq 0$, it follows that $\rho(x_2) = 0$. Similarly, from (5.3.1), we have that

$$x_1x_3x_5 - \frac{3}{2}x_1x_4 - \frac{1}{2}x_2x_5 + \frac{1}{2}x_3^2 = \alpha$$

Apply ρ to this relation to obtain $\rho(x_1) \left(x_3 x_5 - \frac{3}{2} x_4\right) = 0$. Since $x_3 x_5 - \frac{3}{2} x_4 \neq 0$, we must have $\rho(x_1) = 0$. In conclusion, $\rho(x_{\kappa}) = 0$ for all $\kappa \in \{1, \dots, 6\}$.

3. As a result of (1) and (2), we have that $\mathcal{D}(x_{\kappa}) = ham_x(x_{\kappa})$. Consequently, $\mathcal{D} = ham_x$ as desired.

6.3.5 Poisson derivations of $\mathcal{A}_{\alpha,0}$ and $\mathcal{A}_{0,\beta}$. One can observe that the process we went through to compute the Poisson derivations of $\mathcal{A}_{\alpha,\beta}$ (when $\alpha, \beta \neq 0$) is similar to the process we went through to compute the derivations of the non-commutative analogue $A_{\alpha,\beta}$ (see Subsection 4.2.1). Similarly, when α or β is zero, then one can follow procedures similar to Subsection 4.2.5 to compute the Poisson derivations of $\mathcal{A}_{\alpha,0}$ and $\mathcal{A}_{0,\beta}$. The computations have been done, however, for the avoidance of redundancy, we are not going to include them here. We only summarize the results. Before we do that, we compute explicitly the scalar Poisson derivations of $\mathcal{A}_{\alpha,0}$ and $\mathcal{A}_{0,\beta}$.

6.3.6 Lemma. Let $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}$. Suppose that ϑ and $\tilde{\vartheta}$ are linear maps of $\mathcal{A}_{\alpha,0}$ and $\mathcal{A}_{0,\beta}$ respectively, and, are defined by:

$$\vartheta(x_1) = -x_1, \ \vartheta(x_2) = -x_2, \ \vartheta(x_3) = 0, \ \vartheta(x_4) = x_4, \ \vartheta(x_5) = x_5, \ \vartheta(x_6) = 2x_6,$$

and

$$\tilde{\vartheta}(x_1) = -2x_1, \ \tilde{\vartheta}(x_2) = -3x_2, \ \tilde{\vartheta}(x_3) = -x_3, \ \tilde{\vartheta}(x_4) = 0, \ \tilde{\vartheta}(x_5) = x_5, \ \tilde{\vartheta}(x_6) = 3x_6.$$

Then, ϑ and $\tilde{\vartheta}$ are \mathbb{C} -Poisson derivations of $\mathcal{A}_{\alpha,0}$ and $\mathcal{A}_{0,\beta}$ respectively.

Proof. We need to show that ϑ satisfies the following two relations (see (5.3.1) and (5.3.2)):

$$x_1 x_3 x_5 - \frac{3}{2} x_1 x_4 - \frac{1}{2} x_2 x_5 + \frac{1}{2} x_3^2 = \alpha,$$

$$x_2 x_4 x_6 - \frac{2}{3} x_3^3 x_6 - \frac{2}{3} x_2 x_5^3 + 2 x_3^2 x_5^2 - 3 x_3 x_4 x_5 + \frac{3}{2} x_4^2 = \beta,$$

and the Poisson bracket of $\mathcal{A}_{\alpha,\beta}$ (see Section 5.3) when $\alpha \neq 0$ and $\beta = 0$, and do the same for $\tilde{\vartheta}$ when $\alpha = 0$ and $\beta \neq 0$. We will only do this for the relation $x_1x_3x_5 - \frac{3}{2}x_1x_4 - \frac{1}{2}x_2x_5 + \frac{1}{2}x_3^2 = \alpha$ and the Poisson bracket $\{x_6, x_2\} = 3x_2x_6 + 9x_4 - 18x_3x_5$ in both cases, and leave the remaining ones for the reader to verify. We have:

$$\begin{aligned} \vartheta(x_1)x_3x_5 + x_1\vartheta(x_3)x_5 + x_1x_3\vartheta(x_5) &- \frac{3}{2}[\vartheta(x_1)x_4 + x_1\vartheta(x_4)] \\ &- \frac{1}{2}[\vartheta(x_2)x_5 + x_2\vartheta(x_5)] + \vartheta(x_3)x_3 \\ &= 0, \end{aligned}$$

and

$$\begin{aligned} \vartheta(\{x_6, x_2\}) &= \vartheta(3x_2x_6 + 9x_4 - 18x_3x_5) \\ &= 3[\vartheta(x_2)x_6 + x_2\vartheta(x_6)] + 9\vartheta(x_4) - 18[\vartheta(x_3)x_5 + x_3\vartheta(x_5)] \\ &= 3(-x_2x_6 + 2x_2x_6) + 9x_4 - 18x_3x_5 \\ &= 3x_2x_6 + 9x_4 - 18x_3x_5 \\ &= 3x_2x_6 + 9x_4 - 18x_3x_5 \\ &= \{x_6, x_2\} \\ &= \{x_6, x_2\} \\ &= 2\{x_6, x_2\} - \{x_6, x_2\} \\ &= \{2x_6, x_2\} + \{x_6, -x_2\} \\ &= \{\vartheta(x_6), x_2\} + \{x_6, \vartheta(x_2)\}. \end{aligned}$$

When $\alpha=0$ and $\beta\neq 0,$ we show that $\tilde{\vartheta}$ satisfies the same relations as follows:

$$\begin{split} \tilde{\vartheta}(x_1)x_3x_5 + x_1\tilde{\vartheta}(x_3)x_5 + x_1x_3\tilde{\vartheta}(x_5) &- \frac{3}{2}[\tilde{\vartheta}(x_1)x_4 + x_1\tilde{\vartheta}(x_4)] \\ &- \frac{1}{2}[\tilde{\vartheta}(x_2)x_5 + x_2\tilde{\vartheta}(x_5)] + \tilde{\vartheta}(x_3)x_3 \\ &= -2\left(x_1x_3x_5 - \frac{3}{2}x_1x_4 - \frac{1}{2}x_2x_5 + \frac{1}{2}x_3^2\right) \\ &= 0. \end{split}$$

and

$$\begin{split} \tilde{\vartheta}(\{x_6, x_2\}) &= \tilde{\vartheta}(3x_2x_6 + 9x_4 - 18x_3x_5) \\ &= 3[\tilde{\vartheta}(x_2)x_6 + x_2\tilde{\vartheta}(x_6)] + 9\tilde{\vartheta}(x_4) - 18[\tilde{\vartheta}(x_3)x_5 + x_3\tilde{\vartheta}(x_5)] \\ &= 0 \\ &= 3\{x_6, x_2\} - 3\{x_6, x_2\} \\ &= \{3x_6, x_2\} - 3\{x_6, x_2\} \\ &= \{3x_6, x_2\} + \{x_6, -3x_2\} \\ &= \{\tilde{\vartheta}(x_6), x_2\} + \{x_6, \tilde{\vartheta}(x_2)\}. \end{split}$$

- **6.3.7 Remark.** Let $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}$ and $\lambda \in \mathbb{C}$.
 - 1. Every Poisson derivation \mathcal{D} of $\mathcal{A}_{\alpha,0}$ is of the form $\mathcal{D} = \operatorname{ham}_x + \rho_\lambda$, where ham_x is an inner Poisson derivation of $\mathcal{A}_{\alpha,0}$, and ρ is a scalar Poisson derivation of $\mathcal{A}_{\alpha,0}$ defined as follows:

$$\rho_{\lambda}(x_1) = -\lambda x_1 \qquad \rho_{\lambda}(x_3) = 0 \qquad \rho_{\lambda}(x_5) = \lambda x_5$$
$$\rho_{\lambda}(x_2) = -\lambda x_2 \qquad \rho_{\lambda}(x_4) = \lambda x_4 \qquad \rho_{\lambda}(x_6) = 2\lambda x_6.$$

2. Every Poisson derivation \mathcal{D} of $\mathcal{A}_{0,\beta}$ is of the form $\mathcal{D} = \operatorname{ham}_x + \rho_\lambda$, where ham_x is an inner Poisson derivation of $\mathcal{A}_{0,\beta}$, and ρ is a scalar Poisson derivation of $\mathcal{A}_{0,\beta}$ defined as follows:

$$\rho_{\lambda}(x_1) = -2\lambda x_1 \qquad \rho_{\lambda}(x_3) = -\lambda x_3 \qquad \rho_{\lambda}(x_5) = \lambda x_5$$
$$\rho_{\lambda}(x_2) = -3\lambda x_2 \qquad \rho_{\lambda}(x_4) = 0 \qquad \rho_{\lambda}(x_6) = 3\lambda x_6.$$

6.3.8 Theorem. Given that $\mathcal{A}_{\alpha,\beta} = \mathbb{C}[X_1, \cdots, X_6]/\langle \Omega_1 - \alpha, \Omega_2 - \beta \rangle$, with $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0,0)\}$, we have the following results:

- 1. if $\alpha, \beta \neq 0$; then every Poisson derivation \mathcal{D} of $\mathcal{A}_{\alpha,\beta}$ can uniquely be written as $\mathcal{D} = ham_x$, where $x \in \mathcal{A}_{\alpha,\beta}$.
- 2. if $\alpha \neq 0$ and $\beta = 0$, then every Poisson derivation \mathcal{D} of $\mathcal{A}_{\alpha,0}$ can uniquely be written as $\mathcal{D} = ham_x + \lambda \vartheta$, where $\lambda \in \mathbb{C}$ and $x \in \mathcal{A}_{\alpha,0}$.
- 3. if $\alpha = 0$ and $\beta \neq 0$, then every Poisson derivation \mathcal{D} of $\mathcal{A}_{0,\beta}$ can uniquely be written as $\mathcal{D} = ham_x + \lambda \tilde{\vartheta}$, where $\lambda \in \mathbb{C}$ and $x \in \mathcal{A}_{0,\beta}$.
- 4. $HH^1(\mathcal{A}_{\alpha,0}) = \mathbb{C}[\vartheta]$ and $HH^1(\mathcal{A}_{0,\beta}) = \mathbb{C}[\tilde{\vartheta}]$, where $[\vartheta]$ and $[\tilde{\vartheta}]$ respectively denote the classes of ϑ and $\tilde{\vartheta}$ modulo the space of inner Poisson derivations.
- if α, β ≠ 0; then HH¹(A_{α,β}) = {[0]}, where [0] denotes the class of 0 modulo the space of inner Poisson derivations.

Proof. Points (1) is as a result of Lemma 6.3.4. Points (2) and (3) are as a result of Remark 6.3.7. Point (4) is a consequence of Lemma 6.3.6, and (5) is a consequence of (1).

Conclusion

This thesis studied a q-deformation $A_{\alpha,\beta}$ ($(\alpha,\beta) \in \mathbb{C}^2 \setminus \{(0,0)\}$) of a quadratic extension of $A_2(\mathbb{C})$ and compared some properties of $A_{\alpha,\beta}$ to those of $A_2(\mathbb{C})$. Similar to $A_2(\mathbb{C})$, the algebra $A_{\alpha,\beta}$ is simple, noetherian domain, has GKdim of 4 and the central elements are all scalars. In addition, if $\alpha, \beta \neq 0$; then the group of units of $A_{\alpha,\beta}$ is the set of non-zero complex numbers and the derivations of $A_{\alpha,\beta}$ are all inner. However, when either α or β is zero, then the group of units of $A_{\alpha,\beta}$ are non-trivial, and the derivations are the sum of inner and scalar derivations. Therefore, when α and β are non-zero, then the properties of $A_{\alpha,\beta}$ largely reflect those of $A_2(\mathbb{C})$. As a result, $A_{\alpha,\beta}$ gives a better deformation of a quadratic extension of $A_2(\mathbb{C})$ when $\alpha\beta \neq 0$. We also compared the derivations of $A_{\alpha,\beta}$ to the Poisson derivations of a semiclassical limit $\mathcal{A}_{\alpha,\beta}$ of $A_{\alpha,\beta}$. In fact, the derivations of $A_{\alpha,\beta}$ and the Poisson derivations of $\mathcal{A}_{\alpha,\beta}$ are congruent.

All efforts to compute the automorphism group of $A_{\alpha,\beta}$ (i.e. $\operatorname{Aut}(A_{\alpha,\beta})$) were not successful. Nevertheless, we realized some automorphism subgroups of $\operatorname{Aut}(A_{\alpha,\beta})$. Let $\mu, \lambda \in \mathbb{C}^*$; and define $\phi_{\mu,\lambda} : A_{\alpha,\beta} \to A_{\alpha,\beta}$ by $\phi_{\mu,\lambda}(e_1) = \mu e_1$ and $\phi_{\mu,\lambda}(e_6) = \lambda e_6$. We have the following:

- $\{\phi_{\mu,\lambda} \mid \mu^2 \lambda = 1; \ \forall \mu, \lambda \in \mathbb{C}^*\} \subseteq \operatorname{Aut}(A_{\alpha,0}), \text{ where } \alpha \neq 0.$
- $\{\phi_{\mu,\lambda} \mid \mu^3 \lambda^2 = 1; \ \forall \mu, \lambda \in \mathbb{C}^*\} \subseteq \operatorname{Aut}(A_{0,\beta}), \text{ where } \beta \neq 0.$
- $\{\phi_{\mu,\lambda} \mid \mu, \lambda \in \{-1,1\}\} \subseteq \operatorname{Aut}(A_{\alpha,\beta}), \text{ where } \alpha, \beta \neq 0.$

The following questions are worth considering, and are opened for further studies. **Questions.** Let $(\alpha, \beta) \in \mathbb{C}^2 \setminus \{(0, 0)\}$. Are there any automorphisms of $A_{\alpha,\beta}$? Is every endomorphism of $A_{\alpha,\beta}$ an automorphism?

Appendix A

Computations in $U_q^+(G_2)$

In this appendix, we provide the computations of the algebra relations of $U_q^+(G_2)$ in Section A.1, computations of the data of the deleting derivations algorithm (DDA) of $U_q^+(G_2)$ in Section A.2, and finally, provide computations of the generators of the central elements of $U_q^+(G_2)$ in Section A.3.

A.1 Algebra relations of $U_q^+(G_2)$

This section focuses on providing a summary of the computations of the defining relations of $U_q^+(G_2)$ in Subsection A.1.1, and confirming these relations with GAP in Subsection A.1.2. We finally provide some general relations of $U_q^+(G_2)$ in Subsection A.1.3.

A.1.1 Summary of the computations of the defining relations of $U_q^+(G_2)$. Note, the actions of T_1 and T_2 on $U_q^+(G_2)$ are defined in Section 2.1.

- (1) $E_{\beta_1}E_{\beta_6}-q^{(\beta_1,\beta_6)}E_{\beta_6}E_{\beta_1} = \lambda E_{\beta_5}$. This implies that $E_1E_2-q^{-3}E_2E_1 = \lambda T_1T_2T_1T_2(E_1)$ $= \lambda T_2^{-1}(T_2T_1T_2T_1T_2(E_1)) = \lambda T_2^{-1}(E_1)$. Hence, $T_2(E_1)T_2(E_2)-q^{-3}T_2(E_2)T_2(E_1) = \lambda E_1$. One can verify that $T_2(E_1)T_2(E_2) - q^{-3}T_2(E_2)T_2(E_1) = E_1$. As a result, $\lambda = 1$. Therefore, $E_{\beta_1}E_{\beta_6} - q^{-3}E_{\beta_6}E_{\beta_1} = E_{\beta_5}$.
- (2) $E_{\beta_1}E_{\beta_3} q^{(\beta_1,\beta_3)}E_{\beta_3}E_{\beta_1} = \lambda E_{\beta_2}$. This implies that $E_1T_1T_2(E_1) qT_1T_2(E_1)E_1 = \lambda T_1(E_2)$. Hence, $T_1^{-1}(E_1)T_2(E_1) qT_2(E_1)T_1^{-1}(E_1) = \lambda E_2$. One can verify that

$$T_1^{-1}(E_1)T_2(E_1) - qT_2(E_1)T_1^{-1}(E_1) = (q^2 + 1 + q^{-2})E_2. \text{ As a result, } \lambda = (q^2 + 1 + q^{-2}), \text{ and } E_{\beta_1}E_{\beta_3} - qE_{\beta_3}E_{\beta_1} = (q^2 + 1 + q^{-2})E_{\beta_2}.$$
(3) $E_{\beta_3}E_{\beta_5} - q^{(\beta_3,\beta_5)}E_{\beta_5}E_{\beta_3} = \lambda E_{\beta_4}. \text{ This implies that } E_{\beta_3}E_{\beta_5} - qE_{\beta_5}E_{\beta_3} = \lambda E_{\beta_4}.$

This relation is similar to the relation in point (2). Hence, $\lambda = (q^2 + 1 + q^{-2})$, and $E_{\beta_3}E_{\beta_5} - qE_{\beta_5}E_{\beta_3} = (q^2 + 1 + q^{-2})E_{\beta_4}$.

- (4) $E_{\beta_2}E_{\beta_4} q^{(\beta_2,\beta_4)}E_{\beta_4}E_{\beta_2} = \lambda E_{\beta_3}^3$. This implies that $T_1(E_2)T_1T_2T_1(E_2) - q^3T_1T_2T_1(E_2)T_1(E_2) = \lambda T_1T_2(E_1^3)$. Hence, $T_2^{-1}(E_2)T_1(E_2) - q^3T_1(E_2)T_2^{-1}(E_2) = \lambda E_1^3$. One can verify that $T_2^{-1}(E_2)T_1(E_2) - q^3T_1(E_2)T_2^{-1}(E_2) = \frac{q^7 - 2q^5 + q^3}{q^4 + q^2 + 1}E_1^3$. As a result, $\lambda = \frac{q^7 - 2q^5 + q^3}{q^4 + q^2 + 1}$, and $E_{\beta_2}E_{\beta_4} - q^3E_{\beta_4}E_{\beta_2} = \frac{q^7 - 2q^5 + q^3}{q^4 + q^2 + 1}E_{\beta_3}^3$.
- (5) $E_{\beta_4}E_{\beta_6} q^{(\beta_4,\beta_6)}E_{\beta_6}E_{\beta_4} = \lambda E_{\beta_5}^3$. This implies that $E_{\beta_4}E_{\beta_6} q^3 E_{\beta_6}E_{\beta_4} = \lambda E_{\beta_5}^3$. This relation is similar to point (4). Hence, $\lambda = \frac{q^7 - 2q^5 + q^3}{q^4 + q^2 + 1}$, and $E_{\beta_4}E_{\beta_6} - q^3 E_{\beta_6}E_{\beta_4} = \frac{q^7 - 2q^5 + q^3}{q^4 + q^2 + 1}E_{\beta_5}^3$.
- (6) $E_{\beta_1}E_{\beta_5} q^{(\beta_1,\beta_5)}E_{\beta_5}E_{\beta_1} = \lambda E_{\beta_3}$. This implies that $E_{\beta_1}E_{\beta_5} q^{-1}E_{\beta_5}E_{\beta_1} = \lambda E_{\beta_3}$. Now, $E_{\beta_3} = T_1T_2(E_1) = T_1(E_1E_2 - q^{-3}E_2E_1) = T_1(E_2)T_1(E_1) - q^{-3}T_1(E_1)T_1(E_2)$. Further simplification shows that $E_{\beta_3} = T_1(E_2)T_1(E_1) - q_2^{-1}T_1(E_1)T_1(E_2) = \frac{-1}{q+q^{-1}}E_1^2E_2 + \frac{q^{-3}+q^{-1}}{q+q^{-1}}E_1E_2E_1 - \frac{q^{-4}}{q+q^{-1}}E_2E_1^2$. Furthermore, from point (1), $E_{\beta_5} = E_{\beta_1}E_{\beta_6} - q^{-3}E_{\beta_6}E_{\beta_1} = E_1E_2 - q^{-3}E_2E_1$. It follows that:

$$E_{\beta_1}E_{\beta_5} - q^{-1}E_{\beta_5}E_{\beta_1} = E_1(E_1E_2 - q^{-3}E_2E_1) - q^{-1}(E_1E_2 - q^{-3}E_2E_1)E_1$$
$$= E_1^2E_2 - (q^{-3} + q^{-1})E_1E_2E_1 + q^{-4}E_2E_1^2$$
$$= -E_1^2E_2 + (q^{-3} + q^{-1})E_1E_2E_1 - q^{-4}E_2E_1^2$$
$$= (q + q^{-1})E_{\beta_3}.$$

Consequently, $E_{\beta_1}E_{\beta_5} - q^{-1}E_{\beta_5}E_{\beta_1} = (q+q^{-1})E_{\beta_3}$.

- (7) $E_{\beta_3}E_{\beta_6} q^{(\beta_3,\beta_6)}E_{\beta_6}E_{\beta_3} = \lambda E_{\beta_5}^2$. This implies that $E_{\beta_3}E_2 E_2E_{\beta_3} = \lambda E_{\beta_5}^2$. From point (6), one can make substitutions for E_{β_3} and E_{β_5} , and simplify to obtain $\lambda = q^2 1$. Therefore, $E_{\beta_3}E_{\beta_6} E_{\beta_6}E_{\beta_3} = (q^2 1)E_{\beta_5}^2$.
- (8) $E_{\beta_1}E_{\beta_4} q^{(\beta_1,\beta_4)}E_{\beta_4}E_{\beta_1} = \lambda E_{\beta_3}^2$. This implies that $E_{\beta_1}E_{\beta_4} E_{\beta_4}E_{\beta_1} = \lambda E_{\beta_3}^2$. This relation is similar to (7). Hence, $\lambda = q^2 1$, and $E_{\beta_1}E_{\beta_4} E_{\beta_4}E_{\beta_1} = (q^2 1)E_{\beta_3}^2$.
- (9) $E_{\beta_2}E_{\beta_5} q^{(\beta_2,\beta_5)}E_{\beta_5}E_{\beta_2} = \lambda E_{\beta_3}^2$. This implies that $T_1(E_2)E_{\beta_5} E_{\beta_5}T_1(E_2) = \lambda E_{\beta_3}^2$. Similarly, from points (6), we can make substitutions for E_{β_3} and E_{β_5} . Further simplification shows that $\lambda = q^2 1$. As a result, $E_{\beta_2}E_{\beta_5} E_{\beta_5}E_{\beta_2} = (q^2 1)E_{\beta_3}^2$.
- (10) $E_{\beta_2}E_{\beta_6} q^{(\beta_2,\beta_6)}E_{\beta_6}E_{\beta_2} = \lambda E_{\beta_4} + \mu E_{\beta_3}E_{\beta_5}$. This implies that $T_1(E_2)E_2 q^{-3}E_2T_1(E_2)$ $= \lambda E_{\beta_4} + \mu E_{\beta_3}E_{\beta_5}$. Using the expressions for E_{β_3} and E_{β_5} in point (6), and the relation: $E_{\beta_4} = (q^2 + 1 + q^{-2})^{-1}(E_{\beta_3}E_{\beta_5} - qE_{\beta_5}E_{\beta_3})$, one can simplify $T_1(E_2)E_2 - q^{-3}E_2T_1(E_2) = \lambda E_{\beta_4} + \mu E_{\beta_3}E_{\beta_5}$ to obtain $\lambda = -(q + q^{-1} - q^{-3})$ and $\mu = q - q^{-1}$. Consequently, $E_{\beta_2}E_{\beta_6} - q^{-3}E_{\beta_6}E_{\beta_2} = -(q + q^{-1} - q^{-3})E_{\beta_4} + (q - q^{-1})E_{\beta_3}E_{\beta_5}$.

(11)
$$E_{\beta_i}E_{\beta_j} = q^3 E_{\beta_j}E_{\beta_i}$$
, for all $1 \le i, j \le 6$ with $j - i = 1$.

A.1.2 Algebra relations of $U_q^+(G_2)$: GAP Code. The code below confirms the defining relations of $U_q^+(G_2)$. Recall that $E_i = E_{\beta_i}$ (see the comments before the relations of $U_q^+(G_2)$ in Section 2.1). Moreover, $E_3^{(r)} = \frac{E_3^r}{[r]_q!}$ and $E_5^{(r)} = \frac{E_5^r}{[r]_q!}$ for all $r \in \mathbb{N}_{>1}$. brk> U:=QuantizedUEA(RootSystem("G",2));; U:=QuantizedUEA(RootSystem("G",2));; \uparrow brk> T:=GeneratorsOfAlgebra(U); T:=GeneratorsOfAlgebra(U); \uparrow [F1, F2, F3, F4, F5, F6, K1, $(-q+q^{-1})*[$ K1 ; 1]+K1, K2, $(-q^3+q^{-3})*[$ K2 ; 1]+K2, E1, E2, E3, E4, E5, E6] brk> g:=[T[11],T[12],T[13],T[14],T[15],T[16]]; #The generators of U_q^+(G_2) are; g:=[T[11],T[12],T[13],T[14],T[15],T[16]]; \uparrow [E1, E2, E3, E4, E5, E6]

```
brk> #We now compute the algebra relations for U_q^+(G_2)
brk> g[2]*g[1];
(q^-3)*E1*E2
brk> g[3]*g[1];
(q<sup>-1</sup>)*E1*E3+(-q-q<sup>-1</sup>-q<sup>-3</sup>)*E2
brk> g[3]*g[2];
(q^-3)*E2*E3
brk> g[4]*g[1];
E1*E4+(-q^3+q^-1)*E3^(2)
brk> g[4]*g[2];
(q^{-3})*E2*E4+(-q^{3}+q+q^{-1}-q^{-3})*E3^{(3)}
brk> g[4]*g[3];
(q^-3)*E3*E4
brk> g[5]*g[1];
(q)*E1*E5+(-q^2-1)*E3
brk> g[5]*g[2];
E2*E5+(-q<sup>3</sup>+q<sup>-1</sup>)*E3<sup>(2)</sup>
brk> g[5]*g[3];
(q^{-1})*E3*E5+(-q-q^{-1}-q^{-3})*E4
brk> g[5]*g[4];
(q^-3)*E4*E5Page 122
brk> g[6]*g[1];
(q<sup>3</sup>)*E1*E6+(-q<sup>3</sup>)*E5
brk> g[6]*g[2];
(q<sup>3</sup>)*E2*E6+(-q<sup>4</sup>+q<sup>2</sup>)*E3*E5+(q<sup>4</sup>+q<sup>2</sup>-1)*E4
brk> g[6]*g[3];
E3*E6+(-q^3+q^-1)*E5^(2)
brk> g[6]*g[4];
(q<sup>-3</sup>)*E4*E6+(-q<sup>3</sup>+q+q<sup>-1</sup>-q<sup>-3</sup>)*E5<sup>(3)</sup>
brk> g[6]*g[5];
(q^-3)*E5*E6
```

A.1.3 Some general relations of $U_q^+(G_2)$. We have the following selected general relations of $U_q^+(G_2)$.

A.1.4 Lemma. For any $n \in \mathbb{N}$, we have that:

 $1({\rm a}) \; E_j E_i^n = q^{-3n} E_i^n E_j \quad \ ({\rm b}) \; E_j^n E_i = q^{-3n} E_i E_j^n \; {\rm for \; all} \; 1 \leq i,j \leq 6, {\rm with} \; j-i=1.$

$$\begin{aligned} & 2(a) \ E_6 E_4^n = q^{-3n} E_4^n E_6 + d_1[n] E_4^{n-1} E_5^3 \qquad (b) \ E_6^n E_4 = q^{-3n} E_4 E_6^n + d_1[n] E_5^3 E_6^{n-1} \\ & (c) \ E_4 E_2^n = q^{-3n} E_2^n E_4 + d_1[n] E_2^{n-1} E_3^3 \qquad (d) \ E_4^n E_2 = q^{-3n} E_2 E_4^n + d_1[n] E_3^3 E_4^{n-1}, \\ & \text{where } d_1[n] = q^{3(1-n)} d_1[1] \left(\frac{1-q^{-6n}}{1-q^{-6}}\right); \ d_1[1] = -\frac{q^4 - 2q^2 + 1}{q^4 + q^2 + 1} \ \text{and} \ d_1[0] := 0. \end{aligned}$$

$$\begin{aligned} & 3(a) \ E_3 E_1^n = q^{-n} E_1^n E_3 + d_2[n] E_1^{n-1} E_2 \qquad (b) \ E_3^n E_1 = q^{-n} E_1 E_3^n + d_2[n] E_2 E_3^{n-1} \\ & (c) \ E_5 E_3^n = q^{-n} E_3^n E_5 + d_2[n] E_3^{n-1} E_4 \qquad (d) \ E_5^n E_3 = q^{-n} E_3 E_5^n + d_2[n] E_4 E_5^{n-1}, \\ & \text{where } d_2[n] = q^{1-n} d_2[1] \left(\frac{1-q^{-2n}}{1-q^{-2}}\right); \ d_2[1] = -(q+q^{-1}+q^{-3}) \ \text{and} \ d_2[0] := 0. \end{aligned}$$

$$\begin{aligned} & 4(a) \ E_6^n E_3 = E_3 E_6^n + d_3[n] E_5^2 E_6^{n-1} \qquad (b) \ E_5 E_2^n = E_2^n E_5 + d_3[n] E_2^{n-1} E_3^2, \\ & \text{where } \ d_3[n] = d_3[1] \left(\frac{1-q^{-6n}}{1-q^{-6}}\right); \ d_3[1] = 1 - q^2 \ \text{and} \ d_3[0] := 0. \end{aligned}$$

Proof. We prove the lemma by an induction on n. We will only prove for the first non-trivial case (i.e. 2(a)), and leave the remaining ones for the reader to verify.

2(a). The result is clear when n = 0. For $n \ge 0$, suppose that $E_6E_4^n = q^{-3n}E_4^nE_6 + d_1[n]E_4^{n-1}E_5^3$, where $d_1[n] = q^{3(1-n)}d_1[1](1-q^{-6n})/(1-q^{-6})$; $d_1[1] = -(q^4-2q^2+1)/(q^4+q^2+1)$ and $d_1[0] := 0$. Then, $E_6E_4^{n+1} = (q^{-3n}E_4^nE_6+d_1[n]E_4^{n-1}E_5^3)E_4 = q^{-3(n+1)}E_4^{n+1}E_6 + (q^{-3n}d_1[1]+q^{-9}d_1[n])E_4^nE_5^3$. Note, $E_6E_4 = q^{-3}E_4E_6 + d_1[1]E_5^3$. Now, $q^{-3n}d_1[1] + q^{-9}d_1[n] = q^{-3n}d_1[1] + q^{3(1-n)-9}d_1[1](1-q^{-6n})/(1-q^{-6}) = q^{-3n}d_1[1](1-q^{-6(n+1)})/(1-q^{-6}) = d_1[n+1]$ as expected.

A.2 Deleting derivations algorithm of $U_q^+(G_2)$

Given that $\sigma_j \circ \delta_j = q_j \delta_j \circ \sigma_j$ (see the comments after Definition 1.7.1), we have that

- $\sigma_3 \circ \delta_3(E_1) = q^{-2} \delta_3 \circ \sigma_3(E_1)$ hence $q_3 = q^{-2}$,
- $\sigma_4 \circ \delta_4(E_1) = q^{-6} \delta_4 \circ \sigma_4(E_1)$ hence $q_4 = q^{-6}$,
- $\sigma_5 \circ \delta_5(E_1) = q^{-2} \delta_5 \circ \sigma_5(E_1)$ hence $q_5 = q^{-2}$,
- $\sigma_6 \circ \delta_6(E_1) = q^{-6} \delta_6 \circ \sigma_6(E_1)$ hence $q_6 = q^{-6}$.

From (1.7.1), we have the relation:

$$E_{i,j} = \begin{cases} E_{i,j+1} & \text{if } i \ge j \\ \sum_{k=0}^{+\infty} \frac{(1-q_j)^{-k}}{[k]_{q_j}!} \delta_j^k \circ \sigma_j^{-k} (E_{i,j+1}) E_{j,j+1}^{-k} & \text{if } i < j. \end{cases}$$

Given this relation, one can compute the elements of $\mathsf{Fract}(U_q^+(G_2))$ as follows:

$$\begin{split} E_{1,6} &= E_1 + (1-q_6)^{-1} \delta_6 \circ \sigma_6^{-1}(E_1) E_6^{-1} \\ &= E_1 + (1-q^{-6})^{-1} E_5 E_6^{-1} \\ &= E_1 + r E_5 E_6^{-1} . \\ \\ E_{2,6} &= E_2 + (1-q_6)^{-1} \delta_6 \circ \sigma_6^{-1}(E_2) E_6^{-1} + \frac{(1-q_6)^{-2}}{[2]_{g_6}!} \delta_6^2 \circ \sigma_6^{-2}(E_2) E_6^{-2} \\ &= E_2 + (1-q^{-6})^{-1} [(q^{-1}-q) E_3 E_5 + (q+q^{-1}-q^{-3}) E_4] E_6^{-1} \\ &+ \frac{(1-q^{-6})^{-2}}{1+q^{-6}} \left[(q^{-4}-q^{-2})(1-q^2) + \frac{(q^{-2}+q^{-4}-q^{-6})(2q^2-q^4-1)}{(q^4+q^2+1)} \right] E_5^3 E_6^{-2} \\ &= E_2 + \frac{q^{-1}-q}{1-q^{-6}} E_3 E_5 E_6^{-1} + \frac{q+q^{-1}-q^{-3}}{1-q^{-6}} E_4 E_6^{-1} \\ &+ \frac{(q^{-4}-q^{-2})(1-q^2)(q^4+q^2+1) + (q^{-2}+q^{-4}-q^{-6})(2q^2-q^4-1)}{(1-q^{-6})^2(1+q^{-6})(q^4+q^2+1)} E_5^3 E_6^{-2} \\ &= E_2 + t E_3 E_5 E_6^{-1} + u E_4 E_6^{-1} + n E_5^3 E_6^{-2} . \\ \\ E_{3,6} &= E_3 + (1-q_6)^{-1} \delta_6 \circ \sigma_6^{-1}(E_3) E_6^{-1} \\ &= E_3 + \frac{1-q^2}{1-q^{-6}} E_5^2 E_6^{-1} \\ &= E_4 + \frac{-q^7+2q^5-q^3}{(1-q^{-6})(q^4+q^2+1)} E_5^3 E_6^{-1} \\ &= E_4 + b E_5^3 E_6^{-1} . \end{split}$$

$$\begin{split} E_{1,5} &= E_{1,6} + (1-q_5)^{-1} \delta_5 \circ \sigma_5^{-1} (E_{1,6}) E_{5,6}^{-1} + \frac{(1-q_5)^{-2}}{[2]_{g!}!} \delta_5^2 \circ \sigma_5^{-2} (E_{1,6}) E_{5,6}^{-2} \\ &= E_{1,6} + \frac{q+q^{-1}}{q^{-2}-1} E_{3,6} E_{5,6}^{-1} + \frac{q+q^{-1}+q^{-3}}{(1-q^{-2})^2} E_{4,6} E_{5,6}^{-2} \\ &= E_{1,6} + h E_{3,6} E_{5,6}^{-1} + g E_{4,6} E_{5,6}^{-2} \\ E_{2,5} &= E_{2,6} + (1-q_5)^{-1} \delta_5 \circ \sigma_5^{-1} (E_{2,6}) E_{5,6}^{-3} + \frac{(1-q_5)^{-2}}{[2]_{g!}!} \delta_5^2 \circ \sigma_5^{-2} (E_{2,6}) E_{5,6}^{-2} \\ &+ \frac{(1-q_5)^{-3}}{[3]_{g!}!} \delta_5^3 \circ \sigma_5^{-3} (E_{2,6}) E_{5,6}^{-3} \\ &= E_{2,6} + \frac{1-q^2}{1-q^{-2}} E_{3,6}^2 E_{5,6}^{-1} + \frac{(q^{-1}+q^{-3})(q^2-1)(q+q^{-1}+q^{-3})}{(1-q^{-2})^2(1+q^{-2})} E_{3,6} E_{4,6} E_{5,6}^{-2} \\ &+ \frac{(q^{-1}+q^{-3})(1-q^2)(q+q^{-1}+q^{-3})^2}{(1-q^{-2})^3(1+q^{-2})(1+q^{-2}+q^{-4})} E_{4,6}^2 E_{5,6}^{-3} \\ &= E_{2,6} + f E_{3,6}^2 E_{5,6}^{-1} + p E_{3,6} E_{4,6} E_{5,6}^{-2} + e E_{4,6}^2 E_{5,6}^{-3} \\ &= E_{3,6} + (1-q_5)^{-1} \delta_5 \circ \sigma_5^{-1} (E_{3,6}) E_{3,6}^{-1} \\ &= E_{3,6} + a E_{4,6} E_{5,6}^{-1} \\ &= E_{3,6} + a E_{4,6} E_{5,6}^{-1} \\ &= E_{3,6} + a E_{4,6} E_{5,6}^{-1} \\ &= E_{1,5} + \frac{1-q^2}{q^{-2}-1} E_{4,6} E_{5,6}^{-1} \\ &= E_{1,5} + \frac{1-q^2}{1-q^{-6}} E_{3,5}^2 E_{4,5}^{-1} \\ &= E_{1,5} + \frac{1-q^2}{1-q^{-6}} E_{3,5}^2 E_{4,5}^{-1} \\ &= E_{2,5} + \frac{-q^7 + 2q^5 - q^3}{(1-q^{-6})(q^4+q^2+1)} E_{3,5}^3 E_{4,5}^{-1} \\ &= E_{2,5} + b E_{3,5}^3 E_{4,5}^{-1} \\ &= E_{1,4} + \frac{q^2 + 1 + q^{-2}}{q^{-2}-1} E_{2,4} E_{3,4}^{-1} \\ &= E_{1,4} + \frac{q^2 + 1 + q^{-2}}{q^{-2}-1} E_{2,4} E_{3,4}^{-1} \\ &= E_{1,4} + \frac{q^2 + 1 + q^{-2}}{q^{-2}-1} E_{2,4} E_{3,4}^{-1} \\ &= E_{1,4} + \frac{q^2 + 1 + q^{-2}}{q^{-2}-1} E_{2,4} E_{3,4}^{-1} \\ &= E_{1,4} + \frac{q^2 + 1 + q^{-2}}{q^{-2}-1} E_{2,4} E_{3,4}^{-1} \\ &= E_{1,4} + a E_{2,4} E_{3,4}^{-1} . \end{split}$$

Otherwise, $E_{i,j} = E_{i,j+1}$, where $1 \le i,j \le 6$. Note, all the *constant coefficients* (a, b, e, f, g, h, n, p, r, s, t, u) are defined in Appendix C.

A.3 Generators of the center of $U_q^+(G_2)$

Recall from Section 2.2 that $T_i = E_{i,2}$ for all $1 \le i \le 6$, $\Omega_1 = T_1 T_3 T_5$ and $\Omega_2 = T_2 T_4 T_6$. Now, using the data of the DDA of $U_q^+(G_2)$, we have the following:

$$\begin{split} \Omega_1 &= T_1 T_3 T_5 = E_{1,2} E_{3,2} E_{5,2} = E_{1,3} E_{3,3} E_{5,3} \\ &= (E_{1,4} + a E_{2,4} E_{3,4}^{-1}) E_{3,4} E_{5,4} \\ &= (E_{1,4} E_{3,4} + a E_{2,4}) E_{5,4} \\ &= [(E_{1,5} + s E_{3,5}^2 E_{4,5}^{-1}) E_{3,5} + a (E_{2,5} + b E_{3,5}^3 E_{4,5}^{-1})] E_{5,4} \\ &= (E_{1,5} E_{3,5} + q^3 s E_{3,5}^3 E_{4,5}^{-1} + a b E_{3,5}^3 E_{4,5}^{-1} + a E_{2,5}) E_{5,5} \\ &= (E_{1,5} E_{3,5} + a E_{2,5}) E_{5,5} (\text{Note, } q^3 s + a b = 0) \\ &= [(E_{1,6} + h E_{3,6} E_{5,6}^{-1} + g E_{4,6} E_{5,6}^{-2}) (E_{3,6} + a E_{4,6} E_{5,6}^{-1}) + a E_{2,6} + a f E_{3,6}^2 E_{5,6}^{-1} \\ &+ a p E_{3,6} E_{4,6} E_{5,6}^{-2} + a e E_{4,6}^2 E_{5,6}^{-3}] E_{5,6} \\ &= E_{1,6} E_{3,6} E_{5,6} + (hq + af) E_{3,6}^2 + a E_{1,6} E_{4,6} + a E_{2,6} E_{5,6}. \end{split}$$
(A.3.1)

From the data of the DDA of $U_q^+(G_2)$, one can make the necessary substitutions for $E_{i,6}$ (with $1 \le i \le 6$) in (A.3.1), and simplify to obtain

$$\Omega_1 = T_1 T_3 T_5 = E_1 E_3 E_5 + a E_1 E_4 + a E_2 E_5 + a' E_3^2$$

as desired. In a similar manner, we have that:

$$\Omega_{2} = T_{2}T_{4}T_{6} = E_{2,2}E_{4,2}E_{6,2} = E_{2,3}E_{4,3}E_{6,3} = E_{2,4}E_{4,4}E_{6,4}$$

$$= (E_{2,5} + bE_{3,5}^{3}E_{4,5}^{-1})E_{4,5}E_{6,5}$$

$$= E_{2,5}E_{4,5}E_{6,5} + bE_{3,5}^{3}E_{6,5}$$

$$= (E_{2,6} + fE_{3,6}^{2}E_{5,6}^{-1} + pE_{3,6}E_{4,6}E_{5,6}^{-2} + eE_{4,6}^{2}E_{5,6}^{-3})E_{4,6}E_{6,6}$$

$$+ b(E_{3,6} + aE_{4,6}^{3}E_{5,6}^{-1})^{3}E_{6,6}$$

$$= E_{2,6}E_{4,6}E_{6,6} + bE_{3,6}^{3}E_{6,6}.$$
(A.3.2)

Again, from the data of the DDA, one can make the necessary substitutions for $E_{i,6}$ (with $1 \le i \le 6$) in (A.3.2), and simplify to obtain:

$$\Omega_2 = T_2 T_4 T_6 = E_2 E_4 E_6 + b E_2 E_5^3 + b E_3^3 E_6 + b' E_3^2 E_5^2 + c' E_3 E_4 E_5 + d' E_4^2$$

as desired.

Note, the constants a, a', b, b', c' and d' are defined in Appendix C.

A.3.1 The generators of the center of $U_q^+(G_2)$: GAP Code. The code below confirms that Ω_1 and Ω_2 commute with E_1 and E_6 as expected.

```
brk> U:=QuantizedUEA(RootSystem("G",2));;
U:=QuantizedUEA(RootSystem("G",2));;
brk> T:=GeneratorsOfAlgebra(U);
T:=GeneratorsOfAlgebra(U);
[ F1, F2, F3, F4, F5, F6, K1, (-q+q^-1)*[ K1 ; 1 ]+K1, K2,
(-q<sup>3</sup>+q<sup>-3</sup>)*[K2; 1]+K2, E1, E2, E3, E4, E5, E6]
#The generators of U_q^+(G_2) are;
g:=[T[11],T[12],T[13],T[14],T[15],T[16]];
[ E1, E2, E3, E4, E5, E6 ]
brk> a:=-(_q^2+1+_q^-2)/(1-_q^-2);
a:=-(_q^2+1+_q^-2)/(1-_q^-2);
(-q^4-q^2-1)/(q^2-1)
brk> a1:=-_q^6/(1-_q^2);
a1:=-_q^6/(1-_q^2);
(-q^6)/(-q^2+1)Page 123
brk> b:=(_q^7-2*_q^5+_q^3)/((1-_q^-6)*(_q^4+_q^2+1));
b:=-(_q^7-2*_q^5+_q^3)/((1-_q^-6)*(_q^4+_q^2+1));
(q<sup>11</sup>-q<sup>9</sup>)/(q<sup>8</sup>+2*q<sup>6</sup>+3*q<sup>4</sup>+2*q<sup>2</sup>+1)
brk> b1:=(_q^11-_q^13)/(_q^8+2*_q^6+3*_q^4+2*_q^2+1);
b1:=-(_q^11-_q^13)/(_q^8+2*_q^6+3*_q^4+2*_q^2+1);
```

```
(-q<sup>13</sup>+q<sup>11</sup>)/(q<sup>8</sup>+2*q<sup>6</sup>+3*q<sup>4</sup>+2*q<sup>2</sup>+1)
brk> c1:=-_q^9/(_q^4+_q^2+1);
c1:=-_q^9/(_q^4+_q^2+1);
(-q^9)/(q^4+q^2+1)
brk> d1:=-_q^12/(1-_q^6);
d1:=-_q^12/(1-_q^6);
(-q^{12})/(-q^{6+1})
brk> Omega_1:=g[1]*g[3]*g[5]+a*g[1]*g[4]+a*g[2]*g[5]+a1*g[3]^2;
E1*E3*E5+((-q<sup>4</sup>-q<sup>2</sup>-1)/(q<sup>2</sup>-1))*E1*E4+((-q<sup>4</sup>-q<sup>2</sup>-1)/(q<sup>2</sup>-1))*E2*E5+
((-q<sup>7</sup>-q<sup>5</sup>)/(-q<sup>2</sup>+1))*E3<sup>(2)</sup>
brk> Omega_1*g[1]=g[1]*Omega_1;
true
brk> Omega_1*g[6]=g[6]*Omega_1;
true
brk>Omega_2:=g[2]*g[4]*g[6]+b*g[2]*g[5]^3+b*g[3]^3*g[6]+b1*g[3]^2*g[5]^2
+c1*g[3]*g[4]*g[5]+d1*g[4]^2;
E2*E4*E6+((-q<sup>10</sup>+q<sup>6</sup>)/(q<sup>4</sup>+q<sup>2</sup>+1))*E2*E5<sup>(3)</sup>+((-q<sup>9</sup>)/(q<sup>4</sup>+q<sup>2</sup>+1))*E3*E4*E5
+((q<sup>15</sup>+q<sup>13</sup>-q<sup>11</sup>-q<sup>9</sup>)/(q<sup>8</sup>+2*q<sup>6</sup>+3*q<sup>4</sup>+2*q<sup>2</sup>+1))*E3<sup>(2)</sup>*E5<sup>(2)</sup>
+((-q<sup>10</sup>+q<sup>6</sup>)/(q<sup>4</sup>+q<sup>2</sup>+1))*E<sup>3</sup>(3)*E<sup>6</sup>+((-q<sup>15</sup>-q<sup>9</sup>)/(-q<sup>6</sup>+1))*E<sup>4</sup>(2)
brk> Omega_2*g[1]=g[1]*Omega_2;
true
brk> Omega_2*g[6]=g[6]*Omega_2;
true
```

Appendix B

Computations in $\mathcal{A} = \mathbb{C}[X_1, \cdots, X_6]$

Recall that the Poisson algebra $\mathcal{A} = \mathbb{C}[X_1, \cdots, X_6]$ is the semiclassical limit of $U_q^+(G_2)$ (Section 5.2). In this appendix, we define the Poisson bracket of \mathcal{A} in Section B.1, and the data of the Poisson deleting derivations algorithm (PDDA) of \mathcal{A} in Section B.2.

B.1 Poisson bracket of A

Note, $f(z) = z^4 + z^2 + 1$ (see Section 5.2). In addition, set $g(z) := z^2 + z + 1$. We have the following:

$$\begin{split} \{X_3, X_1\} &= \frac{\hat{U}_3 \hat{U}_1 - \hat{U}_1 \hat{U}_3}{z - 1} + (z - 1)\hat{A} = -z^{-1} \hat{U}_1 \hat{U}_3 - z^{-3} \hat{U}_2 + (z - 1)\hat{A} \\ &= -X_1 X_3 - X_2. \\ \{X_4, X_1\} &= \frac{\hat{U}_4 \hat{U}_1 - \hat{U}_1 \hat{U}_4}{z - 1} + (z - 1)\hat{A} = -(z + 1)\hat{U}_3^2 + (z - 1)\hat{A} = -2X_3^2. \\ \{X_4, X_2\} &= \frac{\hat{U}_4 \hat{U}_2 - \hat{U}_2 \hat{U}_4}{z - 1} + (z - 1)\hat{A} = -z^{-3}g(z)\hat{U}_2 \hat{U}_4 - (z + 1)^2 \hat{U}_3^3 + (z - 1)\hat{A} \\ &= -3X_2 X_4 - 4X_3^3. \\ \{X_5, X_1\} &= \frac{\hat{U}_5 \hat{U}_1 - \hat{U}_1 \hat{U}_5}{z - 1} + (z - 1)\hat{A} = \hat{U}_1 \hat{U}_5 - (z^2 + 1)\hat{U}_3 + (z - 1)\hat{A} \\ &= X_1 X_5 - 2X_3. \\ \{X_5, X_2\} &= \frac{\hat{U}_5 \hat{U}_2 - \hat{U}_2 \hat{U}_5}{z - 1} + (z - 1)\hat{A} = -(z + 1)f(z)\hat{U}_3^2 + (z - 1)\hat{A} = -6X_3^2. \end{split}$$

$$\begin{split} \{X_5, X_3\} &= \frac{\hat{U}_5 \hat{U}_3 - \hat{U}_3 \hat{U}_5}{z - 1} + (z - 1)\hat{A} = -z^{-1} \hat{U}_3 \hat{U}_5 - z^{-3} f(z) \hat{U}_4 + (z - 1)\hat{A} \\ &= -X_3 X_5 - 3X_4. \\ \{X_6, X_1\} &= \frac{\hat{U}_6 \hat{U}_1 - \hat{U}_1 \hat{U}_6}{z - 1} + (z - 1)\hat{A} = g(z) \hat{U}_1 \hat{U}_6 - z^3 f(z) \hat{U}_5 + (z - 1)\hat{A} \\ &= 3X_1 X_6 - 3X_5. \\ \{X_6, X_2\} &= \frac{\hat{U}_6 \hat{U}_2 - \hat{U}_2 \hat{U}_6}{z - 1} + (z - 1)\hat{A} = g(z) \hat{U}_2 \hat{U}_6 + (z^4 + z^2 - 1) f(z)^2 \hat{U}_4 \\ &- f(z)^2 (z^2 + z^3) \hat{U}_3 \hat{U}_5 + (z - 1)\hat{A} \\ &= 3X_2 X_6 + 9X_4 - 18X_3 X_5. \\ \{X_6, X_3\} &= \frac{\hat{U}_6 \hat{U}_3 - \hat{U}_3 \hat{U}_6}{z - 1} + (z - 1)\hat{A} = -f(z)(1 + z) \hat{U}_5^2 + (z - 1)\hat{A} = -6X_5^2. \\ \{X_6, X_4\} &= \frac{\hat{U}_6 \hat{U}_4 - \hat{U}_4 \hat{U}_6}{z - 1} + (z - 1)\hat{A} = -z^{-3}g(z) \hat{U}_4 \hat{U}_6 - (z + 1)^2 \hat{U}_5^3 + (z - 1)\hat{A} \\ &= -3X_4 X_6 - 4X_5^3. \\ \{X_j, X_i\} &= \frac{\hat{U}_j \hat{U}_i - \hat{U}_i \hat{U}_j}{z - 1} + (z - 1)\hat{A} = -z^{-3}g(z) \hat{U}_i \hat{U}_j + (z - 1)\hat{A} = -3X_i X_j, \end{split}$$

for all $1 \le i < j \le 6$ with j - i = 1.

B.2 PDDA of the semiclassical limit of $U_q^+(G_2)$

Given that $\delta_j \alpha_j - \alpha_j \delta_j = \eta_j \delta_j$ (Hypothesis 5.1.11), we have the following:

- $(\delta_3\sigma_3 \sigma_3\delta_3)(X_1) = -2X_2$ and $\delta_3(X_1) = -X_2$, hence $\eta_3 = 2$;
- $(\delta_4 \sigma_4 \sigma_4 \delta_4)(X_1) = -12X_3^2$ and $\delta_4(X_1) = -2X_3^2$, hence $\eta_4 = 6$;
- $(\delta_5\sigma_5 \sigma_5\delta_5)(X_2) = -12X_3^2$ and $\delta_5(X_2) = -6X_3^2$, hence $\eta_5 = 2$;
- $(\delta_6 \sigma_6 \sigma_6 \delta_6)(X_1) = -18X_5$ and $\delta_6(X_1) = -3X_5$, hence $\eta_6 = 6$.

We now compute the data of the PDDA of \mathcal{A} using the relation

$$X_{i,j} = \begin{cases} X_{i,j+1} & \text{if } i \ge j \\ \sum_{k=0}^{+\infty} \frac{1}{\eta_j^k k!} \delta_j^k(X_{i,j+1}) X_{j,j+1}^{-k} & \text{if } i < j \end{cases}$$

(which can be found in Subsection 5.1.8) as follows:

- $X_{1,6} = X_1 + \frac{1}{\eta_6} \delta_6(X_1) X_6^{-1} = X_1 \frac{1}{2} X_5 X_6^{-1}.$
- $X_{2,6} = X_2 + \frac{1}{\eta_6} \delta_6(X_2) X_6^{-1} + \frac{1}{2\eta_6^2} \delta_6^2(X_2) X_6^{-2} = X_2 + \frac{3}{2} X_4 X_6^{-1} 3X_3 X_5 X_6^{-1} + X_5^3 X_6^{-2}.$
- $X_{3,6} = X_3 + \frac{1}{\eta_6} \delta_6(X_3) X_6^{-1} = X_3 X_5^2 X_6^{-1}.$
- $X_{4,6} = X_4 + \frac{1}{\eta_6} \delta_6(X_4) X_6^{-1} = X_4 \frac{2}{3} X_5^3 X_6^{-1}.$
- $X_{1,5} = X_{1,6} + \frac{1}{\eta_5} \delta_5(X_{1,6}) X_{5,6}^{-1} + \frac{1}{2\eta_5^2} \delta_5^2(X_{1,6}) X_{5,6}^{-2} = X_{1,6} X_{3,6} X_{5,6}^{-1} + \frac{3}{4} X_{4,6} X_{5,6}^{-2}$
- $X_{2,5} = X_{2,6} + \frac{1}{\eta_5} \delta_5(X_{2,6}) X_{5,6}^{-1} + \frac{1}{2\eta_5^2} \delta_5^2(X_{2,6}) X_{5,6}^{-2} + \frac{1}{3!\eta_5^3} \delta_5^3(X_{2,6}) X_{5,6}^{-3}$ = $X_{2,6} - 3X_{3,6}^2 X_{5,6}^{-1} + \frac{9}{2} X_{3,6} X_{4,6} X_{5,6}^{-2} - \frac{9}{4} X_{4,6}^2 X_{5,6}^{-3}$.
- $X_{3,5} = X_{3,6} + \frac{1}{\eta_5} \delta_5(X_{3,6}) X_{5,6}^{-1} = X_{3,6} \frac{3}{2} X_{4,6} X_{5,6}^{-1}$.
- $X_{1,4} = X_{1,5} + \frac{1}{\eta_4} \delta_4(X_{1,5}) X_{4,5}^{-1} = X_{1,5} \frac{1}{3} X_{3,5}^2 X_{4,5}^{-1}$.
- $X_{2,4} = X_{2,5} + \frac{1}{\eta_4} \delta_4(X_{2,5}) X_{4,5}^{-1} = X_{2,5} \frac{2}{3} X_{3,5}^3 X_{4,5}^{-1}$.
- $X_{1,3} = X_{1,4} + \frac{1}{\eta_3} \delta_3(X_{1,4}) X_{3,4}^{-1} = X_{1,4} \frac{1}{2} X_{2,4} X_{3,4}^{-1}$.
- Otherwise, $X_{i,j} = X_{i,j+1}$, where $1 \le i, j \le 6$.

Appendix C

Definition of scalars used

In this appendix, we define some scalars used in the thesis. Note, for all $n \in \mathbb{N}$, we have already defined the scalars $d_2[n]$ in Lemma A.1.4, hence, we are not going to repeat them here. Any other scalars not defined here must be defined in/before the context in which it is found.

$$\begin{aligned} a &= \frac{q^2 + 1 + q^{-2}}{q^{-2} - 1} & b = -\frac{q^7 - 2q^5 + q^3}{(q^4 + q^2 + 1)(1 - q^{-6})} \\ g &= \frac{q + q^{-1} + q^{-3}}{(1 - q^{-2})^2} & f = \frac{1 - q^2}{1 - q^{-2}} \\ h &= \frac{q + q^{-1}}{q^{-2} - 1} & s = \frac{1 - q^2}{1 - q^{-6}} \\ t &= \frac{q^{-1} - q}{1 - q^{-6}} & u = \frac{q + q^{-1} - q^{-3}}{1 - q^{-6}} \\ p &= \frac{q^4 + q^2 + 1}{q^2 - 1} & r = \frac{-1}{1 - q^{-6}} \\ e &= \frac{-(q^7 + q^5 + q^3)}{q^4 - 2q^2 + 1} & q'' = \frac{q^7 - 2q^5 + q^3}{q^4 + q^2 + 1} \\ n &= \frac{q^{12}}{(q^4 + q^2 + 1)^3} & q' = -(q^2 + 1 + q^{-2}) \\ k_1 &= q^{-3}b_2 + b_6d_2[1] & a' &= af + hq = \frac{q^6}{q^2 - 1} \end{aligned}$$

$$\begin{split} k_2 &= q^{-3}b_3 + b_{12}d_2[1] & b' &= \frac{q^{43} - q^{41}}{(q^4 + q^2 + 1)^2} \\ k_3 &= b_4c_1 & d' &= \frac{q^{42}}{q^6 - 1} \\ k_4 &= b_4c_2 + q^{-3}b_5c_1 + b_7d_2[1] & c' &= -\frac{q^9}{q^4 + q^2 + 1} \\ k_5 &= b_4c_2 + q^{-1}b_6c_1 & c_1 &= \frac{1}{a'} \\ k_6 &= c_3b_4 + q^{-1}b_7 + q^{-3}b_4b_{13}c_2 & c_2 &= -ac_1 \\ k_7 &= q^{-3}c_2b_5 + b_8d_2[1] & c_3 &= -c_1 \\ k_8 &= b_1b_3c_2 & b_1 &= \frac{1}{d'} \\ k_9 &= q^{-4}b_6c_2 + b_9d_2[2] + q^{-3}b_2c_2b_{13} + q^{-3}b_5c_2 & b_2 &= b_1bc_2(q + q^{-1} + q^{-3}) - b_1 \\ k_{10} &= q^{-1}b_6b_8c_2 & b_3 &= -b'b_1c_2 - bb_1 \\ k_{11} &= b_{13}c_1 & b_4 &= -b_1bc_1 \\ k_{12} &= q^{-1}b_6b_7c_2 & b_5 &= b_1b(c_3(q + q^{-1} + q^{-3}) - q^{-3}c_2) \\ k_{13} &= q^{-4}b_6c_3 + q^{-2}b_9 + q^{-3}b_6b_{13}c_2 + q^{-1}b_6b_{13}c_2 & b_6 &= -q^{-1}c_2b_1b \\ k_{14} &= q^{-1}b_1b_6c_2 & b_7 &= -q^{-1}b_1bc_1c_3 \\ k_{15} &= q^{-1}b_2b_6c_2 & b_8 &= b_9 &= -q^{-1}b_1bc_2c_3 \\ k_{16} &= q^{-1}b_3b_6c_2 & b_{13} &= -d^{-1}c_5b_1b \\ k_{17} &= q^{-1}b_4b_6c_2 & b_{11} &= -d^{-1}b_1c_1 \\ k_{18} &= q^{-1}b_5b_6c_2 & b_{12} &= -d^{-1}b_1c_2 \\ k_{20} &= q^{-1}b_6b_1c_2 + q^{-2}b_{10}c_1 + q^{-3}b_7b_{13}c_2 & b_{15} &= q^{-3}c_3 + c_2b_{13} \\ k_{23} &= q^{-1}b_6b_{12}c_2 + q^{-2}b_{10}c_1 + q^{-3}b_7b_{13}c_2 & b_{15} &= q^{-3}c_3 + c_2b_{13} \\ k_{23} &= q^{-1}b_{12}c_1 + b_{11}b_{12}c_2 & k_{24} &= q^{-1}b_6b_1c_2 \\ k_{26} &= q^{-1}b_{12}c_2 + b_{12}b_{13}c_2 & k_{26} &= q^{-1}b_{12}c_2 + b_{13}b_{13}c_2 \\ k_{27} &= q^{-1}b_{12}c_3 + b_{12}b_{13}c_2 & k_{26} &= q^{-1}b_{12}c_2 + b_{13}b_{13}c_2 \\ k_{27} &= q^{-1}b_{12}c_3 + b_{12}b_{13}c_2 & k_{26} &= q^{-1}b_{12}c_2 + b_{13}b_{13}c_2 \\ k_{29} &= b_{13}b_{15}c_1 + d^{-3}b_7b_{13}c_2 & k_{26} &= q^{-1}b_{12}c_2 + b_{13}b_{13}c_2 \\ k_{27} &= q^{-1}b_{12}c_3 + b_{12}b_{13}c_2 & k_{26} &= q^{-1}b_{12}c_2 + b_{13}b_{13}c_2 \\ k_{29} &= b_{13}b_{15}c_1 + d^{-3}b_{15}c_3 + d^{-4}b_8 + b_{10}d_2[2] \\ k_{39} &= b_3b_{13}c_2 + d^{-3}b_{12}c_2 + d^{-3}b_{13}c_2 + d^{-1}b_{12}c_2 \\ k_{29} &= b_{13}b_{15}c_2 + q^{-3}b_{15}c_3 + q^{-4}b_8 + b_{10}d_2[2] \\ k_{39} &=$$

Bibliography

- [1] N. Andruskiewitsch and F. Dumas. On the automorphisms of $U_q^+(\mathfrak{g})$. *IRMA Lectures in Mathematics and Theoritical Physics*, 12:107–133, 2008.
- [2] V. Bavula. Filter dimension of algebras and modules, a simplicity criterion of generalized Weyl algebras. *Communications in Algebra*, 24:1971–1992, 1996.
- [3] V. V. Bavula. Generalized Weyl algebras and their representations. *Algebra i Analiz*, 4:75–97, 1992.
- [4] J. Bell and S. Launois. On the dimension of *H*-strata in quantum algebras. *Algebra & Number Theory*, 4:175–200, 2010.
- [5] A. Belov-Kanel and M. Kontsevich. Automorphisms of the Weyl algebra. Letters in mathematical physics, 74:181–199, 2005.
- [6] A. Belov-Kanel and M. Kontsevich. The Jacobian Conjecture is stably equivalent to the Dixmier Conjecture. *Moscow Mathematical Journal*, 7:209–218, 2007.
- [7] K. A. Brown and K. R. Goodearl. *Lectures on algebraic quantum groups*. Advanced Courses in Mathematics CRM Barcelona (Birkhäuser, Basel, 2002).
- [8] G. Cauchon. Effacement des dérivations et spectres premiers des algebres quantiques.
 J. Algebra, 260:476–518, 2003.
- [9] W. A. De Graaf. Computing with quantized enveloping algebras: PBW-type bases, highest-weight modules and R-matrices. *Journal of Symbolic computation*, 32:475– 490, 2001.

- [10] W. A. De Graaf and T. GAP Team. QuaGroup, computations with quantum groups, Version 1.8.2. https://gap-packages.github.io/quagroup/, Oct 2019. Refereed GAP package.
- [11] J. Dixmier. Sur les algebres de Weyl. Bulletin de la Société mathématique de France, 96:209–242, 1968.
- [12] J. Dixmier. Enveloping algebras. Graduate Studies in Mathemics. American Mathematical Society, Providence (1996). Revised reprint of the 1977 translation, 1996.
- [13] V. G. Drinfel'd. Hopf algebras and the quantum Yang-Baxter equations. Doklady Akademii Nauk SSSR, 283:1060–1064, 1985.
- [14] F. Dumas. Rational equivalence for poisson polynomial algebras. Lecture notes, December, 2011.
- [15] K. Erdmann and M. J. Wildon. Introduction to Lie algebras. Springer Science & Business Media, 2006.
- [16] K. R. Goodearl. A Dixmier-Moeglin equivalence for Poisson algebras with torus actions. *Contemporary Mathematics*, 419:131–154, 2006.
- [17] K. R. Goodearl and S. Launois. Catenarity in quantum nilpotent algebras. Proceedings of the American Mathematical Society, Series B, 7:202–214, 2020.
- [18] K. R. Goodearl, S. Launois, and T. Lenagan. Tauvel's height formula for quantum nilpotent algebras. *Communications in Algebra*, 47:4194–4209, 2019.
- [19] K. R. Goodearl and T. Lenagan. Catenarity in quantum algebras. Journal of Pure and Applied Algebra, 111:123–142, 1996.
- [20] K. R. Goodearl and E. S. Letzter. Prime and primitive spectra of multiparameter quantum affine spaces. *Trends in ring theory (Miskolc, 1996)*, 22:39–58, 1998.

- [21] K. R. Goodearl and E. S. Letzter. The Dixmier-Moeglin equivalence in quantum coordinate rings and quantized Weyl algebras. *Transactions of the American Mathematical Society*, 352:1381–1403, 2000.
- [22] K. R. Goodearl and R. B. Warfield Jr. An introduction to noncommutative noetherian rings. Cambridge University Press, 2004.
- [23] K. R. Goodearl and M. Yakimov. From quantum Ore extensions to quantum tori via noncommutative UFDs. Advances in Mathematics, 300:672–716, 2016.
- [24] M. Gorelik. The prime and the primitive spectra of a quantum Bruhat cell translate.J. Algebra, 227:211–253, 2000.
- [25] J. E. Humphreys. Introduction to Lie algebras and representation theory. Springer, 1972.
- [26] J. C. Jantzen. Lectures on quantum groups. American Mathematical Society, 1996.
- [27] M. Jimbo. A q-difference analogue of $U(\mathfrak{g})$ and the Yang-Baxter equation. Letters in Mathematical Physics, 10:63–69, 1985.
- [28] O. Keller. Ganze cremona-transformationen. Monatshefte für Mathematik und Physik, 47:299–306, 1939.
- [29] A. Kitchin. On the quantum analogues of the Weyl algebras. PhD Thesis, University of Kent, 2016.
- [30] G. R. Krause and T. H. Lenagan. Growth of algebras and Gelfand-Kirillov dimension. American Mathematical Society, 2000.
- [31] T. Y. Lam. A first course in noncommutative rings. Second Edition. Graduate Texts in Mathematics. Springer-Verlag, 2001.
- [32] S. Launois. On the automorphism groups of *q*-enveloping algebras of nilpotent Lie algebras. *arXiv preprint arXiv:0712.0282*, 2007.

- [33] S. Launois. Primitive ideals and automorphism group of $U_q^+(B_2)$. J. Algebra, 6:21–47, 2007.
- [34] S. Launois and C. Lecoutre. A quadratic Poisson Gelfand-Kirillov problem in prime characteristic. *Transactions of the American Mathematical Society*, 368:755–785, 2016.
- [35] S. Launois and C. Lecoutre. Poisson deleting derivations algorithm and Poisson spectrum. *Communications in Algebra*, 45:1294–1313, 2017.
- [36] S. Launois and S. A. Lopes. On the Hochschild cohomology and the automorphism group of $U_q(\mathfrak{sl}_4^+)$. arXiv preprint math/0606134, 2006.
- [37] S. Z. Levendorskii and Y. S. Soibelman. Algebras of functions on a compact quantum group, schubert cells and quantum tori. *Communication in Mathematical Physics*, 139:141–170, 1991.
- [38] G. Lusztig. Introduction to quantum groups. Springer Scienece + Business Media, 2010.
- [39] S.-Q. Oh. Poisson polynomial rings. Communications in Algebra, 34:1265–1277, 2006.
- [40] J. M. Osborn and D. Passman. Derivations of skew polynomial rings. Journal of Algebra, 176:417–448, 1195.
- [41] L. Richard and A. Solotar. Isomorphisms between quantum generalized Weyl algebras. J. Algebras, 5:271–285, 2004.
- [42] C. M. Ringel. PBW-bases of quantum groups. Journal fur die Reine und Angewandte Mathematik, 470:51–88, 1996.
- [43] Y. Tsuchimoto. Preliminaries on Dixmier Conjecture. Mem. Fac. Sci. Kochi Univ. Ser. A Math, 24:43–59, 2003.

- [44] M. Yakimov. Invariant prime ideals in quantizations of nilpotent Lie algebras. London Mathematical Society, 101:454–476, 2010.
- [45] M. Yakimov. Rigidity of quantum tori and the Andruskiewitsch–Dumas Conjecture. Selecta Mathematica, 20:421–464, 2014.